RHEOLOGY AND MICROSTRUCTURE OF STABLE CONCENTRATED IONIC LIQUID COLLOIDAL SUSPENSIONS

by

Jingsi Gao

A dissertation submitted to the Faculty of the University of Delaware in partial fulfillment of the requirements for the degree of Doctor of Philosophy in Chemical Engineering

Winter 2017

© 2017 Jingsi Gao All Rights Reserved ProQuest Number: 10256117

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10256117

Published by ProQuest LLC (2017). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346

RHEOLOGY AND MICROSTRUCTURE OF STABLE CONCENTRATED IONIC LIQUID COLLOIDAL SUSPENSIONS

by

Jingsi Gao

Approved:

Abraham M. Lenhoff, Ph.D. Chair of the Department of Chemical and Biomolecular Engineering

Approved:

Babatunde A. Ogunnaike, Ph.D. Dean of the College of Engineering

Approved:

Ann L. Ardis, Ph.D. Senior Vice Provost for Graduate and Professional Education

Signed:	I certify that I have read this dissertation and that in my opinion it meets the academic and professional standard required by the University as a dissertation for the degree of Doctor of Philosophy.
Signedi	Norman J. Wagner, Ph.D. Professor in charge of dissertation
Circuit I	I certify that I have read this dissertation and that in my opinion it meets the academic and professional standard required by the University as a dissertation for the degree of Doctor of Philosophy.
Signed:	Antony N. Beris, Ph.D. Member of dissertation committee
	I certify that I have read this dissertation and that in my opinion it meets the academic and professional standard required by the University as a dissertation for the degree of Doctor of Philosophy.
Signed:	Eric M. Furst, Ph.D. Member of dissertation committee
	I certify that I have read this dissertation and that in my opinion it meets the academic and professional standard required by the University as a dissertation for the degree of Doctor of Philosophy.
Signed:	Mark B. Shiflett, Ph.D. Member of dissertation committee

ACKNOWLEDGMENTS

This dissertation would not have been possible without the guidance and help from a number of people. First, I would like to thank my advisor Norm Wagner, who brought me to the field of colloids and rheology. His enthusiasm and passion on pursuing new ideas and encouragement to excellence set as an academic role model for me. He is not only my academic mentor, but also providing support and guidance on many aspects of life. Thank you, Norm.

Thank you to all of my committee members: Antony Beris, Eric Furst, April Kloxin, and Mark Shiflett, for their insightful suggestions on my thesis research work. I am also grateful to thank Florian Nettesheim, my supervisor during my summer internship at DuPont. His professional supervision has shaped the project and led to a smooth progress and success on the summer project.

I'd like to thank the members of the Wagner research group of past and present, for their keen insights to my dissertation and being helpful labmates. A special thank you to Dongcui Li, who introduced me into the Wagner group and offered great help to get me started. Thanks to Rose Ndong for training me on particle synthesis, Paul Mwasame for all the helpful discussions on my dissertation. I also would like to thank Paul Butler and Yun Liu at NCNR for their tremendous help with neutron scattering measurements and data analysis. Thank you to a number of colleagues at UD. Thanks to Chaoying Ni for *in situ* TEM measurements, Roddel Remy for TGA analysis, Shu Wang and Wei-Fan Kuan for NMR tests. Thanks to

iv

many DuPont colleagues, especially Mike Pottiger, Mike Wolfe, Hoang Vi Tran, and Hee Hyun Lee for support during my summer internship.

To all of my friends in Delaware and far away in China, thank you for your support and keeping my life fun. Special thanks to Chia-Hung Tsai and Jinlin Jiang for their great help without hesitation, Yan Zhang and Huibo Sheng for accompanying with my daughter Sophia spending great fun time.

At last but not at least, I owe a great debt of gratitude to my family. Thank you to my parents, brother and sister, who have encouraged and supported me on everything over the years. Many thanks to my lovely daughter Sophia for your curiosity to everything and your love. And to my husband Ming Luo, without your love and encouragement, this dissertation would never have been close to possible.

TABLE OF CONTENTS

LIST (OF TA	ABLES			xi
ABST	RAC	Γ	,,		
112011					
Chapte	er				
1	INT	RODUC	CTION		1
	1.1	Motiva	ation		1
	1.2	Shear	Thickenir	g Colloidal Suspensions	3
	1.3	Ionic I	Liquids		7
	1.4	Ionic I	Liquids Co	olloidal Suspensions	14
	1.5	Shear	Thickenir	ng Fluid (STF) – Armor TM	
	1.6	Thesis	Objectiv	es and Overview	
REFEI	RENC	CES			25
2	EXP	ERIME	ENTAL M	ETHODS	
	2.1	Dynan	nic Light	Scattering (DLS)	
	2.2	Scanni	ing/Trans	mission Electron Microscopy (SEM, TEM)	
	2.3	Densit	ometry		
	2.4	Therm	ogravime	tric Analysis (TGA)	41
	2.5	Rheolo	ogical Cha	aracterization	
		251	Falling-l	Ball Viscometry	43
		2.5.2	Rotation	al Rheometry	
				•	
			2.5.2.1	Steady Shear Measurements	46
			2.5.2.2	Small-Amplitude Oscillatory Shear (SAOS)	
				Measurements	47
		2.5.3	Pressure	/Vacuum Cell Geometry	
	2.6	Small-	Angle Ne	eutron Scattering (SANS)	
		2.6.1	Porod's	Law	

		2.6.2	The Form Factor	55
		2.6.3	The Structure Factor	57
	2.7	Small	-Angle X-Ray Scattering (SAXS)	59
	2.8	Quasi	-Static Puncture Test	60
REFE	REN	CES		63
3	EFF	ECT O	F WATER ON PHYSICAL PROPERTIES AND	
	STR	RUCTU	RES OF IONIC LIQUIDS	65
	3.1	Introd	uction	66
		3.1.1 3.1.2	Densities and Viscosities of Ionic Liquid Aqueous Solutions Microstructure and Microphase Transition of [C ₄ mim][BF ₄]	66
			Aqueous Solutions	00
	3.2	Mater	ials and Methods	70
		3.2.1 3.2.2	Materials Fluorine Nuclear Magnetic Resonance Spectroscopy (¹⁹ F	70
		222	NMR)	72
		3.2.3 3.2.4	Viscosity Measurements	73
		3.2.5	Small-Angle Neutron Scattering (SANS)	73
		3.2.6	SANS Data Analysis	74
	3.3	Result	ts and Discussion	75
		3.3.1	Hydrolysis Analysis of Ionic Liquid [C4mim][BF4] with	
		227	Present of Water	75
		5.5.2	riysical rioperues Dependence	70
			3.3.2.1 Density	79
			3.3.2.2 Viscosity	83
			5.5.2.5 Density-viscosity Correlation	00
		3.3.3	Microstructure of Ionic Liquid Aqueous Solutions	96
			3.3.3.1 Ionic Liquid Micellar Formation	96
			3.3.3.2 Water Cluster Formation	. 101
	3.4	Concl	usions	. 111
REFE	REN	CES		. 114

4	PAF	RTICLE	SYNTH	ESIS AND CHARACTERIZATION	
	4.1 4.2 4.3	Mater Partic Partic	ials and M le Synthes le Charact	lethods sis erization	120 123 127
		4.3.1	Surface	Coating	127
			4.3.1.1 4.3.1.2	Proton Nuclear Magnetic Resonance (¹ H NMR) Thermogravimetric Analysis (TGA)) 127 129
		4.3.2	Particle	Size	130
			4.3.2.1 4.3.2.2 4.3.2.3	Scanning Electron Microscopy (SEM) Dynamic Light Scattering (DLS) Small-Angle Neutron Scattering (SANS)	130 133 135
		4.3.3	Particle	Density	139
	4.4	Concl	usions		140
REFE	EREN	CES			
5	FLU ME	JOROC CHANI	ARBON SM AND	COATED PARTICLE SYSTEM – STABILIZA' INTERFACIAL STRUCTURE	TION 143
	5.1 5.2	Introd Mater	uction ials and M	Iethods	143 148
		5.2.1 5.2.2	Particle Derjagu Potentia	Suspension Preparation in-Landau-Verwey-Overbeek (DLVO) Inter-part	148 icle 149
		5.2.3	Characte	erization Techniques	
	5.3	Result	ts and Dis	cussion	154
		5.3.1 5.3.2 5.3.3	Dilute P DLVO I Concent	article Dispersions – Single Particle Properties Potential between Two Silica Particles rated Particle Dispersions – Dispersion Microstru	154 156 ucture158
	5.4	Concl	usions		165
REFE	EREN	CES			168

6	ALC ANI	COHOL D THEF	COATED PARTICLE SYSTEM – SHEAR THICKENING, RMAL RESPONSE	173
	6.1	Introd	uction	174
	6.2	Mater	als and Methods	179
		6.2.1	Materials	179
		6.2.2	Rheology	180
		6.2.3	Dynamic Light Scattering (DLS)	180
		6.2.4	Small-Angle Neutron Scattering (SANS)	180
	6.3	Result	s and Discussion	181
		6.3.1	Steady Shear and Shear Thickening	181
		6.3.2	Linear Viscoelasticity	187
		6.3.3	Inter-Particle Interaction and Dispersion Structure	190
	6.4	Conclu	usions	202
REFE	RENG	CES		204
7	PEN THI	IETRAT CKENI	FION RESISTANCE OF IONIC LIQUIDS BASED SHEAR NG FLUIDS (STFS)-KEVLAR [®] NANOCOMPOSITES	209
	7.1	Introd	uction	209
	7.2	Mater	als and Methods	212
		7.2.1	Materials	212
		7.2.2	Quasi-Static Puncture Tests	
	7.3	Result	s and Discussion	215
		7.3.1	Spike Penetration Measurements	215
			7.3.1.1 Effect of STFs Carrier Medium	215
			7.3.1.2 Effect of Ethanol: STFs Solution Ratio	220
		7.3.2	Needle Penetration Measurements	228
	7.4	Conclu	usions	236
REFE	RENG	CES		238
8	CON	JCI US	IONS AND FUTURE DIRECTIONS	242
0		ICLUD.		272

8.1	Summary		242
	8.1.1 Ionic Liquid	ls-Water Interaction and Phase Transition	243
	8.1.2 Stabilization	Mechanism and Interfacial Structure of Silica	
	Particles in 1	Ionic Liquids	245
	8.1.3 Ionic Liquid	ls Based Shear Thickening Fluids and Thermal	
	Response		247
	8.1.4 Protective P	roperties of Ionic Liquids Based STF-Armor TM ?	248
8.2	Recommendations f	for Future Work	250
REFEREN	ES		259
Appendix			

Α	SUPPORTING INFORMATION FOR CHAPTER 3	261
В	SHEAR RHEOLOGY OF FUMED AMORPHOUS SILICA PARTICLES	
	IN IONIC LIQUID	
С	TEXT AND FIGURE REPRINT PERMISSIONS	. 283

LIST OF TABLES

Table 3.1:	[C ₄ mim][BF ₄]-water mixtures compositions in mass fraction and molar fraction72
Table 3.2:	Density, molecular mass, and SLD of ionic liquid $[C_4mim][BF_4]$ and D_2O at 25 °C75
Table 3.3:	Fitting parameters of the linear relation of density - water content (Figure 3.4A) and density - temperature (Figure 3.4B) for ionic liquid aqueous solutions
Table 3.4:	Fitting parameters (A , B and T_0) of the VFT equation for [C ₄ mim][BF ₄] aqueous solutions with various compositions
Table 3.5:	Summary of validation of semi-empirical linear correlation between excess log viscosity and excess molar volume of various ionic liquids94
Table 3.6:	Summary of ionic liquid micelle size of $[C_4 mim][B_F]-H_2O$ solutions extracted from sphere and ellipsoid form factor model fittings
Table 3.7:	Summary of D_2O water distribution in water dissolved state and microphase separated state for mixtures of $[C_4mim][BF_4]$ and D_2O for water contents above 20 wt% at 25 °C
Table 4.1:	Summary of chemicals used in particle surface coating and characteristics
Table 4.2:	Particle radii for uncoated and coated particles in various solvent measured by DLS
Table 4.3:	The parameters used for SANS form factor fittings
Table 4.4:	Summary of particle radii for uncoated and coated particles with various methods
Table 5.1:	Densities of particles, fluorocarbon and solvents at 25 °C 149
Table 5.2:	Summary of particle hydrodynamic radii for uncoated and coated particles in different solvents measured by various technique methods.156

Table 5.3:	Parameters for the calculation of the DLVO potential for colloidal dispersions in $[C_4mim][BF_4]$ and H_2O /mother liquid at 25 °C 157
Table 5.4:	Summary of structure factor fitting parameters for dispersions coated nanoparticles in [C ₄ mim][BF ₄] corresponding to Figure 5.5
Table 5.5:	Summary of relative viscosity, effective volume fraction and solvation layer thickness calculated from shear flow curves of $\phi = 0.2$ and 0.3 suspensions
Table 6.1:	Summary of particle radii of silica particle in different solvents at 25 °C using various methods
Table 7.1:	Summary of increased areal density (%), penetration peak force with its p-value of t-test (two-sample t-test compared with untreated Kevlar [®] 1148), and dissipated energy with its p-value of t-test (two- sample t-test compared with untreated Kevlar [®] 1148) for Kevlar [®] 1148 before and after treatment by solvents: ethanol, ionic liquid ([C ₄ mim][BF ₄]), and PEG 200
Table 7.2:	Summary of increased areal density (%), penetration peak forces, and dissipated energies during spike penetration measurements for Kevlar [®] 1148 before and after treatment by dry KEP50 particle powders with various volume ratios of ethanol : KEP50
Table 7.3:	Summary of increased areal density (%), penetration peak forces, and dissipated energies during spike penetration measurements of Kevlar [®] 1148 before and after treatment by STFs of $\phi = 0.5$ KEP50 in PEG 200 various volume ratios of ethanol : STFs
Table 7.4:	Summary of increased areal density (%), penetration peak forces, and dissipated energies during spike penetration measurements for Kevlar [®] 1148 before and after treatment by STFs of $\phi = 0.6$ KEP50 in ionic liquid [C ₄ mim][BF ₄] with various volume ratios of ethanol : STFs
Table 7.5:	Summary of increased areal density (%), penetration peak forces, and dissipated energies during spike penetration measurements for Kevlar [®] 1148 before and after treatment by STFs of $\phi = 0.5$ KEP10 in ionic liquid [C ₄ mim][BF ₄] with various volume ratios of ethanol : STFs

Table 7.6:	Summary of increased areal density (%), penetration peak forces, and dissipated energies during spike penetration measurements for Kevlar [®] 1148 before and after treatments
Table 7.7:	Summary of increased areal density (%), penetration peak forces, and dissipated energies during 18G needle penetration tests for Kevlar [®] 1148 before and after treatment by various ionic liquid based and PEG based STFs
Table 7.8:	Summary of increased areal density (%), penetration peak forces, and dissipated energies during 21G needle penetration tests for Kevlar [®] 1148 before and after treatment by various ionic liquid based and PEG based STFs
Table A.1:	Summary of experimental density ρ , viscosity η , density deviation $\Delta \rho$, excess molar volume V^{ex} and excess log viscosity $(log\eta)^{ex}$ of $[C_4 mim][BF_4]$ aqueous solutions at various temperatures
Table A.2:	Fitting parameters of the Redlich-Kister equation for excess molar volume V^{ex} and density deviation $\Delta \rho$ of dependence on water molar fraction x_w for [C ₄ mim][BF ₄] aqueous solutions at varying temperatures. 263
Table A.3:	Fitting parameters of the exponential function Equation 3-4 of the viscosity dependence on water molar fraction for $[C_4mim][BF_4]$ aqueous solutions at different temperatures
Table A.4:	Fitting parameter (activation energy E_a) of the Arrhenius equation of $[C_4 mim][BF_4]$ aqueous solutions with various water contents
Table B.1:	Characteristics of fumed amorphous particles
Table B.2:	Comparison of characteristic parameters for 10 wt% Aerosil 200 - [C ₄ mim][BF ₄] suspensions mixed with various methods
Table B.3:	Comparison of characteristic parameters of 10 wt% hydrophilic silica particle (Aerosil 90, 200, and 380) -[C ₄ mim][BF ₄] suspensions
Table B.4:	Yield stress of 10 wt% suspensions of hydrophilic (Aerosil 200) and hydrophobic silica particle (Aerosil R104, R805) in [C ₄ mim][BF ₄] 280

LIST OF FIGURES

Figure 1.4:	Calculated van der Waals attractive (solid line) and electrostatic	
	repulsive (dash line) potential profiles between two silica particles at	
	T = 25 °C for silica particles in [C ₄ mim][BF ₄] (magenta) and in 0.1	
	mM NaCl aqueous solution (blue).	. 16

- (A) Schematic illustration of the structured solvent around fluorinated Figure 1.5: coated silica nanoparticle in ionic liquid [C₄mim][BF₄]. This structured solvent is initiated by the hydrogen bonding between coated fluorocarbon chains and [BF4]⁻ anion groups. Cations form cations layers around anion through ionic bonds and this ordered layer structure gradually becomes disordered when away from the surface of particles. (B) Solvation force between two macroscopic silica surfaces in [C₄mim][BF₄]: Normal force scaled with surface curvature radius (F/R) as a function of surface separation D. Dotted lines represent van der Waals attractions calculated from Lifshitz theory. Solid and dashed lines in plots indicate stable and unstable regions, respectively. Part (A) is adapted with permission from Gao J., et al. ACS Nano 2015, 9, (3), 3243-3253. Copyright 2015 American Chemical Society; Part (B) is adapted with permission from Ueno, K., et al. Phys. Chem. Chem. Phys. 2010, 12, 4066-4071. Copyright 2010 The Royal Society of Chemistry. 17

Figure 2.3:	The vacuum cell setup for rheological test on ionic liquid based STFs under simulated space vacuum condition. (A) Vacuum cell on the AR- G2 rheometer to measure the rheological response; (B) Turbomolecular pump station to reduce the pressure inside the vacuum cell to 10 ⁻⁶ torr
Figure 2.4:	Cross-section schematic of vacuum/pressure cell. ¹²
Figure 2.5:	Pressure cell rotor assembly with cap on (A) and off (B). ¹²
Figure 2.6:	Illustration of a typical SANS experiment
Figure 2.7:	The graphical description of a sticky hard sphere potential for a coated particle system
Figure 2.8:	Set-up of quasi-static puncture test apparatus used to evaluate the quasi-static puncture resistance of Kevlar fabric with or without STFs treatment. (A) Photo of the modified ASTM F-1342 standard (with needle probe). (B) Optical image of hypodermic needles (18 G and 21 G) and Malco M0 spike on Kevlar 1148. (C) The dimensions of unbacked sample holder. Image courtesy of Dr. Colin D. Cwalina 61
Figure 3.1:	¹⁹ F-NMR spectrum of pure $[C_4mim][BF_4]$ (A), $[C_4mim][BF_4]$ aqueous solution with 5 w% H ₂ O after 30 <i>min</i> at 298.15 K (B) and $[C_4mim][BF_4]$ aqueous solution with 5 w% H ₂ O after density/viscosity measurement for the entire temperature range (C) in CDCl ₃
Figure 3.2:	Comparison of measured densities and dynamic viscosities with literature values for pure $[C_4 mim][BF_4]$ at temperatures from 283.15 K to 373.15 K. The literature data are from: Ries <i>et al.</i> , ⁴ Zhou <i>et al.</i> , ⁵ Taib <i>et al.</i> , ⁶ , Rilo <i>et al.</i> , ^{7,36} Liu <i>et al.</i> , ⁹ Ciocirlan <i>et al.</i> , ¹⁴ Wang <i>et al.</i> , ³⁷ and Zhu <i>et al.</i> ³⁸
Figure 3.3:	Comparison of measured densities and dynamic viscosities with literature values at 298.15 K for $[C_4 mim][BF_4]$ with various water contents. The literature data are from: Ries <i>et al.</i> , ⁴ Taib <i>et al.</i> , ⁶ Rilo <i>et al.</i> , ^{7,36} Liu <i>et al.</i> , ⁹ Wang <i>et al.</i> , ³⁷ and Zhu <i>et al.</i> ³⁸
Figure 3.4:	The densities of [C ₄ mim][BF ₄] aqueous solution for varying water content (A) and temperature (B). Solid lines represent the linear correlations, respectively

Figure 3.5:	The experimental densities of $[C_4mim][BF_4]$ aqueous solution at different water contents compared with calculated densities based on ideal mixing theory. The solid/dash lines represent the linear correlations between experimental/calculated density and water mass fraction
Figure 3.6:	Excess molar volume V^{ex} of [C ₄ mim][BF ₄] aqueous solution dependence on water molar fraction x_w (A) and temperature (B). The symbols represent experimental value. The solid lines are fits to the Redlich-Kister equation
Figure 3.7:	The viscosities of $[C_4mim][BF_4]$ aqueous solution at different water molar fraction (A) and temperatures (B). Solid lines represent an empirical correlation (A) and the Vogel-Fulcher-Tammann (VFT) equation correlation (B), respectively
Figure 3.8:	(A) Viscosities of $[C_4 mim][BF_4]$ aqueous solutions as a function of water mole fraction (solid lines and symbols) compared with calculated viscosities based on ideal mixing theory (-x-). (B) Excess log viscosity, $(log\eta)^{ex}$, vs molar fraction of water x_w for $[C_4 mim][BF_4]$ aqueous mixtures as a function of temperature
Figure 3.9:	The linear correlation between excess log viscosity, $(log\eta)^{ex}$, and excess molar volume, V^{ex} , of [C ₄ mim][BF ₄] aqueous solution at various temperatures
Figure 3.10:	The relative residues of excess log viscosity $(log\eta)^{ex}$ from correlation Equation 3-9 of [C ₄ mim][BF ₄] aqueous solutions at various temperatures and compositions
Figure 3.11:	The semi-empirical linear correlation between excess log viscosity, $(log\eta)^{ex}$, and excess molar volume, V^{ex} , of [Pdmim][BF ₄] (A) and [OcPy][BF ₄] (B) aqueous solutions in salt-rich regime respectively at various temperatures
Figure 3.12:	SAXS intensity profiles of $[C_4mim][BF_4]$ aqueous solutions (in H ₂ O) with various ionic liquid contents at 25 °C
Figure 3.13:	The comparisons of sphere and ellipsoid form factor model fittings for SAXS intensity profiles of mixtures of $[C_4mim][BF_4]$ in H ₂ O at 25 °C with ionic liquid concentrations of 0.6 mol/L to 2.0 mol/L 100
Figure 3.14:	SANS profiles of mixtures of [C ₄ mim][BF ₄] and D ₂ O at 25 °C103

Figure 3.15:	SANS profiles and model fitting for mixtures of $[C_4mim][BF_4]$ and D_2O at 25 °C for higher D_2O concentrations
Figure 3.16:	The volume fraction and size of D_2O clusters as a function of the overall D_2O weight fraction for $[C_4mim][BF_4]$ and D_2O mixtures at 25 °C
Figure 3.17:	Mole fraction of dissolved D_2O and the molar ratio of dissolved D_2O to ionic liquid as a function of total D_2O mole fraction for mixtures of $[C_4mim][BF_4]$ and D_2O at 25 °C
Figure 3.18:	The schematic illustration of structures in mixtures of $[C_4mim][BF_4]$ and D_2O
Figure 4.1:	Chemical structure of fluorocarbon 1H,1H,9H hexadecafluoro-1- nonanol used to functionalize the particle surface
Figure 4.2:	A schematic flow chart of the particle coating procedure
Figure 4.3:	The experimental apparatus for the particle surface coating, consisting of a rotary evaporator and a heating oil bath. (a) Condenser to cool down evaporated water and ethanol, (b) collection flask to collect water/ethanol condensation, (c) reaction flask, (d) oil bath, and (e) condenser cooling water. Image courtesy of Dr. Jung Min Kim 126
Figure 4.4:	¹ H NMR spectra of uncoated silica particles (top), fluorocarbon coated silica particles (middle), and 1H,1H,9H hexadecafluoro-1-nonanol (CAS 376-18-1, $C_9H_4F_{16}O$) (bottom) in CDCl ₃ 128
Figure 4.5:	TGA curves for uncoated and coated particles, showing a weight loss of 2.3 wt% and 4.5 wt% for uncoated and coated particles when temperature increases to ~900 °C, respectively
Figure 4.6:	Particle size analysis of dried, uncoated silica particles. (A) Raw SEM image; (B) SEM image processed by ImageJ software: blue indicates the particles are considered in the area distribution analysis; black indicates the particles are excluded from the analysis (C) Histogram of particle area distribution; (D) Statistical results of the area and radius of the silica particles

Figure 4.7:	Particle size analysis of dried, coated silica particles. (A) Raw SEM image; (B) SEM image analyzed by ImageJ software: blue indicates the particles are considered in the area distribution analysis; black indicates the particles are excluded from the analysis (C) Histogram of particle area distribution; (D) Statistical results of the area and radius of the silica particles
Figure 4.8:	Size distribution of uncoated and coated particles suspended in various solvents from DLS measurement
Figure 4.9:	Measured SANS scattering intensity at 25 °C for three dilute dispersions and corresponding fits: uncoated particles in H ₂ O, coated particles in ethanol and in ionic liquid [C ₄ mim][BF ₄], respectively 137
Figure 4.10:	Results of the densitometry experiments for the uncoated silica particles in mother liquid (secondary y axis, right) and fluorocarbon coated silica particles in $[C_4mim][BF_4]$ (primary y axis, left). The lines represent a fit of equation 4-1 to the inverse of the suspension density and the mass fraction of the disperse phase
Figure 5.1:	Schematic illustration of a coated silica nanoparticle stabilized by solvation layers in ionic liquid $[C_4mim][BF_4]$. Note the ion structure in the layers is schematic
Figure 5.2:	Chemical structure of fluorocarbon 1H,1H,9H hexadecafluoro-1- nonanol used to functionalize the particle surface (A) and ionic liquid 1-butyl-3-methylimidazolium tetrafluoroborate ([C ₄ mim][BF ₄]) (B) 148
Figure 5.3:	Transmission electron microscopy (TEM) images of (A) the dried fluorocarbon-coated particle and (B) fluorocarbon-coated particles in $[C_4mim][BF_4]$
Figure 5.4:	Calculated DLVO interparticle potential profiles at $T = 25$ °C for silica particles in [C ₄ mim][BF ₄] (A) and in 0.1 mM (pH = 9.1 ± 0.1) NaCl aqueous solution (mother liquid) (B)
Figure 5.5:	SANS intensity <i>I versus</i> scattering wave vector q as a function of particle concentration for 25 °C and the corresponding sticky-hard sphere model fits. (The scattering intensities are systematically shifted by a factor of 10 for clarity.)
Figure 5.6:	The SANS intensity of $\phi = 0.44$ coated particle [C ₄ mim][BF ₄] dispersion with sticky-hard sphere and hard sphere model fits

Figure 5.7:	Relative viscosity (with error bar) <i>versus</i> shear stress for $\phi = 0.2$ and 0.3 stable suspension of fluorocarbon-coated silica nanoparticles in [C ₄ mim][BF ₄]. Closed symbols are ascending stress curves and the open symbols are descending stress curves. The solid lines are the corresponding relative zero shear viscosity; the dashed lines are the corresponding relative high shear viscosity
Figure 6.1:	Steady shear viscosity as a function of shear rate for colloidal dispersion $= 0.4$ silica particle $- [C_4 mim][BF_4]$ at (a) low temperatures and (b) high temperatures. Viscosity data are acquired with ascending shear rate
Figure 6.2:	Steady shear relative viscosity as a function of <i>Pe</i> and shear stress of $\phi = 0.4$ silica particle – [C ₄ mim][BF ₄] dispersion at different experimental temperatures. The relative viscosity curves are compared with the flow curve of $\phi = 0.4$ dispersion of same silica particles in polyethylene glycol (PEG-200) at 25 °C. ³⁵
Figure 6.3:	Comparison of steady shear relative viscosity of KE-P10 – $[C_4 mim][BF_4]$ dispersion, KE-P10 – polyethylene glycol (PEG-200) dispersion, ³⁵ and fluorinated silica particle – $[C_4 mim][BF_4]$ dispersion ⁶ of same volume fraction $\phi = 0.4$ at 25 °C
Figure 6.4:	(a) Dynamic moduli of $\phi = 0.4$ silica particle – [C ₄ mim][BF ₄] dispersion at different temperatures; and (b) shifted dynamic moduli curves according to time-temperature superposition (TTS) described by Shikata and Pearson. ⁴⁷ The closed symbols represent storage modulus G' and open symbols represent loss modulus G''(a) or subtracted G''- $\omega \eta_{\infty}$ (b)
Figure 6.5:	Shift factors a_T and viscosity change of $[C_4 \text{mim}][BF_4]$ referenced to 25 °C (left y axis) and relative high frequency limiting viscosity $\eta'_{r,\infty}$ (right y axis) of $\phi = 0.4$ silica particle – $[C_4 \text{mim}][BF_4]$ dispersion at various temperatures
Figure 6.6:	SANS profiles of (a) dilute silica particles in D_2O and $[C_4mim][BF_4]$ solvents with Schulz-distributed sphere form factor model fitting; (b) silica particles in $[C_4mim][BF_4]$ of various concentrations with "sticky" hard sphere structure factor model fitting at 25 °C

Figure 6.7:	Hydrodynamic radius (intensity average) of KE-P10 silica particle or aggregations in $[C_4mim][BF_4]$ measured by DLS at the temperatures noted. Error bar is the standard deviation of measured particle size, indicating the polydispersity of particle or aggregations
Figure 6.8:	(a) SANS profiles of $\phi = 0.4$ silica particle – [C ₄ mim][BF ₄] dispersion at various temperatures with "sticky" hard sphere structure factor model fittings; (b) Extracted solvation layers thickness and stickiness parameter τ_b of different temperatures from SANS fitting of $\phi = 0.4$ silica particle – [C ₄ mim][BF ₄] dispersion
Figure 6.9:	State positions of $\phi = 0.4$ silica particle – [C ₄ mim][BF ₄] dispersion at different temperatures in state diagram of AHS particle system. ⁵² The dark-yellow circle points are the experimental gel points of particles of 100 nm diameter size ⁵⁵ and a gel line shown as the dash line is determined from the experimental gel points
Figure 7.1:	The increased areal density (%) of Kevlar [®] 1148 fabric (3 inch \times 3 inch) after treated by various solvents: ethanol, ionic liquid ([C ₄ mim][BF ₄]), and PEG 200
Figure 7.2:	The force curves (force vs. displacement) of Kevlar [®] 1148 before and after treated by solvents: ethanol, ionic liquid ($[C_4mim][BF_4]$), and PEG 200 with twelve repeats. 218
Figure 7.3:	Box charts of penetration peak forces and dissipated energies for Kevlar [®] 1148 with or without solvents treatments
Figure 7.4:	The increased areal density (%) of Kevlar [®] 1148 fabric (3 inch × 3 inch) after treated by dry KEP50 particle powder, and different STFs ($\phi = 0.5$ KEP50 in PEG 200, $\phi = 0.6$ KEP50 in ionic liquid [C ₄ mim][BF ₄], and $\phi = 0.5$ KEP10 in ionic liquid [C ₄ mim][BF ₄]) with different volume ratios of ethanol : treating materials (particles or STFs).
Figure 7.5:	Box charts of penetration peak forces and dissipated energies during spike penetration measurements for Kevlar [®] 1148 before and after treated by dry KEP50 particle powders with various volume ratios of ethanol : KEP50

Figure 7.6:	Box charts of penetration peak forces and dissipated energies during spike penetration measurements for Kevlar [®] 1148 before and after treated by STFs of $\phi = 0.5$ KEP50 in PEG 200 with various volume ratios of ethanol : STFs
Figure 7.7:	Box charts of penetration peak forces and dissipated energies during spike penetration measurements for Kevlar [®] 1148 before and after treated by STFs of $\phi = 0.6$ KEP50 in ionic liquid [C ₄ mim][BF ₄] with various volume ratios of ethanol : STFs
Figure 7.8:	Box charts of penetration peak forces and dissipated energies during spike penetration measurements for Kevlar [®] 1148 before and after treated by STFs of $\phi = 0.5$ KEP10 in ionic liquid [C ₄ mim][BF ₄] with various volume ratios of ethanol : STFs
Figure 7.9:	The force curves (force vs. displacement) of Kevlar [®] 1148 before and after treated with various STFs punctured by 18G needles with twelve repeats
Figure 7.10:	The force curves (force vs. displacement) of Kevlar [®] 1148 before and after treated with various STFs punctured by 21G needles with twelve repeats
Figure 7.11:	The averaged force curves (force vs. displacement) of Kevlar [®] 1148 before and after treated with various STFs punctured by needles of 18G (A) and 21G (B) with twelve repeats
Figure 7.12:	Box charts of penetration peak forces during 18G (A) and 21G (B) needle penetration tests for Kevlar [®] 1148 before and after treated by various ionic liquid based and PEG based STFs
Figure 7.13:	Box charts of dissipated energy during 18G (A) and 21G (B) needle penetration tests for Kevlar [®] 1148 before and after treated by various ionic liquid based and PEG based STFs
Figure 8.1:	The rheological response of ionic liquid $[C_4mim][BF_4]$ using temperature sweep measurements with shear rate 2 s ⁻¹ under atmospheric condition. Temperature starts from 10 °C and cools down to -80 °C at a cooling rate of 5 °C/min
Figure 8.2:	Steady state viscosity (A) and relative viscosity (B) as a function of shear stress at various temperature from 25 °C to -40 °C for $\phi = 0.4$ KE-P10 silica particles in [C ₄ mim][BF ₄] under atmospheric condition.253

Figure 8.3:	A) The pressure change inside the vacuum cell chamber as a function of time when the turbo-pump station starts to work at 25 °C. B) Comparison of the steady state viscosity of $\phi = 0.4$ KE-P10 silica particles in [C ₄ mim][BF ₄] at 25 °C under atmospheric and 10 ⁻⁶ hPa vacuum conditions
Figure A.1:	Density deviations $\Delta \rho$ of [C ₄ mim][BF ₄] aqueous solutions for varying water molar fraction x_w (A) and temperature (B). The symbols represent experimental value. The solid lines represent Redlich-Kister equation fitting
Figure A.2:	The viscosities of [C ₄ mim][BF ₄] aqueous solutions at different temperatures with Arrhenius equation fitting
Figure A.3:	Plots of excess log viscosity, $(log\eta)^{ex}$, and excess molar volume, V^{ex} , of [C ₂ mim][BF ₄] of salt-rich regime ² (water molar fraction up to 40%) (A), [BuPy][BF ₄] ³ (B), [C ₂ mim][OTf] ⁴ (C), [C ₂ mim][EtSO ₄] ⁴ (D), [C ₂ mim][TFA] ⁴ (E), EAN ⁵ (F) aqueous solutions over the entire range of water compositions. 268
Figure B.1:	Viscosity shear flow curves of silica particle Aerosil 200 in $[C_4mim][BF_4]$ with various particle contents and comparison with literature results by Ueno <i>et al.</i> ²
Figure B.2:	Steady state shear flow curves of 10 wt% Aerosil 200 in [C ₄ mim][BF ₄] suspensions mixed by various methods276
Figure B.3:	Steady state shear flow curves of 10 wt% Aerosil hydrophilic silica particle suspensions of various particle sizes
Figure B.4:	Steady state flow curves of silica particle suspensions with various surface chemical in $[C_4 mim][BF_4]$ with 10 wt% particle content 279
Figure B.5:	Schematic illustration of hydrophilic Aerosil particles stabilization in [C ₄ mim][BF ₄] by solvation layer

ABSTRACT

Astronauts are constantly threatened by impact from micrometeorite and orbital debris (MMOD) when conducting extra-vehicular activities (EVA) in low Earth orbit (LEO). These threats have already become a major challenge to long-term missions and deep space exploration. Shear thickening fluids (STFs) demonstrate an abrupt increase in viscosity with applied high shear stress, improving their ability to dissipate energy and making them good candidates for protective body armor. In my thesis work, a novel STF formulation in ionic liquid has been developed to improve the resistance of EVA suit against threats from ballistic, puncture, and hypervelocity MMOD impacts. Ionic liquids serve as the solvent phase for the STF formulations because of their low volatility and stability over a broad range of temperatures. However, dispersing colloidal particles in ionic liquids can be challenging because the high ionic strength of ionic liquids screens the electrostatic stabilizing forces that are typically important for stabilizing colloidal dispersions in polar solvents.

Stable nanoparticle dispersions in the ionic liquid $[C_4mim][BF_4]$ are created through surface coatings (*e.g.*, fluorinated alkyl chains, alcohol), which induce solvation layering around the particles. Solvation layers are initiated by hydrogen bonds between the anion groups $[BF_4]^-$ and the functionalized particle surface, providing a stabilizing steric repulsive inter-particle force. Rheology, electron microscopy, dynamic light scattering (DLS), and small-angle neutron scattering (SANS) are employed to determine the thickness of the solvation layers and the

xxiv

microstructure of dispersions for different coating systems. A quantitative model based on analysis of SANS data is developed to evaluate the inter-particle interactions and the thickness of the solvation layers. Additionally, the rheological behavior of dispersions is controlled by tuning the strength of surface hydrogen bonding with different surface chemistry. The influence of temperature on the thickness of solvation layers and particle interactions is also investigated through rheology, DLS, and SANS studies. Destabilization phenomena (from stable dispersion to unstable gel) are identified due to the change of interfacial structure with increasing temperature. Furthermore, the influence of impurities (i.e., water) on the microstructure and thermodynamic properties of ionic liquid are studies using SANS and small-angle x-ray scattering (SAXS) techniques. A phase diagram for ionic liquid aqueous solutions (microphase separation, phase inversion, and micelle formation) is constructed, revealing similarities to traditional oil-water-surfactant systems. This understanding of ionic liquid phase behavior and formation of solvation layers is critical for the formulation of colloidal dispersions in ionic liquids with a specific rheological profile. Ionic liquids based STF-Kevlar[®] nanocomposites are shown to provide superior puncture resistance in lab scale quasi-static puncture tests. The fabricated nanocomposites are proven to provide better protection than traditional Kevlar[®] without compromising the flexibility. The results of the present research demonstrate the feasibility of STF-Kevlar[®] nanocomposites for astronaut protection and identify technological challenges that still need to be addressed.

Chapter 1

INTRODUCTION

1.1 Motivation

Millions of micrometeoroids and orbital debris (MMOD) particles are traveling at hypervelocity speeds averaging 10 km/s in low-earth orbit.¹ MMOD particles in the near-Earth space environment are comprised of naturally occurring micrometeoroids originated from comets and asteroids, and man-made debris of fragmented rocket bodies and spacecraft parts created during the space exploration in the past 60 years. MMOD particles continually collide with spacecraft and satellites. Although spacecraft and satellites can avoid collisions with large and trackable MMOD particles (usually larger than 10 cm) by maneuvering around,² smaller MMOD particles (1 cm to 10 cm in diameter) are not easy to track; collision with these particles can cause severe damage. Furthermore, extra-vehicular activity (EVA) exposes astronauts to potential MMOD impacts ranging from secondary lunar ejecta at a speed of 0.1-1 km/s to orbital debris at a speed of 1-15 km/s to micrometeoroids at a speed of up to 70 km/s,¹ which becomes a life threatening risk to astronauts performing EVA in space. What's worse, the MMOD threat is still growing. With more satellites being launched, explosion of non-functional satellites, and high-speed collision between existing orbital debris, more and more debris is created. The threat of MMOD impact on active spacecraft and EVA suits has already become a major challenge to long-term missions and deep space exploration.

In addition to MMOD impact, space suits also are exposed to damages arising from other physical hazards such as tools, sharp edges on handrails or surface elements.³ It may be possible to add bulky or stiff individual layers to protect against each of the different types of threat for specific cases,⁴ this approach will detrimentally increase the weight and reduce the flexibility of the EVA suit, which is definitely undesirable. Thus, flexible, lightweight, and multi-threat protective materials are imperative to provide enhanced protection against multi-threats without compromising the weight or flexibility of the space suit.

Shear thickening fluid (STF)-nanocomposites (typically referred as STF-ArmorTM) are a promising candidate for use in advanced EVA suits to address these challenges because of their flexibility, lightweight and ability to protect against multiple mechanical threats. STF-ArmorTM is a protective nanocomposite where STF is intercalated into the protective textile. This material has been demonstrated to provide enhanced protection against simulated hypervelocity MMOD impacts, with higher energy absorption capability than prototype lay-ups of the standard thermal micrometeoroid garment (TMG).⁵ However, the current generation of STF-ArmorTM uses STF formulation of dispersed colloidal particles in polyethylene glycol as the carrier medium, which solidifies at low temperature and evaporates under vacuum. Therefore, it is crucial to develop new STF formulations to withstand vacuum and extreme low temperature space environment. Ionic liquids are emerging candidates for the solvent phase of STF formulations for space applications because of their thermal stability and low volatility.⁶ However, achieving stable ionic liquid-based STFs can be challenging because the high ionic strength of the ionic liquids screens the electrostatic stabilizing forces that are typically important for dispersing particles

in polar solvents. Hence, understanding the particle interactions and engineering the particle phase chemistry to achieve stable and desirable shear thickening fluids in ionic liquids are important.

1.2 Shear Thickening Colloidal Suspensions

Shear thickening is one of the most significant changes evident in stable colloidal dispersions, which is marked by an increasing viscosity with increasing shear stress or shear rate.⁷⁻⁹ As opposite to shear-induced aggregation, shear thickening herein is reversible and relevant to numerous industrial products and processes, such as paints, pharmaceuticals, lubricants, microabrasives, paper coating, and spraying which are often processed at high shear rate and shear tress. Shear thickening phenomena oftentimes need to be avoided as they can cause equipment damage and failure, as well as irreversible aggregation of dispersion particles. However, they are found in applications for damping and control devices,¹⁰⁻¹² electrorheological fluids,¹³ stab and puncture resistance flexible composites,¹⁴⁻¹⁵ and enhanced soft body armor.¹⁵

Three distinguished rheological behaviors: Newtonian, shear thinning, and shear thickening, exist for the concentrated near hard-sphere colloidal dispersions during shear flow, and their corresponding microstructures are shown in Figure 1.1. The non-Newtonian flow behavior is the direct consequence of the microstructural rearrangement of the near hard-sphere particles during shear flow. At low shear rate, the suspension is at equilibrium. Colloidal particles are driven by Brownian motion governed by the thermodynamic forces. The random collisions between particles give rise to the natural resistance to flow. Thus, an elastic stress from Brownian motion provides the driving force to move the microstructure back to equilibrium.¹⁶⁻¹⁸ The Brownian stress is linear with the shear rate, which results in a Newtonian, zero shear

viscosity in the linear regime. When the shear rate increases to moderate values, the microstructure exhibits anisotropic behavior as particles spatially rearrange to reduce their interaction so as to flow with less resistance. The dispersion exhibits shear thinning, as the stress contributions from the thermodynamic forces no longer increase linearly with increasing shear rate. At even higher shear rates, the suspension shear thickens as the viscosity increases reversibly, due to an increasing contribution from the hydrodynamic interactions between the particles in suspensions.

These non-Newtonian flow behaviors can be understood through the analogy to traffic with cars acting as particles.⁹ The equilibrium state is analogous to driving cars in a parking lot. The traffic flow is limited as individual cars are watching out for other cars from any direction. With increasing shear rates, particles behave like cars merging on the highway. The particles flow streamlined as cars follow each other and the increasing efficient transport of colloidal particles reduces the viscosity of dispersions. However, when the shear rate reaches the non-linear regime, the microstructure of particles is similar to traffic collisions or traffic jam, in which the well organized and low resistance flow is disrupted. The difficulty of cars (particles) flowing around each other increases the resistance to flow and the viscosity increases.



Figure 1.1: Illustration the microstructure of shear thickening fluids explaining the transition to shear thinning and shear thickening during shear flow. The change of microstructure corresponds to three distinguish regions: (1) equilibrium: the particles collide randomly due to thermodynamic forces; this Brownian motion of particles makes them naturally resistant to flow. (2) shear thinning: particle become organized along the shear flow direction at moderate shear rate or shear stress, which lowers their viscosity. (3) shear thickening: hydrodynamic interactions between particles dominate over thermodynamic forces, resulting in transient fluctuation of particles associating as "hydroclusters". An abrupt increase in viscosity is resulted from the difficulty of particles flowing around each other in the flow field. Adapted with permission from Wagner, N. J.; Brady, J. F. *Physics Today* 2009, 62 (10), 27-32. Copyright 2009 American Physical Society.

The mechanism of reversible shear thickening in stable colloidal dispersions have been studied through numerous pioneering simulations^{16, 19-22} and experiments combining rheology with scattering techniques, such as rheo-optics,¹⁸ and small-angle

neutron scattering (SANS).²³⁻²⁷ The competition between the conservative thermodynamic force and the viscous hydrodynamic force acting between the particles in suspension during shear flow is critically important to understand and predict the non-equilibrium shear rheology. The shear thickening response is the result of an increasing contribution from the hydrodynamic interactions between the particles in suspensions. At high shear rate/stress, colloidal suspensions are dominated by short-range lubrication interactions which lead to density fluctuations of particle and ultimately, stress-bearing clusters named as "hydroclusters".^{16, 24, 28} The transient density fluctuations are first predicted by Stokesian dynamic simulations,^{16, 19-22} and then observed experimentally by rheo-optical,¹⁸ neutron scattering techniques,²³⁻²⁷ and stress jump measurements.²⁹⁻³⁰ Note that these clusters caused by lubrication hydrodynamics are not permanent aggregates; they are transient density fluctuations, which form and break during shear.

The shear thickening rheological properties of the concentrated colloidal dispersions have been investigated by both experimental and theoretical works.^{16, 18-27} The relationship between particle properties and shear thickening has provided insight into the prediction of dispersion properties. Barnes summarized many parameters that would influence shear thickening, including particle size, particle concentration, polydispersity, shape, particle interaction, and the properties of the solvent.⁷ The shear thickening behavior strongly depends on the particle size, with the critical stress for shear thickening scaled as the inverse of the particle radius squared.²⁸ The non-Newtonian flow behavior can be observed at volume fraction above 0.2 for monodispersed spherical colloidal dispersions, while discontinuous shear thickening starts from 0.5 volume fraction. It is found that the required particle concentrations

for these transitions increase for polydisperse systems,³¹ while anisotropy in particle shape decreases the required concentration or increases the strength of shear thickening at a given concentration.³²⁻³³ As the separation between hydroclustered particles is of the order of a few nanometers, shear thickening behavior directly reflects the short-range particle interaction.^{22, 34} It has been found that when the range of the repulsive force between particles approaches 10% of the particle radius, shear thickening is effectively eliminated. Thus the shear thickening can be controlled by tailoring particle properties and particle interactions.

1.3 Ionic Liquids

Room temperature ionic liquids (ILs) are defined as salts that are liquids at or near room temperature. Similar to traditional salts, such as sodium chloride (NaCl), ionic liquids are composed only of ions: organic cations and inorganic or organic anions.^{6, 35} ILs are pure molten salts at room temperature without the presence of any molecular solvents; this makes them fundamentally different from ionic salt solutions, which are the aqueous solutions of salts dissolved in water. In traditional salts, small and symmetrical cations and anions are packed together tightly, resulting in extremely high melting temperature (800 to 1000 °C). In contrast, the organic cations and anions of ILs are typically big and unsymmetrical, which cannot get close enough to crystallize, as shown in Figure 1.2. This inability to form a solid is known as frustrated crystallization, and as a result, many ionic liquids are fluids at room temperature.³⁶



Room Temperature Ionic Liquids: Molten Salt at or near to Room Temperature



Figure 1.2: Schematic illustration of structural difference between traditional salts and ionic liquids. Traditional salts are crystal solid with small and symmetrical ions at room temperature. They can form ionic salt solution when dissolved in water or be melted to liquid form when heated up to around 1000 °C. Room temperature ionic liquids are molten salts at or near room temperature. They are composed of only ions, which are typically large and unsymmetrical, resulting in non-crystallized structure. Ionic liquids exhibit unique physicochemical properties, such as negligible vapor pressure, non-flammability, high thermal and chemical stability, a broad electrochemical window, a wide range of liquid state, and good electrical conductivity.^{6, 37} The most fundamental characteristic of ionic liquids is their low volatility. Molecular solvents, such as water and alcohol, are composed of neutral molecules with van der Waals attraction force. When the molecules at the surface obtain enough energy to overcome the van der Waals attraction through heating, the molecules enter the gas phase and evaporate into the atmosphere. Unlike the molecular solvents, the interactions between ions in ionic liquids are through ionic bonds, which are several orders of magnitude stronger than the van der Waals force. It is unlikely for the ions to overcome such stronger attraction to evaporate into the atmosphere under normal condition when heated.³⁶ Thus, the vapor pressure of ionic liquids is negligible in comparison to molecular solvents. This low vapor pressure induces the non-flammability of ionic liquids; thus it is much safer to use ionic liquids in industrial processes when large quantities of solvents are needed.

In addition to the low volatility and non-flammability compared with traditional volatile organic compounds (VOCs), another important advantage of ionic liquids is their great capability of dissolving numerous substances due to their high and tunable polarity. Cellulose, polymers and metal salts, which are otherwise only soluble in a few solvents, can be dissolved in ionic liquids. Ionic liquids are the only solvents that are able to dissolve coal.³⁶ Their capability to dissolve substances that are traditionally considered insoluble enables the access to new chemistry and novel chemical reactions, which were impossible before due to the insolubility of reactants into the same liquid phase. More importantly, ionic liquids do not dissolve glass or

stainless steel.³⁶ Thus, they are stored in glass bottles and stainless steel rheometer tools are used when measuring their rheological properties in this dissertation.

Another key advantage of ionic liquids is that their properties can be tailored as desired. Ionic liquids offer great flexibility in the design of both the cationic and anionic structures and their combinations, thus they have been termed as "designer solvents".³⁶ Through manipulation of their cations, anions, or both, ionic liquids can be designed to possess specific properties. In contrast, the application range of conventional molecular solvents is limited because their properties cannot be easily tuned. Additionally, there are only about 600 conventional molecular solvents in use, while there are at least a million possible simple ionic liquids through the different combination of various cations and anions.³⁵⁻³⁶ The most commonly used cations and anions are shown in Figure 1.3. The wide variety of available cations and anions provide a high flexibility in material design. Furthermore, at least a billion binary ionic liquids and 10¹⁸ ternary ionic liquids are possible when considering the mixtures of ionic liquids.³⁶


Figure 1.3: List of commonly used cations and anions in ionic liquid systems.³⁸ The abbreviation $[C_n mpyr]^+$ represents the 1-alkyl-1-methylpyrrolidinium cation, where the index *n* represents the number of carbon atoms in the linear alkyl chain. $[P_{wxyz}]^+$, $[N_{wxyz}]^+$, and $[S_{xyz}]^+$ represent tetraalkylphosphonium, tetraalkylammonium and trialkylsulfonium cations, respectively, where the indices *w*, *x*, *y* and *z* indicate the length of the corresponding linear alkyl chains. Adapted with permission from Plechkova, N. V.; Seddon, K. R. Chem. Soc. Rev. 2008, 37 (1), 123-150. Copyright 2008 The Royal Society of Chemistry.

This incredible diversity enables the solvent to be designed and tuned to obtain optimal substrate solubility, selectivity, yield, product separation, and even enantioselectivity.^{6, 39} Ionic liquids can serve as media for a wide range of organic and inorganic reactions,⁶ can dissolve enzymes,⁴⁰ can form versatile biphasic systems for

separations.⁴¹ Ionic liquids can be highly conductive, and form both polymers and gels for device applications.⁴²⁻⁴³ Ionic liquids have received considerable interest in various fields including separations,⁶ catalysis,^{6, 37, 44-45} electrolytes,^{43, 46} heat transfer,⁴⁷⁻⁴⁸ and drug delivery.⁴⁹ Their unique physicochemical properties suggest they can be used as an environmentally friendly "green" alternative for conventional organic solvents,^{6, 37, 44} as lubricants for space and high vacuum environments,⁵⁰⁻⁵² as separation media for batteries, fuel cells,⁴⁵ dye sensitized solar cells,⁴³ and more recently, for carbon capture from flue gases in industrial scale.⁵³

In many of these applications, the ionic liquids are exposed to water and nonideal thermodynamic properties in mixture properties are evident.⁵⁴ Thus, the physicochemical properties of ionic liquids and their mixture with organic/inorganic solvents are interesting both from practical and theoretical point of view and are of importance for their applications.^{38, 55-56} Non-ideal thermodynamic properties in mixture properties such as density and viscosity have been reported for the ionic liquids [C_n mim][BF₄] with various molecular solvents,^{9, 54, 57-65} such as water, and alcohols, through systematic studies. Rodriguez *et al.* studied the temperature and composition dependence of the density and viscosity of aqueous solutions of a series of 1-ethyl-3-methylimidazoloium with varying anions. A trend between excess volume and viscosity is suggested, which can be used for prediction of physical property.⁶⁶

Water is a common contaminant in ionic liquids,⁶⁷ especially for hydrophilic ionic liquids, which have the tendency to mix with, or dissolve in water, such as $[C_4mim][BF_4]$. The thermodynamic properties of ionic liquids can be significantly altered by addition of trace amount of water.⁶⁸⁻⁶⁹ Therefore, knowledge of excess

12

thermodynamic properties of ionic liquids aqueous solutions provides insight into the molecular interactions in these mixtures and aids in developing structure-property relationships. With these knowledge, the properties of ionic liquids can be optimized for specific applications.^{38, 70-71}

For example, the density and viscosity of binary mixtures of $[C_4mim][BF_4]$ with water have been investigated and found to be non-ideal over the entire composition range.^{54, 57-58} These mixture non-idealities are attributed to the hydrogen bonds between water molecules and the $[BF_4]^-$ anions, which affects the microstructure of the ionic liquid.⁶⁸ The molecular structure and interactions in mixtures of ionic liquids and water have been studied by experimental techniques,⁷²⁻⁷⁵ and complemented by molecular simulations.⁷⁶⁻⁷⁷

Because of the amphiphilic structure of ionic liquids, similar to surfactant, they can self-assemble into micelle-like structures when diluted in solvents. The aggregation of ionic liquids is evident and characterized using various techniques, including SANS,^{75, 78} NMR,^{63, 69} surface tension and conductivity measurements.⁷⁵ The conformational changes in a variety of ionic liquids induced by the self-aggregation of ionic liquid are shown to depend on the aromatic ring, alkyl chain, counter ions, and their interactions with water.⁷³

In addition, the opposite end of the composition spectrum, rich in ionic liquids, is equally interesting, however less well understood.^{76, 79-80} The interaction between the anion and water is found to generally dominate for salt-rich region of $[C_4mim][BF_4]$ -water mixtures. Individual water molecules form hydrogen bonds with two $[BF_4]^-$ anions without substantially disrupting the nanoscale structuring of the ionic liquid.^{72, 75, 80-81} However, water dimers and even larger clusters start to form

with increasing water content. Furthermore, the strong electrostatic interactions between the cations and anions is gradually disrupted with increasing water addition, resulting in the disturbed nanostructure of the ionic liquid.⁷⁶ Further addition of water leads to a micro-phase transition from ionic liquid polar networks to water clusters in the ionic liquid, then to ionic liquid self-aggregates in water, and finally to dilute ionic liquid electrolytes in water. These phase transitions and properties are revealed by molecular simulation⁷⁶ and experiments,^{75, 79-81} including FTIR,⁷⁹ Raman and IR spectroscopy⁸⁰ *et al.* It is confirmed that water aggregation in ionic liquids is mainly determined by association of the water with ionic groups in the ionic liquids and less by the hydrophobicity of the cations.

1.4 Ionic Liquids Colloidal Suspensions

Colloidal suspensions in ionic liquids have attracted increasing attention in recent decades as many applications of ionic liquids require successfully dispersing nanoparticles across a broad range of conditions. There are numerous research studies of colloidal systems in ionic liquids including nanoparticle synthesis,⁸²⁻⁸³ phase transfer,⁸⁴ catalytic reactions medium as a "green" solvent,^{6, 37, 44, 85} and the self-assembly of surfactants.⁸⁶⁻⁸⁹ The stability of nanoparticle and the associated suspension microstructure is critical in determining the suspension rheology and controlling aggregation, gel and glass formation. However, particle stabilization in ILs is challenging because their ionic nature screens electrostatic repulsive forces.

Particle dispersion and stability in ILs depends on the specific chemical composition of the ionic liquids as well as the surface chemistry of the particles.⁹⁰⁻⁹⁴ Hydrophilic fumed silica particles are reported to be well dispersed in $[C_4mim][BF_4]$ displaying shear thickening.⁹¹ However, unstable dispersion with strong shear

14

thinning and yielding behavior is discovered for the same hydrophilic fumed silica particles in 1-ethyl-3-methylimidazolium bis(trifluoromethylsulfonyl)imide $([C_2mim][NTf_2])$.⁹¹ In addition, aggregation and gelation are reported for hydrophobic fumed silica particles in $[C_4mim][BF_4]$.⁹¹ These dispersion phenomena demonstrate that it is non-trivial to disperse particles into ionic liquids.

The interaction between particles in dispersions is typically evaluated by Derjaguin-Landau-Verwey-Overbeek (DLVO) theory, which accounts for van der Waals attraction and electrostatic repulsion between particles in solution.⁹⁵ Due to the high ionic strength in ionic liquids, the electrostatic repulsion between silica particles is effectively screened. Thus, the electrostatic stabilization is not sufficient to stabilize particles in ionic liquids against the van der Waals attraction between particles.⁹⁶ Figure 1.4 gives an example of screened electrostatic repulsion between two silica particles of ~100 nm size in ionic liquid [C₄mim][BF₄]. Compared with silica particles in salt aqueous solution of low salt concentration, silica particles in [C₄mim][BF₄] have similar van der Waals attraction, but significantly screened electrostatic repulsion. The screening of the electrical potential and electrical double layers at charged solid surfaces in ionic liquids and molten salts⁹⁷ has been investigated using experiments⁹⁸ and simulations.⁹⁹⁻¹⁰⁰



Figure 1.4: Calculated van der Waals attractive (solid line) and electrostatic repulsive (dash line) potential profiles between two silica particles at T = 25 °C for silica particles in [C₄mim][BF₄] (magenta) and in 0.1 mM NaCl aqueous solution (blue).

Despite the screening of the electrical repulsion, hydrophilic fumed silica particles are reported to be dispersed into an aprotic ionic liquid, $[C_4mim][BF_4]$ in the absence of any stabilizer.^{91, 101-102} Altin points to the unique properties of ILs with respect to their nanoscale structure, such that particles can be stabilized in solution by structured solvent surrounding the particles.⁹⁰ This couples to the nanostructure in the ionic liquid to create alternating layers of anions and cations that are known as solvation layers. Such ionic ordering of the solvent propagates some distance into the ionic liquid and provides repulsive force acting between two particle surfaces dispersed in the ionic liquid. The oscillating distant-dependent solvation forces have

been observed between macroscopic mica and silica surfaces in both protic and aprotic ionic liquids by surface force measurements as shown in Figure 1.5,¹⁰³⁻¹⁰⁸ which are consistent with molecular dynamics simulation.¹⁰⁹ Discrete friction across ionic liquid thin films between two mica sheets have been discovered to be a function of surface separation distance or integer number of ion layers in the film. These measurements and simulations indicate that the ion pairs are arranged in a layered structure near the solid surfaces.¹⁰⁷



Figure 1.5: (A) Schematic illustration of the structured solvent around fluorinated coated silica nanoparticle in ionic liquid [C₄mim][BF₄]. This structured solvent is initiated by the hydrogen bonding between coated fluorocarbon chains and [BF4]⁻ anion groups. Cations form cations layers around anion through ionic bonds and this ordered layer structure gradually becomes disordered when away from the surface of particles. (B) Solvation force between two macroscopic silica surfaces in [C₄mim][BF₄]: Normal force scaled with surface curvature radius (*F/R*) as a function of surface separation *D*. Dotted lines represent van der Waals attractions calculated from Lifshitz theory. Solid and dashed lines in plots indicate stable and unstable regions, respectively. Part (A) is adapted with permission from Gao J., et al. *ACS Nano* 2015, 9, (3), 3243-3253. Copyright 2015 American Chemical Society; Part (B) is adapted with permission from Ueno, K., et al. Phys. Chem. Chem. Phys. 2010, 12, 4066-4071. Copyright 2010 The Royal Society of Chemistry.

Besides the solvation layering, the suspension stability in ionic liquids can be achieved by polymer grafting to the particle surface when IL is a good solvent for the polymer.^{96, 110} Poly(methyl methacrylate) (PMMA) is soluble in 1-butyl-3methylimidazolium hexafluorophosphate ([C₄mim][PF₆]) and 1-alkyl-3methylimidazolium bis(trifluoromethylsulfonyl)imide ($[C_n mim][NTf_2]$), the PMMAgrafted silica nanoparticles were reported to show good dispersion in these two ionic liquids. However, such coated particles show poor dispersion if PMMA is insoluble in other ionic liquids (e.g., [C₄mim][BF₄]).⁹⁶ Such particle stability is mainly a result of the compatibility of solvent with coated polymer chains, which can be controlled by tuning the solvent quality through pH,⁷¹ temperature,^{62, 110} etc. The mechanism of such particle stability is afforded by the steric repulsion between the grafted layers and is system dependent, *i.e.*, the system will become unstable when the brush is immersed into a poor solvent and collapses.¹¹¹ The stability provided by surface grafted polymers through steric repulsion is reported to be thermal-sensitive and a temperature-induced colloidal glass-to-gel transition was observed.^{62, 110} However, such polymer-stabilized dispersions do not exhibit shear thickening rheology in the regimes studied presumably due to the strong nanoscale repulsive forces afforded by the well-solvated polymer brush.¹¹²⁻¹¹³

1.5 Shear Thickening Fluid (STF) – ArmorTM

The ability of STFs to behave as a liquid or solid under different shear rate or shear stress can be beneficial for applications where rate-sensitive behavior or energy adsorption is critical. STFs have been used as an additive material and impregnated into protective fabrics, such as Kevlar[®], to improve the performance of regular body armor made by Kevlar[®] fabric.¹⁵ The scanning electron microscopy (SEM) images of

the treated STF–Kevlar[®] material are shown in Figure 1.6.¹⁴ It can be found that the STF fully wets the fabric, penetrating between fibers within a yarn. As the size of the silica particles are much smaller than the diameter of the Kevlar[®] fiber, the STF can be treated as a continuum relative to the fabric.



Figure 1.6: SEM images of woven Kevlar[®] impregnated with STF between fibers. Adapted with permission from Decker, M., *et al.* Composites Science and Technology 2007, 67 (3), 565-578. Copyright 2007 Elsevier.

This enhanced protective nanocomposite material after intercalated with STF are often referred as STF-ArmorTM, which can be used for ballistic, puncture, and blast protection for the military, police, and first responders. The STF-ArmorTM is as

flexible as the bare Kevlar[®] fabric because STF has liquid-like properties at the shear rates associated with normal movements. When impacted by a bullet or fragment traveling at high speed, the STF rigidizes almost instantaneously to aid in stopping the projectiles. STF enhances the performance of Kevlar[®] by restricting the relative movement of adjacent fibers when impacted, thereby taking full advantage of the high tensile strength of Kevlar[®].¹¹⁴⁻¹¹⁷ Additionally, it has been shown that the STF can distribute stress away from the impact site to neighboring fibers, engaging a larger portion of the fabric in energy absorption. STF-nanocomposites have been proven to provide enhanced protection against puncture and ballistic threats which would otherwise easily penetrate the bare Kevlar[®] with the same or reduced mass and more flexibility than conventional materials.^{14, 117} Figure 1.7 demonstrates one example of the superior puncture resistance of STF-ArmorTM nanocomposites as compared to conventional Kevlar[®].¹⁴ STF-Kevlar[®] exhibits significantly better stab resistance with less layer penetration and higher energy dissipation as compared with the neat Kevlar® of equal areal density. Furthermore, neat Kevlar[®] shows significant puncture damage, while there is little obvious damage to the STF-Kevlar[®].

Besides the superior protection against puncture and ballistic threats, STF-ArmorTM have been shown to provide enhanced protection against simulated hypervelocity MMOD impacts, which occurs with extreme high energy and short time scale.⁵ With the neoprene-coated nylon absorber layers of the standard thermal micrometeoroid garment (TMG) are substituted with STF-ArmorTM, the prototype layups provide meaningful hypervelocity impact protection from MMOD threats with higher energy absorption.



Figure 1.7: (a) Results of National Institute of Justice (NIJ) 0115.0 spike stab resistance test for body armor on 15 layers of neat Kevlar[®] (270 g/) and 12 layers of STF-Kevlar[®] (271 g/). The plot shows the number of witness papers placed behind the target that are penetrated. Penetration of all 5 layers indicates failure. (b) photographs of Kevlar[®] and STF-Kevlar[®] fabric damage after NIJ 0115.0 spike puncture test at m = 2.33 kg and h = 0.75 m. Adapted with permission from Decker, M., *et al.* Composites Science and Technology 2007, 67 (3), 565-578. Copyright 2007 Elsevier.

1.6 Thesis Objectives and Overview

This dissertation has three goals. The primary goal of my dissertation is to develop shear thickening fluid formulations that are stable under simulated space conditions and to intercalate them with aramid fabrics Kevlar[®] to formulate STF nanocomposites that are appropriate for space applications. Materials made for the environment of low earth orbit must be able to withstand vacuum and extreme temperatures.¹¹⁸ An STF formulation for space applications requires a carrier medium that will remain flowable under vacuum and extreme low temperatures. However, the shear thickening fluid formulations of the current generation of multi-threat protective composites use polyethylene glycol as the carrier medium, which will solidify at low temperature and evaporate under vacuum. Thus, ionic liquids are chosen due to their low volatility and thermal stability especially under low temperature, which will meet the formulation requirements.¹¹⁹

Unfortunately, to date, there have been only a few literature reports of shear thickening colloidal dispersions formulated in ionic liquids.^{96, 110} Colloidal particles are reported to aggregate in ionic liquids due to the screened electrostatic repulsion under high ionic strength. Consequently, research into the mechanisms by which particles can be dispersed into ionic liquids such that shear thickening rheology can be achieved is warranted. To address this in part, the second goal of this dissertation is to study the nanostructure, stability, and dynamics of model, surface-functionalized spherical silica particles in a common aprotic ionic liquid $[C_4mim][BF_4]$. Direct measurements of surface structure around nanoparticles and a predictive, quantitative knowledge of inter-particle interactions in ionic liquids will be crucial to predict and control colloidal stability in ionic liquids. With these knowledge, the desired shear

22

thickening response and transition rate and strength will be achieved by controlling the structure on the particle surface through engineering particle phase chemistry.

In order to successfully complete the first and second goals, the rheology of ionic liquid colloidal dispersions needs to be accurately measured. However, water is a common contaminant in ionic liquids due to their hygroscopic nature,⁶⁷ especially for hydrophilic ionic liquids. Furthermore, ionic liquids can easily absorb water from atmospherically humidity during the rheological measurement in lab, affecting the measured viscosity and microstructure. Thus, it is crucial to understand the interaction between water molecules and ionic liquids. The last goal of this dissertation is to study this water-ionic liquid interaction and any changes in microstructure cover the whole composition range.

The structure of this dissertation is organized as follows. Chapter 2 outlines the relevant experimental techniques used in this dissertation. Chapter 3 presents a study on the effect of addition of water molecules on the physical properties (density and viscosity) and microstructure of an ionic liquid $[C_4mim][BF_4]$. The phase transition of ionic liquid $[C_4mim][BF_4]$ -water mixtures covering the full composition range from high salt to low salt is investigated. Densities, viscosities and their semiempirical correlation of the homogenous binary mixtures of $[C_4mim][BF_4]$ with water are reported for a wide temperature range. This study provides a guidance on the rheological measurements for the rest of this dissertation. Chapter 4 describes the materials and the protocol used to synthesize fluorocarbon coated silica particles, which will be used and further discussed in Chapter 5. The particle properties, such as the chemical coating, particle size, and density, are evaluated. Chapter 5 demonstrates the critical role of solvation forces in dispersing and stabilizing nanoparticles and

23

colloids in $[C_4mim][BF_4]$. A stable dispersion of solid silica nanoparticles in ionic liquid $[C_4mim][BF_4]$ is achieved by particle surface chemical functionalization with a fluorinated alcohol. In Chapter 6, a successful attempt to formulate stable nanoparticle dispersions in ionic liquids that exhibit significant shear thickening through designing the surface coating of particles (alcohol coating) is described. Their rheological response to temperature is investigated and interpreted in terms of particle interactions and interfacial structure. With the successful development of a shear thickening fluid formulation with strong shear thickening strength, the new ionic liquid based formulation is intercalated with Kevlar[®] and the penetration resistance of the ionic liquids based STFs- Kevlar[®] nanocomposites are described in Chapter 7. The effect of changing from the traditional STF carrier medium, PEG, to ionic liquid $[C_4mim][BF_4]$ has been studied. Optimal areal density, penetration peak force, and dissipated energy are obtained through design of the treatment condition. Chapter 8 summarizes the major conclusions of this dissertation and identifies directions for future work.

REFERENCES

1. Christiansen, E. L.; Arnold, J.; Davis, A.; Hyde, J.; Lear, D.; Liou, J.; Lyons, F.; Prior, T.; Ratliff, M.; Ryan, S. Handbook for designing MMOD protection. *NASA Johnson Space Center, NASA/TM-2009-214785* **2009**.

2. Rodriguez, K. Micrometeoroids and Orbital Debris (MMOD). https://www.nasa.gov/centers/wstf/laboratories/hypervelocity/mmod.html (accessed 12/5).

3. Bergin, C. EVA-3 terminated due to Mastracchio glove damage. https://www.nasaspaceflight.com/2007/08/eva-3-terminated-due-to-mastracchio-glove-damage/ (accessed 12/5).

4. Bergin, C. Another EMU glove modification to debut on STS-126. https://www.nasaspaceflight.com/2008/10/new-emu-glove-modification-debut-sts-126/ (accessed 12/5).

5. Cwalina, C. D.; Dombrowski, R. D.; McCutcheon, C. J.; Christiansen, E. L.; Wagner, N. J. MMOD Puncture Resistance of EVA Suits with Shear Thickening Fluid (STF) – Armortm Absorber Layers. *Procedia Engineering* **2015**, *103*, 97-104.

6. Welton, T. Room-temperature ionic liquids. Solvents for synthesis and catalysis. *Chem. Rev.* **1999**, *99* (8), 2071-2084.

7. Barnes, H. Shear - thickening ("Dilatancy") in suspensions of nonaggregating solid particles dispersed in Newtonian liquids. *Journal of Rheology* (1978-present) **1989**, *33* (2), 329-366.

8. Lee, Y. S.; Wagner, N. J. Dynamic properties of shear thickening colloidal suspensions. *Rheol. Acta* **2003**, *42* (3), 199-208.

9. Wagner, N. J.; Brady, J. F. Shear thickening in colloidal dispersions. *Physics Today* **2009**, *62* (10), 27-32.

10. Helber, R.; Doncker, F.; Bung, R. Vibration attenuation by passive stiffness switching mounts. *Journal of sound and vibration* **1990**, *138* (1), 47-57.

11. Laun, H.; Bung, R.; Schmidt, F. Rheology of extremely shear thickening polymer dispersionsa)(passively viscosity switching fluids). *Journal of Rheology* (1978-present) **1991**, *35* (6), 999-1034.

12. Fischer, C.; Plummer, C. J.; Michaud, V.; Bourban, P.-E.; Månson, J.-A. E. Pre-and post-transition behavior of shear-thickening fluids in oscillating shear. *Rheol. Acta* **2007**, *46* (8), 1099-1108.

13. Shenoy, S. S.; Wagner, N. J.; Bender, J. W. E-FiRST: Electric field responsive shear thickening fluids. *Rheol. Acta* **2003**, *42* (4), 287-294.

14. Decker, M.; Halbach, C.; Nam, C.; Wagner, N.; Wetzel, E. Stab resistance of shear thickening fluid (STF)-treated fabrics. *Composites Science and Technology* **2007**, *67* (3), 565-578.

15. Lee, Y. S.; Wetzel, E. D.; Wagner, N. J. The ballistic impact characteristics of Kevlar® woven fabrics impregnated with a colloidal shear thickening fluid. *J. Mater. Sci.* **2003**, *38* (13), 2825-2833.

16. Bossis, G.; Brady, J. F. The rheology of Brownian suspensions. *The Journal of Chemical Physics* **1989**, *91* (3), 1866-1874.

17. Brady, J. F. The rheological behavior of concentrated colloidal dispersions. *The Journal of Chemical Physics* **1993**, *99* (1), 567-581.

18. Bender, J. W.; Wagner, N. J. Optical measurement of the contributions of colloidal forces to the rheology of concentrated suspensions. *J. Colloid Interface Sci.* **1995**, *172* (1), 171-184.

19. Phung, T. N.; Brady, J. F.; Bossis, G. Stokesian dynamics simulation of Brownian suspensions. *J. Fluid Mech.* **1996**, *313*, 181-207.

20. Foss, D. R.; Brady, J. F. Structure, diffusion and rheology of Brownian suspensions by Stokesian dynamics simulation. *J. Fluid Mech.* **2000**, *407*, 167-200.

21. Sierou, A.; Brady, J. F. Accelerated Stokesian dynamics simulations. *J. Fluid Mech.* **2001**, *448*, 115-146.

22. Melrose, J. R.; Ball, R. C. Continuous shear thickening transitions in model concentrated colloids—The role of interparticle forces. *Journal of Rheology (1978-present)* **2004**, *48* (5), 937-960.

23. Newstein, M. C.; Wang, H.; Balsara, N. P.; Lefebvre, A. A.; Shnidman, Y.; Watanabe, H.; Osaki, K.; Shikata, T.; Niwa, H.; Morishima, Y. Microstructural

changes in a colloidal liquid in the shear thinning and shear thickening regimes. *The Journal of Chemical Physics* **1999**, *111* (10), 4827-4838.

24. Maranzano, B. J.; Wagner, N. J. Flow-small angle neutron scattering measurements of colloidal dispersion microstructure evolution through the shear thickening transition. *The Journal of Chemical Physics* **2002**, *117* (22), 10291-10302.

25. Lee, Y. S.; Wagner, N. J. Rheological Properties and Small-Angle Neutron Scattering of a Shear Thickening, Nanoparticle Dispersion at High Shear Rates. *Ind. Eng. Chem. Res.* **2006**, *45* (21), 7015-7024.

26. Kalman, D. P.; Wagner, N. J. Microstructure of shear-thickening concentrated suspensions determined by flow-USANS. *Rheol. Acta* **2009**, *48* (8), 897-908.

27. Gurnon, A. K.; Wagner, N. J. Microstructure and rheology relationships for shear thickening colloidal dispersions. *J. Fluid Mech.* **2015**, *769*, 242-276.

28. Maranzano, B. J.; Wagner, N. J. The effects of particle size on reversible shear thickening of concentrated colloidal dispersions. *The Journal of Chemical Physics* **2001**, *114* (23), 10514-10527.

29. Kaffashi, B.; O'Brien, V. T.; Mackay, M. E.; Underwood, S. M. Elastic-like and viscous-like components of the shear viscosity for nearly hard sphere, Brownian suspensions. *J. Colloid Interface Sci.* **1997**, *187* (1), 22-28.

30. O'Brie, V. T.; Mackay, M. E. Stress components and shear thickening of concentrated hard sphere suspensions. *Langmuir* **2000**, *16* (21), 7931-7938.

31. Bender, J.; Wagner, N. J. Reversible shear thickening in monodisperse and bidisperse colloidal dispersions. *Journal of Rheology (1978-present)* **1996,** *40* (5), 899-916.

32. Egres, R. G.; Wagner, N. J. The rheology and microstructure of acicular precipitated calcium carbonate colloidal suspensions through the shear thickening transition. *Journal of Rheology (1978-present)* **2005,** *49* (3), 719-746.

33. Egres, R. G.; Nettesheim, F.; Wagner, N. J. Rheo-SANS investigation of acicular-precipitated calcium carbonate colloidal suspensions through the shear thickening transition. *Journal of Rheology (1978-present)* **2006**, *50* (5), 685-709.

34. Maranzano, B. J.; Wagner, N. J. The effects of interparticle interactions and particle size on reversible shear thickening: Hard-sphere colloidal dispersions. *Journal of Rheology (1978-present)* **2001,** *45* (5), 1205-1222.

35. Rogers, R. D.; Seddon, K. R. Ionic Liquids--Solvents of the Future? *Science* **2003**, *302* (5646), 792-793.

36. Seddon, K.; Holbrey, J. Ionic liquids. *Clean Products and Processes* **1999**, *1*, 223-236.

37. Sheldon, R. Catalytic reactions in ionic liquids. *Chem. Commun.* **2001**, (23), 2399-2407.

38. Plechkova, N. V.; Seddon, K. R. Applications of ionic liquids in the chemical industry. *Chem. Soc. Rev.* **2008**, *37* (1), 123-150.

39. Hallett, J. P.; Welton, T. Room-Temperature Ionic Liquids: Solvents for Synthesis and Catalysis. 2. *Chem. Rev.* **2011**, *111* (5), 3508-3576.

40. Sheldon, R. A.; Lau, R. M.; Sorgedrager, M. J.; van Rantwijk, F.; Seddon, K. R. Biocatalysis in ionic liquids. *Green Chem.* **2002**, *4* (2), 147-151.

41. Gutowski, K. E.; Broker, G. A.; Willauer, H. D.; Huddleston, J. G.; Swatloski, R. P.; Holbrey, J. D.; Rogers, R. D. Controlling the aqueous miscibility of ionic liquids: Aqueous biphasic systems of water-miscible ionic liquids and water-structuring salts for recycle, metathesis, and separations. *J. Am. Chem. Soc.* **2003**, *125* (22), 6632-6633.

42. Ohno, H. Ionic Liquids: The front and future of material developments. *CMC*, *Tokyo* **2003**.

43. Lee, S.-Y.; Ogawa, A.; Kanno, M.; Nakamoto, H.; Yasuda, T.; Watanabe, M. Nonhumidified intermediate temperature fuel cells using protic ionic liquids. *J. Am. Chem. Soc.* **2010**, *132* (28), 9764-9773.

44. Dupont, J.; de Souza, R. F.; Suarez, P. A. Ionic liquid (molten salt) phase organometallic catalysis. *Chem. Rev.* **2002**, *102* (10), 3667-3692.

45. Wasserscheid, P.; Keim, W. Ionic liquids - New "solutions" for transition metal catalysis. *Angew. Chem. Int. Ed.* **2000**, *39* (21), 3772-3789.

46. Zakeeruddin, S. M.; Graetzel, M. Solvent - Free Ionic Liquid Electrolytes for Mesoscopic Dye - Sensitized Solar Cells. *Adv. Funct. Mater.* **2009**, *19* (14), 2187-2202.

47. Crosthwaite, J. M.; Muldoon, M. J.; Dixon, J. K.; Anderson, J. L.; Brennecke, J. F. Phase transition and decomposition temperatures, heat capacities and viscosities of pyridinium ionic liquids. *J. Chem. Thermodyn.* **2005**, *37* (6), 559-568.

48. Valkenburg, M. E. V.; Vaughn, R. L.; Williams, M.; Wilkes, J. S. Thermochemistry of ionic liquid heat-transfer fluids. *Thermochim. Acta* **2005**, *425* (1), 181-188.

49. Hough, W. L.; Smiglak, M.; Rodriguez, H.; Swatloski, R. P.; Spear, S. K.; Daly, D. T.; Pernak, J.; Grisel, J. E.; Carliss, R. D.; Soutullo, M. D.; Davis, J. H.; Rogers, R. D. The third evolution of ionic liquids: active pharmaceutical ingredients. *New J. Chem.* **2007**, *31* (8), 1429-1436.

50. Street, K. W.; Morales, W.; Koch, V. R.; Valco, D. J.; Richard, R. M.; Hanks, N. Evaluation of Vapor Pressure and Ultra-High Vacuum Tribological Properties of Ionic Liquids. *Tribol. Trans.* **2011**, *54* (6), 911-919.

51. Morales, W.; Street, K. W.; Richard, R. M.; Valco, D. J. Tribological Testing and Thermal Analysis of an Alkyl Sulfate Series of Ionic Liquids for Use as Aerospace Lubricants. *Tribol. Trans.* **2012**, *55* (6), 815-821.

52. Liu, W. M.; Ye, C. F.; Gong, Q. Y.; Wang, H. Z.; Wang, P. Tribological performance of room-temperature ionic liquids as lubricant. *Tribology Letters* **2002**, *13* (2), 81-85.

53. Seo, S.; Quiroz-Guzman, M.; DeSilva, M. A.; Lee, T. B.; Huang, Y.; Goodrich, B. F.; Schneider, W. F.; Brennecke, J. F. Chemically Tunable Ionic Liquids with Aprotic Heterocyclic Anion (AHA) for CO2 Capture. *J. Phys. Chem. B* **2014**, *118* (21), 5740-5751.

54. Zhou, Q.; Wang, L. S.; Chen, H. P. Densities and Viscosities of 1-butyl-3methylimidazolium tetrafluoroborate + H2O binary mixtures from (303.15 to 353.15) K. J. Chem. Eng. Data **2006**, *51* (3), 905-908.

55. Keskin, S.; Kayrak-Talay, D.; Akman, U.; Hortacsu, O. A review of ionic liquids towards supercritical fluid applications. *J. Supercrit. Fluids* **2007**, *43* (1), 150-180.

56. COSMO-RS GUI Tutorials_Ionic Liquids. https://www.scm.com/documentation/Tutorials/COSMO-RS/Ionic_Liquids/ (accessed 4/26/2016).

57. Ries, L. A. S.; do Amaral, F. A.; Matos, K.; Martini, E. M. A.; de Souza, M. O.; de Souza, R. F. Evidence of change in the molecular organization of 1-n-butyl-3-methylimidazolium tetrafluoroborate ionic liquid solutions with the addition of water. *Polyhedron* **2008**, *27* (15), 3287-3293.

58. Taib, M. M.; Murugesan, T. Density, Refractive Index, and Excess Properties of 1-Butyl-3-methylimidazolium Tetrafluoroborate with Water and Monoethanolamine. *J. Chem. Eng. Data* **2012**, *57* (1), 120-126.

59. Rilo, E.; Vila, J.; Pico, J.; Garcia-Garabal, S.; Segade, L.; Varela, L. M.; Cabeza, O. Electrical Conductivity and Viscosity of Aqueous Binary Mixtures of 1-Alkyl-3-methyl Imidazolium Tetrafluoroborate at Four Temperatures. *J. Chem. Eng. Data* **2010**, *55* (2), 639-644.

60. Liu, W.; Cheng, L.; Zhang, Y.; Wang, H.; Yu, M. The physical properties of aqueous solution of room-temperature ionic liquids based on imidazolium: Database and evaluation. *J. Mol. Liq.* **2008**, *140* (1-3), 68-72.

61. Liu, W.; Zhao, T.; Zhang, Y.; Wang, H.; Yu, M. The physical properties of aqueous solutions of the ionic liquid BMIM BF4. *J. Solution Chem.* **2006**, *35* (10), 1337-1346.

62. Ueno, K.; Inaba, A.; Ueki, T.; Kondoh, M.; Watanabe, M. Thermosensitive, soft glassy and structural colored colloidal array in ionic liquid: colloidal glass to gel transition. *Langmuir* **2010**, *26* (23), 18031-18038.

63. Gomez, E.; Gonzalez, B.; Calvar, N.; Tojo, E.; Dominguez, A. Physical properties of pure 1-ethyl-3-methylimidazolium ethylsulfate and its binary mixtures with ethanol and water at several temperatures. *J. Chem. Eng. Data* **2006**, *51* (6), 2096-2102.

64. Gonzalez, B.; Calvar, N.; Gomez, E.; Dominguez, A. Physical properties of the ternary system (ethanol + water+1-butyl-3-methylimidazolium methylsulphate) and its binary mixtures at several temperatures. *J. Chem. Thermodyn.* **2008**, *40* (8), 1274-1281.

65. Ciocirlan, O.; Croitoru, O.; Iulian, O. Densities and Viscosities for Binary Mixtures of 1-Butyl-3-Methylimidazolium Tetrafluoroborate Ionic Liquid with Molecular Solvents. *J. Chem. Eng. Data* **2011**, *56* (4), 1526-1534.

66. Rodriguez, H.; Brennecke, J. F. Temperature and composition dependence of the density and viscosity of binary mixtures of water plus ionic liquid. *J. Chem. Eng. Data* **2006**, *51* (6), 2145-2155.

67. Spickermann, C.; Thar, J.; Lehmann, S. B. C.; Zahn, S.; Hunger, J.; Buchner, R.; Hunt, P. A.; Welton, T.; Kirchner, B. Why are ionic liquid ions mainly associated in water? A Car-Parrinello study of 1-ethyl-3-methyl-imidazolium chloride water mixture. *J. Chem. Phys.* **2008**, *129* (10).

68. Seddon, K. R.; Stark, A.; Torres, M. J. Influence of chloride, water, and organic solvents on the physical properties of ionic liquids. *Pure Appl. Chem.* **2000**, *72* (12), 2275-2287.

69. Menon, S. V. G.; Manohar, C.; Rao, K. S. A new interpretation of the sticky hard sphere model. *The Journal of Chemical Physics* **1991**, *95* (12), 9186-9190.

70. Meot-Ner, M. The ionic hydrogen bond. *Chem. Rev.* **2005**, *105* (1), 213-284.

71. Guibert, C.; Dupuis, V.; Fresnais, J.; Peyre, V. Controlling nanoparticles dispersion in ionic liquids by tuning the pH. *J. Colloid Interface Sci.* **2015**, *454*, 105-111.

72. Lopez-Pastor, M.; Ayora-Canada, M. J.; Valcarcel, M.; Lendl, B. Association of methanol and water in ionic liquids elucidated by infrared spectroscopy using twodimensional correlation and multivariate curve resolution. *J. Phys. Chem. B* **2006**, *110* (22), 10896-10902.

73. Singh, T.; Kumar, A. Aggregation Behavior of Ionic Liquids in Aqueous Solutions: Effect of Alkyl Chain Length, Cations, and Anions. *J. Phys. Chem. B* **2007**, *111* (27), 7843-7851.

74. Sarkar, A.; Ali, M.; Baker, G. A.; Tetin, S. Y.; Ruan, Q.; Pandey, S. Multiprobe Spectroscopic Investigation of Molecular-level Behavior within Aqueous 1-Butyl-3-methylimidazolium Tetrafluoroborate. *J. Phys. Chem. B* **2009**, *113* (10), 3088-3098.

75. Bowers, J.; Butts, C. P.; Martin, P. J.; Vergara-Gutierrez, M. C.; Heenan, R. K. Aggregation behavior of aqueous solutions of ionic liquids. *Langmuir* **2004**, *20* (6), 2191-2198.

76. Zhong, X.; Fan, Z.; Liu, Z.; Cao, D. Local structure evolution and its connection to thermodynamic and transport properties of 1-butyl-3-methylimidazolium tetrafluoroborate and water mixtures by molecular dynamics simulations. *J. Phys. Chem. B* **2012**, *116* (10), 3249-3263.

77. Moreno, M.; Castiglione, F.; Mele, A.; Pasqui, C.; Raos, G. Interaction of Water with the Model Ionic Liquid [bmim][BF4]: Molecular Dynamics Simulations and Comparison with NMR Data. *J. Phys. Chem. B* **2008**, *112* (26), 7826-7836.

78. Almasy, L.; Turmine, M.; Perera, A. Structure of aqueous solutions of ionic liquid 1-butyl-3-methylimidazolium tetrafluoroborate by small-angle neutron scattering. *J. Phys. Chem. B* **2008**, *112* (8), 2382-2387.

79. López-Pastor, M.; Ayora-Cañada, M. J.; Valcárcel, M.; Lendl, B. Association of Methanol and Water in Ionic Liquids Elucidated by Infrared Spectroscopy Using Two-Dimensional Correlation and Multivariate Curve Resolution. *J. Phys. Chem. B* **2006**, *110* (22), 10896-10902.

80. Masaki, T.; Nishikawa, K.; Shirota, H. Microscopic Study of Ionic Liquid–H2O Systems: Alkyl-Group Dependence of 1-Alkyl-3-Methylimidazolium Cation. *J. Phys. Chem. B* **2010**, *114* (19), 6323-6331.

81. Malham, I. B.; Letellier, P.; Turmine, M. Evidence of A Phase Transition in Water-1-butyl-3-methylimidazolium tetrafluoroborate and Water-1-butyl-2, 3-dimethylimidazolium tetrafluoroborate mixtures at 298 K: Determination of the Surface Thermal Coefficient, b T, P. *J. Phys. Chem. B* **2006**, *110* (29), 14212-14214.

82. Antonietti, M.; Kuang, D.; Smarsly, B.; Zhou, Y. A. C., Int. Ed. 2004, 43, 4988. Ionic Liquids for the Convenient Synthesis of Functional Nanoparticles and Other Inorganic Nanostructures. *Angew. Chem.* **2004**, *43*.

83. Scheeren, C. W.; Machado, G.; Dupont, J.; Fichtner, P. F.; Texeira, S. R. Nanoscale Pt (0) particles prepared in imidazolium room temperature ionic liquids: synthesis from an organometallic precursor, characterization, and catalytic properties in hydrogenation reactions. *Inorg. Chem.* **2003**, *42* (15), 4738-4742.

84. Wei, G.-T.; Yang, Z.; Lee, C.-Y.; Yang, H.-Y.; Wang, C. R. C. Aqueous-Organic Phase Transfer of Gold Nanoparticles and Gold Nanorods Using an Ionic Liquid. *J. Am. Chem. Soc.* **2004**, *126* (16), 5036-5037.

85. Migowski, P.; Dupont, J. Catalytic applications of metal nanoparticles in imidazolium ionic liquids. *Chemistry-A European Journal* **2007**, *13* (1), 32-39.

86. Lopez-Barron, C. R.; Basavaraj, M. G.; DeRita, L.; Wagner, N. J. Sponge-to-Lamellar Transition in a Double-Tail Cationic Surfactant/Protic Ionic Liquid System: Structural and Rheological Analysis. *J. Phys. Chem. B* **2012**, *116* (2), 813-822.

87. Lopez-Barron, C. R.; Li, D. C.; DeRita, L.; Basavaraj, M. G.; Wagner, N. J. Spontaneous Thermoreversible Formation of Cationic Vesicles in a Protic Ionic Liquid. *J. Am. Chem. Soc.* **2012**, *134* (51), 20728-20732.

88. López-Barrón, C. R.; Li, D.; Wagner, N. J.; Caplan, J. L. Triblock Copolymer Self-Assembly in Ionic Liquids: Effect of PEO Block Length on the Self-Assembly of PEO–PPO–PEO in Ethylammonium Nitrate. *Macromolecules* **2014**, *47* (21), 7484-7495. 89. Bai, Z.; Lodge, T. P. Polymersomes with Ionic Liquid Interiors Dispersed in Water. J. Am. Chem. Soc. 2010, 132 (45), 16265-16270.

90. Altin, E.; Gradl, J.; Peukert, W. First Studies on the Rheological Behavior of Suspensions in Ionic Liquids. *Chem. Eng. Technol.* **2006**, *29* (11), 1347-1354.

91. Ueno, K.; Imaizumi, S.; Hata, K.; Watanabe, M. Colloidal interaction in ionic liquids: Effects of ionic structures and surface chemistry on rheology of silica colloidal dispersions. *Langmuir* **2009**, *25* (2), 825-831.

92. Wittmar, A.; Ruiz-Abad, D.; Ulbricht, M. Dispersions of silica nanoparticles in ionic liquids investigated with advanced rheology. *Journal of Nanoparticle Research C7 - 651* **2012**, *14* (2), 1-10.

93. Wittmar, A.; Gajda, M.; Gautam, D.; Dörfler, U.; Winterer, M.; Ulbricht, M. Influence of the cation alkyl chain length of imidazolium-based room temperature ionic liquids on the dispersibility of TiO2 nanopowders. *Journal of nanoparticle research* **2013**, *15* (3), 1-12.

94. Gao, J.; Ndong, R. S.; Shiflett, M. B.; Wagner, N. J. Creating nanoparticle stability in ionic liquid [C4mim][BF4] by inducing solvation layering. *ACS Nano* **2015**, *9* (3), 3243-3253.

95. Israelachvili, J. N. *Intermolecular and Surface Forces, 3rd Edition*; Elsevier Academic Press Inc: San Diego, 2011. p 1-674.

96. Ueno, K.; Inaba, A.; Kondoh, M.; Watanabe, M. Colloidal stability of bare and polymer-grafted silica nanoparticles in ionic liquids. *Langmuir* **2008**, *24* (10), 5253-5259.

97. Lanning, O. J.; Madden, P. A. Screening at a Charged Surface by a Molten Salt. J. Phys. Chem. B 2004, 108 (30), 11069-11072.

98. Baldelli, S. Probing Electric Fields at the Ionic Liquid–Electrode Interface Using Sum Frequency Generation Spectroscopy and Electrochemistry. *J. Phys. Chem. B* **2005**, *109* (27), 13049-13051.

99. Lynden-Bell, R. M.; Del Pópolo, M. G.; Youngs, T. G. A.; Kohanoff, J.; Hanke, C. G.; Harper, J. B.; Pinilla, C. C. Simulations of ionic liquids, solutions, and surfaces. *Acc. Chem. Res.* **2007**, *40* (11), 1138-1145.

100. Feng, G.; Qiao, R.; Huang, J.; Dai, S.; Sumpter, B. G.; Meunier, V. The importance of ion size and electrode curvature on electrical double layers in ionic liquids. *Phys. Chem. Chem. Phys.* **2011**, *13* (3), 1152-1161.

101. Nordström, J.; Aguilera, L.; Matic, A. Effect of lithium salt on the stability of dispersions of fumed silica in the ionic liquid BMImBF4. *Langmuir* **2012**, *28* (9), 4080-4085.

102. Smith, J. A.; Werzer, O.; Webber, G. B.; Warr, G. G.; Atkin, R. Surprising particle stability and rapid sedimentation rates in an ionic liquid. *J. Phys. Chem. Lett.* **2009**, *1* (1), 64-68.

103. Hayes, R.; Warr, G. G.; Atkin, R. At the interface: solvation and designing ionic liquids. *Phys. Chem. Chem. Phys.* **2010**, *12* (8), 1709-1723.

104. Atkin, R.; Warr, G. G. Structure in confined room-temperature ionic liquids. *J. Phys. Chem. C* **2007**, *111* (13), 5162-5168.

105. Ueno, K.; Kasuya, M.; Watanabe, M.; Mizukami, M.; Kurihara, K. Resonance shear measurement of nanoconfined ionic liquids. *Phys. Chem. Chem. Phys.* **2010**, *12* (16), 4066-4071.

106. Segura, J. J.; Elbourne, A.; Wanless, E. J.; Warr, G. G.; Voïtchovsky, K.; Atkin, R. Adsorbed and near surface structure of ionic liquids at a solid interface. *Phys. Chem. Chem. Phys.* **2013**, *15* (9), 3320-3328.

107. Perkin, S.; Crowhurst, L.; Niedermeyer, H.; Welton, T.; Smith, A. M.; Gosvami, N. N. Self-assembly in the electrical double layer of ionic liquids. *Chem. Commun.* **2011**, *47* (23), 6572-6574.

108. Smith, A. M.; Lovelock, K. R. J.; Perkin, S. Monolayer and bilayer structures in ionic liquids and their mixtures confined to nano-films. *Faraday Discuss.* **2013**, *167* (0), 279-292.

109. Pinilla, C.; Del Pópolo, M. G.; Lynden-Bell, R. M.; Kohanoff, J. Structure and dynamics of a confined ionic liquid. Topics of relevance to dye-sensitized solar cells. *J. Phys. Chem. B* **2005**, *109* (38), 17922-17927.

110. Ueno, K.; Watanabe, M. From colloidal stability in ionic liquids to advanced soft materials using unique media. *Langmuir* **2011**, *27* (15), 9105-9115.

111. Mewis, J.; Wagner, N. J. *Colloidal suspension rheology*; Cambridge University Press: New York, **2012**.

112. Krishnamurthy, L.-n.; Wagner, N. J. The influence of weak attractive forces on the microstructure and rheology of colloidal dispersions. *Journal of Rheology* **2005**, *49* (2), 475-499.

113. Krishnamurthy, L.-N.; Wagner, N. J.; Mewis, J. Shear thickening in polymer stabilized colloidal dispersions. *Journal of Rheology* **2005**, *49* (6), 1347-1360.

114. Kirkwood, K. M.; Kirkwood, J. E.; Lee, Y. S.; Egres, R. G.; Wagner, N. J.; Wetzel, E. D. Yarn pull-out as a mechanism for dissipating ballistic impact energy in Kevlar® KM-2 fabric part I: quasi-static characterization of yarn pull-out. *Text. Res. J.* **2004**, *74* (10), 920-928.

115. Kirkwood, J. E.; Kirkwood, K. M.; Lee, Y. S.; Egres, R. G.; Wagner, N. J.; Wetzel, E. D. Yarn pull-out as a mechanism for dissipating ballistic impact energy in Kevlar® KM-2 fabric part II: predicting ballistic performance. *Text. Res. J.* **2004**, *74* (11), 939-948.

116. Houghton, J.; Schiffman, B.; Kalman, D.; Wetzel, E.; Wagner, N. Hypodermic needle puncture of shear thickening fluid (STF)-treated fabrics. *Proceedings of SAMPE* **2007**, *3*, 1-11.

117. Kalman, D. P.; Merrill, R. L.; Wagner, N. J.; Wetzel, E. D. Effect of particle hardness on the penetration behavior of fabrics intercalated with dry particles and concentrated particle– fluid suspensions. *ACS applied materials & interfaces* **2009**, *1* (11), 2602-2612.

118. NASA. http://www.nasa.gov/audience/forstudents/5-8/features/what-is-a-spacesuit-58.html.

119. Suojiang Zhang, X. L., Qing Zhou, Xiaohua Li, Xianping Zhang, Shucai Li. *Ionic Liquids: Physicochemical Properties*; 1 ed.; Elsevier **2009**.

Chapter 2

EXPERIMENTAL METHODS

In this chapter, an overview of the experimental methods and techniques used throughout this dissertation is described. Unless otherwise stated, experiments are conducted and results are analyzed according to the procedures described herein. These methods include dynamic light scattering (DLS), scanning/transmission electron microscopy, densitometry, thermogravimetric analysis (TGA), rheological characterization, small angle neutron scattering (SANS), X-ray scattering, and quasistatic puncture test on textile materials. The designed vacuum cell setup, which is used to measure the rheological response of shear thickening fluids (STFs) under simulated space vacuum condition, is described in Section 2.6.3, and the primary results are discussed in Chapter 9. Materials and sample preparations are outlined in the "Materials and Methods" Section of the corresponding chapter.

2.1 Dynamic Light Scattering (DLS)

The effective hydrodynamic size and the size distribution of particles are evaluated by DLS through measuring the diffusion coefficient of the particles in dilute suspension undergoing Brownian motion.

In a typical measurement, a monochromatic light source, usually a laser, is projected through a polarizer and into a sample cell. The diffracted light from all of the particles can either interfere constructively or destructively. The interaction between the incident beam and the random Brownian motion of the particles in the liquid sample causes random fluctuations of intensity as a function of time. The intensity autocorrelation function is recorded as a function of time, given by¹⁻²

$$G_{2}(q,\tau) = \frac{\left\langle I(q,t)I(q,t+\tau) \right\rangle}{\left\langle I(q,t) \right\rangle^{2}}$$
 2-1

$$q = \frac{4\pi n_0}{\lambda} \sin(\frac{\theta}{2}), \qquad 2-2$$

where $G_2(q,\tau)$ is the intensity autocorrelation function, q is the momentum transfer vector, τ is the delay time, I(q,t) is the scattering intensity at the scattering vector q and time t, n_0 is the refractive index of the sample, λ is the wavelength of the laser, and θ is the angle of the photodetector. The autocorrelation function $G_2(q,\tau)$ can be converted to the first-order autocorrelation function $G_1(q,\tau)$ through the Siegert relation as shown below¹

$$G_2(q,\tau) = B_0 + \xi |G_1(q,\tau)|^2,$$
 2-3

where B_0 is the baseline, which is close to unity, and ξ is the instrument parameter, usually is on the order of 0.1-1 and is close to unity for an optimum experiment.

Analysis of the intensity correlation function is performed using the cumulant method,^{1, 3}

$$G_1(q,\tau) = \exp(-\Gamma\tau)(1 + \frac{\sigma}{2}\tau^2 + \cdots), \qquad 2-4$$

where Γ is the average decay rate, which characterizes the average relaxation time of the particle, and σ is the coefficient of variation, which gives the polydispersity index of the sample. Through a non-linear fitting to the intensity decay data using Equation 2-4, the average decay rate Γ and the coefficient of variation σ are obtained.

For spherical particle system, the average decay rate Γ is related to the diffusion coefficient *D* by

$$\Gamma = q^2 D \,. \qquad 2-5$$

The hydrodynamic radius R_H of the particles is then calculated from the Stokes-Einstein-Sutherland relation^{2,4}

$$D = \frac{k_B T}{6\pi\mu R_H},$$
 2-6

where μ is the viscosity of the solvent, k_B is Boltzmann's constant, which is 1.38×10^{-23} J/K, and *T* is the absolute temperature. The polydispersity index (PDI) is calculated from the average decay rate Γ and the coefficient of variation σ by¹

$$PDI = \frac{\sigma}{\Gamma^2}.$$
 2-7

Herein, the DLS measurements were performed on a ZetaPALS instrument (Brookhaven Instruments). The instrument optimized the power of the incident laser at the beginning of each measurement with an incident photon wavelength $\lambda = 658$ nm. The scattered light intensity was recorded by a photodetector at a scattering angle $\theta =$ 90°. For the system investigated in this dissertation, q = 0.025 nm⁻¹, and $n_0 \sim 1.4$ (e.g., [C₄mim][BF₄], $n_0 = 1.42$, silica dioxide, $n_0 = 1.46$). The auto-correlation function was measured over a period of 120 s and the average decay time was extracted by fitting to a second-order cumulant expansion using the ZetaPALS software.

Prior to the DLS measurements, the instrument was warmed up for 30 min, allowing it to equilibrate to the experimental temperature and stabilize the laser. The cuvette cells were rinsed by ethanol followed by compressed air to dry the surface. All the raw materials were filtered by a 0.2 um Nalgene syringe filter when preparing samples, no further filtration performed on the final sample before loading into the cuvette cell. On each DLS measurement, 10 repeated data collections were carried out with each spanned 2 mins in order to get a good statistical average value and standard error. Number-averaged particle size and polydispersity were extracted from the software analysis.

2.2 Scanning/Transmission Electron Microscopy (SEM, TEM)

Advanced microscopy techniques such as SEM and TEM are essential techniques to investigate the microstructure, morphology, particle size, and particle coatings of the sample. SEM produces images of a sample surface by raster scanning it with a focused beam of electrons. Through collecting the secondary electrons emitted by the atoms excited by the electron beam, an image displaying the topography of the surface is generated. TEM is a microscopy technique in which a beam of electrons is transmitted through an ultra-thin specimen, interacting with the specimen as it passes through it. The difference of the electron densities between the specimen and its background enables imaging.

In this dissertation, a high resolution SEM (JEOL JSM-7400F) was used to capture the dried silica particles, and their size distributions were analyzed through ImageJ software.⁵ Samples were prepared on aluminum stubs (Electron Microscopy Sciences) with an adhesive carbon tab (Electron Microscopy Sciences). A dilute suspension of silica particles was pipetted onto the aluminum stubs, and dried by atmosphere air. Then the dried samples were sputter-coated with an Au/Pd layer under vacuum for 40 s to increase their conductivity. After that, the samples were loaded into the SEM chamber and digital images were recorded.

39

TEM images of the nanoparticles were taken on a JEOL JEM-2000FX (LaB6) TEM operated at 200 kV. A droplet of the fluorocarbon-coated particle/ethanol solution was dispensed onto a carbon-sputtered copper grid, and the grid was loaded into the TEM chamber after the ethanol completely evaporated. High resolution TEM images then were recorded. The fluorinated particles dispersed in $[C_4mim][BF_4]$ were imaged directly in their solution phase. Samples were prepared by placing a small droplet of dilute particle/ $[C_4mim][BF_4]$ dispersion onto a carbon-sputtered copper grid. Extra solution was removed by blotting with filter paper, leaving a thin film of dispersion on the grid. Liquid nitrogen was constantly supplied during the experiment to eliminate static electricity which could blur the micrographs.

2.3 Densitometry

Gravimetric densitometry is used to determine the density of the liquid samples (such as suspensions and aqueous solutions). The sample is introduced into a Ushaped glass tube which is being excited to vibrate at its characteristic frequency. Through oscillating the U-tube, a correlation between the oscillation frequency and the density of the sample enables the determination of the density of the liquid sample.

The skeletal density of particle in solution is estimated based on the densities of a series of dilute particle suspensions with particle mass fraction $0 \le x < 0.02$. Assuming ideal mixing of the dispersion constituents, the particle density can be extracted using the following equation:

$$\frac{1}{\rho_{solution}} = \left(\frac{1}{\rho_{particles}} - \frac{1}{\rho_{solvent}}\right)x + \frac{1}{\rho_{solvent}}$$
2-8

where, $\rho_{solution}$, $\rho_{particles}$, and $\rho_{solvent}$ are the density of the solution, particle, and solvent, respectively; x is the mass fraction of particles in solution.

Measurements of the particle dispersions densities and solution mixtures at atmospheric pressure were conducted on an Anton Paar DMA-4500M densitometer. The temperature of the sample inside the U-shaped glass tube is controlled within the range of 10 °C to 90 °C by a Peltier temperature control system (accuracy of ± 0.03 °C). The accuracy of the measured density is reported to be $\pm 5 \times 10^{-5}$ g/cm³ by the instrument. The densitometer was calibrated with dry air and ultra-pure water at atmospheric pressure before each measurement. For density measurements, a liquid sample was injected into the U-shaped glass tube using a 3 mL syringe slowly to eliminate the formation of air bubble inside the sample. The temperature of the sample was equilibrated for at least 5 minutes at set temperature before taking measurement, thus minimizing the effect of temperature fluctuation on the oscillation of U-tube. Between each measurement, the U-shaped glass tube was flushed with excess solvents (water, ethanol, and soapy water) gently and then dried by an air fan. "Air check" and "water check" were performed after cleaning to ensure no sample residual in the U-tube, which otherwise would have significant influence on the density measurement.

2.4 Thermogravimetric Analysis (TGA)

Thermogravimetric analysis is a method of thermal analysis in which the mass of a substance is monitored as a function of temperature or time as the sample specimen is subjected to a controlled temperature program in a controlled atmosphere.⁶ TGA is commonly used to determine selected characteristics of materials that exhibit either mass loss or gain due to decomposition, oxidation, or loss of volatiles. In this dissertation, TGA is applied to evaluate the decomposition of surface coating of silica particle, from which the grafting density of coated fluorocarbon can be derived. The TGA instrument continuously weighs a sample as it is heated in a furnace. During a heating experiment, various components of the sample are decomposed as the temperature goes up, and the weight percentage of each resulting mass change can be measured. Results are displayed with temperature on the X-axis and mass loss on the Y-axis. The relative weight fraction of different components can be calculated based on the transition temperature during experiment and the material's specific combustion temperature. For example, the decomposition temperature of fluorocarbon 1H,1H,9H hexadecafluoro-1-nonanol is ~300 °C, while the melting temperature of the silica core is over 2000 °C. Because of the large difference between the combustion temperature of the grafted fluorocarbon chain and the silica particle core, the decomposition signal of fluorocarbon can be easily distinguished. The grating density of the fluorocarbon on the surface of silica particles, which is the number of fluorocarbon chain per nm² of the particle surface, can be readily obtained.

In this thesis work, TGA measurements were performed using a TA Instruments Q500. Two samples: dried uncoated silica particle and fluorocarbon coated silica particle were tested. Prior to the TGA measurement, both of the uncoated and coated silica particles were dried under a stream of nitrogen gas for 5 hours and then in a vacuum oven for 24 hours; however, there was still physically adsorbed water residual left in the pores of the particle. Thus, the physical absorbed water needs to be considered when designing the temperature heated-up program. Specifically, the temperature was increased from 25 °C to 80 °C at a rate of 10 °C/min and equilibrated at 80 °C for 20 min to remove physically adsorbed water. Then, the

42

temperature was increased to 900 °C at the same rate as before and equilibrated at 900 °C for 15 min to ensure complete decomposition of coated fluorocarbon. Nitrogen gas was flowing through the sample pan over the entire measurement. By comparing the final mass loss of uncoated and coated silica particles, the grafting density of fluorocarbon can be deduced.

2.5 Rheological Characterization

2.5.1 Falling-Ball Viscometry

The falling-ball viscometer typically measures the viscosity of gases and Newtonian liquids of low viscosity. A falling-ball viscometer has a spherical ball falling down along a tube containing the sample liquid to be measured. The temperature of sample is controlled by a concentrically surrounded tubular jacket. Newton's law of motion is applied to the force balance of a falling sphere ball when it reaches its terminal steady velocity,⁷ that is a ball falling under the force of gravity in a fluid attains a terminal steady velocity when the net force of frictional (drag) force, gravity force, and buoyancy force reaches zero, such that

$$F_D + F_B = m_s g , \qquad 2-9$$

where m_s is the weight of spherical ball, F_B is the buoyancy force, and F_D is the drag force. In the case of a boundless fluid medium and a negligible inertial effect (Reynolds number $Re \ll 1$), the drag force F_D can be obtained from Stokes' law,⁷⁻⁸

$$\operatorname{Re} = \frac{\rho_f V d}{\mu}$$
 2-10

$$F_D = 3\pi\mu V d , \qquad 2-11$$

where ρ_f is the density of the measured fluid sample, *V* is the terminal velocity of the falling ball, *d* is the diameter of the falling ball, μ is the viscosity of the measured fluid. Thus, the force balance becomes:

$$3\pi\mu Vd + \left(\frac{\pi d^3}{6}\right)\rho_f = \left(\frac{\pi d^3}{6}\right)\rho_s. \qquad 2-12$$

From this equation, the viscosity of fluid sample can be determined when the density difference between the spherical ball and the measured fluid, the diameter and terminal velocity of the spherical ball are known, yielding

$$\mu = \frac{(\rho_s - \rho_f)d^2}{18V}.$$
 2-13

Equation 2-13 can be simplified to the following form:

$$\mu = K(\rho_s - \rho_f)t$$
 where $K = \frac{gd^2}{18l}$. 2-14

In this equation, t is the time needed for the spherical ball to pass the falling length l.

In this dissertation, the dynamic viscosities of ionic liquid and the mixture with water solutions were measured using an Anton Paar AMVn automated microviscometer at temperatures from 20 °C to 100 °C under atmospheric pressure condition. The temperature was controlled by a built-in Peltier thermometer with an accuracy of 0.05 K. The time t_0 of a ball falling over a defined distance in a capillary is determined by a laser detector. As the glass capillary tube is displaying at an angle <90° instead of vertical, Equation 2-14 can be modified to yield

$$\mu = K(\alpha)(\rho_s - \rho_f)t_0. \qquad 2-15$$

Where $K(\alpha)$ is the calibration constant of the measuring system at α angle. Measurements were conducted at three different angles using calibrated capillary tube (d = 3.0 mm) and stainless steel balls (d = 2.5 mm) following standard protocols,⁹ and for each angle measurements were repeated four times. The instrument accuracy for time is reported to be 0.002 s.

2.5.2 Rotational Rheometry

Rheological measurements were conducted on two types of rheometers in this dissertation: an AR-G2 Rheometer and a Discovery Hybrid Rheometer (DHR) from TA Instruments. Both of them are stress controlled rheometers with Peltier plate temperature control within 0.01 °C. Their magnetic drive motor and optical encoder are located in the head of the rheometer as shown in Figure 2.1B. As ionic liquid $[C_4 mim][BF_4]$ corrodes aluminum coating at high temperature, stainless steel tools were used for rheological measurements. Moreover, cone and plate geometry was used to ensure homogeneous shear rate (Figure 2.1A). Conversion between the applied and the measured parameters (torque *M* and strain γ) and the rheological material functions were processed using the TA Instruments Trios software. To erase any previous shear histories and maintain a consistency between measurements, a steady pre-shear was applied at a shear stress of 10 Pa for 2 min prior to each dynamic rheological measurement. Two standard shear experiments: steady shear and oscillatory shear rheometry were used in this dissertation.



Figure 2.1: Schematics of (A) the cone and plate geometry with sample (marked in red color) and (B) stress-controlled AR-G2/DHR rheometer from TA Instruments.

2.5.2.1 Steady Shear Measurements

Steady shear measurements were employed to evaluate the shear viscosity of materials at different shear stress/shear rate. Cone and plate geometry (40 mm stainless steel cone with 2° cone angle) was applied with homogeneous shear rate throughout the radius of cone geometry. The shear rate $\dot{\gamma}$ is given by

$$\dot{\gamma} = \frac{\Omega}{\beta}$$
, 2-16

where Ω is the angular velocity, and β is the cone angle in radians.

For the steady shear tests, a series of steady flow sweeps were performed. The flow sweeps were always conducted by increasing stress first followed by decreasing stress. Multiple forward and backward sweeps were performed to check for hysteresis. At each shear stress, during steady flow sweep, the sample was equilibrated at this stress for 5 seconds, and then the viscosity value was measured for a duration of 60
seconds. Peak hold tests, constant shear stress or shear rate for a period of time, were conducted at certain shear stresses to ensure a true steady state was reached during the flow sweep measurements.

2.5.2.2 Small-Amplitude Oscillatory Shear (SAOS) Measurements

As depicted in Figure 2.2, in a small amplitude oscillatory shear measurement, a sinusoidal strain $\gamma(t)$ or shear rate $\dot{\gamma}(t)$ at an oscillatory frequency ω is applied to a sample, and the responsive sinusoidal stress is measured:¹⁰

$$\gamma(t) = \gamma_0 \sin(\omega t)$$
 2-17

$$\dot{\gamma}(t) = \gamma_0 \omega \cos(\omega t)$$
 2-18

$$\sigma(t) = G^* \gamma_0 \sin(\omega t), \qquad 2-19$$

where *t* is the time, γ_0 is the maximum strain, $G^*=(G'+iG'')$ is the viscoelastic complex modulus.¹⁰⁻¹¹ The linear viscoelastic storage /elastic (*G'*) represents the elasticity of the material, and the loss/viscous moduli (*G''*) characterizes the ability of a material to dissipate energy. They are defined as

$$\sigma(t) = (G'(\omega) + iG''(\omega)\gamma(t)$$
 2-20

$$G'(\omega) = \frac{\sigma_0}{\gamma_0} \cos \delta$$
 2-21

$$G''(\omega) = \frac{\sigma_0}{\gamma_0} \sin \delta, \qquad 2-22$$

where δ is the phase angle (the phase difference between the stress and the stain). δ is 0° and 90° for perfect elastic solids and viscous liquids, respectively. The phase angle δ is defined as the ratio of the viscoelastic moduli, and is a useful parameter to quantify the elasticity of a material:

$$\tan(\delta) = \frac{G''(\omega)}{G'(\omega)}.$$
 2-23



Figure 2.2: Example of strain deformation, the linear stress response, and phase shift as a function of time in a small amplitude oscillatory shear experiment.

A strain sweep is utilized to determine the linear viscoelastic regime (LVE) for each sample, where G' and G'' are independent on the strain applied, before frequency sweep tests. As the LVE changes with temperature, strain sweep measurements need to be conducted for each temperature at interest. In the following frequency sweep measurements, an oscillatory strain, which is sufficiently large to perturb a sample without disturbing the structure of the sample at equilibrium state, is chosen in the linear regime. For the time-temperature superposition measurement, the sample was equilibrated at target temperature for at least 10 min to ensure a homogenous temperature profile inside the sample.

2.5.3 Pressure/Vacuum Cell Geometry

In this dissertation, a vacuum cell was developed based on the pressure cell provided by TA Instruments and it has been used to study the rheology of ionic liquid based shear thickening fluids under vacuum condition. The vacuum cell setup is shown in Figure 2.3. Turbomolecular pump station (HiCubeTM 80 Eco from Pfeiffer Vacuum) is connected to the vacuum cell through flexible stainless steel tubes. The pressure inside the vacuum cell can reach 1×10^{-6} torr within one hour, which is monitored by a Full Range® Complete Pressure Gauge PKR361.



Figure 2.3: The vacuum cell setup for rheological test on ionic liquid based STFs under simulated space vacuum condition. (A) Vacuum cell on the AR-G2 rheometer to measure the rheological response; (B) Turbomolecular pump station to reduce the pressure inside the vacuum cell to 10⁻⁶ torr.

The vacuum cell has a standard concentric cylinder geometry, with temperature controlled by a Peltier-controlled heating jacket (-10 °C to 300 °C). A copper sheath is fitted to the cell to ensure good heat transfer between the jacket and the cell. The cell consists of three main component assemblies, including the cell cup, the concentric cylinder rotor, and the magnet assembly as shown in Figure 2.4. The rotor assembly contains the concentric cylinder rotor, which is mounted on a shaft attaching to a four-pole magnet located under the rotor assembly cap, as shown in Figure 2.5. The cell cup contains the sample fluid and is inserted into the Peltier jacket. The inner radius of the cup is 14.00 mm, the outer radius of the rotor is 13.00 mm, and 9.5 ± 0.5 mL volume of sample is needed for each test. The magnet assembly is attached to the rheometer's rotating spindle, and then it is brought down over the rotor assembly. The measured fluid samples are sealed inside the cell cup. Samples are sheared by the rotor inside the cell, which is controlled by the magnet assembly through magnetic force. The stress-strain response is recorded and analyzed using Trios software.



Figure 2.4: Cross-section schematic of vacuum/pressure cell.¹²



Figure 2.5: Pressure cell rotor assembly with cap on (A) and off (B).¹²

2.6 Small-Angle Neutron Scattering (SANS)

Small angle neutron scattering is a well-established characterization technique and has been widely applied to investigate the microstructure of various materials. Hence, SANS is employed to study the microstructure of ionic liquid aqueous solutions and silica particle-ionic liquid colloidal suspensions. The quiescent microstructures of the complex fluids are investigated using static SANS, whereas their microstructures induced by shear flow are studied by rheo-SANS and flow-SANS.

A schematic illustration of SANS is shown in Figure 2.6. A defined incident neutron beam of wavelength λ passes through a sample, interacts with the sample, giving a wave vector q at a scattering angle θ , which is defined as

$$q = \left|\vec{q}\right| = \left|\vec{k}_s - \vec{k}_i\right| = \frac{4\pi}{\lambda} \sin\left(\frac{\theta}{2}\right).$$
 2-24

Intuitively, a measurement made at a given q allows to investigate the density fluctuations in the sample on a distance scale $D = 2\pi/q$. With neutrons of a typical wavelength of 6 Å, the length scale probed by SANS measurements is O(1-100 nm) depends on the geometry configurations. This covers the range of structure (micellar, emulsion, and particle) discussed in this dissertation. More than 10^5 scattering counts are recorded for each single SANS measurement to obtain good statistics when analyzing the results.



Figure 2.6: Illustration of a typical SANS experiment.

For the SANS measurement, the scattering intensity is related to the structure and the scattering length density (SLD) distribution of the sample. Assuming the coherent scattered intensity measured by SANS is equal to the scattering cross-section, the total scattering intensity is given by:

$$I(q,\phi) = n_p P(q,\phi) S(q,\phi) + I_b,$$
 2-25

where n_p is the number density of the scattering objects, and I_b is the intensity of incoherent scattering background. $P(q,\phi)$ is the form factor, which describes the scattering from individual particle. $S(q,\phi)$ is the structure factor, which contains information of interparticle interactions and spatial arrangement of particles. The incoherent background, I_b , is determined from the slope of a linear regression of the Porod plot $(Iq^4 \text{ vs. } q^4)$.¹³

For spherical particle suspensions, the total scattering intensity can be simplified as:

$$I(q) = \phi_{p} V_{p} (\Delta \rho_{SLD})^{2} p(q) S(q) + I_{b}, \qquad 2-26$$

where ϕ_p is the volume fraction of the particles in the suspension, V_p is the volume of single particle as $V_p = \frac{4\pi r}{3}$, $\Delta \rho_{SLD}$ is the SLD contrast (difference) between the particle and the suspension medium. The form factor P(q) and structure factor S(q) can be decoupled by using both dilute and concentrated particle suspensions. For dilute dispersions, the interactions between particles can be ignored and the structure factor $S(q) \sim 1$; hence information about the form factor can be extracted from the scattering intensity profile.

Static SANS measurements were performed at the National Institute of Standards and Technology (NIST) Center for Neutron Research (NCNR) in Gaithersburg, MD on both NG3 and NG7 beamlines. Samples were loaded into demountable titanium cells with a 1 mm path length. The temperatures of the samples were maintained at target temperature ± 0.1 °C using the 10 CB, 10 position sample cell holder with fluid bath. A wide range of scattering wavevector q was obtained by combining scattering data from four different instrument configurations at detector distances of 1, 4, 13, and 15.3 m with lens. The wavelength for 1, 4, and 13 m configurations is 6.0 Å and 8.0 Å for 15.3 Å with lens. The SANS scattering intensity profiles were reduced to absolute scale and analyzed with the NIST data reduction and analysis macros in IGOR Pro available from NIST.¹⁴

2.6.1 Porod's Law

In the high-q domain, the scattering probes the internal configuration of the mass within the scattering objects. As $q \rightarrow \infty$, there is scattering contrast only at the interface between two media of different SLD. Porod derived the following relationship for the scattering intensity from the interfaces:¹³

$$I(q) \approx \frac{A}{q^4} + I_b, \qquad 2-27$$

where *A* is the Porod constant, and I_b is the incoherent scattering cross-section, which is typically obtained by a "Porod plot" of Iq^n vs. q^n . n = 4 for objects with sharp interface, n = 3-4 for diffuse or rough interfaces.¹³

2.6.2 The Form Factor

In general, the form factor P(q) is defined to describe the distribution of mass within a scattering object. It provides information about a single particle, such as shape, size, and polydispersity. Mathematically, form factor P(q) can be obtained from the direct Fourier transform of the radial pair correlation function, $p(\vec{r}_{ij})$:^{13, 15}

$$P(q) = \int_{V_p} p(\vec{r}_{ij}) \exp(-iq \cdot \vec{r}_{ij}) d\vec{r}_{ij} , \qquad 2-28$$

where the integral is taken over the volume of the scattering object. The pair correlation function $p(\vec{r}_{ij})$ describe the density probability of find a point of mass *i* within the object from a reference point *j*. In the literature, form factors have been derived for many prototypical and complex objects.¹⁵ For example, the model of monodispersed spherical objects for analyzing the structure of ionic liquid-water mixture and the model of polydispersed core-shell object for silica colloidal suspensions are discussed below.

Consider a monodisperse sphere of radius r and uniform density, the form factor is given by¹⁶

$$p(q) = \left(\Delta \rho_{SLD} \frac{3J_1(qr)}{qr}\right)^2, \qquad 2-29$$

where the spherical Bessel function $J_1(x)$ is defined as:

$$J_1(x) = \frac{\sin x}{x^2} - \frac{\cos x}{x}.$$
 2-30

The expression for the form factor of a core-shell object with Schultz distributed core and constant shell is

$$\overline{P}(q,\rho_i,\phi,\overline{r},p,\delta) = \int_0^\infty f(R,\overline{r},p) \Big[\big(\rho_{shell} - \rho_{solvent}\big) F(q,R+\delta) + \big(\rho_{core} - \rho_{shell}\big) F(q,R) \Big]^2 dR$$
2-31

in which the interference factor F(q, R) for a spherical object in the Rayleigh–Gans– Debye limit is¹⁷

$$F(q,R) = 3V_p \frac{\sin(qR) - qR\cos(qR)}{(qR)^3}.$$
 2-32

The Schultz distribution function f(R) is defined as

$$f(R) = (z+1)^{z+1} \left(\frac{R}{\bar{r}}\right)^2 \frac{\exp\left(-(z+1)\frac{R}{\bar{r}}\right)}{\bar{r}\Gamma(z+1)}$$
 2-33

$$z = \left(\frac{\bar{r}}{\sigma}\right)^2 - 1, \qquad 2-34$$

where \bar{r} is the average particle radius, σ is the standard deviation of radius, and $\Gamma(z+1)$ is the Gamma function.

In this dissertation, form factor model of monodispersed sphere is used to describe the properties of single micelle and water cluster in ionic liquid aqueous solutions. Schultz distributed spherical model and the associated core-shell models are fitted to the SANS data for dilute uncoated and coated silica particle dispersions.

2.6.3 The Structure Factor

The structure factor S(q) generally is related to the arrangement and interactions between individual scattering objects. The structure factor $S(q, \bar{r})$ can be expressed as the Fourier transform of the pair distribution function,^{13, 15} $g(\bar{r})$, which is related to the potential of interaction between objects from statistical mechanics, and can be obtained using a liquid state theory such as the Ornstein–Zernike¹⁸⁻¹⁹ (OZ) equation:

$$S(q) = 1 + \overline{N} \int [g(\vec{r}) - 1] \exp(-iq \cdot \vec{r}) d\vec{r}$$
 2-35

$$g(\vec{r}) - 1 = c(\vec{r}) + \overline{N} \int c \left(|\vec{r} - \vec{r}_1| \right) \left(g(\vec{r}_1) - 1 \right) d\vec{r}_1, \qquad 2-36$$

where \overline{N} is the number density of particles, $c(\vec{r})$ is the direct correlation function, which is related to the potential of mean force, $\Phi(r)$, between interacting particles. Closer approximation: Percus–Yevick²⁰ approximation is used to calculate the direct correlation function $c(\vec{r})$ for short-ranged interactions between particles, as shown below

$$c(r) = g(r) \left[1 - \exp\left(-\frac{\Phi(r)}{k_b T}\right) \right].$$
 2-37

Sticky-hard sphere structure factor model is applied to evaluate the interaction between particles for a hard sphere fluid with a narrow attractive well. The strength of the attractive well is described in terms of "stickiness" parameter or the Baxter parameter is defined as

$$\tau_{B} = \frac{1}{12\varepsilon} \exp(\Phi_{0} / k_{B}T)$$
 2-38

$$\varepsilon = \Delta/(a + \Delta),$$
 2-39

where Δ and a are separation distance.

The interaction potential between particles, $\Phi(r)$, as shown in Figure 2.7, is defined as²¹

$$\Phi(r) = \begin{cases} \infty & r < a \\ -\Phi_0 & a < r < a + \Delta \\ 0 & r > a + \Delta \end{cases}$$
 2-40



Figure 2.7: The graphical description of a sticky hard sphere potential for a coated particle system.

The perturbation parameter, ε , is between 0.01 and 0.1. The stickiness, τ_B , is representative of the attraction strength, and is related to the reduced second virial coefficient, B_2^* , by $\tau_B = \frac{1}{4(1-B_2^*)}$. From the definition, it is clear that smaller τ_B

means stronger attraction.

2.7 Small-Angle X-Ray Scattering (SAXS)

X-rays are electromagnetic waves and have shorter wavelength than light. It is able to probe the material structure on the length scale of 1-100 nm.²² SAXS method was used to characterize the microstructure of liquid solutions, complementary to SANS and other light scattering methods. SAXS shares the same experimental theory as for SANS with scattering intensity depending on the electron density contrast between structure and background using x-ray instead of neutron beam.¹³

SAXS measurements were conducted at the F2 BioSAXS beamline ($\lambda = 1.29$ Å) at the Cornell High Energy Synchrotron Source (CHESS) facility. The incident X-ray beam was aligned to a 250 × 250 exposure area. The path tubes were kept under vacuum to reduce scattering from air. The scattered signal was recorded by a Quantum 1CCD detector. Calibration was conducted using a silver behenate powder sample. Samples were pipetted manually into the funnel connected to the end of the capillary. BioXTAS RAW program²³ was used to reduce the 2D data to 1D data. The absolute intensity was obtained by normalizing to the incident beam intensity, and subtracting the background scattering of the solvent.

2.8 Quasi-Static Puncture Test

The puncture resistance of the Kevlar textile with or without shear thickening fluids treatment was evaluated by quasi-static puncture test. During the measurement, the ASTM F-1342 standard was followed. An A0 6-1/4" awl spike from Malco Products, Inc. was used to simulate the threat from round-edge objects and hypodermic needles of varying gauges were used to simulate the threat from sharpedge objects. Between each experiment, the needle is replaced by a fresh one to ensure the same sharpness.²⁴ The spike or needle was held within the chunk that was mounted to a 500 N load cell on an Instron 5965. The single layer of Kevlar textile of 3 inch \times 3 inch was mounted in a frame as shown in Figure 2.8, such that a circle of 6.35 diameter unbacked fabric was exposed. The spike or needle was loaded at a constant quasi-static speed approaching the textile. Zero displacement and zero force were defined when the probe nearly touched the fabric. The force experienced by the probe was recorded as a function of displacement. The maximum value of the force in the force-displacement curve was used to evaluate the force resistance of the tested

fabric. Twelve repeated experiments were conducted on each sample at different spots to obtain a good statistical value.





Figure 2.8: Set-up of quasi-static puncture test apparatus used to evaluate the quasi-static puncture resistance of Kevlar fabric with or without STFs treatment.
(A) Photo of the modified ASTM F-1342 standard (with needle probe).
(B) Optical image of hypodermic needles (18 G and 21 G) and Malco M0 spike on Kevlar 1148. (C) The dimensions of unbacked sample holder. Image courtesy of Dr. Colin D. Cwalina.

Because of the geometry of probe such as the spike and the needle, a typical force-displacement curve²⁵ initially shows gradual loading as the probe begins to push against the textile, with one or more dips in curves due to the partial tip breakthrough. In addition to that, yarns are deflected at a contact point at low displacements when using needle probe. After the force reaches the maximum value, a sharp force drop occurs as the probe tip penetrates through the textile completely. The point of puncture is defined as the point at which the force-displacement curve goes through a maximum value.²⁵ After the point of puncture, a long tail of low force is observed. For needle probe test, this low force is the result of sliding friction between the yarns and the barrel of the needle. For the spike probe test, however, there is a slightly increase in force after puncture point because the textile is pushed through and the diameter of the main shaft gradually increases.

REFERENCES

1. Berne, B. J.; Pecora, R. *Dynamic light scattering: with applications to chemistry, biology, and physics*; Courier Corporation, **1976**.

2. Hiemenz, P. C.; Rajagopalan, R. *Principles of Colloid and Surface Chemistry, revised and expanded*; CRC Press, **1997**; Vol. 14.

3. Koppel, D. E. Analysis of macromolecular polydispersity in intensity correlation spectroscopy: the method of cumulants. *The Journal of Chemical Physics* **1972**, *57* (11), 4814-4820.

4. Einstein, A. Über die von der molekularkinetischen Theorie der Wärme geforderte Bewegung von in ruhenden Flüssigkeiten suspendierten Teilchen. *Annalen der physik* **1905**, *322* (8), 549-560.

5. ImageJ. https://imagej.nih.gov/ij/ (accessed 11/12/2016).

6. Coats, A. W.; Redfern, J. P. Thermogravimetric analysis. A review. *Analyst* **1963**, 88 (1053), 906-924.

7. Batchelor, G. K. *An Introduction to Fluid Dynamics*; Cambridge University Press: Cambridge, **2000**.

8. Lamb, H. *Hydrodynamics*; Cambridge university press1932.

9. Rolling Ball Viscometer Standard Operation Procedure. In *Rolling Ball Viscometer Standard Operation Procedure*; Anton Paar, **2015**.

10. Macosko, C. W. Rheology: principles, measurements, and applications. **1994**.

11. Stastna, J.; De Kee, D.; Powley, M. Complex viscosity as a generalized response function. *Journal of Rheology (1978-present)* **1985,** *29* (4), 457-469.

12. Pressure Cell. http://www.tainstruments.com/pressure-cell/ (accessed 11/16).

13. Roe, R.-J. *Methods of X-ray and neutron scattering in polymer science*; Oxford University Press, **2000**; Vol. 739.

14. Kline, S. R. Reduction and analysis of SANS and USANS data using IGOR Pro. *J. Appl. Crystallogr.* **2006**, *39* (6), 895-900.

15. Hammouda, B. Probing nanoscale structure - the SANS toolbox [Online],2008.

16. Fournet, G.; Guinier, A. Small angle scattering of X-rays. *Translated by Walker, CB and Yudowitch, KL In: New York: John Wiley & Sons* **1955**, 7-78.

17. Glatter, O.; Kratky, O. *Small angle X-ray scattering*; Academic press1982.

18. Ornstein, L. S.; Zernike, F. Acculental deviations of density and opalescence at the critical point of a simple substance. *Proceedings of the Koninklijke Akademie Van Wetenschappen Te Amsterdam* **1914**, *17*, 793-806.

19. Hansen, J.-P.; McDonald, I. R. *Theory of simple liquids*; Elsevier1990.

20. Percus, J. K.; Yevick, G. J. Analysis of Classical Statistical Mechanics by Means of Collective Coordinates. *Physical Review* **1958**, *110* (1), 1-13.

21. Menon, S. V. G.; Manohar, C.; Rao, K. S. A new interpretation of the sticky hard sphere model. *The Journal of Chemical Physics* **1991**, *95* (12), 9186-9190.

22. Goodenough, J. B.; Manthiram, A. A perspective on electrical energy storage. *MRS Communications* **2014**, *4* (4), 135.

23. Nielsen, S.; Toft, K. N.; Snakenborg, D.; Jeppesen, M. G.; Jacobsen, J.; Vestergaard, B.; Kutter, J. P.; Arleth, L. BioXTAS RAW, a software program for high-throughput automated small-angle X-ray scattering data reduction and preliminary analysis. *J. Appl. Crystallogr.* **2009**, *42* (5), 959-964.

24. Houghton, J.; Schiffman, B.; Kalman, D.; Wetzel, E.; Wagner, N. Hypodermic needle puncture of shear thickening fluid (STF)-treated fabrics. *Proceedings of SAMPE* **2007**, *3*, 1-11.

25. Richard, D. D. N. J. W. Measurement of needle puncture resistance using an electronic puncture detection system. In *Proceedings of SAMPE*: Baltimore, MD, **2012**.

Chapter 3

EFFECT OF WATER ON PHYSICAL PROPERTIES AND STRUCTURES OF IONIC LIQUIDS

The purpose of this chapter is to describe the effect of addition of water molecules on the physical properties (density and viscosity) and microstructure of an ionic liquid [C₄mim][BF₄]. The phase transition of ionic liquid [C₄mim][BF₄]-water mixtures covering the full composition range from high salt to low salt is investigated. At high salt levels, water molecularly dissolves into the network of ionic liquid $[C_4 mim][BF_4]$ through hydrogen bond, and a water-ionic liquid hydrogen bond network is formed without formation of microphase. Densities and viscosities of the homogenous binary mixtures of [C₄mim][BF₄] with water are reported for a temperature range of (283.15 to 373.15) K. A correlation between the excess mixture viscosity and the excess molar volume in the ionic liquid rich regime is identified, which is motivated by the free volume theory. As the water concentration increases (over the salt-rich regime), a microphase transition to water nanoclusters resembling an inverse microemulsion is identified and characterized using small angle neutron scattering (SANS). On the other hand, ionic liquid micelles form at low salt concentration due to the amphiphilic properties of cation $[C_4 mim]^+$, as shown by small angle x-ray scattering (SAXS). A phase diagram for ionic liquid aqueous solutions (microphase separation, phase inversion, and micelle formation) is constructed, revealing similarities to traditional oil-water-surfactant systems. These results provide direct confirmation for recent simulations, as well insight into the source of nonidealities in some thermophysical and transport properties (e.g., density and viscosity) of salt-rich aqueous mixtures reported in literature. Text and figures are reproduced and adapted with permission from Gao J., et al. *Langmuir* 2016, 32, (20), 5078–5084, and Gao J., et al. *Journal of Molecular Liquids* 2016, 223, 678-686.

3.1 Introduction

3.1.1 Densities and Viscosities of Ionic Liquid Aqueous Solutions

The physicochemical properties of ionic liquids and their mixture with organic/inorganic solvents are of importance for their industrial applications.¹⁻³ Therefore, thermodynamic properties such as density, viscosity, surface tension, refractive index, conductivity, and the speed of sound in binary mixtures of ionic liquids with molecular solvents are of great interest both from practical and theoretical point of view. Recently, systematic studies of the density, viscosity of ionic liquids $[C_nmim][BF_4]$ with water,⁴⁻¹¹ alcohols,¹²⁻¹³ dimethyl sulfoxide (DMSO),¹⁴ ethylene glycol (EG)¹⁴ *et al.* over broad composition ranges have been reported. Of particular relevance for the work here, Rodriguez *et al.*¹⁵ studied the temperature and composition dependence of the density and viscosity of aqueous solutions of a series of 1-ethyl-3-methylimidazoloium with varying anions, whereupon they suggested a trend between excess volume and viscosity.¹⁵ In this chapter, I have explored the thermodynamic relationships between transport and state properties in ionic liquid and water mixtures, and identified a new, semi-empirical relationship between non-idealities in viscosity and excess free volume that is motivated by free volume theory.

1-Butyl-3-methylimidazolium tetrafluoroborate, [C₄mim][BF₄], is among the most prevalently used and commonly investigated ionic liquids. Water contamination

is well known in ionic liquids,¹⁶ especially for ionic liquids, which have the tendency to mix with or dissolve in water, such as $[C_4mim][BF_4]$. The addition of trace amount of water into ionic liquids significantly changes the thermodynamic properties by affecting the microstructure.¹⁷⁻¹⁸ Therefore, in addition to the inherent importance of understanding the density and viscosity of binary mixture of $[C_4mim][BF_4]$ with water, knowledge of excess properties provides insight into the molecular interactions in these mixtures and aids in developing structure-property relationships necessary for optimizing the properties of ionic liquids for specific applications.^{1, 19-20}

Herein, in section 3.3.2, I report the change of density and viscosity upon addition of water for [C₄mim][BF₄] in salt-rich regime (up to 40% water molar fraction) over the temperature range from 283.15 to 373.15 K and demonstrate a fundamentally new correlation between density and viscosity that aids in understanding the non-idealities in a wider range of ionic liquid aqueous mixtures. Neutron scattering measurements, which will be discussed in section 3.3.3.2, show solubility of water in the ionic liquid without microphase separation in this composition regime.²¹ The temperature dependence of density and dynamic viscosity of the aqueous mixtures are described by an empirical, linear relationship and by the Vogel-Fucher-Tammann (VFT) equation, respectively. Their excess properties are interpreted to evaluate the molecular interaction between the ionic liquid and water. The excess molar volumes are calculated using the measured experimental data and fitted to the Redlich-Kister equation. From this analysis, I discover a new correlation between the so-called "excess mixture viscosity"¹⁹ and the excess molar volume in the ionic liquid rich regime that is consistent with free volume theory. The non-ideal densities and viscosities are understood in terms of the interactions between cations

67

and anions of the ionic liquid and the water molecules. Finally, I explore the generality of this observation by analyzing published data on aqueous solutions of related ionic liquids.

3.1.2 Microstructure and Microphase Transition of [C₄mim][BF₄] Aqueous Solutions

As mentioned before, the density and viscosity of $[C_4mim][BF_4]$ aqueous mixtures are non-ideal across the entire composition range. These non-idealities were attributed to specific water effects on the microstructure of the room temperature ionic liquids (RTIL),¹⁷ whereby water molecules hydrogen bond to the $[BF_4]^-$ anions as reported by simulation²²⁻²³ and experiment.²⁴⁻²⁷

The molecular structure and interactions in mixtures of ionic liquids and water have been studied by both experimental techniques (e.g., NMR,²⁸ IR/Raman,²⁴⁻²⁵ fluorescence spectroscopy,²⁹ and SANS^{26, 30}) and molecular simulations.²² Due to their amphiphilicity, RTILs can self-assembly into micelle-like structures when diluted in solvents, similar to surfactants. This feature was evident in SANS measurements on dilute [C₄mim][BF₄], [C₈mim][I] and [C₈mim][Cl] aqueous solutions as reported by Bowers²⁶ and Almasy.³⁰ Surface tension and conductivity measurements for aqueous solutions of [C₄mim][BF₄] also display distinct break points characteristic of aggregation of the ionic liquid salts.²⁶ Singh *et al.* employed NMR and conducted a detailed analysis of chemical shifts of water and various protons of ionic liquids ([C₄mim][BF₄], [C₄mim][Cl], [C₈mim][Cl], N-butyl-3methypyridinium chloride ([C₄mpy][Cl])) aqueous solutions to probe the aggregation structure.^{28, 31} The conformational changes in a variety of RTILs induced by the selfaggregated of ionic liquid are shown to depend on the aromatic ring, alkyl chain, counter ions, and their interactions with water.²⁸ In Section 3.3.3.1, I study the formation and growth of micellar structure of $[C_4mim][BF_4]$ in water solutions with various concentrations at room temperature through SAXS. The shape and size of micellar structure are evaluated and compared quantitatively using monodispersed sphere and ellipsoid form factor model fittings. The growth of micelle is illustrated to be along one direction forming "rod" like shape micelle.

The opposite end of the composition spectrum, rich in RTIL, is equally interesting, but perhaps less understood.^{22, 24-25} Research shows that the interaction between the anion and water generally dominates for salt-rich region of [C₄mim][BF₄]-water mixtures, with individual water molecules forming hydrogen bonds with two [BF₄]⁻ anions without substantially disrupting the nanoscale structuring present in the ionic liquid.^{22, 17, 18, 19, 20} However, with increasing water addition, water molecules begin to associate, forming dimers and eventually larger clusters. Furthermore, the ionic liquid's nanostructure, which is regulated by strong electrostatic interactions between the cations and anions, is gradually disrupted with increasing water addition.²² Further addition of water leads to a sequence of microphase transitions from ionic liquid polar networks to water clusters in the RTIL, which self-aggregates in water, and finally to dilute ionic liquid electrolytes in water. These phase transitions and properties are revealed by molecular simulation²² and experiments.²⁴⁻²⁷

Water and methanol associations in ionic liquids have been studied by FTIR spectroscopy.²⁴ At low water content, spectral features in the OH stretching region reveals that most of the water and methanol molecules tend to be isolated from each other and interact with the anion of the IL through hydrogen bonding. Self-association

69

of water (and methanol) molecules is evident in $[C_2mim][BF_4]$ and $[C_4mim][BF_4]$ at higher water (methanol) concentrations. Shirota *et al.*²⁵ investigated the state of molecular association in $[C_nmim][BF_4]-H_2O$ and $[C_nmim][NTf_2]-H_2O$ mixtures by Raman and IR spectroscopy and reported that water structure is similar for mixture with RTILs with varying cation alkyl length. This result confirms that water aggregation in RTILs is mainly determined by association of the water with ionic groups in the RTILs and less by the hydrophobicity of the cations.

As discussed, the presence of water clusters in salt-rich RTIL aqueous mixtures has been proposed in experiments²⁴⁻²⁷ and predicted by simulations;²²⁻²³ however, the nanoscale structure of such water clusters has not been measured directly. In section 3.3.3.2, I investigate the effects of addition of deuterated water on the nanoscale structure of $[C_4mim][BF_4]$ at room temperature. SANS measurements quantitatively determine the microstructure in the salt-rich mixtures with varying water content. Distinct water structures are revealed in these aqueous solutions with increasing water content, indicating that water clusters in the form of an inverse microemulsion form from the excess water once the ions are fully hydrated, confirming prior hypotheses and simulations.

3.2 Materials and Methods

3.2.1 Materials

The RTIL [C₄mim][BF₄] (>99% pure, 226 g/mol, from Iolitec, batch# K00119.1.Inc, The reported purities from manufacture are: cation 99.8%, anion 99.9%, halides <100 ppm, water 151 ppm.), was dehydrated under vacuum condition at 70 °C for 24 *h* prior to use. The water content was measured in our lab and determined to be 373 ppm by Karl Fischer titration. Millipore deionized water with resistivity of 18.2 $M\Omega \cdot cm$ (25 °C) was used to prepare aqueous ionic liquid solutions for density and viscosity measurements. Deuterated water (D₂O, 99.8% pure, Cambridge Isotope Lab. Inc) was used for enhancing contrast in SANS measurement. Both the ionic liquid and water were filtered using 200 nm nylon membrane syringe filter before mixing. The samples were prepared gravimetrically and the results reported on both weight and molar fraction basis. The uncertainty in the samples' mass fraction is within 3% of mass fraction for each sample. The mass fractions of water in samples and the corresponding molar fractions are listed in Table 3.1. All samples were prepared freshly right before measurements. Density and viscosity measurements were completed within 30 min. Samples were sealed in glass tube which prevented the water absorption during measurements. SANS measurement for each sample was finished within 10 min and samples were loaded to SANS titanium cell sealed with oring to prevent any possible humidity effects in samples during measurement. The effect of hydrolysis of [C₄mim][BF₄] in aqueous solution on our SANS, density and viscosity measurements was proved to be negligible based on the ¹⁹F NMR measurements in Figure 3.1.

Density & viscosity measurements						
Sample #	H ₂ O mass fraction in mixture*	* H ₂ O molar fraction in mixture				
1	0.28 wt%	0.03				
2	0.52 wt%	0.06				
3	1.01 wt%	0.11				
4	2.03 wt%	0.21				
5	3.01 wt%	0.28				
6	5.00 wt% 0.40					
SANS measurements						
Sample #	D_2O mass fraction in mixture* D_2O molar fraction in m					
1	3.02 wt%	0.26				
2	4.97 wt% 0.37					
3	11.01 wt%	0.58				
4	14.97 wt% 0.67					
5	20.00 wt%	0.74				
6	22.01 wt%	0.76				
7	25.03 wt%	0.79				
8	26.98 wt% 0.81					
9	30.02 wt&	0.83				

Table 3.1: $[C_4mim][BF_4]$ -water mixtures compositions in mass fraction and molar fraction.

* The uncertainty of samples' mass fraction is within 3% of mass fraction for each sample.

3.2.2 Fluorine Nuclear Magnetic Resonance Spectroscopy (¹⁹F NMR)

¹⁹F NMR spectra were measured on Bruker AV400. Ionic liquid
[C₄mim][BF₄] and its aqueous solution were dissolved in deuterated chloroform
(CDCl₃) before measurement. MestReNova NMR software was used to perform spectra analysis.

3.2.3 Density Measurements

The densities of pure ionic liquid $[C_4mim][BF_4]$ and of its aqueous mixtures at atmospheric pressure were measured with an oscillating U-tube densitometer Anton Paar DMA 4500 M at temperatures from 283.15 K to 363.15 K with Peltier temperature control with an accuracy of ± 0.03 K. The densitometer was calibrated with dry air and ultra-pure water at atmospheric pressure.

3.2.4 Viscosity Measurements

Kinematic viscosities were measured using an Anton Paar AMVn automated microviscometer at temperatures from 293.15 K to 373.15 K at atmospheric pressure condition. The temperature was controlled by a built-in Peltier within 0.05 K. Measurements were made with three different angles using calibrated capillary of d = 3.0 mm and stainless steel balls of d = 2.5 mm following standard protocols ³² and repeated four times for each angle. The viscosities and standard deviations were reported as the uncertainty in Appendix A (Table A.1) for each composition and temperature. The instrument's accuracy for measuring time is reported to be 0.002 s. The reproducibility of the instrument used for viscosity measurement was within 0.35% and the reproducibility between samples was within 0.7%. Dynamic viscosities are reported here, which are the product of the measured kinematic viscosities and the density.

3.2.5 Small-Angle Neutron Scattering (SANS)

SANS measurements were performed at the National Institute of Standards and Technology (NIST) Center for Neutron Research (NCNR) in Gaithersburg, MD on both NGB 30 and NG7 beamlines. Samples were loaded into demountable titanium cells with a 1 mm path length. The cell temperatures were maintained to ± 0.1 °C

73

using the 10 CB, 10 position sample cell holder with fluid bath. A high scattering wavevector q range was achieved by obtaining scattering intensity from one instrument configuration at detector distances of 1 m. The wavelength for 1 m configurations is 6.0 Å. The SANS scattering intensity profiles were reduced to absolute scale and analyzed with the NIST data reduction and analysis macros in IGOR Pro available from NIST.³³

3.2.6 SANS Data Analysis

The properties of the ionic liquid aqueous solution with water clusters were determined by modeling the measured SANS intensity with a monodisperse sphere form factor using software packages provided the Center for Neutron Research.³³ The scattering length density (SLD) of water and pure ionic liquid [C₄mim][BF₄] were calculated using the measured water density (Table 3.2) at 25 °C by NIST NCNR online tool: Neutron activation and scattering calculator. I assumed that the deuterated water clusters could be modeled as monodisperse spheres with the properties of pure D₂O. The volume fraction of water clusters determined from the fitting was used to calculate the volume fraction and molar fraction of deuterated water molecularly dissolved in the ionic liquid. For simplicity, ideal mixing was assumed to calculate the water distributions in dissolved state and microphase separated state. The SLD of ionic liquid-water mixture in dissolved state were calculated using the additive equation:

$$SLD_m = \phi_{IL}SLD_{IL} + \phi_w SLD_w$$
 3-1

where ϕ_{IL} and ϕ_w are the volume fraction of ionic liquid [C₄mim][BF₄] and deuterated water in dissolved state, respectively; SLD_{IL} and SLD_w are the SLD of pure ionic liquid $[C_4mim][BF_4]$ and deuterated water at 25 °C, respectively (Table 3.2). This calculated SLD of ionic liquid-water mixture in dissolved state was compared with the SLD of the same phase from SANS fitting to evaluate the validation of SANS fitting results.

Materials ρ (g/cm³, 25 °C)Mw (g/mol)SLD (Å-2)[C_4mim][BF_4]1.201226 1.40×10^{-6} D_2O1.10720 6.38×10^{-6}

Table 3.2: Density, molecular mass, and SLD of ionic liquid $[C_4mim][BF_4]$ and D_2O at 25 °C.

3.3 **Results and Discussion**

3.3.1 Hydrolysis Analysis of Ionic Liquid [C₄mim][BF₄] with Present of Water

Aqueous solutions of tetrafluoroborate ($[BF_4]^-$) and hexafluorophoshate ($[PF_6]^-$) based ionic liquids are reported to hydrolyze, liberating side products such as $[BF_3OH]^-$ and hydrogen fluoride.³⁴⁻³⁵ However, the degradation rate depends on water content, temperature, pH, and time since preparation. Freire *et al.*³⁵ reported that $[C_4mim][BF_4]$ aqueous solutions with particularly high water content (~ 50 wt%, 93 mol%) decompose to ~2% side products after heated to 373 K for 30 min.

The density or viscosity measurements herein were conducted at temperature range from 283.15 K to 373.15K and the tests were completed within 30 min. Considering the low water contents (\leq 5 wt%) and low temperature range, the decomposition rate is likely lower than 2%. Our ionic liquid-D₂O aqueous solution samples were prepared freshly right before SANS measurement and loaded to SANS

titanium cell sealed with o-ring to prevent possible humidity effects in samples during measurement. Our SANS measurement on each sample were able to be completed within 10 min at 298 K, and measurements on all samples were finished in 80 min. Based on the low water contents, low temperature, and short time, the decomposition rate would be even lower than 2%. I performed ¹⁹F-NMR measurements for pure [C₄mim][BF₄], [C₄mim][BF₄] aqueous solution with 5 w% H₂O after 30 min at 298.15 K, and [C₄mim][BF₄] aqueous solution with 5 w% H₂O measured after density and viscosity measurements for the entire temperature range in deuterated chloroform (CDCl₃) and shown in Figure 3.1. The ¹⁹F-NMR spectra of the pure IL show no other peaks besides the one at -151.6 ppm, which corresponds to the anion $[BF_4]^-$. In addition to this peak, for the [C₄mim][BF₄] aqueous solution with 5 w% H₂O at 298.15 K and for the one taken after density and viscosity measurements, these two aqueous samples show a small peak at -150.7 ppm, which indicates the decomposition of the anion $[BF_4]^-$. However, as seen, the decomposition peak intensities are very low, such that the decomposition rate is negligible, when compared with the un-hydrolyzed $[BF_4]^-$ anion. The effect of $[C_4 mim][BF_4]$ hydrolysis on density/viscosity/SANS measurement results can be neglected.



Figure 3.1: ¹⁹F-NMR spectrum of pure [C₄mim][BF₄] (A), [C₄mim][BF₄] aqueous solution with 5 w% H₂O after 30 *min* at 298.15 K (B) and [C₄mim][BF₄] aqueous solution with 5 w% H₂O after density/viscosity measurement for the entire temperature range (C) in CDCl₃.

3.3.2 Physical Properties Dependence

Density and viscosity measurements for the pure ionic liquid $[C_4mim][BF_4]$ over the temperature range 283.15 K to 373.15 K are compared with the available literature data in Figure 3.2. The corresponding comparisons for $[C_4mim][BF_4]$ water systems with various water contents at 298.15 K are shown in Figure 3.3. A satisfactory agreement is found. The densities and viscosities results show minor differences (with a few outlier points in the literature data), which may be attributable to differences in sample purity, i.e., the accuracy of the reported data is generally limited by the sample purity.



Figure 3.2: Comparison of measured densities and dynamic viscosities with literature values for pure [C₄mim][BF₄] at temperatures from 283.15 K to 373.15 K. The literature data are from: Ries *et al.*,⁴ Zhou *et al.*,⁵ Taib *et al.*,⁶, Rilo *et al.*,^{7,36} Liu *et al.*,⁹ Ciocirlan *et al.*,¹⁴ Wang *et al.*,³⁷ and Zhu *et al.*³⁸



Figure 3.3: Comparison of measured densities and dynamic viscosities with literature values at 298.15 K for [C₄mim][BF₄] with various water contents. The literature data are from: Ries *et al.*,⁴ Taib *et al.*,⁶ Rilo *et al.*,^{7, 36} Liu *et al.*,⁹ Wang *et al.*,³⁷ and Zhu *et al.*³⁸

3.3.2.1 Density

The densities of $[C_4mim][BF_4]$ aqueous solutions with respect to water weight fraction and temperature are shown in Figure 3.4 and in Table A.1 (see Appendix A). The addition of small amounts of water decreases the mixture density, which also decreases with temperature. Figure 3.4A shows that the density decreases linearly with the water mass fraction at each temperature, and the values of the linear fits are given in Table 3.3. The density decreases with increasing temperature, as shown in Figure 3.4B, but the relative effect of water addition is nearly insensitive to temperature. A linear correlation between density and temperature is observed for each composition as shown in Figure 3.4B and the fitting parameters are also given in Table 3.3. These linear correlations between density and water weight fraction and/or temperature are also useful for determining water content from density experiments.



Figure 3.4: The densities of [C₄mim][BF₄] aqueous solution for varying water content (A) and temperature (B). Solid lines represent the linear correlations, respectively.

Table 3.3:	Fitting parameters of the linear relation of density - water content (Figure
	3.4A) and density - temperature (Figure 3.4B) for ionic liquid aqueous
	solutions.

Density-Water content (Figure 3.4A)							
T(K)	Slope	Intercept (g/cm^3)	T(K)	Slope	Intercept (g/cm^3)		
283.15	3.21×10 ⁻³	1.212	323.15	3.30×10 ⁻³	1.183		
293.15	3.22×10 ⁻³	1.205	333.15	3.35×10 ⁻³	1.176		
298.15	3.23×10 ⁻³	1.201	343.15	3.39×10 ⁻³	1.169		
303.15	3.24×10 ⁻³	1.197	353.15	3.43×10 ⁻³	1.162		
313.15	3.27×10 ⁻³	1.190	363.15	3.40×10 ⁻³	1.155		
Density-Temperature (Figure 3.4B)							
x_w (wt%)	Slope	Intercept (g/cm^3)	$\begin{array}{c} x_w \\ (\text{wt\%}) \end{array}$	Slope	Intercept (g/cm^3)		
0.00	-7.03×10 ⁻⁴	1.412	2.03	-7.08×10 ⁻⁴	1.406		
0.28	-7.10×10 ⁻⁴	1.411	3.01	-7.12×10 ⁻⁴	1.403		
0.52	-7.04×10^{-4}	1.409	5.00	-7.20×10^{-4}	1.400		
1.01	-7.06×10 ⁻⁴	1.408					

The experimental densities for the binary mixtures are compared with calculated densities based on ideal mixing theory in Figure 3.5, where negative deviations from ideal mixing are observed for all mixtures. The negative deviations indicate an expansion of the ionic liquid microstructure with the addition of small amounts of water. The excess molar volumes V^{ex} are calculated as:

$$\underline{V}^{ex} = \frac{x_{IL}M_{IL} + x_{w}M_{w}}{\rho_{m}} - \frac{x_{IL}M_{IL}}{\rho_{IL}} - \frac{x_{w}M_{w}}{\rho_{w}}$$
3-2

where ρ_m is the density of the mixture; x_{ll} and x_w are the molar fraction of ionic liquid and water, respectively; M_{IL} and M_w are the molecular weight of pure ionic liquid and water. The values of \underline{V}^{ex} shown in Figure 3.6A at several temperatures are fitted to the Redlich-Kister polynomial equation:

$$\underline{V}^{ex} = x_{w} x_{IL} \sum_{i=0}^{m} A_{i} (x_{w} - x_{IL})^{i}$$
3-3

Where A_i are the fitting coefficients, *m* is the order of the Redlich-Kister polynomial equation, x_w and x_{IL} are the molar fraction of water and ionic liquid in the aqueous mixtures. Good fits are achieved with m = 3 and the fitting parameters are reported in Table A.2 (See Appendix A). The excess molar volume of $[C_4mim][BF_4]$ aqueous solutions with water content in the range 0-5 wt% (0-0.4 molar fraction) can be predicted using these parameters. Figure 3.6A shows that the values of excess molar volume are positive at all temperatures and over the solution compositions studied. The values of \underline{V}^{ex} are expected to eventually decrease for compositions $x_w > 0.4$ as reported for other ionic liquids mixture systems.^{4, 6, 12}



Figure 3.5: The experimental densities of $[C_4mim][BF_4]$ aqueous solution at different water contents compared with calculated densities based on ideal mixing theory. The solid/dash lines represent the linear correlations between experimental/calculated density and water mass fraction.

The measurement of positive excess molar volumes is consistent with the reported structure of isolated water molecules hydrogen bonding to two anions via hydrogen bonds for trace water in ionic liquid as reported by simulation²² and experiments.²⁴ The intercalation of water into the ionic bonds in the ionic liquid is expected to expand the ionic liquid. Further evidence for this can be found in the work of Ries *et al.*,⁴ whose study of the conductivity and infrared spectra (IR) of $[C_4mim]BF_4]$ water solutions at room temperature shows that the solution conductivities increase due to the weakening of the intermolecular interaction between the anionic and cationic species of $[C_4mim][BF_4]$ upon the addition of water. The
magnitude of \underline{V}^{ex} increases with increasing temperature, as shown in Figure 3.6B. This increase is expected due to the further weakening of the ionic interactions at increasing temperatures. Our data show that the \underline{V}^{ex} of mixture solutions with higher water concentrations are more strongly affected by temperature.



Figure 3.6: Excess molar volume \underline{V}^{ex} of $[C_4mim][BF_4]$ aqueous solution dependence on water molar fraction x_w (A) and temperature (B). The symbols represent experimental value. The solid lines are fits to the Redlich-Kister equation.

3.3.2.2 Viscosity

The viscosities of the ionic liquid aqueous mixtures are observed to decrease strongly with the addition of water and increase in temperature as shown in Figure 3.7A. Importantly, the viscosity decreases significantly with the addition of small amount of water. The viscosity dependence on water molar fraction shown in Figure 3.7A can be correlated by an exponential function:

$$\eta = \eta_0 \exp(-x_w / a) \tag{3-4}$$

where η is the viscosity of the aqueous solution at certain temperature, η_0 is the viscosity of pure ionic liquid at the same temperature, x_w is the molar fraction of water and *a* is a fitting constant reported in Table A.3 (See Appendix A). The sensitivity to water addition increases with temperature, as evident by the values of the fitting constant *a*, which increases from 0.24 at 293.15 K to 0.54 at 373.15 K.



Figure 3.7: The viscosities of [C₄mim][BF₄] aqueous solution at different water molar fraction (A) and temperatures (B). Solid lines represent an empirical correlation (A) and the Vogel-Fulcher-Tammann (VFT) equation correlation (B), respectively.

The mixture viscosities decrease with temperature as shown in Figure 3.7B. This behavior can be accurately described by the Vogel-Fulcher-Tammann (VFT) equation:

$$\eta = AT^{1/2} \exp[B/(T - T_0)]$$
 3-5

where A, B and T_0 are the fitting parameters. T_0 is the ideal transition temperature, which is often associated with but lower than the glass transition temperature,¹⁴ and the glass transition temperature is reported to be 193.6 K for the pure ionic liquid [C₄mim][BF₄] as determined using differential scanning calorimetry.³⁹ Domanska *et* al.⁴⁰ and Ferry⁴¹ proposed using the ideal transition temperature for T_0 , for glassforming ionic liquids and their mixtures with organic solvents. Ciocirlan et al.¹⁴ set the value of T_0 to be 170 K for the [C₄mim][BF₄]/ dimethyl sulfoxide (DMSO) and $[C_4 mim][BF_4]$ ethylene glycol (EG) mixtures to correlate the temperature dependence of the viscosity. For the ionic liquid aqueous solution system studied here, the ideal transition temperature T_0 is set as one fitting parameter and determined empirically with other two parameters A and B for the aqueous $[C_4mim][BF_4]$ mixtures investigated. Equation 3-5 provides a successful correlation of the viscosity measurements, as shown in Figure 3.7B. The fitting parameters A, B and T_0 are listed in Table 3.4. The ideal transition temperature T_0 decreases with the addition of water as shown in Table 3.4. The obtained ideal transition temperatures T_0 at various water contents are consistent (slightly smaller) with corresponding glass transition temperatures of [C₄mim][BF₄]-water solutions measured by differential thermal analysis (DTA).⁴² Moreover, Yoshimura⁴² found the measured glass transition temperature of [C₄mim][BF₄]-water solutions decreases with water content in the saltrich regime until microphase separation. In the water-rich regime, the formation of water clusters has led to an increased glass transition temperature. Previous study using neutron scattering measurements have illustrated that the composition regime studied herein have no microphase separation.²¹ Thus, the ideal transition temperatures obtained from the VFT fitting are in agreement with the glass transition

85

temperature measured by DTA in the same salt-rich regime. I note that the VFT equation provides a better correlation of the viscosity data than a simple Arrhenius plot (see Appendix A - Figure A.2), which was suggested by Haghani *et al.*, who modeled an extensive variety of binary ionic liquid mixtures.⁴³

Water content	A	В	T_{0}
wt %	$mPa.s.K^{-0.5}$	K	K
0	0.0106	734.0	181.8
0.28	0.0097	760.4	176.8
0.52	0.0089	780.1	173.1
1.01	0.0095	757.0	171.3
2.03	0.0102	716.1	169.0
3.01	0.0099	707.0	164.9
5.00	0.0096	675.7	160.4

Table 3.4: Fitting parameters (A, B and T_0) of the VFT equation for [C₄mim][BF₄] aqueous solutions with various compositions.

Although there is no established theory for the viscosity of ideal binary mixtures, there have been many models proposed and experimental studies directed towards validating these models (see⁴⁴⁻⁴⁵ for example). The viscosity of ideal mixtures can be calculated according to the Arrhenius ideal solution model,⁴⁶⁻⁴⁸ which is equivalent to the fluidity additivity rule:

$$\log(\eta_{ideal}) = x_{IL} \log(\eta_{IL}) + x_w \log(\eta_w)$$
3-6

where η_{ideal} , η_{IL} and η_w are the viscosities of the ideal mixture, pure ionic liquid and water, respectively. To quantify the viscosity deviation from ideal behavior for the ionic liquid 1-hexyl-3-methylimidazolium bis(trifluoromethylsulfonyl)amide

([Hmim][Tf₂N]) and 1-octene mixtures, Ahosseini and Scurto¹⁹ introduced the "excess viscosity" as the difference between the mixture viscosity and the ideal viscosity. I adopt this nomenclature convention in our study of the [C₄mim][BF₄] and water mixtures. Here, the excess log viscosity $(log\eta)^{ex}$ is defined as the difference between the logarithm of the mixture viscosity and ideal mixture viscosity defined by Arrhenius:⁴⁶⁻⁴⁸

$$(\log \eta)^{ex} = \log(\eta) - \log(\eta_{ideal})$$

The ideal mixture viscosities calculated from Equation 3-6 are compared with experimental results in Figure 3.8A and the excess viscosities are plotted vs water molar fraction in Figure 3.8B. Positive deviations from ideal mixing are an indication of strong interactions between the water and the $[C_4mim][BF_4]$, such that water does not exhibit its bulk viscosity as anticipated in the Arrhenius ideal mixing theory.



Figure 3.8: (A) Viscosities of $[C_4mim][BF_4]$ aqueous solutions as a function of water mole fraction (solid lines and symbols) compared with calculated viscosities based on ideal mixing theory (-x-). (B) Excess log viscosity, $(log\eta)^{ex}$, vs molar fraction of water x_w for $[C_4mim][BF_4]$ aqueous mixtures as a function of temperature.

3.3.2.3 Density-Viscosity Correlation

Doolittle *et al.*⁴⁹ related the dynamic viscosity of simple liquids with the free volume in solution \underline{V}_{f} , with an empirical equation that has successfully modeled the viscosity of ordinary liquids:

$$\log \eta = \log A + B \frac{\underline{V} - \underline{V}_f}{\underline{V}_f}$$
 3-8

where *A* and *B* are empirical constants, \underline{V} and \underline{V}_f are the molar volume and molar free volume of the liquid. Based on the Doolittle equation, I propose the following, semi-empirical relationship between the excess viscosity and the excess volume:

$$(\log \eta)^{ex} = C\underline{V}^{ex}$$

where *C* is an empirical constant, which is taken to be independent of temperature and composition.

The physical basis for this semi-empirical relationship lies in the expectation that the excess molar volume arises from the strong interaction between the water and the ionic liquid at lower water content. Thus, the positive excess molar volume reflects the binding of water into the ionic domains of the IL. Hence, the viscosity of the water in this bound state is significantly greater as compared to water in the freestate. According to the Doolittle application of free volume theory, this excess molar volume acts to increase the viscosity above what would be expected for ideal mixing. Note that a possible relationship between the excess mixing properties, including the viscosity and molar volume, and free volume theory has been explored for binary liquid mixtures for over forty years,⁴⁸ but a formal relationship of the form of Equation 3-9 has not been derived or validated. Therefore, I propose it as a semi-empirical relationship to be tested against experimental data.

For the $[C_4 \text{mim}][BF_4]$ aqueous solutions, it is found that excess log viscosity $(log\eta)^{ex}$ is indeed linearly correlated with the excess molar volume \underline{V}^{ex} ($\mathbb{R}^2 = 0.996$) over the range of compositions and temperatures studied as shown in Figure 3.9. This new correlation provides insight into the significant effect of water on the viscosity of ionic liquids. The relative residues of the excess log viscosity $(log\eta)^{ex}$ from the correlation fitting for different temperatures and various water compositions are shown in Figure 3.10. The relative residues in Figure 3.10 indicate that significant deviations occur at low water content and at lower temperature, with the maximum relative deviation at water molar fraction $x_w = 0.06$ composition with temperature 293.15 K. Correlation improves with increasing water content and temperature.

89



Figure 3.9: The linear correlation between excess log viscosity, $(log\eta)^{ex}$, and excess molar volume, \underline{V}^{ex} , of [C₄mim][BF₄] aqueous solution at various temperatures.



Figure 3.10: The relative residues of excess log viscosity $(log\eta)^{ex}$ from correlation Equation 3-9 of [C₄mim][BF₄] aqueous solutions at various temperatures and compositions.

Finally, I explore whether this semi-empirical correlation is applicable to aqueous solutions of other ionic liquids. Ionic liquids of similar structures, such as 1-ethyl-3-methylimidazolium tetrafluoroborate ($[C_2mim][BF_4]$) aqueous solutions ⁵⁰ are investigated. Their excess molar volume and excess log viscosity are positive for the complete composition, and a linear correlation is evident ($R^2 = 0.809$) for salt-rich regime up to water molar fraction 40% (Appendix A, Figure A.3A). However, this linear correlation does not go through the "zero" point, which suggests an inconsistency in the experimentally reported pure component properties. Indeed, the reported density values for [C_2mim][BF_4] are found to be systemically higher than

those reported by others,⁵¹⁻⁵⁴ which could lead to this offset. For the other imidazolium-based ionic liquids with the common anion [BF₄], such as 1-propyl-2,3dimethylimidazolium tetrafluoroborate ([Pdmim][BF₄]),⁵⁵ the excess molar volumes and excess log viscosity are positive for the entire water compositions, and a good linear correlation ($R^2 = 0.959$) is observed for the salt-rich regime (water molar fraction up to 30%) as shown in Figure 3.11A. The density and viscosity of pyridinium-based ionic liquid aqueous solutions have been reported by Mokhtarani et $al.^{56}$ In contrast to [C₄mim][BF₄], the excess log viscosity of 1-butylpyridinium tetrafluoroborate ([BuPy][BF₄]) aqueous solutions are negative for certain water compositions and temperatures in the salt rich regime and no linear correlation between excess molar volume and excess log viscosity is observed for this system (See Appendix A - Figure A.3). However, for the 1-octylpyridinium tetrafluoroborate ([OcPy][BF₄]), which has a longer alkyl chain, the excess log viscosity is positive for the entire water compositions, and as shown in Figure 3.11B, a good linear correlation between excess molar volume and excess log viscosity of aqueous solution with water molar fraction up to 35% is observed with $R^2 = 0.955$.



Figure 3.11: The semi-empirical linear correlation between excess log viscosity, $(log\eta)^{ex}$, and excess molar volume, \underline{V}^{ex} , of [Pdmim][BF₄] (A) and [OcPy][BF₄] (B) aqueous solutions in salt-rich regime respectively at various temperatures.

Imidazolium-based ionic liquids aqueous systems with anions other than $[BF_4]^$ have also been studied. The density and viscosity values for mixtures of water and three ionic liquids: 1-ethyl-3-methylimidazolium ethylsulfate ($[C_2mim][EtSO_4]$), 1ethyl-3-methylimidazolium trifluoroacetate ($[C_2mim][TFA]$), and 1-ethyl-3methylimidazoliumn trifluoromethanesulfonate ($[C_2mim][OTf]$) are reported by Rodriguez *et al.*¹⁵ Their excess molar volumes are negative for the entire compositions (*e.g.* [C_2mim][TFA] aqueous solutions) or partial compositions (*e.g.* [C_2mim][EtSO₄] and [C_2mim][OTf]). No obvious linear correlations are observed for these systems (See Appendix A – Figure A.3). The same for the widely used protic ionic liquid ethylammonium nitrate (EAN) water system.⁵⁷ The excess molar volumes change from positive to negative with increasing water content and the semi-empirical linear correlation between excess molar volume and excess log viscosity cannot be applied to this system.

Table 3.5:	Summary of validation of semi-empirical linear correlation between
	excess log viscosity and excess molar volume of various ionic liquids.

Ionic liquid	Cation structure	Anion structure	$(log\eta)^{ex}$	$\underline{V}^{ex}*$	Linear correlation	Ref.
1-butyl-3- methylimidazolium tetrafluoroborate ([C ₄ mim][BF ₄])	H ₃ C C ₄ H ₉	BF4	Positive	Positive	Yes	
1-ethyl-3- methylimidazolium tetrafluoroborate ([C ₂ mim][BF ₄])	H ₃ C N ⁺ N ⁻ C ₂ H ₅	BF4	Positive	Positive	Yes [§]	50
1-propyl-2,3- dimethylimidazolium tetrafluoroborate ([Pdmim][BF ₄])	$H_3C_N^+$	BF4	Positive	Positive	Yes	55
1-octylpyridinium tetrafluoroborate ([OcPy][BF ₄])	C ₈ H ₁₇	BF4	Positive	Positive	Yes	56
1-butylpyridinium tetrafluoroborate ([BuPy][BF ₄])	√N ⁺ C ₄ H ₉	BF4	Mostly positive	Positive	No	56
1-ethyl-3- methylimidazoliumn trifluoromethanesulfonate ([C ₂ mim][OTf])	H ₃ C N ⁺ N ⁻ C ₂ H ₅	$F_3C \stackrel{O}{\substack{\parallel\\ \\ \parallel\\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\ \\$	Positive	Mostly positive	No	15



* The positive or negative values of excess log viscosity, $(log\eta)^{ex}$, and excess molar volume, \underline{V}^{ex} of entire water compositions. "Positive" indicates the values are positive for the entire water compositions, and "Mostly positive" indicates the values are positive for most of the water concentration and only for small part of water compositions are negative, "Negative" and "Mostly negative" can be deduced by analogy.

§ Note that the linear correlation does not go through the origin as expected.

The survey of the available literature and our research, as summarized in Table 3.5, shows that this semi-empirical correlation is valid for $[C_4mim][BF_4]$ aqueous solutions and other aqueous solutions of ionic liquids containing the $[BF_4]^-$ anion. The cations include another imidazolium ([Pdmim][BF₄]) and a pyridinium ([OcPy][BF₄]). Note this semi-empirical linear correlation applies only for aqueous ionic liquid solutions in the salt-rich regime ($x_w < 50 \text{ mol}\%$). It is known that the interactions between ionic liquids and water lead to microphase separation and phase inversion for higher water concentrations,^{22, 24-25} which would suggest the model is not applicable.

As summarized in Table 3.5, both the excess molar volume and excess log viscosity must be positive over the entire water composition range of interest for this correlation to apply. The excess properties are determined by the specific interactions

between ions and water molecules.¹⁷ It is reported that anions have the dominant effect on the ionic liquid-water interactions, while water hydrogen bonding with the cations is a secondary effect.¹⁷ The anion $[OTf]^{-}$ (*i.e.*, CF₃SO₃⁻) has a slightly larger water solubility than $[BF_4]^-$ group, ⁵⁸⁻⁵⁹ and it can form strong acid CF₃SO₃H with the present of proton and react with water molecules to form ion-ligand complex $CF_3SO_3^{-1}$ • H_2O or $CF_3SO_3^-$ • $2H_2O_3^{-60}$ The strength of H-bonding or water solubility increases in the order $[BF_4]^- < [OTf]^-$, $[EtSO_4]^- < [NO_3]^- < [TFA]^{-.59}$ The water interactions with $[OTf]^{-}$, $[EtSO_4]^{-}$, $[TFA]^{-}$ and $[NO_3]^{-}$ differ from that with $[BF_4]^{-}$, which results in different excess properties that are not captured by our semi-empirical linear correlation. The specific cation plays a secondary effect on the ionic liquids and water molecules interactions, and the cations [Pdmim]⁺, and [OcPy]⁺ have similar hydrophobicity as $[C_4 mim]^+$, but $[BuPy]^+$, and $[C_2 mim]^+$ are less hydrophobic.⁵⁸ Consequently, this semi-empirical linear correlation is not valid for [BuPy][BF₄] aqueous solutions due to differing cation-water interactions. This empirical survey of the available literature provides hints that the development of a molecular thermodynamic theory for water interactions with ionic liquids in the salt rich regime may be able to predict the broad range of composition specific nonideal behaviors reported to date.

3.3.3 Microstructure of Ionic Liquid Aqueous Solutions

3.3.3.1 Ionic Liquid Micellar Formation

SAXS measurement was taken to evaluate the micellar formation of ionic liquid $[C_4mim][BF_4]$ in water solutions. The absolute intensity of SAXS spectra (scattering intensity, *I*, versus scattering wavevector *q*) of $[C_4mim][BF_4]$ aqueous

solutions with different ionic liquid concentrations at 25 °C are shown in Figure 3.12. For low ionic liquid concentration solutions from 0.2 mol/L to 0.6 mol/L, no significant structure peak is observed over the *q* range studied. The addition of ionic liquid [C₄mim][BF₄] uniformly increases the overall spectra intensity as expected, due to comparatively higher incoherent scattering of [C₄mim][BF₄]. However, starting from 0.8 mol/L, a slight increase in the intensity at low *q* range is observed indicating the formation of ionic liquid micellar structure, and the intensity peak become more evident due to the growth of micellar structure as the ionic liquid concentration increases. Because of the amphiphilic structure property of ionic liquid cation [C₄mim]⁺, a structure peak which indicating the formation of micellar structure is expected at higher ionic liquid concentration for ionic liquid aqueous solutions. Thus, the CMC of ionic liquid [C₄mim][BF₄] in H₂O is determined to be 0.8 mol/L by SAXS measurement, which is consistent with previous micellar formation study of [C₄mim][BF₄] aqueous solutions.^{26, 30}



Figure 3.12: SAXS intensity profiles of [C₄mim][BF₄] aqueous solutions (in H₂O) with various ionic liquid contents at 25 °C.

In order to further understand the micellar structure of ionic liquid aqueous solutions, quantitatively accurate model fits of the measured SAXS intensity are used to obtain the shape and size of micellar structure of ionic liquid. Monodispersed sphere form factor model and monodispersed ellipsoid form factor model are used to evaluate the growth of ionic liquid micelle with addition of ionic liquid, and their model fitting results are shown in Figure 3.13. The average size of micellar structure extracted from both of the sphere and ellipsoid form factor models are listed in Table 3.6. As shown in Figure 3.13, for aqueous solution of 0.6 mol/L, there is no scattering

intensity peak at low q value indicating no micellar structure formation, which is confirmed by the negative value of micelle volume fraction extracted from model fitting in Table 3.6. Both of the sphere and ellipsoid model fits are able to capture the intensity peak at low q value for 0.8 mol/L solution as shown in Figure 3.13. The same R_a and R_b values were obtained for the ellipsoid model (same as the radius for the sphere model). This result indicates that the ionic liquid $[C_4mim][BF_4]$ forms spherical micelles with radius of 1.8 nm, which is close to the molecular size of ionic liquid. However, for higher ionic liquid concentration >1.0 mol/L, the sphere model no longer captures the scattering profile and the deviation from a scattering intensity profile increases with ionic liquid concentration. The change of ionic liquid micellar structure shape from sphere to ellipsoid is expected as micelle grows with higher micellar concentration at high ionic liquid concentrations. The growth of ionic liquid micelle with ionic liquid concentration can be illustrated by the volume fraction and size of micelle determined from the ellipsoid model fitting shown in Table 3.6. The volume fraction of ionic liquid micelle in solutions increases, as expected, due to more ionic liquid involving in the formation of micelle at higher ionic liquid concentration. The size of the micellar structure increases with ionic liquid concentration, especially for the R_a direction, size along the R_a direction doubled with ionic liquid concentration increased from 0.8 mol/L to 2.0 mol/L, leaving the size along the R_b direction nearly unchanged. Thus, ionic liquid spherical micelles form initially at 0.8 mol/L and grow along the Ra direction resulting in a "rod" like shape micelle with increasing of ionic liquid concentration.

99



Figure 3.13: The comparisons of sphere and ellipsoid form factor model fittings for SAXS intensity profiles of mixtures of [C₄mim][BF₄] in H₂O at 25 °C with ionic liquid concentrations of 0.6 mol/L to 2.0 mol/L.

	Sphere model fitting		Ellipsoid model fitting		
[C ₄ mim][BF ₄] concentration	Volume fraction	R	Volume fraction	R_a	R_b
mol/L		nm		nm	
0.6	-1.7×10 ⁻⁴		-2.3×10 ⁻⁴		
0.8	8.4×10^{-5}	1.8	8.4×10 ⁻⁵	1.8	1.8
1.0	0.7×10^{-3}	1.6	1.0×10^{-3}	2.3	1.1
1.2	1.3×10^{-3}	1.6	2.2×10^{-3}	2.5	1.0
1.5	2.6×10^{-3}	1.7	4.7×10^{-3}	2.9	1.0
2.0	4.3×10 ⁻³	1.8	7.0×10 ⁻³	3.3	1.2

Table 3.6:Summary of ionic liquid micelle size of $[C_4mim][B_F]$ -H2O solutions
extracted from sphere and ellipsoid form factor model fittings.

3.3.3.2 Water Cluster Formation

Absolute SANS spectra (scattering intensity, *I*, versus scattering wavevector *q*) for varying water compositions at 25 °C are shown in Figure 3.14. The pure ionic liquid and [C₄mim][BF₄]/D₂O solutions with water content up to 11 wt% (0.60 D₂O mole fraction) show no significant structure over the *q* range studied. A slight increase is observed towards higher *q* values consistent with the expected molecular structure in the ionic liquid⁶¹ and the addition of D₂O uniformly lowers the overall spectra, as expected, because D₂O has comparatively lower incoherent scattering. For aqueous solution of 15 wt% D₂O content (0.67 mole fraction), there is a slight decrease in intensity for higher *q* (*q* = 0.4~0.5 Å⁻¹), which is indicative of water association. However, this intensity change is not significant enough to enable reliable determination of water association from model fitting. A significant change in SANS intensity is evident starting at 20 wt% D₂O content (0.74 D₂O mole fraction), whereby the relative growth of scattering intensity at low *q* signals the onset of microstructure formation in the mixtures on length-scales greater than the molecular size of the

constituents. Further addition of water leads to large increases in scattering intensity at lower q values, consistent with the formation of larger, nanoscale structures.

These observations help explain previous experimental results²⁴⁻²⁷ and verify simulations,²²⁻²³ which can be analyzed as follows. For low water concentrations, water molecules are incorporated into the ionic liquid cation-anion network without significantly disrupting the structure.²² Katayanagi *et.al.*⁶² measured the excess chemical potential, μ_i^E , the excess partial molar enthalpy, H_i^E , and excess partial molar entropies, S_i^E in [C₄mim][BF₄]-H₂O system at 25 °C. They found that H_{IL}^E and S_{IL}^{E} of [C₄mim][BF₄] are almost zero with the net μ_{IL}^{E} also zero in the IL-rich region, $x_{IL} > 0.5$ -0.6. These results reveal that the structure of the ionic liquid is not disrupted by the addition of small amounts of water. Furthermore, homogeneous mixing was proposed in the [C₄mim][BF₄]-H₂O mixture in the salt mole fraction range of 0.4 - 1 by Almasy³⁰ and confirmed using concentration fluctuations from vapor pressure measurements.³⁰ However, for water addition beyond the IL-rich boundary (around $x_{II} = 0.4$), the ions are fully hydrated and excess water molecules aggregate and form distinct water clusters. The mixture phase transforms from IL-rich to water-rich mixture beyond this concentration.^{22, 30} Water aggregation and cluster formation were also observed by MD simulations.²²⁻²³ Our SANS scattering intensity profiles of 20 wt% water concentrations and above confirm the formation of distinct nanostructures. The relatively high molar water concentration required for nanocluster formation is consistent with the propensity of water to associate with the ionic groups in the IL.



Figure 3.14: SANS profiles of mixtures of $[C_4mim][BF_4]$ and D_2O at 25 °C.



Figure 3.15: SANS profiles and model fitting for mixtures of [C₄mim][BF₄] and D₂O at 25 °C for higher D₂O concentrations.

Quantitatively accurate model fits of the measured SANS intensity are shown in Figure 3.15. The radius and volume fraction, $\phi_{cluster}$, of water clusters are determined through model fitting, and these parameters are summarized in Table 3.7. The measured scattering intensities are fitted to a monodisperse, sphere form factor, with the water cluster size, $r_{cluster}$, volume fraction, $\phi_{cluster}$, and scattering length density of solvent, SLD_{solvent}, allowed to vary within constraints. In the fitting, the volume fraction of the D₂O clusters is restricted by mass balance that accounts for the fraction of the D₂O that is dissolved in molecular form in the ionic liquid, such that only the excess D₂O is available to forms clusters. The scattering length density of the solvent is constrained within the range of the SLD of pure ionic liquid [C₄mim][BF₄]. In the fitting procedure, the D_2O mass balance is enforced to select a unique fit. From these fits, I extract the average D_2O cluster size and volume fraction as shown in Table 3.7. In this manner, SANS provides a method to quantitatively measure the molecularly dissolved D_2O and the amount that forms distinct nanoclusters, along with the size of these clusters.

Table 3.7: Summary of D₂O water distribution in water dissolved state and microphase separated state for mixtures of [C₄mim][BF₄] and D₂O for water contents above 20 wt% at 25 °C.

Properties	IL+20 wt%	IL+22 wt%	IL+25 wt%	IL+27 wt%	IL+30 wt%
$x_w (total)^a$	0.74	0.76	0.79	0.81	0.83
$\phi_{_{\!W}}$ (total) ^a	0.21	0.24	0.27	0.29	0.32
$\phi_{cluster}$ a	0.043	0.045	0.048	0.053	0.063
$\mathbf{r}_{cluster}(\mathrm{\AA})$	5.5	6.4	6.8	8.2	8.9
ϕ^s_w (in dissolved state) ^b	0.18	0.20	0.23	0.25	0.28
SLD _{dissolved phase}	$2.29 imes 10^{-6}$	$2.40 imes 10^{-6}$	$2.54\times10^{\text{-}6}$	2.66×10^{-6}	$2.79 imes 10^{-6}$
x_w^s (in dissolved state) ^b	0.69	0.72	0.76	0.78	0.80
x_w^s / x_{IL} (in dissolved state) ^b	2.3	2.6	3.1	3.5	4.0

^a x_w (total), ϕ_w (total), $\phi_{cluster}$, and ϕ_w^s are the molar fraction or volume fraction of total water, water cluster or dissolved water in the system.

^b ϕ_w^s (*in dissolved state*), x_w^s (*in dissolved state*), and x_w^s / x_{IL} (*in dissolved state*) are the water and ionic liquid distribution in the molecularly dissolved state



Figure 3.16: The volume fraction and size of D_2O clusters as a function of the overall D_2O weight fraction for $[C_4mim][BF_4]$ and D_2O mixtures at 25 °C.

The D₂O cluster size, *r*, and volume fraction, ϕ , extracted from SANS fitting are plotted in Figure 3.16. The D₂O cluster size and volume fraction both increase with overall D₂O mole fraction for D₂O content of 20 wt% (0.74 D₂O mole fraction) and above, and this behavior is in reasonable quantitative agreement with molecular dynamic simulations.²² Indeed, the simulations of Liu *et.al.*²² suggest water nanoclusters with a well-defined structure, as opposed to weak aggregates, only exist for water mole fractions of ~70% and above, and that the aggregation number increases with increasing water content until the water percolates the material. Furthermore, at 70 mole% the average aggregation number is ~160 water molecules, which corresponds to a droplet of water of approximately 10 Å in radius. Our experimental observations show well-defined water clusters above 70 mole% that are about half the average radius as predicted. These clusters have properties consistent with those of inverse microemulsions, but are novel in that they are present in a binary mixture, consistent with the surfactant-like properties of ILs. Some of the deviation in the simulation prediction may be due to the simulations showing a very broad size distribution that approaches the limited size of the simulation box, or may be specific to isotope effects. Nevertheless, the trends in the simulations are in good qualitative agreement with our experimental observations.



Figure 3.17: Mole fraction of dissolved D₂O and the molar ratio of dissolved D₂O to ionic liquid as a function of total D₂O mole fraction for mixtures of [C₄mim][BF₄] and D₂O at 25 °C.

SANS measurements enable measurement of the distribution of D₂O between being molecularly dissolved and microphase separated in the mixtures with [C₄mim][BF₄]. These distributions are summarized in Table 3.7 and plotted in Figure 3.17 as mole fractions and mole ratios. The volume fraction of D₂O cluster, $\phi_{cluster}$, shown in Table 3.7 was obtained from SANS model fitting, the constraint of mass balance for the D₂O and Equation 3-1. The mole fraction water distributions were calculated based on the volume fraction of water in the molecular dissolved state and cluster microphase separated state. As seen in Table 3.7, the volume fraction of dissolved D₂O, ϕ_w^s , increases slightly with added D₂O, along with the growth of D₂O clusters. This indicates that D₂O partitions between the molecularly dissolved state and the microphase separated state, and both of these increase nearly linearly with added D₂O, as evident in Figure 3.17.

Returning to the lower D_2O concentrations, the data shows that D_2O molecularly dissolves in the ionic liquid. Complementary measurements of the viscosity and density⁴⁻⁶ of mixtures show significant non-idealities, which is consistent with a specific interaction of the water with the ionic groups. Such hydration at low water content has been confirmed by experiments²⁴⁻²⁷ and simulations.²²⁻²³ Shirota *et al.*²⁵ investigated the state of molecular association in $[C_nmim][BF_4]-H_2O$ by Raman spectroscopy and showed that the symmetric F-B stretching mode of $[BF_4]^-$ in the $[C_nmim][BF_4]-H_2O$ shifts to higher frequency with increase water concentration, indicating hydrogen bonding between water and the anions. This experimental observation is consistent with molecular dynamic (MD) simulations by Moreno *et al.*.²³ Three dimensional probability distribution of water around $[C_4mim]^+$ cation, as simulated through MD by Liu *et al.*,²² indicates interactions between water and the cation as well. However, due to the asymmetry and larger size of the cation, the correlation between the cation and water is found to be weaker than that between water and the anion.²² This was also confirmed by experiment as well as predicted using UNIFAC and COSMO-RS models for gas solubility in ionic liquids.^{3,44} Furthermore, water was found to interact with the anions by bridging between anions. As the $[BF_4]^-$ anion has four fluorine atoms and water has two hydrogens, the coordination ratio is 2:1 for water/IL for full hydration as illustrated in Figure 3.17. As seen in Figure 3.14, water clusters are evident in the SANS spectra for water content above 15 wt% (0.67 mole fraction), consistent with this interpretation. Figure 3.17 shows that this corresponds to a mole ratio of about 2:1 for the molecular dissolved D₂O in the ionic liquid as determined by SANS, which is quantitatively consistent with this molecular interpretation.

Taib *et al.* measured the density of $[C_4mim][BF_4]$ aqueous solutions at 25 °C of full water concentration range and plotted the excess molar volume as a function of water molar fraction in Figure 6 of reference⁶, which is consistent with the earlier data reported by Ries *et al.*⁴ The excess molar volume of $[C_4mim][BF_4]$ aqueous solution is positive and increases with addition of water due to the volume expansion caused by ordered hydrogen bonds between water and anions. However, the excess molar volume reaches a maximum value at water mole fraction of 0.6~0.7, and then decreases monotonically with the addition of more water. This maximum in excess molar volume at 0.6~0.7 water mole fraction is consistent with the phenomenon of water cluster formation as revealed by our SANS measurements. At 0.6~0.7 water mole fraction, water clusters start to form and grow with D₂O addition as seen in Figure 3.17. This added water is in the form of microphase separated water clusters,

109

which is in the form of bulk water and hence, the overall mixture excess molar volume will decrease. The fraction of bulk water (water cluster) continually increases with addition of water, as shown in Table 3.7, resulting in the continued decrease of the apparent excess molar volume for this microphase separated mixture. In summary, our SANS measurements confirm the mechanism whereby water addition to this ionic liquid leads to significant non-idealities in mixture thermophysical and transport properties, *e.g.* mixture density and viscosity, in the salt-rich regime due to this strong association between water and the $[BF_4]^-$ anion.



Figure 3.18: The schematic illustration of structures in mixtures of $[C_4mim][BF_4]$ and D_2O .

With this compilation of literature data, simulations and the new microstructural measurements presented here, a coherent picture emerges for the microphase solution structure in $[C_4 mim][BF_4]$ water mixtures over the entire composition range, as illustrated in Figure 3.18. Water is soluble in the ionic liquid until ~ 2 water molecules per ionic liquid ion pair, whereupon microphase separation of water is observed. The addition of water to ionic liquid [C₄mim][BF₄] above ~80 mole % results in water percolation.¹⁵ Above this concentration, there is a phase inversion and the IL forms aggregates consistent with its amphiphilic character. This conventional cationic surfactant-like behavior has been well studied.^{4, 22, 26-28, 30, 62} At $x_w = 0.925$ SANS³⁰ and vapor pressure measurements³⁰ indicates the most significant micellization of the ionic liquid, in agreement with MD simulation.²² The strong concentration fluctuations suggest that the system is quite close to a de-mixing spinodal at this composition.³⁰ The critical aggregation concentration (CAC) or the critical micellar concentration (CMC) of [C₄mim][BF₄] in this water-rich regime was determined to be $x_w = 0.985$ through vapor pressure,⁶² surface thermal coefficient,²⁷ surface tension,²⁰ SANS,^{26, 30} conductivity,^{4, 26} NMR²⁸ and fluorescence spectroscopy²⁸ measurements. For water concentrations beyond this critical concentration, the ionic liquid dissociates in the water as dilute electrolyte solution.

3.4 Conclusions

The densities and viscosities of binary mixtures of 1-butyl-3-methlimidazolium tetrafluoroborate ($[C_4mim][BF_4]$) with water are reported for the temperature range of 283.15 to 373.15 K over the salt-rich composition regime. Excess molar volumes and non-idealities of the viscosities are calculated and can be accurately correlated, which are provided for reference. A new semi-empirical correlation between the excess

mixture viscosities and the excess molar volumes in the salt-rich regime is identified, which is consistent with free volume theory. This correlation enables predicting the viscosity of ionic liquids with added water directly from density measurements, and is also found to be valid for other ionic liquids aqueous solutions with similar ion structures. A survey of the literature identifies ionic liquids for which this correlation is valid, and provides evidence for the role of specific molecular interactions that govern the observed non-idealities in salt-rich aqueous solutions of ionic liquids.

Micellar formation of ionic liquid $[C_4mim][BF_4]$ in aqueous solution is investigated through SAXS measurements. The CMC of [C4mim][BF4] in H₂O is determined to be 0.8 mol/L, which is consistent with previous literature results.^{26, 30} Spherical micelles form at ionic liquid concentrations above CMC; however, micelles grow to ellipsoid rod shape as ionic liquid concentration increases. A "rod" like micellar structure is observed at higher concentration.

A microphase transition to water nanoclusters resembling an inverse microemulsion is observed in the salt-rich regime of mixtures of the room temperature ionic liquid $[C_4mim][BF_4]$ with the addition of D₂O above ~ 70 mol%. This transition is consistent with previous reports on related ionic liquid mixtures as also deduced from SANS measurements, vapor pressure, NMR, Raman, IR spectroscopy measurement, and dynamic molecular simulation. In the IL-rich region, water molecules are isolated and accommodated in the ionic liquid cation-anion polar network by interacting with anions and cations through hydrogen bonds. Such mixtures are a homogenous solution, with no significant changes in microstructure upon water addition as detected by SANS measurement. Water form clusters when the water/ionic liquid molecular ratio exceeds ~ 2:1, as expected from molecular

112

structure determination. Distinct nanometer-sized water clusters form and grow with water content, and water continues to partition between the dissolved state and microphase separated state. These observations complement measurements of watercontinuous mixtures to complete a picture that displays microphase separation and phase inversion in this binary mixture which shows similarities to that of traditional oil-water-surfactant systems. These microstructure measurements of water dissolution in the salt-rich regime also confirm proposed mechanism for the observed nonidealities in mixture thermophysical and transport properties such as density and viscosity.

REFERENCES

1. Plechkova, N. V.; Seddon, K. R. Applications of ionic liquids in the chemical industry. *Chem. Soc. Rev.* **2008**, *37* (1), 123-150.

2. Keskin, S.; Kayrak-Talay, D.; Akman, U.; Hortacsu, O. A review of ionic liquids towards supercritical fluid applications. *J. Supercrit. Fluids* **2007**, *43* (1), 150-180.

3. COSMO-RS GUI Tutorials_Ionic Liquids. https://www.scm.com/documentation/Tutorials/COSMO-RS/Ionic_Liquids/ (accessed 4/26/2016).

4. Ries, L. A. S.; do Amaral, F. A.; Matos, K.; Martini, E. M. A.; de Souza, M. O.; de Souza, R. F. Evidence of change in the molecular organization of 1-n-butyl-3-methylimidazolium tetrafluoroborate ionic liquid solutions with the addition of water. *Polyhedron* **2008**, *27* (15), 3287-3293.

5. Zhou, Q.; Wang, L. S.; Chen, H. P. Densities and Viscosities of 1-butyl-3methylimidazolium tetrafluoroborate + H2O binary mixtures from (303.15 to 353.15) K. J. Chem. Eng. Data **2006**, *51* (3), 905-908.

6. Taib, M. M.; Murugesan, T. Density, Refractive Index, and Excess Properties of 1-Butyl-3-methylimidazolium Tetrafluoroborate with Water and Monoethanolamine. *J. Chem. Eng. Data* **2012**, *57* (1), 120-126.

7. Rilo, E.; Vila, J.; Pico, J.; Garcia-Garabal, S.; Segade, L.; Varela, L. M.; Cabeza, O. Electrical Conductivity and Viscosity of Aqueous Binary Mixtures of 1-Alkyl-3-methyl Imidazolium Tetrafluoroborate at Four Temperatures. *J. Chem. Eng. Data* **2010**, *55* (2), 639-644.

8. Liu, W.; Cheng, L.; Zhang, Y.; Wang, H.; Yu, M. The physical properties of aqueous solution of room-temperature ionic liquids based on imidazolium: Database and evaluation. *J. Mol. Liq.* **2008**, *140* (1-3), 68-72.

9. Liu, W.; Zhao, T.; Zhang, Y.; Wang, H.; Yu, M. The physical properties of aqueous solutions of the ionic liquid BMIM BF4. *J. Solution Chem.* **2006**, *35* (10), 1337-1346.

10. Ueno, K.; Inaba, A.; Ueki, T.; Kondoh, M.; Watanabe, M. Thermosensitive, soft glassy and structural colored colloidal array in ionic liquid: colloidal glass to gel transition. *Langmuir* **2010**, *26* (23), 18031-18038.

11. Wagner, N. J.; Brady, J. F. Shear thickening in colloidal dispersions. *Physics Today* **2009**, *62* (10), 27-32.

12. Gomez, E.; Gonzalez, B.; Calvar, N.; Tojo, E.; Dominguez, A. Physical properties of pure 1-ethyl-3-methylimidazolium ethylsulfate and its binary mixtures with ethanol and water at several temperatures. *J. Chem. Eng. Data* **2006**, *51* (6), 2096-2102.

13. Gonzalez, B.; Calvar, N.; Gomez, E.; Dominguez, A. Physical properties of the ternary system (ethanol + water+1-butyl-3-methylimidazolium methylsulphate) and its binary mixtures at several temperatures. *J. Chem. Thermodyn.* **2008**, *40* (8), 1274-1281.

14. Ciocirlan, O.; Croitoru, O.; Iulian, O. Densities and Viscosities for Binary Mixtures of 1-Butyl-3-Methylimidazolium Tetrafluoroborate Ionic Liquid with Molecular Solvents. *J. Chem. Eng. Data* **2011**, *56* (4), 1526-1534.

15. Rodriguez, H.; Brennecke, J. F. Temperature and composition dependence of the density and viscosity of binary mixtures of water plus ionic liquid. *J. Chem. Eng. Data* **2006**, *51* (6), 2145-2155.

16. Spickermann, C.; Thar, J.; Lehmann, S. B. C.; Zahn, S.; Hunger, J.; Buchner, R.; Hunt, P. A.; Welton, T.; Kirchner, B. Why are ionic liquid ions mainly associated in water? A Car-Parrinello study of 1-ethyl-3-methyl-imidazolium chloride water mixture. *J. Chem. Phys.* **2008**, *129* (10).

17. Seddon, K. R.; Stark, A.; Torres, M. J. Influence of chloride, water, and organic solvents on the physical properties of ionic liquids. *Pure Appl. Chem.* **2000**, *72* (12), 2275-2287.

18. Menon, S. V. G.; Manohar, C.; Rao, K. S. A new interpretation of the sticky hard sphere model. *The Journal of Chemical Physics* **1991**, *95* (12), 9186-9190.

19. Meot-Ner, M. The ionic hydrogen bond. *Chem. Rev.* **2005**, *105* (1), 213-284.

20. Guibert, C.; Dupuis, V.; Fresnais, J.; Peyre, V. Controlling nanoparticles dispersion in ionic liquids by tuning the pH. *J. Colloid Interface Sci.* **2015**, *454*, 105-111.

21. Bergenholtz, J.; Brady, J.; Vicic, M. The non-Newtonian rheology of dilute colloidal suspensions. *J. Fluid Mech.* **2002**, *456*, 239-275.

22. Zhong, X.; Fan, Z.; Liu, Z.; Cao, D. Local structure evolution and its connection to thermodynamic and transport properties of 1-butyl-3-methylimidazolium tetrafluoroborate and water mixtures by molecular dynamics simulations. *J. Phys. Chem. B* **2012**, *116* (10), 3249-3263.

23. Moreno, M.; Castiglione, F.; Mele, A.; Pasqui, C.; Raos, G. Interaction of Water with the Model Ionic Liquid [bmim][BF4]: Molecular Dynamics Simulations and Comparison with NMR Data. *J. Phys. Chem. B* **2008**, *112* (26), 7826-7836.

24. López-Pastor, M.; Ayora-Cañada, M. J.; Valcárcel, M.; Lendl, B. Association of Methanol and Water in Ionic Liquids Elucidated by Infrared Spectroscopy Using Two-Dimensional Correlation and Multivariate Curve Resolution. *J. Phys. Chem. B* **2006**, *110* (22), 10896-10902.

25. Masaki, T.; Nishikawa, K.; Shirota, H. Microscopic Study of Ionic Liquid–H2O Systems: Alkyl-Group Dependence of 1-Alkyl-3-Methylimidazolium Cation. *J. Phys. Chem. B* **2010**, *114* (19), 6323-6331.

26. Bowers, J.; Butts, C. P.; Martin, P. J.; Vergara-Gutierrez, M. C.; Heenan, R. K. Aggregation behavior of aqueous solutions of ionic liquids. *Langmuir* **2004**, *20* (6), 2191-2198.

27. Malham, I. B.; Letellier, P.; Turmine, M. Evidence of A Phase Transition in Water-1-butyl-3-methylimidazolium tetrafluoroborate and Water-1-butyl-2, 3-dimethylimidazolium tetrafluoroborate mixtures at 298 K: Determination of the Surface Thermal Coefficient, b T, P. J. Phys. Chem. B 2006, 110 (29), 14212-14214.

28. Singh, T.; Kumar, A. Aggregation Behavior of Ionic Liquids in Aqueous Solutions: Effect of Alkyl Chain Length, Cations, and Anions. *J. Phys. Chem. B* **2007**, *111* (27), 7843-7851.

29. Sarkar, A.; Ali, M.; Baker, G. A.; Tetin, S. Y.; Ruan, Q.; Pandey, S. Multiprobe Spectroscopic Investigation of Molecular-level Behavior within Aqueous 1-Butyl-3-methylimidazolium Tetrafluoroborate. *J. Phys. Chem. B* **2009**, *113* (10), 3088-3098.

30. Almasy, L.; Turmine, M.; Perera, A. Structure of aqueous solutions of ionic liquid 1-butyl-3-methylimidazolium tetrafluoroborate by small-angle neutron scattering. *J. Phys. Chem. B* **2008**, *112* (8), 2382-2387.

31. Singh, T.; Kumar, A. Self-aggregation of ionic liquids in aqueous media: A thermodynamic study. *Colloids Surf.*, A **2008**, *318* (1-3), 263-268.

32. Rolling Ball Viscometer Standard Operation Procedure. In *Rolling Ball Viscometer Standard Operation Procedure*; Anton Paar, 2015.

33. Kline, S. R. Reduction and analysis of SANS and USANS data using IGOR Pro. J. Appl. Crystallogr. **2006**, *39* (6), 895-900.

34. Archer, D. G.; Widegren, J. A.; Kirklin, D. R.; Magee, J. W. Enthalpy of Solution of 1-Octyl-3-methylimidazolium Tetrafluoroborate in Water and in Aqueous Sodium Fluoride. *J. Chem. Eng. Data* **2005**, *50* (4), 1484-1491.

35. Freire, M. G.; Neves, C. M. S. S.; Marrucho, I. M.; Coutinho, J. A. P.; Fernandes, A. M. Hydrolysis of Tetrafluoroborate and Hexafluorophosphate Counter Ions in Imidazolium-Based Ionic Liquids. *J. Phys. Chem. A* **2010**, *114* (11), 3744-3749.

36. Rilo, E.; Pico, J.; García-Garabal, S.; Varela, L. M.; Cabeza, O. Density and surface tension in binary mixtures of CnMIM-BF4 ionic liquids with water and ethanol. *Fluid Phase Equilib.* **2009**, *285* (1–2), 83-89.

37. Wang, S.; Jacquemin, J.; Husson, P.; Hardacre, C.; Gomes, M. F. C. Liquidliquid miscibility and volumetric properties of aqueous solutions of ionic liquids as a function of temperature. *J. Chem. Thermodyn.* **2009**, *41* (11), 1206-1214.

38. Zhu, A.; Wang, J.; Han, L.; Fan, M. Measurements and correlation of viscosities and conductivities for the mixtures of imidazolium ionic liquids with molecular solutes. *Chem. Eng. J.* **2009**, *147* (1), 27-35.

39. Qi, M.; Wu, G.; Li, Q.; Luo, Y. γ-Radiation effect on ionic liquid [bmim][BF4]. *Radiat. Phys. Chem.* **2008**, *77* (7), 877-883.

40. Domanska, U.; Rekawek, A.; Marciniak, A. Solubility of 1-alkyl-3ethylimidazolium-based ionic liquids in water and 1-octanol. *J. Chem. Eng. Data* **2008**, *53* (5), 1126-1132.

41. Ferry, J. D. Viscoelastic properties of polymers; John Wiley & Sons1980.

42. Chiew, Y. C.; Glandt, E. D. Percolation behaviour of permeable and of adhesive spheres. J. Phys. A: Math. Gen. **1983**, 16 (11), 2599.

43. Haghani, A.; Saeva, D. R.; Iloukhani, H.; Hoffmann, M. M. Comparing Composition- and Temperature-Dependent Viscosities of Binary Systems Involving Ionic Liquids. *J. Chem. Eng. Data* **2015**, *60* (11), 3272-3288.

44. Lei, Z.; Dai, C.; Chen, B. Gas Solubility in Ionic Liquids. *Chem. Rev.* 2014, *114* (2), 1289-1326.

45. Kendall, J.; Monroe, K. P. The viscosity of liquids. II. The viscosity-composition curve for ideal liquid mixtures. *J. Am. Chem. Soc.* **1917**, *39*, 1787-1802.

46. Eirich, F. *Rheology V2: Theory and Applications*; Elsevier2012; Vol. 2.

47. Arrhenius, S. Ueber die Dissociation der in Wasser gelösten Stoffe. Z. phys. Chem. **1887**, (1), 631-648.

48. Bloomfield, V. A.; Dewan, R. K. Viscosity of liquid mixtures. *J. Phys. Chem.* **1971**, 75 (20), 3113-3119.

49. Doolittle, A. K.; Doolittle, D. B. Studies in Newtonian Flow. V. Further Verification of the Free - Space Viscosity Equation. *J. Appl. Phys.* **1957**, *28* (8), 901-905.

50. Zhang, S. J.; Li, M.; Chen, H. P.; Wang, J. F.; Zhang, J. M.; Zhang, M. L. Determination of physical properties for the binary system of 1-ethyl-3-methylimidazolium tetrafluoroborate + H2O. *J. Chem. Eng. Data* **2004**, *49* (4), 760-764.

51. Sanmamed, Y. A.; Gonzalez-Salgado, D.; Troncoso, J.; Romani, L.; Baylaucq, A.; Boned, C. Experimental methodology for precise determination of density of RTILs as a function of temperature and pressure using vibrating tube densimeters. *J. Chem. Thermodyn.* **2010**, *42* (4), 553-563.

52. Currás, M. R.; Husson, P.; Pádua, A. A. H.; Costa Gomes, M. F.; García, J. High-Pressure Densities of 2,2,2-Trifluoroethanol + Ionic Liquid Mixtures Useful for Possible Applications in Absorption Cycles. *Ind. Eng. Chem. Res.* **2014**, *53* (26), 10791-10802.

53. Seki, S.; Tsuzuki, S.; Hayamizu, K.; Umebayashi, Y.; Serizawa, N.; Takei, K.; Miyashiro, H. Comprehensive Refractive Index Property for Room-Temperature Ionic Liquids. *J. Chem. Eng. Data* **2012**, *57* (8), 2211-2216.

54. Klomfar, J.; Součková, M.; Pátek, J. Buoyancy density measurements for 1alkyl-3-methylimidazolium based ionic liquids with tetrafluoroborate anion. *Fluid Phase Equilib.* **2009**, *282* (1), 31-37.
55. Ge, M.-L.; Ren, X.-G.; Song, Y.-J.; Wang, L.-S. Densities and Viscosities of 1-Propyl-2,3-dimethylimidazolium Tetrafluoroborate + H2O at T = (298.15 to 343.15) K. *J. Chem. Eng. Data* **2009**, *54* (4), 1400-1402.

56. Mokhtarani, B.; Sharifi, A.; Mortaheb, H. R.; Mirzaei, M.; Mafi, M.; Sadeghian, F. Density and viscosity of pyridinium-based ionic liquids and their binary mixtures with water at several temperatures. *J. Chem. Thermodyn.* **2009**, *41* (3), 323-329.

57. Zarrougui, R.; Dhahbi, M.; Lemordant, D. Transport and Thermodynamic Properties of Ethylammonium Nitrate–Water Binary Mixtures: Effect of Temperature and Composition. *J. Solution Chem.* **2015**, *44* (3-4), 686-702.

58. Ranke, J.; Othman, A.; Fan, P.; Müller, A. Explaining Ionic Liquid Water Solubility in Terms of Cation and Anion Hydrophobicity. *Int. J. Mol. Sci.* **2009**, *10* (3), 1271.

59. Cammarata, L.; Kazarian, S. G.; Salter, P. A.; Welton, T. Molecular states of water in room temperature ionic liquids. *Phys. Chem. Chem. Phys.* **2001**, *3* (23), 5192-5200.

60. Blades, A. T.; Klassen, J. S.; Kebarle, P. Free Energies of Hydration in the Gas Phase of the Anions of Some Oxo Acids of C, N, S, P, Cl, and I. *J. Am. Chem. Soc.* **1995**, *117* (42), 10563-10571.

61. Hayes, R.; Warr, G. G.; Atkin, R. Structure and Nanostructure in Ionic Liquids. *Chemical Reviews* **2015**, *115* (13), 6357-6426.

62. Katayanagi, H.; Nishikawa, K.; Shimozaki, H.; Miki, K.; Westh, P.; Koga, Y. Mixing Schemes in Ionic Liquid–H2O Systems: A Thermodynamic Study. *J. Phys. Chem. B* **2004**, *108* (50), 19451-19457.

Chapter 4

PARTICLE SYNTHESIS AND CHARACTERIZATION

The purpose of this chapter is to describe the materials and the protocol used to synthesize fluorocarbon coated silica particles. The particle properties, such as the chemical coating on particle surface, particle size, and density, will be discussed in this chapter. Text and figures are reproduced and adapted with permission from Gao J., et al. *ACS Nano* 2015, 9, (3), 3243-3253.

4.1 Materials and Methods

Colloidal silica nanoparticles (NexSil 125-40, nominal radius 42.5 nm) were purchased from Nyacol Nano Technologies, Inc. and used as received (40 wt% solid in H₂O with sodium stabilizing counterion). Fluorocarbon 1H,1H,9H hexadecafluoro-1-nonanol (CAS 376-18-1, C₉H₄F₁₆O) was selected for surface modification for the silica particles (Figure 4.1). The ionic liquid [C₄mim][BF₄] (Iolitec, >99% pure) was dehydrated at 70 °C under vacuum condition for 24 *h* prior to use. The final water content was 373 ppm as determined by the Karl Fischer titration. The basic information of chemicals used in particle synthesis and characterization are listed in Table 4.1.



Figure 4.1: Chemical structure of fluorocarbon 1H,1H,9H hexadecafluoro-1-nonanol used to functionalize the particle surface.

Table 4.1: Summary of chemicals used in particle surface coating and characteristics.

Materials	Manufacturer	Durity
Waterfais	Wallulactulei	1 unity
NexSil 125-40	Nyacol	40 wt% solid in H ₂ O
1H,1H,9H hexadecafluoro-	Matrix Scientific	97%
1-nonanol		
$[C_4 mim][BF_4]$	Iolitec	99%
Ethanol	Decon	200 proof
Sulfuric acid	Fisher Chemical	98%
deuterated chloroform	Fisher Chemical	99.9%
(CDCl ₃)		

After coating procedure, the surface chemical on the silica particle was confirmed by proton nuclear magnetic resonance (¹H NMR). ¹H NMR spectra were measured on a Bruker AV400. Uncoated silica particles, fluorinated silica particles, and fluorocarbon 1H,1H,9H hexadecafluoro-1-nonanol were dissolved in deuterated chloroform (CDCl₃) separately before measurements. MestReNova NMR software was used to perform spectra analysis.

The grafting density of fluorocarbon on the silica particle surface was measured by thermogravimetric analysis (TGA) using a TA Instruments Q500. Nitrogen gas was flowing through the sample pan over the entire measurement. The temperature increased from 25 °C to 80 °C at a rate of 10 °C/min and equilibrated at 80 °C for 20 min to remove physically adsorbed water. Then, the temperature was increased to 900 °C at the same rate as before and equilibrated at 900 °C for 15 min to ensure the complete decomposition of coated fluorocarbon.

A high resolution SEM (JSM-7400F) was used to image the dried silica particles, their size distributions were analyzed through ImageJ software.¹ To prepare for SEM imaging, a dilute suspension ($\phi = 0.001$) of silica particles was pipetted onto the alumina stubs, allowing it to completely dry by atmosphere air. Then, the dried samples were sputter coated with an Au/Pd layer under vacuum for 40 s before being loaded into the SEM chamber and digital images were recorded.

The diffusivity of uncoated silica particles and coated silica particles suspension in various solvents were measured using a Brookhaven Instruments ZetaPals dynamic light scattering (DLS). Dilute (0.1 wt%) particle dispersions were used for DLS measurements. Single particle size was determined from diffusion coefficient measured using Stokes-Einstein-Sutherland correlation²⁻³ as described in Chapter 2.

Small Angle Neutron Scattering (SANS) measurements were performed for the dilute silica particle dispersions ($\phi = 0.004$) at the National Institute of Standards and Technology (NIST) Center for Neutron Research (NCNR) in Gaithersburg, MD on the NG7 beamline. Samples were loaded into demountable titanium cells with a 1 mm path length. Three different detector distances: 1m, 4m, and 13m were used to probe the complete length scale. Scattering intensities were reduced to absolute scale⁴ and analyzed following the protocol described in Chapter 2.

The densities of dispersions were measured using an Anton Paar DMA-4500M densitometer. Particle density in solution was extracted from a series of six diluted concentrations with particle mass fraction $0 \le x < 0.02$. For the uncoated silica, the

stock NexSil solution ($x = 0.415 \pm 0.006$) was diluted with a solution of 0.1 mM NaCl nanopure H₂O solution (pH = 9.1 ± 0.1) to maintain particle stability and match the mother liquid. The fluorocarbon-coated silica was dispersed into [C₄mim][BF₄], a good solvent for the fluorocarbon brush. Assuming ideal mixing of the dispersion constituents, the particle density can be calculated using the following equation based on ideal mixing:⁵

$$\frac{1}{\rho_{solution}} = \left(\frac{1}{\rho_{particles}} - \frac{1}{\rho_{solvent}}\right)x + \frac{1}{\rho_{solvent}}$$
4-1

where, $\rho_{solution}$, $\rho_{particles}$, and $\rho_{solvent}$ are the solution density, particle density, and solvent density, respectively; x is the mass fraction of particles in solution.

4.2 Particle Synthesis

The 9-carbon chain fluorocarbon 1H,1H,9H hexadecafluoro-1-nonanol was covalently grafted to the nanoparticle surface *via* a dehydration reaction following the protocol of van Helden *et al.*⁶ According to literature, a model adhesive hard sphere (AHS) particle was synthesized using this procedure to chemically graft 1-octadecanol chain layer onto the surface of silica particles.⁷ Thus, a similar coating protocol was used for the fluorocarbon coating system. The main chemical reaction during the surface coating is the dehydration reaction between silanol group on the surface of silica particle and the hydroxyl group at the end of the fluorinated alcohol chain (*i.e.* -Si-OH + R-OH \rightarrow -Si-O-R + H2O), whereas, other side chemical reactions such as -Si-O-Si- + R-OH \rightarrow -Si-O-R + Si-OH and -Si-O-Si- + H₂O \rightarrow 2-Si-OH (in which -R is fluorinated hydrocarbon), can also happen depending on the silica particle surface chemistry.⁸⁻⁹ Colloidal silica nanoparticles NexSil 125-40 were used for surface coating and the surface coating procedure is shown in Figure 4.2. First, 40 g of the 40 wt% silica particle suspension was poured into a beaker (250 mL) and agitated with a magnetic stirring bar. Four drops of sulfuric acid were added to the particle suspension while stirring to induce protonation of the silica surface, promoting the dehydration chemical reaction. The mixture was stirred for 15 min to ensure a homogenous solution was reached. Then, 70 g of ethanol was added into the mixture solution under stirring in order to help the dissociation of the H⁺ from the surface of particle. The mixture was stirred for 15 min followed by a DLS measurement of particle size to ensure the silica particles were well dispersed in this mixture. After that, 13 g of fluorocarbon 1H,1H,9H hexadecafluoro-1-nonanol was added into a separate round bottom flask. Then, 156 g ethanol was added into the flask, and the mixture was stirred for 5 min to dissolve the fluorocarbon. After complete dissolving, all the materials in the beaker are transferred into the round bottom flask slowly under stirring.



Figure 4.2: A schematic flow chart of the particle coating procedure.

The round bottom flask contained the fluorocarbon, NexSil silica particle suspension, ethanol, and sulfuric acid was mounted onto a rotary evaporator (RE211 from Buchi), shown in Figure 4.3. The round bottom flask was immersed into an oil bath with a high-precision temperature control system. Rotary evaporator condenser with 15 °C circulating cold water was used to condense the water and ethanol vapor, which was then collected into a separate flask. The mixture solution was heated up to 70 °C and equilibrated for 15 min in oil bath. Then the temperature was raised up to 222°C by 10 °C every 15 min. During this process, the mixture started to boil off when the temperature reached 90 °C and then condensation of water/ethanol vapor mixture was observed at 100~110 °C. After the temperature reached to 222 °C, the reaction was allowed to proceed for 5 h at this temperature, after which, the round bottom flask was lifted up from the oil bath and allowed to cool down to room temperature overnight.



Figure 4.3: The experimental apparatus for the particle surface coating, consisting of a rotary evaporator and a heating oil bath. (a) Condenser to cool down evaporated water and ethanol, (b) collection flask to collect water/ethanol condensation, (c) reaction flask, (d) oil bath, and (e) condenser cooling water. Image courtesy of Dr. Jung Min Kim.

After the reaction, the mixture was cooled down to the room temperature. The obtained silica particle "cake" at the bottom of reaction flask was dispersed in ethanol using water bath sonication for 2 h. After ensuring all the particles were well dispersed, and no particle agglomerates left, the particle suspension was centrifuged at 7500 g_s (g_s is the standard gravitational acceleration) for 1.5 h using a Sorvall RC6+ centrifuge. After the centrifugation, the supernatant liquid was discarded. The particle sediment "cake" was dispersed in fresh ethanol again in a glass bottle under sonication until all the particles were dispersed. Then, another centrifugation was performed. This cleaning procedure was repeated 3 times to make sure all the excess

fluorocarbon was removed by ethanol. Finally, the cleaned particles were stored in ethanol until use and no visual sedimentation was observed, even after 4 months.

To prepare dispersions of the fluorocarbon coated silica particles, concentrated coated particle dispersion in ethanol was dried by evaporating ethanol using a nitrogen flowing stream for 4 h at room temperature. Then, the dried silica particles were placed under vacuum at -30 inHg for 24 h to remove the residual ethanol and ready to be used to prepare particle dispersions in various solvents including ionic liquids.

4.3 Particle Characterization

The properties of single silica particle before and after coating, such as surface coating chemical, particle size, and density, have been evaluated and characterized through various methods. The successful grafting of fluorocarbon onto the silica particle surface is confirmed by ¹H NMR and the grafting density is measured by TGA. The size distributions of dried uncoated and coated silica particles are investigated using SEM. The hydrodynamic radii of uncoated and coated silica particles dispersed in various solvents are measured and compared using different techniques, such as DLS and SANS. The skeletal particle densities in solutions at room temperature are evaluated for both uncoated and coated particles using densitometer.

4.3.1 Surface Coating

4.3.1.1 Proton Nuclear Magnetic Resonance (¹H NMR)

The ¹H NMR spectra of uncoated silica particle, fluorocarbon coated silica particle and fluorocarbon 1H,1H,9H hexadecafluoro-1-nonanol suspended in deuterated chloroform (CDCl₃) are shown in Figure 4.4. Compared with the uncoated

silica particles, the NMR spectrum of the fluorocarbon coated silica particles shows chemical shift peaks at 4.1 ppm and 5.9-6.2 ppm, which corresponds to the chemical shift peaks of fluorocarbon, indicating the success of coating fluorocarbon onto the surface of silica particles through covalent bonding.



Figure 4.4: ¹H NMR spectra of uncoated silica particles (top), fluorocarbon coated silica particles (middle), and 1H,1H,9H hexadecafluoro-1-nonanol (CAS 376-18-1, C₉H₄F₁₆O) (bottom) in CDCl₃.

4.3.1.2 Thermogravimetric Analysis (TGA)

The grafting density of fluorocarbon onto particle surface is measured by TGA. The TGA results of uncoated and coated silica particles are shown in Figure 4.5 in the form of mass fraction versus temperature. The uncoated particles show a weight loss of about 2.3 wt%, which is attributed to the desorption of physically adsorbed water. The coated particles have a weight loss of about 4.5 wt %, which indicates the additional presence of the surface modifier. From this weight loss, the graft density is calculated to be 1.4 chain/nm² of particles based on a particle average radius of 54.2 nm obtained by SANS. This value is lower than the graft density for octadecyl alcohol (2.9/nm²) reported previously.¹⁰ It is calculated that 28-35% surface silanol groups are covalent bonded with fluorinated group based on the estimation of 4-5 silanol group/nm^{2.9} This lower graft density is attributed to the larger molecular size of the fluorinated compound. The calculated graft density corresponds to ~ 60 % of the theoretical maximum surface coverage, which is calculated based on the molecular size derived from the bond lengths, angles, and atom sizes of fluorocarbon.



Figure 4.5: TGA curves for uncoated and coated particles, showing a weight loss of 2.3 wt% and 4.5 wt% for uncoated and coated particles when temperature increases to ~900 °C, respectively.

4.3.2 Particle Size

4.3.2.1 Scanning Electron Microscopy (SEM)

The dried silica particles before and after surface coating are recorded by SEM and the particle size distribution is analyzed by ImageJ software.¹ For example, Figure 4.6A is the raw SEM image and this SEM image is analyzed by software ImageJ to calculate the number-averaged nanoparticle radius as shown in Figure 4.6B. Particles of blue color in the image are accounted in the statistical analysis while particles of black color are excluded due to overlapping issues. Area distribution of the blue-color particles is shown in Figure 4.6C, representative of a Gaussian distribution function. The number-average radius and standard deviation of the uncoated particles are then calculated to be 45.3 ± 5.3 nm, which is in a good agreement with manufacturer.



Figure 4.6: Particle size analysis of dried, uncoated silica particles. (A) Raw SEM image; (B) SEM image processed by ImageJ software: blue indicates the particles are considered in the area distribution analysis; black indicates the particles are excluded from the analysis (C) Histogram of particle area distribution; (D) Statistical results of the area and radius of the silica particles.

After coating, the size distribution of the coated silica particle is evaluated using the same method as for uncoated particle and is shown in Figure 4.7. The

number-averaged particle size and standard deviation for coated particles are 53.6 ± 10.0 nm, which are slightly larger than for the uncoated silica particles. This larger average particle size could result from the surface coating as well as lower statistical accuracy. In Figure 4.7B, fewer single particles (blue-color) are considered in the statistical analysis as most of the particles are overlapped with each other in this SEM image analysis. Due to the limited particle population studied using SEM image analysis, other independent analytical methods involving great particle population, such as DLS and SANS measurements, are employed to characterize the particle size distribution.



Figure 4.7: Particle size analysis of dried, coated silica particles. (A) Raw SEM image; (B) SEM image analyzed by ImageJ software: blue indicates the particles are considered in the area distribution analysis; black indicates the particles are excluded from the analysis (C) Histogram of particle area distribution; (D) Statistical results of the area and radius of the silica particles.

4.3.2.2 Dynamic Light Scattering (DLS)

The size distributions of uncoated and coated particles in various polar solvents and in $[C_4mim][BF_4]$ were measured by DLS and are shown in Figure 4.8. Uncoated silica particles are well dispersed in H₂O and fluorocarbon coated silica particles are well dispersed in ethanol with a narrow size distribution. The shape of the size distribution of the coated silica particles is similar to that for the uncoated particles with a slight shift to larger sizes, which likely is due to the addition of surface coating. It is evident that the uncoated silica particles aggregate in ionic liquid $[C_4mim][BF_4]$ even after sonication and extensive mixing, exhibiting a broader size distribution and a much larger averaged particle size. In contrast, the coated particles disperse in $[C_4mim][BF_4]$ as evinced by a much narrower size distribution.

The values of the number-averaged particle size from the DLS measurements are listed in Table 4.2. The results of coated particles in ethanol demonstrate that the addition of the fluorocarbon coating increases the radius of particles by an amount consistent with the molecular size of the fluorocarbon (~14 Å), as estimated from the bond lengths and angles. However, there is an anomalous increase in hydrodynamic radius observed for the particles dispersed in the ionic liquid, which is the consequence of solvation layers forming near the particle surface. This solvation layering will be discussed in details in Chapter 5.



Figure 4.8: Size distribution of uncoated and coated particles suspended in various solvents from DLS measurement.

Table 4.2:	Particle radii for uncoated and coated particles in various solvent
	measured by DLS.

	Uncoated	Uncoated	coated	
	particle in	particle in	Coated particle	particle in
Solutions	H_2O	[C ₄ mim][BF ₄]	in ethanol	[C ₄ mim][BF ₄]
Radius (nm)	53.8 ± 4.8	298.6 ± 28.1	55.2 ± 3.8	60.4 ± 3.0

4.3.2.3 Small-Angle Neutron Scattering (SANS)

SANS measurements of dilute nanoparticle suspensions are performed to further quantify the particle size, polydispersity, and coating layer structure. The measured SANS intensities for dilute dispersions ($\phi = 0.004$) of uncoated and coated particles in different solvents at 25 °C are shown in Figure 4.9. The general shapes of

the scattering spectra are similar as expected, given that the particles have the same core, which dominates the scattering intensity. The differences in scattering intensities reflect the differences in the neutron scattering length densities of the solvents, which is also evident in the different background intensities observed at high values of scattering vector q. The scattering intensity from the dispersion of uncoated particles in H_2O is fitted by varying particle size, polydispersity, and volume fraction with a Schulz-distributed sphere model⁴ using the known scattering length densities (SLD) for the core and solvent. The measured scattering intensities from dispersions of coated particles in ethanol and [C₄mim][BF₄] are fitted by a core-shell model and varying the shell thickness, volume fraction, and scattering length density of the shell under constraints. The fitting parameters are summarized in Table 4.3. In the fitting, the volume fraction of the dispersion is restricted with the particle concentration calculated from mass with uncertainty, the shell thickness is constrained by the extended molecular size of fluorocarbon, and the scattering length density of the shell is constrained within the range of the SLD of solvent and fluorocarbon. From the fits, I extract individual particle properties such as core size, polydispersity, and coating thickness. As the density of ionic liquid solvation layers is close to the bulk ionic liquid, there is no significant scattering length density difference between the solvation layers and the ionic liquid. Therefore, the solvation layers are not detected directly by neutron scattering. Note that the shell SLD determined from the fit is close to the value calculated for the pure fluorocarbon. This suggests that the large molecular size of fluorine in the fluorocarbon chains (fluorocarbon chains occupy 60 vol% of the shell) prevents significant solvent penetration into the coating. Analysis of the SANS form factor measurements, summarized in Table 4.4, yield a particle radius of $54.2 \pm$

5.4 nm for the core particles with 10.5% polydispersity (p) and shell thickness of 0.9 nm and 1.2 nm for particles dispersed in ethanol and [C₄mim][BF₄], respectively. These measurements further confirm the presence of the fluorocarbon coating and show that the fluorocarbon brush is more extended in [C₄mim][BF₄], which suggests a better compatibility with [C₄mim][BF₄] than with ethanol.



Figure 4.9: Measured SANS scattering intensity at 25 °C for three dilute dispersions and corresponding fits: uncoated particles in H_2O , coated particles in ethanol and in ionic liquid [C₄mim][BF₄], respectively.

Parameters	Uncoated particles	Coated particles in	Coated particles in
	in H ₂ O	ethanol	[C ₄ mim][BF ₄]
ϕ	0.0036 ^b	0.0048 ^b	0.0032 ^b
\overline{r} (nm)	54.2 ^b	54.2 ^a	54.2 ^a
р	0.105 ^b	0.105 ^a	0.105 ^a
δ (shell, nm)		0.9 ^b	1.2 ^b
$SLD_{core}(A^{-2})$	3.47×10^{-6} a	3.47×10^{-6} a	$3.47 imes 10^{-6}$ a
$SLD_{shell}(A^{-2})$		3.24×10^{-6} b	$3.24 \times 10^{-6 b}$
$SLD_{solvent}(A^{-2})$	$-5.60 imes 10^{-6}$ a	$-3.45 imes 10^{-7}$ a	1.40×10 ^{-6 a}
Background (cm ⁻¹)	0.76 ^a	0.99 ^a	0.62 ^a

 Table 4.3:
 The parameters used for SANS form factor fittings.

a- The parameters are fixed as constant.

b- The parameters are fitted under constraints.

The average radii of the uncoated and coated particles obtained from DLS and SANS are summarized and compared in Table 4.4. An estimate of the solvation layer of ~ 5 nm can be made by comparing the difference between the hydrodynamic radius in $[C_4mim][BF_4]$ measured by DLS and the core-shell radius determined by SANS measurements. A solvation layer is not detected by SANS fits to the form factor, but it will increase the particle radius and thus, slow the diffusion and yield an apparently larger particle size in DLS. More detailed analysis of solvation layers can be found in Chapter 5.

 Table 4.4:
 Summary of particle radii for uncoated and coated particles with various methods.

Measurement	Uncoated particle	Coated particle in	Coated particle in
Techniques	in H ₂ O	ethanol	[C ₄ mim][BF ₄]
DLS	$53.8\pm4.8~nm$	$55.2 \pm 3.8 \text{ nm}$	$60.4 \pm 3.0 \text{ nm}$
SANS	$54.2 \pm 5.4 \text{ nm}$	Core: 54.2 ± 5.4 nm	Core: 54.2 ± 5.4 nm
	p = 10.5 %	p = 10.5 %	<i>p</i> = 10.5 %
Brush		0.9 nm	1.2 nm

4.3.3 Particle Density

To calculate the accurate volume fraction of particles in dispersion, it is crucial to obtain an accurate particle density. The densities of both uncoated and coated silica particles are obtained through gravimetric densitometry using equation 4-1. Densitometry was performed on the uncoated particles suspended in 0.1 mM NaCl/H₂O (pH = 9.1 ± 0.1) solution, and the coated particles suspended in [C₄mim][BF₄] at 25 °C using an Anton Paar DMA-4500M densitometer. The experimental results can be found in Figure 4.10. The skeletal particle density in solution is extracted from a linear regression of equation 4-1 to the data and found to be $\rho = 2.210 \pm 0.005$ g/cm³ and $\rho = 2.167 \pm 0.007$ g/cm³ for the uncoated and coated particles, respectively. The reduction in solution density of the coated particles is attributed to the contribution from the surface layer, where the fluorocarbon coating is expected to have a lower density than the silica nanoparticle.



Figure 4.10: Results of the densitometry experiments for the uncoated silica particles in mother liquid (secondary y axis, right) and fluorocarbon coated silica particles in [C₄mim][BF₄] (primary y axis, left). The lines represent a fit of equation 4-1 to the inverse of the suspension density and the mass fraction of the disperse phase.

4.4 Conclusions

Fluorocarbon 1H,1H,9H hexadecafluoro-1-nonanol coated silica particles are synthesized through dehydration reaction following the procedure of van Helden *et al.* The surface coating is confirmed by ¹H NMR and the grafting density is measured by TGA determined to be 1.4 chain/nm² with 60 % surface coverage. The particle size distributions of uncoated and coated silica particles are measured by independent techniques, such as SEM, DLS, and SANS measurements, resulted consistent agreement with each other, indicating the size of the particles is on the order of ~ 100

nm with 10% polydispersity. The thickness of coated brush is determined to be ~ 1 nm, which is consistent with the molecular size of the fluorocarbon (~1.4 nm) as estimated from the bond lengths and angles. The coated silica particles disperse well in the ionic liquid [C₄mim][BF₄] due to the formation of solvation layers, which will be discussed in detail in Chapter 5, while, uncoated particles aggregate in the same ionic liquid. The skeletal particle densities of uncoated and coated silica particles are measured by densitometry and are used to convert the particle mass fraction loading to volume fraction in this dissertation.

REFERENCES

1. ImageJ. https://imagej.nih.gov/ij/ (accessed 11/12/2016).

2. Einstein, A. Über die von der molekularkinetischen Theorie der Wärme geforderte Bewegung von in ruhenden Flüssigkeiten suspendierten Teilchen. *Annalen der physik* **1905**, *322* (8), 549-560.

3. Hiemenz, P. C.; Rajagopalan, R. *Principles of Colloid and Surface Chemistry, revised and expanded*; CRC Press1997; Vol. 14.

4. Kline, S. R. Reduction and analysis of SANS and USANS data using IGOR Pro. *J. Appl. Crystallogr.* **2006**, *39* (6), 895-900.

5. Gao, J.; Ndong, R. S.; Shiflett, M. B.; Wagner, N. J. Creating nanoparticle stability in ionic liquid [C4mim][BF4] by inducing solvation layering. *ACS Nano* **2015**, *9* (3), 3243-3253.

6. Van Helden, A.; Jansen, J. W.; Vrij, A. Preparation and characterization of spherical monodisperse silica dispersions in nonaqueous solvents. *J. Colloid Interface Sci.* **1981**, *81* (2), 354-368.

7. Eberle, A. P. R.; Wagner, N. J.; Castañeda-Priego, R. Dynamical Arrest Transition in Nanoparticle Dispersions with Short-Range Interactions. *Phys. Rev. Lett.* **2011**, *106* (10), 105704.

8. Ballard, C. C.; Broge, E. C.; Iler, R. K.; John, D. S. S.; McWhorter, J. R. ESTERIFICATION OF THE SURFACE OF AMORPHOUS SILICA. *J. Phys. Chem.* **1961**, *65* (1), 20-25.

9. Iller, R. The chemistry of silica: solubility, polymerization, colloid and surface properties, and biochemistry. *New York, Cap* **1979**, *1* (2).

10. Eberle, A. P.; Castañeda-Priego, R. n.; Kim, J. M.; Wagner, N. J. Dynamical arrest, percolation, gelation, and glass formation in model nanoparticle dispersions with thermoreversible adhesive interactions. *Langmuir* **2012**, *28* (3), 1866-1878.

Chapter 5

FLUOROCARBON COATED PARTICLE SYSTEM – STABILIZATION MECHANISM AND INTERFACIAL STRUCTURE

In this chapter, stable solid silica nanoparticle dispersion in ionic liquid 1butyl-3-methylimidazolium tetrafluoroborate [C₄mim][BF₄] is achieved and the stabilization mechanism of silica particle in ionic liquid is investigated. The critical role of solvation forces in dispersing and stabilizing nanoparticles and colloids in [C₄mim][BF₄] is demonstrated. Stable silica nanoparticle suspensions over 60 wt% solids are obtained by particle surface chemical functionalization with a fluorinated alcohol. A combination of techniques including rheology, dynamic light scattering (DLS), transmission electron microscopy (TEM), and small angle neutron scattering (SANS) are employed to determine the mechanism of colloidal stability. Solvation layers of ~ 5 nm at room temperature are measured by multiple techniques, and are thought to be initiated by hydrogen bonds between the anion [BF₄]⁻ and the fluorinated group on the surface coating. Inducing structured solvation layering at particle surfaces through hydrogen bonding is demonstrated as a method to stabilize particles in ionic liquids. Text and figures are reproduced and adapted with permission from Gao J., et al. *ACS Nano* 2015, 9, (3), 3243-3253.

5.1 Introduction

Many applications of ionic liquids require successfully dispersing nanoparticles across a broad range of conditions. Studies of colloidal systems in ionic liquids include nanoparticle synthesis,¹⁻² phase transfer,³ catalytic reactions medium as

a "green" solvent,⁴⁻⁵ the self-assembly of surfactants,⁶⁻⁷ and block copolymer mesophases.⁸⁻⁹ However, there are relatively few studies of the stability of colloidal particles in ionic liquids and these have been primary empirical,¹⁰⁻¹⁷ despite the fact that nanoparticle stability is of critical importance in determining the suspension rheology and controlling aggregation, gel and glass formation, as well as in the performance of ionic liquid based lubricants. It is apparent from the limited reports in the literature that particle dispersion and stability in ionic liquids depends on the specific chemical composition of the ionic liquid as well as the surface chemistry of the particles.^{12, 16-17}

The shear viscosity of hydrophilic and hydrophobic fumed silica colloidal dispersions in various ionic liquids were measured, in which shear thinning, shear thickening, and gelation were reported.¹² The effect of particle size and surface hydrophobicity of fumed silica particle on particle stability and suspension properties in [C₄mim][BF₄] have been investigated and can be found in Appendix B. An increase of the alkyl chain length of the cation has been reported as beneficial for stabilizing TiO₂ nanopowders in 1-alkyl-3-methylimidazolium tetrafluoroborate ([C_nmim][BF₄]) ionic liquids.¹⁷ Spherical silica nanoparticles were reported to aggregate in imidazolium-based ionic liquids by Ueno *et al.*,¹⁰ while grafting polymer such as poly(methyl methacrylate) (PMMA) to the particle surface improved suspension stability in ionic liquids.^{10, 13} For example, PMMA-grafted silica nanoparticles show good dispersion in 1-butyl-3-methylimidazolium because PMMA is itself soluble in these solvents; however, such coated particles show poor dispersion if

PMMA is insoluble in the ionic liquids (*e.g.*, $[C_4mim][BF_4]$).¹⁰ Here the mechanism of stability is afforded by the steric repulsion between the grafted layers of significant extent and as is typical with such systems, the system will become unstable when the brush is immersed into a poor solvent and collapses.¹⁸

Particle stability in dispersions is typically treated within the classical framework of colloid science, namely the Derjaguin-Landau-Verwey-Overbeek (DLVO) theory, which accounts for van der Waals attraction and electrostatic repulsion between particles in solution.¹⁹ Due to the ionic nature of the solvent, electrostatic stabilization is not sufficient to stabilize particles in ionic liquids against the ubiquitous attraction from dispersion forces between the particles.¹⁰ The high ionic strength in ionic liquids effectively screens the electrostatic repulsion between silica particles. The screening of the electrical potential and electrical double layers at charged solid surfaces in ionic liquids and molten salts²⁰ has been investigated by experiments²¹ and simulations.²²⁻²³ The electrical double layer in ionic liquid [C₄mim][BF₄] has been determined to be one-ion thick layer (about 4 Å thickness) using sum frequency generation vibrational spectroscopy (SFG), electrochemical impedance spectroscopy (EIS), and the vibrational Stark effect.²¹

Due to this screening effect, surface-adsorbed or grafted surfactants and polymers, which are soluble in the ionic liquid, are introduced to provide stability.¹⁸ However, there are rare cases where "bare" particles disperse into ionic liquids, *e.g.*, hydrophilic fumed silica particles can be dispersed into an aprotic ionic liquid, $[C_4mim][BF_4]$ in the absence of any stabilizer.^{12, 14, 24} This unexpected stability is hypothesized to be induced by solvation forces arising from solvent structuring near the particle surfaces, and the fact that the particles are often porous aggregates such

that van der Waals attractions are weak.^{13-14, 24} There is evidence for such solvation forces occurring at macroscopic surfaces: Oscillating solvation forces have been observed between macroscopic mica and silica surfaces in both protic and aprotic ionic liquids as detected by surface force measurements,²⁵⁻³⁰ and these observations have been qualitatively reproduced by molecular dynamics simulation.³¹ These measurements and simulations indicate that the ion pairs are arranged in a layered structure near the solid surfaces.²⁹ Other evidence supporting the existence of solvation layers in particle dispersions include the increase in fumed silica particle stability in [C₄mim][BF₄] upon addition of lithium salt which is presumed to stabilize the solvation layers.²⁴ Furthermore, silica particle suspensions lose stability in the presence of small amounts of water, which decreases the structuring of solvation layers.^{14, 32} However, to date, there are no direct measurements of the solvationinduced structuring in ionic liquids around nanoparticle surfaces or a predictive, quantitative knowledge of inter-particle interactions in ionic liquids that can be used to predict and control colloidal stability.

To directly test this hypothesis and to develop a strategy for dispersing nanoparticles in ionic liquids more generally, I study the nanostructure, stability, and dynamics of model spherical silica particles surface-functionalized in a common aprotic ionic liquid [C₄mim][BF₄]. Surface modification using a small fluorocarbon molecule 1H,1H,9H hexadecafluoro-1-nonanol induces solvation layering around silica particles, leading to dispersion stability. The choice of this surface modifier is motivated by the research of Shiflett and Yokozeki,³³ who studied the behavior of fluorocarbons in room temperature ionic liquids and found a wide range of solubility that correlated with hydrogen bonding. I propose that hydrogen bonding between the

fluorinated anion in the ionic liquid [C₄mim][BF₄] and the terminal hydrogen group on the surface coating drives the formation of solvation layers that impart colloidal stability by steric repulsion. Figure 5.1 shows a schematic representation of such solvation layers, based on the dimensions determined in this work. Note that my methods cannot determine the molecular orientation in the layers and the molecular details in the schematic are based on related literature work.²⁹⁻³⁰ This study presents the first direct measurements of the extent of the formation of these solvation layers around nanoparticles in ionic liquid [C₄mim][BF₄] using TEM, rheology, DLS, and SANS.



Figure 5.1: Schematic illustration of a coated silica nanoparticle stabilized by solvation layers in ionic liquid $[C_4mim][BF_4]$. Note the ion structure in the layers is schematic.

5.2 Materials and Methods

5.2.1 Particle Suspension Preparation

The ionic liquid $[C_4mim][BF_4]$ (>99% pure, Iolitec, Figure 5.2B), was dehydrated under vacuum condition at 70 °C for 24 h prior to use. The final water content was 373 ppm as determined by the Karl Fischer titration. Fluorocarbon 1H,1H,9H hexadecafluoro-1-nonanol (Figure 5.2A) coated silica particles are synthesized and stored following the protocol described in Chapter 4.



Figure 5.2: Chemical structure of fluorocarbon 1H,1H,9H hexadecafluoro-1-nonanol used to functionalize the particle surface (A) and ionic liquid 1-butyl-3-methylimidazolium tetrafluoroborate ([C₄mim][BF₄]) (B).

Dispersions of coated particles were prepared by first drying the fluorocarboncoated silica under a nitrogen stream and then under vacuum for 24 h. The particles were dispersed in dehydrated ionic liquid at varying concentrations from dilute through concentrated. For the uncoated silica particle dispersions, the particles were thoroughly washed 3 times with water using a Sorvall RC6+ centrifuge to remove the sodium stabilizer before sample preparation. All the dispersions were mixed using a vortex mixer (3 min at 3000 RPM), sonication water bath (4 h at room temperature) and roll mixer (72 h at room temperature) to ensure homogeneity, followed by a 5 min degassing to remove air bubbles in the samples. As the presence of water decreases the structure of solvation layer and reduce the particle stability,^{32, 34} the samples were dehydrated again in a vacuum oven at 70 °C for 24 h prior to each measurement and the water content was 1194 ppm as measured by the Karl Fischer titration. All of the dispersion samples were prepared by weight and converted to volume fraction using the measured particle and solvent densities in Table 5.1. The coated particles in $[C_4mim][BF_4]$ dispersion samples are stable at room temperature and no sedimentation is observed after 6 months.

Table 5.1: Densities of particles, fluorocarbon and solvents at 25 °C.

	silica	coated silica	fluorocarbon	[C ₄ mim][BF ₄]	H ₂ O	ethanol
ρ (g/cm ³)	$\begin{array}{c} 2.210 \pm \\ 0.005 \end{array}$	2.167 ± 0.007	1.646*	1.201 ± 0.001	0.997 ± 0.001	0.789*

^{*} The densities of fluorocarbon and ethanol are from material safety data sheet. All the others are measured or calculated from densitometry as discussed in Chapter 4.

5.2.2 Derjaguin-Landau-Verwey-Overbeek (DLVO) Inter-particle Potential Calculation

According to the DLVO theory, the total interparticle potential $\Phi_{total}(d)$ can be written as the sum of an attractive London-van der Waals potential $\Phi_{vdw}(d)$ and a repulsive electrostatic potential $\Phi_{vdw}(d)$ due to the charge of the particles¹⁹

$$\Phi_{total}(d) = \Phi_{vdw}(d) + \Phi_{ele}(d), \qquad 5-1$$

where *d* is the distance between the particle surfaces. $\Phi_{vdw}(d)$ between two identical spherical silica particles of radius *r* is given by³⁵

$$\Phi_{vdw} = -\frac{1}{6}A\left(\frac{2r^2}{a^2 - 4r^2} + \frac{2r^2}{a^2} + \ln\frac{a^2 - 4r^2}{a^2}\right),$$
 5-2

where *a* is the two particles center-center separation, a = 2r + d. *A* is the Hamaker constant (two silica dioxide phases interacting across solvent) and can be calculated from the relative dielectric constant ε and refractive index *n* of silica dioxide and solvent based on the Lifshitz theory

$$A = \frac{3}{4} k_{B} T \left(\frac{\varepsilon_{p} - \varepsilon_{s}}{\varepsilon_{p} + \varepsilon_{s}} \right)^{2} + \frac{3hv_{e}}{16\sqrt{2}} \frac{\left(n_{p}^{2} - n_{s}^{2}\right)^{2}}{\left(n_{p}^{2} + n_{s}^{2}\right)^{3/2}},$$
 5-3

where k_B is the Bolzmann constant, *T* is the absolute temperature, *h* is Planck's constant, which is 6.63×10^{-34} J·s, and v_e is the frequency of the main electronic absorption for the dielectric permittivity, which can be taken to be 3×10^{15} s⁻¹ for water and [C₄mim][BF₄]. The Hamaker constant of silica particles in mother liquid is assumed to be identical as for that in H₂O.

The repulsive electrostatic potential $\Phi_{vdw}(d)$ of spheres with thin doublelayers in ionic liquids and with constant potential in H₂O are given respectively by³⁶

$$\Phi_{ele} = 32\pi r \varepsilon_s \varepsilon_0 \left(\frac{k_B T}{ze}\right)^2 \tanh^2 \left(\frac{\psi_s ez}{4k_B T}\right) \exp(-\kappa d), \qquad 5-4$$

$$\Phi_{ele} = 2\pi r \varepsilon_s \varepsilon_0 \left(\frac{k_B T}{ze}\right)^2 \left(\frac{\psi_s ez}{k_B T}\right)^2 \ln(1 + e^{-\kappa d}), \qquad 5-5$$

where ε_s is the relative dielectric permittivity of solvent, ε_0 is dielectric constant, which is $8.85 \times 10^{-12} \text{ C}^2 / (J \cdot m)$, *z* is the symmetric electrolyte of valence, *e* is unit charge $1.60 \times 10^{-19} \text{ C}$, ψ_s is the surface potential, and κ is the Debye reciprocal length parameter. Debye length κ^{-1} is calculated as follows³⁵

$$\kappa^{-1} = \sqrt{\frac{\varepsilon_s \varepsilon_0 k_B T}{1000 e^2 N_A \left(\sum_i z_i^2 M_i\right)}}, \qquad 5-6$$

where z_i is the electrolyte valence of solvent, M_i is the molar concentration of electrolyte, and N_A is the Avogadro's number, which is 6.02×10^{-23} mol⁻¹.

The Zeta potential is used as an estimation of nanoparticle surface potential for the electrostatic force calculation. Because of the high ionic strength and non-aqueous feature of ionic liquid [C₄mim][BF₄], it was difficult to measure the zeta potential with a conventional electrokinetic method in the ionic liquid colloid system.¹⁰ The zeta potential of silica particles in [C₄mim][BF₄] is calculated by assuming the charge density of silica nanoparticles in [C₄mim][BF₄] is the same as that measured in the sodium chloride (NaCl) aqueous solution of 1.25M concentration. The Debye length in ionic liquid [C₄mim][BF₄] is estimated from electrical double layer measurement using sum frequency generation vibrational spectroscopy (SFG), electrochemical impedance spectroscopy (EIS).³⁷ Uncoated silica particles are dispersed in aqueous solution (0.1 mM (pH = 9.1 ± 0.1) NaCl solution: mother liquid) to measure the zeta potential of silica nanoparticles in the aqueous solution using a Brookhaven ZetaPals instrument.

5.2.3 Characterization Techniques

TEM images of the nanoparticles were recorded using a JEOL JEM-2000FX (LaB6) TEM operated at 200 kV in the Keck Microscope facility at the University of Delaware. The dried, coated particles were imaged after the ethanol of coated particle/ethanol solution drops dried on the carbon-sputtered copper grid. The coated particles dispersed in $[C_4mim][BF_4]$ were imaged directly in their solution phase. Samples were prepared by placing a small droplet of dilute coated

particle/[C₄mim][BF₄] dispersion onto a carbon-sputtered copper grid and extra solution was removed by blotting with filter paper, leaving a thin film of dispersion on the grid.

Rheological measurements were performed on a TA Instruments AR-G2 stress controlled rheometer at 25 °C with Peltier plate temperature control, and a 40 mm stainless steel cone with 2° cone angle with solvent trap. Conversion between applied and measured parameters (torque *M* and strain γ) and the rheological material functions were performed using the TA Instruments Trios software. To erase any previous shear histories and maintain a consistency between measurements, a steady pre-shear was applied at a shear stress of 10 Pa for 2 min before each dynamic rheological measurement. A steady state shear measurement is performed as a function of shear stress for two different concentrations (volume fraction = 0.2 and 0.3) suspensions. For the steady shear tests, the viscosity responses are reversible. An ascending and descending shear stress loop was performed, and there is no observed hysteresis or thixotropic effects during steady shear. Each measurement was repeated three times for accuracy.

SANS measurements were performed at the National Institute of Standards and Technology (NIST) Center for Neutron Research (NCNR) in Gaithersburg, MD on both NG3 and NG7 beamlines. Samples were loaded into demountable titanium cells with a 1 mm path length. The cell temperatures were maintained to \pm 0.1 °C using the 10 CB, 10 position sample cell holder with fluid bath. A wide range in scattering wavevector *q* was obtained by combining scattering intensity from four different instrument configurations at detector distances of 1, 4, 13, and 15.3 m with lens. The wavelength for 1, 4, and 13 m configurations is 6.0 Å and 8.0 Å for 15.3 m with lens.

The SANS scattering intensity profiles were reduced to absolute scale and analyzed using the standard protocol discussed in Chapter 2.

SANS spectra are analyzed following the protocol in Chapter 2. The scattering length density (SLD) of particles and solvents were calculated using measured densities (Table 5.1) by NIST NCNR online tools: Neutron activation and scattering calculator.³⁸ Properties of the dispersions were determined by modeling the measured SANS intensity using a form factor model for polydisperse core-shell particles (Core-shell with Schultz distributed core) and a monodisperse structure factor model that includes hard core repulsion with contact attraction (Sticky hard sphere model). The SANS spectra of particle solutions are described by the function

$$I(q) = n_p \overline{P}(q, \rho_{i(SLD)}, \phi, \overline{r}, p, \delta) S(q, \phi_{eff}, \overline{r}_{eff}, \tau, \varepsilon) + bkg, \qquad 5-7$$

where n_p is the particle number density, $n_p = \phi/\langle V \rangle$, $\langle V \rangle$ is the average of single particle volume, and *bkg* is the background level. $\overline{P}(q, \rho_i, \phi, \overline{r}, p, \delta)$ is the form factor, which describes the scattering from individual particles, and is a function of wavevector q, particle, solvent, and shell scattering length density $\rho_{i(SLD)}$, particle volume fraction ϕ , core particle average size \overline{r} , polydispersity p, and shell thickness $\delta \cdot S(q, \phi_{eff}, \overline{r}_{eff}, \tau, \varepsilon)$ is the structure factor, which contains information of interparticle interactions and spatial arrangement of particles; and it is a function of wavevector q, effective particle volume fraction ϕ_{eff} , effective particle average size \overline{r}_{eff} , interaction parameter τ (Baxter temperature) and ε (Perturbation parameter). The structure factor is calculated using the Ornstein–Zernike³⁹ (OZ) equation with the Percus–Yevick⁴⁰ (PY) closure approximation scaled by the mean effective particle diameter. For dilute dispersions, the interactions between particles can be ignored and the structure factor $S(q, \phi_{eff}, \overline{r}_{eff}, \tau, \varepsilon) \sim 1$. The expression for the core-shell form factor with Schultz distributed core and constant shell is

$$\overline{P}(q,\rho_i,\phi,\overline{r},p,\delta) = \int_0^\infty f(R,\overline{r},p) \Big[\big(\rho_{shell} - \rho_{solvent}\big) F(q,R+\delta) + \big(\rho_{core} - \rho_{shell}\big) F(q,R) \Big]^2 dR$$
5-8

where f(R) is the Schultz distribution function. The interference factor F(q, R) is that for a spherical object in the Rayleigh–Gans–Debye limit:⁴¹

$$F(q,R) = 3V_p \frac{\sin(qR) - qR\cos(qR)}{(qR)^3}.$$
 5-9

5.3 Results and Discussion

5.3.1 Dilute Particle Dispersions – Single Particle Properties

The silica nanoparticles as supplied are dispersed in water, but will aggregate if they are dispersed into $[C_4mim][BF_4]$. This is in direct contrast to the surface functionalized nanoparticles - fluorocarbon-coated nanoparticles, which readily disperse and remain stable in $[C_4mim][BF_4]$ but do not disperse in water. TEM images comparing the dried fluorocarbon-coated silica particles and fluorocarboncoated silica nanoparticles dispersed in $[C_4mim][BF_4]$ are shown in Figure 5.3. The synthesis procedure for coating the fluorocarbon to the surface of the silica nanoparticle as discussed in Chapter 2 has no effect on the silica nanoparticle core shape or size. Figure 5.3B shows the dispersion of the particles by direct observation using *in situ* TEM, which is possible in the ionic liquid medium despite the high vacuum condition and electron beam irradiation of electron microscope chamber.² The lower resolution for the image in the ionic liquid is due in part to the Brownian
motion of the particles and possible flow of the dispersion on the grid during the imaging.



Figure 5.3: Transmission electron microscopy (TEM) images of (A) the dried fluorocarbon-coated particle and (B) fluorocarbon-coated particles in [C₄mim][BF₄].

The hydrodynamic radii of uncoated and coated silica nanoparticles in polar solvents (e.g., water, ethanol) and the ionic liquid are measured by dynamic light scattering (see Chapter 4 Section 4.3.2.2). The results in ethanol demonstrate that addition of the fluorocarbon coating increases the radius of particles (Table 5.2) by an amount consistent with the molecular size of the fluorocarbon (~1.4 nm) as calculated from the bond lengths and angles. However, there is an anomalous increase in hydrodynamic radius observed for the particles dispersed in the ionic liquid. Detailed analysis of the size distributions in Chapter 4 Section 4.3.2.2, coupled with the direct TEM observations, show that this increase in hydrodynamic size is not due to a small amount of aggregation, but rather, is the consequence of solvation layers forming near the particle surface that increase the hydrodynamic size of the particles. Note that the uncoated silica particles will not disperse in $[C_4mim][BF_4]$ even after sonication and

extensive mixing; DLS measurements on these systems yield a broader size distribution and a much larger particle size indicative of aggregation as discussed in Chapter 4 Section 4.3.2.2.

1		•	-
Measurement	Uncoated particle	Coated particle in	Coated particle in
Techniques	in H ₂ O	ethanol	$[C_4 mim][BF_4]$
DLS	$53.8\pm4.8~\text{nm}$	$55.2 \pm 3.8 \text{ nm}$	$60.4 \pm 3.0 \text{ nm}$
SANS	$54.2 \pm 5.4 \text{ nm}$	Core: 54.2 ± 5.4 nm	Core: $54.2 \pm 5.4 \text{ nm}$
	<i>p</i> = 10.5 %	<i>p</i> = 10.5 %	<i>p</i> = 10.5 %
Brush		0.9 nm	1.2 nm
Schematic		A11255 6	-TILLISS.
illustration		Jose Contraction	State and the second
		185% 2018	
		Cartiness werent	Contraction of the second

Table 5.2:Summary of particle hydrodynamic radii for uncoated and coated
particles in different solvents measured by various technique methods.

5.3.2 DLVO Potential between Two Silica Particles

The interaction potentials between two silica particles in solution can be calculated from the DLVO theory. The parameters used to calculate the van der Waals, electrostatic, and DLVO potentials are listed in Table 5.3. The DLVO potential profiles of two uncoated silica nanoparticles (R = 54.2 nm) in [C₄mim][BF₄] and mother liquid at 25 °C are shown in Figure 5.4 as a function of the surface separation distance *d*. The DLVO potential for silica nanoparticles in the aqueous mother liquid is positive for *d* > 0.1 nm (Figure 5.4B), indicating long-range repulsion leading to particle stabilized by electrostatic forces, in agreement with observation.

However, for silica nanoparticles dispersed in $[C_4mim][BF_4]$ as shown in Figure 5.4A, the electrostatic repulsion is screened and the DLVO potential is negative, indicating the particles are attracted to each other at all separation distances. Thus, the DLVO calculations predict that the uncoated silica particles can be electrostatically stabilized in mother liquid, but not in $[C_4mim][BF_4]$, which agrees with experimental results.

The addition of the fluorocarbon coating will provide a steric barrier to keep particles apart. However, the calculations in Figure 5.4A show that the particles with the fluorocarbon will still have a weak attraction $O(k_BT)$ sufficient to induce weak flocculation if the coating acts as a ~ 1 nm steric layer keeping particle surfaces ~ 2 nm apart. However, DLS measurements show that the silica nanoparticles are surrounded by ~ 5 nm solvation layers such that minimum separations of $d \approx 10$ nm are anticipated. At this separation distance, the DLVO attractive potential between nanoparticles is negligible (Figure 5.4A), confirming that these coated nanoparticles can be stabilized by inducing the formation of solvation layers in [C₄mim][BF₄].

Table 5.3:Parameters for the calculation of the DLVO potential for colloidal
dispersions in $[C_4mim][BF_4]$ and H_2O /mother liquid at 25 °C.

Materials	Е	n	A $(10^{-21}J)$	$\psi_s(\mathrm{mV})$	M (mol/L)	κ^{-1} (nm)
Silica	4.42	1.46				
[C ₄ mim][BF ₄]	11.70	1.42	1.04	-9.48		0.4
H ₂ O / mother liquid	78.54	1.33	6.97	-36.57	1×10 ⁻⁴	30.5



Figure 5.4: Calculated DLVO interparticle potential profiles at T = 25 °C for silica particles in [C₄mim][BF₄] (A) and in 0.1 mM (pH = 9.1 ± 0.1) NaCl aqueous solution (mother liquid) (B).

5.3.3 Concentrated Particle Dispersions – Dispersion Microstructure

SANS measurements of dilute nanoparticle suspensions discussed in Chapter 4 Section 4.3.2.3 further quantify the particle size, polydispersity, and coating layer structure and are listed in Table 5.2. As the density of ionic liquid solvation layers is close to the bulk ionic liquid, there is no significant scattering length density difference between solvation layers and the ionic liquid. Therefore, the solvation layers are not detected directly by neutron scattering. However, SANS measurements of a concentration series enable detecting the particle stabilization mechanism through direct measurement of the structure factor.

SANS measurements on more concentrated dispersions of the coated particles in [C₄mim][BF₄] are reported in Figure 5.5, with scattering intensity I(q) versus scattering wave vector q systematically shifted vertically for clarity. With increasing particle concentration, a structure peak becomes evident for q < 0.01 Å⁻¹. The scattering intensities are analyzed by fitting to Equation 5-7 (defined in materials and methods section, Section 5.2.3) with an effective structure factor. From these fits, I determine the potential parameters that include an effective particle size and strength of attraction. In the analysis, the particle form factor is accurately represented by a polydisperse core-shell form factor model⁴² with all of the associated parameters measured from the dilute sample for single particle size discussed in Chapter 4.



Figure 5.5: SANS intensity *I versus* scattering wave vector q as a function of particle concentration for 25 °C and the corresponding sticky-hard sphere model fits. (The scattering intensities are systematically shifted by a factor of 10 for clarity.)



Figure 5.6: The SANS intensity of $\phi = 0.44$ coated particle [C₄mim][BF₄] dispersion with sticky-hard sphere and hard sphere model fits.

The solvation layers detected by DLS should also be evident in the structure factor as they provide steric stability and thus, prevent particles from approaching closely. Consequently, solvation layers will lead to a dispersion structure reflecting this increased, larger particle size. Examination of Figure 5.6 shows that an effective hard sphere potential with an effective radius that includes the solvation layer can capture the location of the intensity peak, but is still not sufficient to describe the suspension microstructure as there are still weak interparticle attractions. To improve the description, I use the simplest model that includes both excluded volume and attractions - the "sticky" effective hard sphere model - and calculate the structure

factor S(q) accordingly.⁴² This structure factor depends on the effective volume fraction (defined in Equation 5-7 in materials and method section, Section 5.2.3) in a self-consistent manner as the number of particles and their effective volume is known in terms of the effective radius. The particles' effective volume fraction can be extracted from the structure factor modeling, keeping the form factor parameters fixed. The sticky hard sphere structure factor fitting parameters and the resultant thicknesses of the solvation layer at each volume fraction are shown in Table 5.4. The solvation layers thickness determined in this manner is slightly less than, but comparable to, the value of 5 nm ascertained from the DLS measurements. Note that the accuracy of the analysis improves with increasing particle concentration as the structure becomes more prominent. Attempts to fit the data with various potential models but without including the solvation shell were unsuccessful to capture the neutron scattering profiles. The increase in effective solvation layer thickness with increasing particle concentration trend is expected in part as close approach of surfaces creates confinement, which can induce more ordering in ionic liquids. It is reported in literature that the solvation layers between confined flat surfaces become more ordered as the surface separation decreases, resulting in a thicker solvation layer.⁴³

Parameters	$\phi = 0.100$	$\phi = 0.207$	$\phi = 0.295$	$\phi = 0.352$	$\phi = 0.436$
Weight percent	17%	32%	44%	50%	61%
$\phi_{\!\!eff}$	0.106	0.225	0.340	0.441	0.545
\overline{r} (nm)	54.2	54.2	54.2	54.2	54.2
р	0.105	0.105	0.105	0.105	0.105
δ (shell, nm)	1.2	1.2	1.2	1.2	1.2
$SLD_{core}(A^{-2})$	3.47×10^{-6}				
$SLD_{shell}(A^{-2})$	3.24×10^{-6}				
$SLD_{solvent}(A^{-2})$	1.40×10^{-6}				
Perturbation ε	0.01	0.01	0.01	0.01	0.01
Baxter parameter					
τ	0.70	0.59	0.32	0.16	0.46
Background(cm ⁻¹)	0.56	0.50	0.45	0.39	0.32
$\overline{r}_{eff}(nm)$	56.3	56.9	58.1	59.7	59.7
Δr_{solv} (nm)	0.9	1.5	2.7	4.3	4.3

Table 5.4:Summary of structure factor fitting parameters for dispersions coated
nanoparticles in [C4mim][BF4] corresponding to Figure 5.5.

* The effective volume fraction ϕ_{eff} and stickiness parameter τ are set to float during fitting and the other parameters are set to be fixed value. The average effective particle radii \bar{r}_{eff} and solvation layer thickness Δr_{solv} are calculated from effective volume fraction.

The rheological properties of dispersions depend strongly on nanoparticle stability and interparticle interactions, such as solvation layers.¹⁸ Rheological measurements were performed to further explore the effect of the solvation layer formation in the ionic liquid. A suspension of coated nanoparticles dispersed in [C₄mim][BF₄] at moderate concentrations ($\phi = 0.2$) displays near Newtonian behavior, as shown in Figure 5.7. This further confirms their excellent dispersion due to the effect of the coating. The corresponding zero shear and high shear relative viscosity are $\eta_{r,0} = 2.65 \pm 0.05$ and $\eta_{r,\infty} = 2.51 \pm 0.03$, respectively. These viscosities are larger than expected for a hard sphere dispersion based on the silica volume fraction,⁴⁴⁻⁴⁵ as shown in Figure 5-7. Mode coupling theory with $\phi_g = 0.57$ is

employed for hard sphere zero shear relative viscosity,⁴⁵ while the van der Werff correlation with $\phi_m = 0.71$ is used for hard sphere high shear relative viscosity.⁴⁴ The extent of the solvation layer responsible for this increase in viscosity can be deduced from effective particle sizes obtained by fitting the measured zero and high shear viscosities to hard sphere correlations. These values are shown in Table 5.5 and the corresponding solvation layer thicknesses are calculated to be 5.2 ± 0.2 nm and 4.3 ± 0.3 nm, respectively for the dispersion. The results are in good agreement with the previous discussed solvation layer thickness determined by DLS and SANS measurements. Further, the reduction in solvation layer thickness at high shear stress is expected as these layers can be displaced by force; however, the accuracy of this method is not sufficient to draw further conclusions about this reduction.



Figure 5.7: Relative viscosity (with error bar) *versus* shear stress for $\phi = 0.2$ and 0.3 stable suspension of fluorocarbon-coated silica nanoparticles in [C₄mim][BF₄]. Closed symbols are ascending stress curves and the open symbols are descending stress curves. The solid lines are the corresponding relative zero shear viscosity; the dashed lines are the corresponding relative high shear viscosity.

Interestingly, at higher concentration ($\phi = 0.3$), the suspension shows slightly shear thickening for shear stresses exceeding 4 Pa. This is to be expected for such dispersions with stabilizing layers that are only slightly permeable to the suspending medium, such as solvation layers.⁴⁶⁻⁴⁷ Further, the zero shear viscosity is not well defined and there is some hysteresis in the experiments, both of which are evidence for weak attractive interactions. The same analysis is applied only for the high shear relative viscosity, which is taken as the minimum of the viscosity curve. The thickness of the solvation layer is estimated to be 5.3 ± 0.2 nm. The corresponding zero shear viscosity is predicted using the effective volume fraction obtained from fitting the high shear viscosity as shown in Figure 5.7, and this value agrees with measured values to within the variation in the data.

These solvation layer thicknesses calculated from rheological measurements are in good agreement with the hydrodynamic radius from DLS and those obtained from SANS structure factor measurements. Interestingly, both the SANS and the rheological measurements suggest the presence of weak attractive interactions that are not expected based on the simple DLVO interaction, suggesting that the van der Waals forces are underestimated for this system.

Table 5.5: Summary of relative viscosity, effective volume fraction and solvation layer thickness calculated from shear flow curves of $\phi = 0.2$ and 0.3 suspensions.

	$\phi = 0.209$		$\phi = 0.309$		
Parameters	Zero Shear	High Shear	Zero Shear	High Shear	
$\eta_r(HS)$	1.95	2.00	3.25	3.13	
η_r (Measured)	2.65 ± 0.05	2.51 ± 0.03	$7.34\pm0.49^*$	5.46 ± 0.20	
$\phi_{_{e\!f\!f}}$	0.274 ± 0.003	0.262 ± 0.002	$0.406 \pm 0.006 *$	0.406 ± 0.006	
$\overline{r}_{eff}(nm)$	60.6 ± 0.1	59.7 ± 0.2	$60.7\pm0.1*$	60.7 ± 0.1	
$\Delta r_{solv}(nm)$	5.2 ± 0.2	4.3 ± 0.3	$5.3\pm0.2*$	5.3 ± 0.2	

* ϕ_{eff} is calculated from high shear relative viscosity of $\phi = 0.309$ dispersion and is used to predict the corresponding zero shear relative viscosity η_r (*Measured*).

5.4 Conclusions

I have demonstrated a method to spontaneously disperse silica particles in ionic liquid [C₄mim][BF₄] by inducing solvent structuring around the surface of the particle. Extensive characterizations of fluorocarbon (1H,1H,9H hexadecafluoro-1nonanol) coated silica nanoparticles suspended in ionic liquid [C₄mim][BF₄] using DLS, SANS, TEM, and rheology all confirm the presence of solvation layers of approximately 5 nm. This layer is sufficient to provide steric stabilization against the dispersion attractions. The thickness of this layer is quantitatively similar to that measured for this ionic liquid structuring near flat mica surfaces using surface forces apparatus.^{13, 29} The 5 nm, layer thickness corresponds to seven ion-pair layers, each with 0.7 to 0.8 nm thickness as measured using surface forces.²⁷ The number of layers is also consistent with the measurements of Perkin et al.^{29, 48} on the structuring of [C₄mim][NTf₂] confined between mica.

In much of the analyses presented here it is assumed for simplicity that the solvation layers form a well-defined, uniform shell surrounding the nanoparticles. However, the solvation layers near flat substrates are expected to become less organized and smoothly transition to the bulk liquid, as shown by surface force apparatus (SFA),²⁹⁻³⁰ atomic force microscope (AFM) measurements,^{13, 25, 43} and simulations⁴⁹. Closer examination of the variations in shell thickness determined at varying particle concentrations and shear rates, and comparison of dynamic and static measurements, suggests that a more detailed analysis can provide further information about the nanostructure of this solvation layer. The molecular structure for confined layers of related ionic liquids have been proposed based on surface forces measurements and molecular packing parameters.³⁰ Such an analysis will require a full theory for the dispersion forces and electrostatic forces acting between highly curved surfaces, such as those of the nanoparticles, with such nanostructured solvation layers, which goes well beyond the scope of this work. The comprehensive data

166

presented here, collected using a variety of measurement techniques, not only provides a database with which such theories can be rigorously tested, but also acts as guidance for the development of future theories.

Molecular design of a surface coating to induce specific molecular interactions between the particle surface and the ionic liquid with the intent to create solvent structuring is demonstrated as a method to create spontaneous dispersion and stabilization of nanoparticles that are otherwise not dispersible in ionic liquids. The resultant solvation layers impart colloidal stability, but also lead to larger effective volume fractions in solution. The presence of ubiquitous dispersion forces and the solvation layer will ultimately limit the total concentration of particles dispersible in the ionic liquids; however, I have successfully dispersed these particles up to 61 wt% (corresponding to 45 vol%), which is a significant improvement on the current published maximum of 20 wt% silica particle content.

REFERENCES

1. Antonietti, M.; Kuang, D.; Smarsly, B.; Zhou, Y. A. C., Int. Ed. 2004, 43, 4988. Ionic Liquids for the Convenient Synthesis of Functional Nanoparticles and Other Inorganic Nanostructures. *Angew. Chem.* **2004**, *43*.

2. Scheeren, C. W.; Machado, G.; Dupont, J.; Fichtner, P. F.; Texeira, S. R. Nanoscale Pt (0) particles prepared in imidazolium room temperature ionic liquids: synthesis from an organometallic precursor, characterization, and catalytic properties in hydrogenation reactions. *Inorg. Chem.* **2003**, *42* (15), 4738-4742.

3. Wei, G.-T.; Yang, Z.; Lee, C.-Y.; Yang, H.-Y.; Wang, C. R. C. Aqueous-Organic Phase Transfer of Gold Nanoparticles and Gold Nanorods Using an Ionic Liquid. *J. Am. Chem. Soc.* **2004**, *126* (16), 5036-5037.

4. Welton, T. Room-temperature ionic liquids. Solvents for synthesis and catalysis. *Chem. Rev.* **1999**, *99* (8), 2071-2084.

5. Migowski, P.; Dupont, J. Catalytic applications of metal nanoparticles in imidazolium ionic liquids. *Chemistry-A European Journal* **2007**, *13* (1), 32-39.

6. Lopez-Barron, C. R.; Basavaraj, M. G.; DeRita, L.; Wagner, N. J. Sponge-to-Lamellar Transition in a Double-Tail Cationic Surfactant/Protic Ionic Liquid System: Structural and Rheological Analysis. *J. Phys. Chem. B* **2012**, *116* (2), 813-822.

7. Lopez-Barron, C. R.; Li, D. C.; DeRita, L.; Basavaraj, M. G.; Wagner, N. J. Spontaneous Thermoreversible Formation of Cationic Vesicles in a Protic Ionic Liquid. *J. Am. Chem. Soc.* **2012**, *134* (51), 20728-20732.

8. López-Barrón, C. R.; Li, D.; Wagner, N. J.; Caplan, J. L. Triblock Copolymer Self-Assembly in Ionic Liquids: Effect of PEO Block Length on the Self-Assembly of PEO–PPO–PEO in Ethylammonium Nitrate. *Macromolecules* **2014**, *47* (21), 7484-7495.

9. Bai, Z.; Lodge, T. P. Polymersomes with Ionic Liquid Interiors Dispersed in Water. J. Am. Chem. Soc. **2010**, 132 (45), 16265-16270.

10. Ueno, K.; Inaba, A.; Kondoh, M.; Watanabe, M. Colloidal stability of bare and polymer-grafted silica nanoparticles in ionic liquids. *Langmuir* **2008**, *24* (10), 5253-5259.

11. Ueno, K.; Hata, K.; Katakabe, T.; Kondoh, M.; Watanabe, M. Nanocomposite ion gels based on silica nanoparticles and an ionic liquid: ionic transport, viscoelastic properties, and microstructure. *J. Phys. Chem. B* **2008**, *112* (30), 9013-9019.

12. Ueno, K.; Imaizumi, S.; Hata, K.; Watanabe, M. Colloidal interaction in ionic liquids: Effects of ionic structures and surface chemistry on rheology of silica colloidal dispersions. *Langmuir* **2009**, *25* (2), 825-831.

13. Ueno, K.; Watanabe, M. From colloidal stability in ionic liquids to advanced soft materials using unique media. *Langmuir* **2011**, *27* (15), 9105-9115.

14. Smith, J. A.; Werzer, O.; Webber, G. B.; Warr, G. G.; Atkin, R. Surprising particle stability and rapid sedimentation rates in an ionic liquid. *The Journal of Physical Chemistry Letters* **2009**, *1* (1), 64-68.

15. Smith, J.; Webber, G. B.; Warr, G. G.; Atkin, R. Silica Particle Stability and Settling in Protic Ionic Liquids. *Langmuir* **2014**, *30* (6), 1506-1513.

16. Wittmar, A.; Ruiz-Abad, D.; Ulbricht, M. Dispersions of silica nanoparticles in ionic liquids investigated with advanced rheology. *Journal of Nanoparticle Research C7 - 651* **2012**, *14* (2), 1-10.

17. Wittmar, A.; Gajda, M.; Gautam, D.; Dörfler, U.; Winterer, M.; Ulbricht, M. Influence of the cation alkyl chain length of imidazolium-based room temperature ionic liquids on the dispersibility of TiO2 nanopowders. *Journal of nanoparticle research* **2013**, *15* (3), 1-12.

18. Mewis, J.; Wagner, N. J. *Colloidal suspension rheology*; Cambridge University Press: New York, 2012.

19. Israelachvili, J. N. *Intermolecular and Surface Forces, 3rd Edition*; Elsevier Academic Press Inc: San Diego, 2011. p 1-674.

20. Lanning, O. J.; Madden, P. A. Screening at a charged surface by a molten salt. *J. Phys. Chem. B* **2004**, *108* (30), 11069-11072.

21. Baldelli, S. Probing electric fields at the ionic liquid–electrode interface using sum frequency generation spectroscopy and electrochemistry. *J. Phys. Chem. B* **2005**, *109* (27), 13049-13051.

22. Lynden-Bell, R. M.; Del Pópolo, M. G.; Youngs, T. G. A.; Kohanoff, J.; Hanke, C. G.; Harper, J. B.; Pinilla, C. C. Simulations of ionic liquids, solutions, and surfaces. *Acc. Chem. Res.* **2007**, *40* (11), 1138-1145.

23. Feng, G.; Qiao, R.; Huang, J.; Dai, S.; Sumpter, B. G.; Meunier, V. The importance of ion size and electrode curvature on electrical double layers in ionic liquids. *Phys. Chem. Chem. Phys.* **2011**, *13* (3), 1152-1161.

24. Nordström, J.; Aguilera, L.; Matic, A. Effect of lithium salt on the stability of dispersions of fumed silica in the ionic liquid BMImBF4. *Langmuir* **2012**, *28* (9), 4080-4085.

25. Hayes, R.; Warr, G. G.; Atkin, R. At the interface: solvation and designing ionic liquids. *Phys. Chem. Chem. Phys.* **2010**, *12* (8), 1709-1723.

26. Atkin, R.; Warr, G. G. Structure in confined room-temperature ionic liquids. *The Journal of Physical Chemistry C* **2007**, *111* (13), 5162-5168.

27. Ueno, K.; Kasuya, M.; Watanabe, M.; Mizukami, M.; Kurihara, K. Resonance shear measurement of nanoconfined ionic liquids. *Phys. Chem. Chem. Phys.* **2010**, *12* (16), 4066-4071.

28. Segura, J. J.; Elbourne, A.; Wanless, E. J.; Warr, G. G.; Voïtchovsky, K.; Atkin, R. Adsorbed and near surface structure of ionic liquids at a solid interface. *Phys. Chem. Chem. Phys.* **2013**, *15* (9), 3320-3328.

29. Perkin, S.; Crowhurst, L.; Niedermeyer, H.; Welton, T.; Smith, A. M.; Gosvami, N. N. Self-assembly in the electrical double layer of ionic liquids. *Chem. Commun.* **2011**, *47* (23), 6572-6574.

30. Smith, A. M.; Lovelock, K. R. J.; Perkin, S. Monolayer and bilayer structures in ionic liquids and their mixtures confined to nano-films. *Faraday Discuss.* **2013**, *167* (0), 279-292.

31. Pinilla, C.; Del Pópolo, M. G.; Lynden-Bell, R. M.; Kohanoff, J. Structure and dynamics of a confined ionic liquid. Topics of relevance to dye-sensitized solar cells. *J. Phys. Chem. B* **2005**, *109* (38), 17922-17927.

32. Szilagyi, I.; Szabo, T.; Desert, A.; Trefalt, G.; Oncsik, T.; Borkovec, M. Particle aggregation mechanisms in ionic liquids. *Phys. Chem. Chem. Phys.* **2014**, *16* (20), 9515-9524.

33. M.B.Shiflett; Yokozeki, A. Solubility of Fluorocarbons in Room Temperature Ionic Liquids. In *Ionic Liquids: From Knowledge to Application*, Plechkova, N. V.; Rogers, R. D.; Seddon, K. R., Eds.; American Chemical Society, **2009**, Vol. 1030, pp 21-42.

34. Smith, J. A.; Werzer, O.; Webber, G. B.; Warr, G. G.; Atkin, R. Surprising particle stability and rapid sedimentation rates in an ionic liquid. *J. Phys. Chem. Lett.* **2009**, *1* (1), 64-68.

35. Hiemenz, P. C.; Rajagopalan, R. *Principles of Colloid and Surface Chemistry, revised and expanded*; CRC Press1997; Vol. 14.

36. Russel, W. B.; Saville, D. A.; Schowalter, W. R. *Colloidal dispersions*; Cambridge university press **1992**.

37. Baldelli, S. Probing Electric Fields at the Ionic Liquid–Electrode Interface Using Sum Frequency Generation Spectroscopy and Electrochemistry. *J. Phys. Chem. B* **2005**, *109* (27), 13049-13051.

38. Jacquemin, J.; Ge, R.; Nancarrow, P.; Rooney, D. W.; Gomes, M. F. C.; Padua, A. A. H.; Hardacre, C. Prediction of ionic liquid properties. I. Volumetric properties as a function of temperature at 0.1 MPa. *J. Chem. Eng. Data* **2008**, *53* (3), 716-726.

39. Ornstein, L. S.; Zernike, F. Acculental deviations of density and opalescence at the critical point of a simple substance. *Proceedings of the Koninklijke Akademie Van Wetenschappen Te Amsterdam* **1914**, *17*, 793-806.

40. Percus, J. K.; Yevick, G. J. Analysis of Classical Statistical Mechanics by Means of Collective Coordinates. *Physical Review* **1958**, *110* (1), 1-13.

41. Glatter, O.; Kratky, O. Small angle X-ray scattering; Academic press 1982.

42. Kline, S. R. Reduction and analysis of SANS and USANS data using IGOR Pro. *J. Appl. Crystallogr.* **2006**, *39* (6), 895-900.

43. Atkin, R.; Warr, G. G. Structure in confined room-temperature ionic liquids. *J. Phys. Chem. C* **2007**, *111* (13), 5162-5168.

44. Van der Werff, J.; De Kruif, C. Hard - sphere colloidal dispersions: The scaling of rheological properties with particle size, volume fraction, and shear rate. *J. Rheol.* **1989**, *33* (3), 421-454.

45. Russel, W. B.; Wagner, N. J.; Mewis, J. Divergence in the low shear viscosity for Brownian hard-sphere dispersions: At random close packing or the glass transition? *J. Rheol.* **2013**, *57* (6), 1555-1567.

46. Shenoy, S. S.; Wagner, N. J. Influence of medium viscosity and adsorbed polymer on the reversible shear thickening transition in concentrated colloidal dispersions. *Rheol. Acta* **2005**, *44* (4), 360-371.

47. Lakshmi-narasimhan, K.; Wagner, N. J. The influence of weak attractive forces on the microstructure and rheology of colloidal dispersions. *Journal of Rheology* **2005**, *49* (2), 475-499.

48. Perkin, S. Ionic liquids in confined geometries. *Phys. Chem. Chem. Phys.* **2012**, *14* (15), 5052-5062.

49. Pinilla, C.; Del Pópolo, M. G.; Lynden-Bell, R. M.; Kohanoff, J. Structure and dynamics of a confined ionic liquid. Topics of relevance to dye-sensitized solar cells. *J. Phys. Chem. B* **2005**, *109* (38), 17922-17927.

Chapter 6

ALCOHOL COATED PARTICLE SYSTEM – SHEAR THICKENING, AND THERMAL RESPONSE

In this chapter, I describe a successful attempt to formulate stable nanoparticle dispersions in ionic liquids that exhibit significant shear thickening through designing the surface coating of particles (alcohol coating). A stable dispersion of silica nanoparticles in ionic liquid [C₄mim][BF₄] that exhibits significant shear thickening is formulated by controlling the strength of hydrogen bonding between the nanoparticle surface and the anion of $[C_4 mim][BF_4]$. Colloidal stability is demonstrated to be caused by the formation of solvation layers, the properties of which are determined from scattering measurements. Time-temperature superposition measurements and the onset of yielding behavior in steady shear indicate the increase of particle-particle attraction and loss of stability of colloidal dispersions above 30 °C. Small angle neutron scattering (SANS) and dynamic light scattering (DLS) measurements confirm the reduction in solvation layering at elevated temperatures, leading to a transition from stable dispersion to unstable, attractive gel at increased temperature (*i.e.*, an inverse melting transition). Thus, controlling specific surface-anion interactions through chemical functionality can be used to formulate nanoparticle dispersions in ionic liquids with specific rheological properties. Text and figures are reproduced and adapted with permission from Gao J., et al. Journal of Rheology 2016 (submitted).

6.1 Introduction

Stable, concentrated colloidal suspensions that exhibit a reversible increase in viscosity when subjected to increasing shear rate are known as shear thickening fluids (STFs).¹⁻² This shear thickening transition is the result of a competition between hydrodynamic and stabilizing forces, such that a rearrangement of the microstructure occurs during strong shear flow. At high shear rate/stress, colloidal suspensions are dominated by short-range lubrication interactions that lead to density fluctuations of particle during shear and ultimately, stress-bearing clusters named as "hydroclusters".³ Control of shear thickening in colloidal dispersions is relevant to numerous industrial products and processes, such as paints, lubricants, pharmaceutical, as well as protective materials such as "soft" body armor.⁴ Typical shear thickening fluids are formulated in molecular solvents, such as polyethylene glycol (PEG) or ethylene glycol (EG) as carrier medium. Recently, it was reported that stable concentrated silica particles in room temperature ionic liquids (ILs) dispersions display reversible shear thickening under high shear rate/stress.⁵⁻⁶ Room temperature ILs are organic salts in the liquid state at room temperature, which are comprised entirely of ions.⁷⁻⁸ Due to the unique physicochemical properties of room temperature ILs, such as low melting temperature and negligible vapor pressure, IL dispersed shear thickening fluids can be used in extreme environments compared to PEG/EG based STFs, such as applications as field-responsive nanocomposites for protecting astronauts against hazards from micrometeorite and orbital debris (MMOD).⁹⁻¹⁰

The rheological response of colloidal dispersions strongly depends on particle stability and microstructure. Particle stabilization in ILs is challenging because their ionic nature screens electrostatic repulsive forces. Thus, not surprisingly, it has been reported that particle dispersion and stability in ILs depends on the specific chemical

174

composition of the ILs as well as the surface chemistry of the particles.^{5-6, 11-13} The first report of suspension rheology in ILs was for Hematite colloids in ethylmethylimidazolonium ethylsulfate, where a stable, Newtonian dispersion was observed for lower particle loadings, but a yielding dispersion was observed at higher loadings, without shear thickening.¹¹ Hydrophilic fumed silica particles were investigated to be well dispersed in $[C_4mim][BF_4]$ displaying shear thickening.⁵ However, such dispersions were unstable in 1-ethyl-3-methylimidazolium bis(trifluoromethane sulfonyl)-amide ($[C_2mim][NTf_2]$) and the dispersions exhibited a yielding behavior and strong shear thinning.⁵ In addition, aggregation and gelation were reported for hydrophobic fumed silica particles in $[C_4mim][BF_4]$,⁵ demonstrating that it is not trivial to disperse particles into ILs.

In the original work by Altin *et al.*, the authors point to the unique properties of ILs with respect to their nanoscale structure, such that particles, *i.e.* the hematite, can be stabilized in solution by structured solvent surrounding the particles.¹¹ This picture was formalized by Ueno *et al.*⁵ and proven experimentally by Gao *et al.*⁶ The stability of fumed hydrophilic silica particles in $[C_4mim][BF_4]$ in the absence of any stabilizer⁵, ¹⁴⁻¹⁵ is due to the presence of solvation forces arising from solvent structuring near the particle surfaces.^{6, 14-16} Derjaguin-Landau-Verwey-Overbeek (DLVO) theory, which accounts for van der Waals attraction and electrostatic repulsion between particles in solution, ¹⁷ was used to evaluate the interaction between particles in dispersions. The high ionic strength in ILs effectively screens the electrostatic repulsion between silica particles, ¹⁸⁻²⁰ resulting in loss of stability of the particles in ILs by the van der Waals attraction between particles in ILs by the van der Waals attraction between particles in ILs by the van der Waals attraction between particles in ILs by inducing solvation layers to form around particles through suitably functionalizing the surface

to generate hydrogen bonds with the anion of the IL. This couples to the nanostructure in the IL to create alternating layers of anions and cations that are known as solvation layers. Such ionic ordering of the solvent propagates some distance into the IL and provides a repulsive force acting between two particle surfaces dispersed in the IL. This was hypothesized as the mechanism for stabilizing fumed silica dispersed in $[C_4mim][BF_4]$, where surface silanol groups hydrogen bond with the anion and create solvation layers provide sufficient repulsion against the weak van der Waals attractions of porous aggregated fumed silica particles.⁵⁻⁶ Indeed, solvation forces have been confirmed and measured both between macroscopic surfaces and colloidal particles in ILs. Distant-dependent solvation forces have been observed between macroscopic mica or silica surfaces in both protic and aprotic ILs by surface force measurements.²²⁻²⁷ These reports are also consistent with molecular dynamics simulation results of solvation layers formation driven by specific ionsurface interactions.²⁸ Both measurements and simulations indicate that the ion pairs are arranged in a layered structure near the solid surfaces.^{6, 26} Solvation structuring in ILs around nanoparticle surfaces was also measured directly on spherical silica particles purposefully surface functionalized with fluorinated hydrocarbons to disperse in an aprotic IL.⁶ The hydrogen bonding between the fluorinated anion of the IL $[C_4 mim][BF_4]$ and the terminal hydrogen atom on the surface coating drives the formation of solvation layers that impart colloidal stability by steric repulsion. The thickness of solvation layers was determined to be approximately 5 nm at room temperature by transmission electron microscopy (TEM), rheology, DLS, and SANS.⁶ Importantly, such surface solvation layers, being just surface-organized structures comparable to the sponge-like structure in the IL itself, are thought to be fluid-like in

176

the direction parallel to the solid surface.²⁹ Consequently, they act to stabilize the dispersion but mitigate the shear thickening response.

Grafting polymer to the particle surface can also improve suspension stability in ILs^{16, 21} if the IL is a good solvent for the polymer. It was reported that poly(methyl methacrylate) (PMMA)-grafted silica nanoparticles show good dispersion in 1-butyl-3-methylimidazolium hexafluorophosphate ($[C_4mim][PF_6]$) and 1-alkyl-3methylimidazolium bis(trifluoromethylsulfonyl)imide ($[C_n mim][NTf_2]$), presumably because PMMA is itself soluble in this IL; however, such coated particles show poor dispersion if PMMA is insoluble in the IL (e.g., $[C_4mim][BF_4]$).²¹ Similar phenomenon can be found for grafting polyelectrolytes, such as poly(acrylic acid) (PAA) coated maghemite nanoparticles in a protic IL ethylammonium nitrate (EAN).³⁰ The stability of the nanoparticles was controlled by tuning the pH through changing the solvation quality of EAN for the polymer chains comprising the coating on the particles.³⁰ PAA-coated particles are stable at acidic pH, when coated polymer chains are fully protonated and can provide steric repulsive interactions. At basic pH, polymer chains are deprotonated due to the poor solvation quality of EAN, resulting in the aggregation of coated particles. For both of the cases, the mechanism of stability is afforded by the steric repulsion between the grafted layers of significant extent and as is typical with such systems, the system will become unstable when the brush is immersed into a poor solvent and collapses.² The stability provided by surface grafted polymers through steric repulsion is reported to be thermal-sensitive. Poly(benzyl methacrylate) (PBnMA)-grafted silica nanoparticles^{16, 31} were stabilized in [C₂mim][NTf₂] through steric forces. Highly concentrated suspensions formed stable soft glassy colloidal dispersion at low temperature; however, a temperature-induced

177

colloidal glass-to-gel transition was observed with a discrete change in the color of dispersion with increasing temperature. The particle interaction changed from repulsive to attractive with increasing temperature, as deduced from the non-monotonic behavior of the viscosity.^{16, 31} However, such polymer-stabilized dispersions do not exhibit shear thickening rheology in the regimes studied presumably due to the strong nanoscale repulsive forces afforded by the well-solvated polymer brush.³²⁻³³

This concise literature summary of particle dispersions in ILs suggests a hypothesis that by controlling the strength of anion-colloid interactions, I can induce solvation layering sufficient to provide steric stability but not to an extent that suppresses shear thickening at reasonable shear rates. In this study, I test this hypothesis by studying the temperature dependence of solvation layering and the rheology of a model dispersion of coated silica particles in a typical, aprotic IL $[C_4 mim][BF_4]$. The effect of solvation layers thickness on shear rheology is investigated by comparing silica particles with varying surface functionality designed to vary the strength of hydrogen bonding with the IL anion.⁶ Furthermore, high temperature is used to weaken this hydrogen bonding and hence, reduce the solvation stabilization, leading to a loss of shear thickening and eventually, gelation and an inverse melting transition. Particle aggregation due to the loss of stability at high temperatures is confirmed directly by DLS measurements. SANS measurements are used to investigate the corresponding solvation layers and particle interaction change with temperature. As will be shown, this dramatic transition in rheological behavior is explained by the strength of attraction between the particles as controlled by the extent of solvation layering induced by temperature sensitive hydrogen bonds between surface coating and anion group $[BF_4]^-$ of $[C_4mim][BF_4]$.

6.2 Materials and Methods

6.2.1 Materials

The IL [C₄mim][BF₄] (99% pure, Iolitec, $\rho_m = 1.201 \text{ g/cm}^3$, $\eta_m = 101.4 \text{ mPa} \cdot s$), was dehydrated in vacuum oven at 70 °C for 24 h prior to use. The final water content was 373 ppm as determined by the Karl Fischer titration. The rheological properties of this IL and its sensitivity to water have been reported elsewhere.³⁴ Silica nanoparticles (Seahostar KE-P10, Nippon Shokubai Co., Tokyo, Japan) of radius *a* = 60 nm and particle density $\rho_p = 1.89 \pm 0.02 \text{ g/cm}^{3.35}$ were used to prepare colloidal dispersions. The volume fraction is calculated using these densities. KE-P10 silica particles contain ~ 10% organic alcohol as surface functionalization according to the manufacture.³⁶ The nanoparticles were dispersed in the dehydrated IL at varying concentrations from dilute through concentrated. The colloidal dispersions were mixed using a roll mixer for one week to insure homogeneity throughout the sample, followed by a 5 min degassing to remove air bubbles in the samples. As the particle stability and structure of solvation layer are sensitive to water,³⁷⁻⁴⁰ silica particles were dried in oven at 80 °C for 10 h before dispersing in [C₄mim][BF₄] and the dispersions were dehydrated again in a vacuum oven at 50 °C for 24 h prior to each measurement. The water contents of dispersions were about 1000 ppm as measured by Karl Fischer titration.

6.2.2 Rheology

Rheological measurements were performed on a stress controlled Discovery Hybrid Rheometer (DHR) (TA Instruments, New castle, DE) with Peltier plate temperature control, and a 40 mm stainless steel cone with 2° cone angle with solvent trap. To erase any previous shear histories and maintain a consistency between measurements, a steady pre-shear was applied at a shear stress of 10 Pa for 2 min before each dynamic rheological measurement. Samples were stable and reproducible during the measurements as determined by comparing the properties during this preshear step. Prolonged exposure to the atmosphere (~days) will induce significant rheological changes primarily due to the strong effect of small amounts of water on the rheological properties of the IL.³⁴

6.2.3 Dynamic Light Scattering (DLS)

The hydrodynamic radius R_H of the particles are determined from the intensity autocorrelation function (IACF) of dilute dispersions measured using a Brookhaven Instruments ZetaPals DLS instrument at 10 °C to 60 °C. The IACF is modeled using the Stokes-Einstein-Sutherland equation and assuming diffusive behavior. Measurements are performed after 10 min temperature equilibrium at each temperature. The incident photon laser wavelength is $\lambda = 658$ nm and the scattered light intensity was detected at a scattering angle $\theta = 90$ °. The DLS instrument setup, measurement protocol and analysis method follow prior protocols.⁶

6.2.4 Small-Angle Neutron Scattering (SANS)

SANS measurements were performed at the National Institute of Standards and Technology (NIST) Center for Neutron Research (NCNR) in Gaithersburg, MD on NGB30 beamline. Samples were loaded into demountable titanium cells with a 1 mm path length. The cell temperatures were maintained to ± 0.1 °C using the 10 position sample cell holder with fluid bath. A wide range in scattering wavevector q was obtained by combining scattering intensity from four different instrument configurations at detector distances of 1, 4, 13, and 15.3 m with lens. The wavelength for 1, 4, and 13 m configurations is 6.0 Å and 8.0 Å for 15.3 m with lens. The SANS scattering intensity profiles were reduced to absolute scale and analyzed with the NIST data reduction and analysis macros in IGOR Pro available from NIST.⁴¹ The total background was calculated from the slope of the Porod plot (Iq^4 versus q^4), and subtracted from all SANS spectra.

The SANS profiles are analyzed by IGOR software.⁴¹ Properties of the dispersions were determined by fitting the measured SANS intensity using a form factor model for Schultz distributed sphere particles and a monodisperse structure factor model that includes hard core repulsion with contact attraction (sticky-hard sphere model). More details about the models employed are provided in Chapter 5.⁶

6.3 Results and Discussion

6.3.1 Steady Shear and Shear Thickening

The ascending steady state viscosity curves of the $\phi = 0.4$ dispersion in $[C_4 \text{mim}][BF_4]$ over a broad range of temperatures are compared in Figure 6.1. I observed that the ascending and descending flow curves overlap without hysteresis at low temperatures (10 °C to 30 °C). However, at high temperatures above 40 °C, the flow curves no longer overlap and exhibit hysteresis. Nonetheless, the flow properties are thermal-reversible and no permanent particle aggregation is observed during thermal cycling.

At lower temperatures, from 10 °C to 30 °C, the colloidal dispersion displays a shear rheology characteristic of a stable dispersion of near-hard sphere particles, as reported previously for dispersions of these particles in various molecular solvents.³⁵ A marked shear thickening behavior is observed, which is more pronounced than for related particles with different surface functionality dispersed in the same IL medium.⁶ With increasing temperature, the dispersion viscosity decreases systematically due to the decrease of IL viscosity, but the shear thickening property is still evident. However, at 40 °C, this shear thickening behavior becomes less evident as there is a substantial increase in viscosity at low shear rates. Above 40 °C, shear thickening is not observed for this colloidal dispersion and the low shear viscosity grows dramatically, signaling the onset of a weak yield stress. The presence of shear thinning and yielding at these higher temperatures is characteristic of gelation or inverse melting,⁴² which suggests stronger attractions between particles leading to particle aggregation at these higher temperatures.



Figure 6.1: Steady shear viscosity as a function of shear rate for colloidal dispersion = 0.4 silica particle $- [C_4 mim][BF_4]$ at (a) low temperatures and (b) high temperatures. Viscosity data are acquired with ascending shear rate.

The variation in the thermal-rheological response is more evident by plotting the relative viscosity as a function of Péclet number (*Pe*) and shear stress, as shown in Figure 6.2. At low temperatures, the flow curves (10 °C to 30 °C) overlap with each other, as expected for Brownian suspensions governed largely by hydrodynamic interactions.⁴³ It is evident that the onset of shear thickening occurs at a nearly constant value of shear stress independent of temperature over this range. The onset of shear thickening in colloidal dispersions has been well studied and is shown to be stress-controlled.^{3, 44} Furthermore, the stress at the onset of shear thickening is a strong function of the particle size, such that the steady shear rheological data suggest the same particle size at these low temperatures. Remarkably, these overlapping curves also are in quantitatively consistent with the flow curve of the $\phi = 0.4$ dispersion of KE-P10 silica particle in PEG-200,³⁵ which is shown in Figure 6.2. That the flow curves are nearly identical for the silica particles dispersed in both PEG-200 and [C₄mim][BF₄] is an evidence of a stable silica nanoparticle dispersion in this IL over this temperature range.

When shown in a dimensionless plot such as Figure 6.2, the viscosity behavior of the dispersion above 30 °C is observed to increase dramatically relative to that expected for dispersions of near hard-spheres. Strong shear thinning behavior is evident at these higher temperatures. This is typical of weak gelation,² and is characteristic of strong inter-particle attractions at higher temperatures.⁴⁵ The onset of gelation and the loss of shear thickening in the regime of flow explored is consistent with a loss of stability of the silica particles in [C₄mim][BF₄] at these elevated temperatures. This is postulated to be induced by a loss of the solvation layer structure, which will be discussed in Section 6.3.3. Note, however, that decreasing the temperature restores the dispersion to its flowable, shear thickened state such that the gelation is thermo-reversible, but with an inverse temperature behavior of 'normal' gelation.



Figure 6.2: Steady shear relative viscosity as a function of *Pe* and shear stress of $\phi = 0.4$ silica particle – [C₄mim][BF₄] dispersion at different experimental temperatures. The relative viscosity curves are compared with the flow curve of $\phi = 0.4$ dispersion of same silica particles in polyethylene glycol (PEG-200) at 25 °C.³⁵

Importantly, these dispersions exhibit a stronger shear thickening than previously reported for fluorocarbon coated silica nanoparticles⁶ dispersed in the same IL at the same volume fraction $\phi = 0.4$ at 25 °C, as shown in the comparison plotted

in Figure 6.3. The critical shear stress for shear thickening of the dispersions of fluorocarbon coated silica particles in $[C_4mim][BF_4]$ is consistent with the KE-P10 silica particle dispersions in $[C_4 mim][BF_4]$ as well as for dispersions in PEG-200, which is expected due to the similar particle sizes (~ 100 nm in diameter).³ However, it is evident that the shear thickening strength of fluorocarbon coated silica particle dispersion is weaker than the KE-P10 silica particles in $[C_4mim][BF_4]$ and PEG-200 of volume fraction $\phi = 0.4$. The separation between hydroclustered particles in the shear thickened state is on the order of a nanometer for these colloidal dispersions, such that shear-thickening directly probes the nanoscale forces acting between particle surfaces.⁴⁶ For example, it has been reported that polymer grafted or adsorbed onto the particle surface can delay or even suppress shear thickening.^{1, 32-33, 37} In our systems, solvation layers of highly organized ionic liquid forming on the particle surface provide steric repulsion and particle stability in a manner similar to the steric repulsion arising from a grafted/adsorbed polymer brush. Consequently, it can be postulated that difference in shear thickening strength, compared to the fluorocarbon coated silica particle system, is a consequence of differences in the range and strength of the repulsive forces created by the solvation layers on the nanoscale. To further explore this hypothesis, some further insights about the thermal properties of the solvation layers can be obtained from linear viscoelastic measurements, as described next.

186



Figure 6.3: Comparison of steady shear relative viscosity of KE-P10 – $[C_4mim][BF_4]$ dispersion, KE-P10 – polyethylene glycol (PEG-200) dispersion,³⁵ and fluorinated silica particle – $[C_4mim][BF_4]$ dispersion⁶ of same volume fraction $\phi = 0.4$ at 25 °C.

6.3.2 Linear Viscoelasticity

The dynamic linear viscoelastic moduli of the $\phi = 0.4$ silica particle – $[C_4 mim][BF_4]$ dispersion are shown in Figure 6.4a for the temperatures corresponding to Figure 6.2. The loss modulus G" is always larger than the storage modulus G', indicating the dispersion is governed by the hydrodynamic forces for all frequencies studied in this paper.⁴⁷ Using time-temperature superposition method for colloids as described by Shikata and Pearson,⁴⁷ after subtracting off the high frequency limiting viscosity η_{∞} , the moduli are shifted by a horizontal shift factor a_T onto a master curve with the reference temperature $(T_{ref} = 25 \text{ °C})$, as shown in Figure 6.4b. The horizontal shift factors a_T for each temperature are defined in Figure 6.5 and compared with the ratio of the viscosity of dispersion medium $[C_4 mim][BF_4]$ at the experimental temperature to the viscosity of $[C_4 mim][BF_4]$ at the reference temperature $T_{ref} = 25 \text{ °C}$.

It has been demonstrated that hard-sphere colloidal suspensions governed by thermodynamic and hydrodynamic forces can be scaled by the viscosity of the suspending medium.⁴⁷ As shown in Figure 6.5, the shift factor a_T scales with the viscosity of the medium $[C_4 mim][BF_4]$ over the temperature range from 10 °C to 30 °C. This confirms that the linear response of the dispersion scales with the viscosity of the dispersion medium, and that the inter-particle interactions are not affected by changing temperature from 10 °C to 30 °C. However, for 40 °C and above, the shift factors are nearly constant with temperature, and importantly, significantly larger than would be expected from the thermal dependence of the medium viscosity. These deviations indicate a fundamental change in the particle-particle interaction with increasing temperature above 30 °C, which is consistent with the steady shear results. Importantly, the relative high frequency viscosity $\eta'_{r,\infty}$ exhibits the same qualitative trends, as evident in Figure 6.5. From 10 °C to 30 °C, the relative high frequency viscosities are nearly constant due to the liquid state of dispersion at these temperatures, while the relative high frequency viscosity $\eta'_{r,\infty}$ increases with increasing temperature above 40 °C. This is attributable to the presence of aggregation and gelation of the dispersion at high temperatures.

Both the steady shear and small amplitude oscillation shear rheology results indicate a stable and near-hard sphere silica colloidal dispersion in $[C_4mim][BF_4]$ at temperatures between 10 °C to 30 °C, and loss of stability above 30 °C. At lower temperatures, this IL based dispersion displays strong shear thickening, which is comparable to that for a similar dispersion in PEG-200 at the same solid volume fraction. In the following, the dispersion microstructure and inter-particle interactions will be explored by SANS measurement to understand the mechanism behind the strong shear thickening properties at room temperature and the cause of the loss of stability of silica colloids at higher temperatures.



Figure 6.4: (a) Dynamic moduli of $\phi = 0.4$ silica particle – [C₄mim][BF₄] dispersion at different temperatures; and (b) shifted dynamic moduli curves according to time-temperature superposition (TTS) described by Shikata and Pearson.⁴⁷ The closed symbols represent storage modulus G' and open symbols represent loss modulus G''(a) or subtracted G''- $\omega \eta_{\infty}$ (b).



Figure 6.5: Shift factors a_T and viscosity change of $[C_4 mim][BF_4]$ referenced to 25 °C (left y axis) and relative high frequency limiting viscosity $\eta'_{r,\infty}$ (right y axis) of $\phi = 0.4$ silica particle – $[C_4 mim][BF_4]$ dispersion at various temperatures.

6.3.3 Inter-Particle Interaction and Dispersion Structure

The form factor of the KE-P10 silica particle is measured by small angle neutron scattering (SANS) using dilute dispersions in deuterated water D_2O and $[C_4mim][BF_4]$ at 25 °C and plotted in Figure 6.6a. As expected, the general shapes of the scattering profiles are similar due to the same dispersed silica particle core. Both scattering intensity profiles are fit well by a Schulz - distributed polydisperse sphere model using the same protocol to fit the particle size, polydispersity, and volume fraction as used previously for the fluorocarbon coated silica particle system.⁶ These fits yield a particle radius of 62 nm and polydispersity of 10%. As stated in our
previous study for the fluorocarbon coated silica particle system, the solvation layers around the particle surface are not detectable by neutron scattering due to the similar density of the solvation layers as compared with the bulk IL. However, the presence of solvation layers increases the effective hydrodynamic particle size of the particles, which leads to a slower diffusion rate when dispersed in $[C_4mim][BF_4]$ that can be measured by dynamic light scattering (DLS).⁶ The hydrodynamic radius of silica particles in H₂O and $[C_4mim][BF_4]$ as measured using DLS are listed in Table 6.1. There is an obvious increase in the hydrodynamic radius when the particles are dispersed in $[C_4mim][BF_4]$, as compared with water. The increase of hydrodynamic radius (~3.5 nm) in $[C_4mim][BF_4]$ as compared to water is a consequence of solvation layers forming around silica particle surface.

The solvation layers detected by DLS should also be evident in the structure factor as they provide steric stability and thus, prevent particles from approaching closely. Consequently, solvation layers will lead to a dispersion structure reflecting this larger effective particle size due to the solvation layers. The structure factor of KE-P10 silica particles in [C₄mim][BF₄] is measured using SANS applied to concentrated particle dispersions. The scattering profiles of KE-P10 silica particles in [C₄mim][BF₄] at various concentrations at 25 °C are plotted in Figure 6.6b. A structure peak at q < 0.01 Å⁻¹ becomes evident with increasing particle concentration, as expected. The scattering intensities are fitted by a "sticky" effective hard sphere structure factor model and the particle form factor model with parameters determined from the dilute particle solutions, as shown in Figure 6.6a. The effective volume fraction of particles in suspensions can be extracted from the structure factor, and thus, the effective particle size can be determined. The fitting method follows the same

procedure as for fluorocarbon coated silica particle system.⁶ From these measurements, the solvation layer thickness is determined to be 2 - 3 nm, which is slightly smaller but comparable to the layer thickness of 3.5 nm from DLS experiments.

The 3 nm-thick solvation layers on KE-P10 silica particles in $[C_4mim][BF_4]$ are thinner than the solvation layers formed for fluorocarbon coated silica particles in the same IL. This decrease in solvation layer thickness can be attributed to weaker hydrogen bonding between the anion $[BF_4]^-$ and the KE-P10 particle surface. KE-P10 silica particles are coated by an alcohol.³⁶ The resultant carbon chain coating is expected to form weak hydrogen bonds with the anion group $[BF_4]^-$. For example, the strength of hydrogen bonds between 4-carbon chain alcohol and the ion F⁻ is ~42 kJ/mol, however, the strength increases by 3 times (~127 kJ/mol) for a 2-carbon chain fluorocarbon.⁴⁸ In general, the strength of hydrogen bonds also increases with longer fluorocarbon chain length.⁴⁸ Thus, I can estimate the strength of hydrogen bonds for KE-P10 particles in the IL to be at least 3 times weaker than those of the fluorocarbon coated particles. The weaker hydrogen bonds between alcohol surface coating and $[BF_4]^-$ anion group of $[C_4mim][BF_4]$ will result in a thinner solvation layer.

Importantly, the thinner solvation layers (~ 3 nm) are responsible for the stronger shear thickening strength as compared to fluorocarbon coated silica particle dispersions with thicker solvation layers (~ 5 nm). It was found that shear thickening can be suppressed by grafting polymer layer or brush onto the particle surfaces, which imposes a repulsive force that prevents the particles from approaching each other.^{1, 37} This effect increases progressively as the brush thickness on the particle surface increases. When the thickness of brush approaches 10% of the particle radius, the

shear thickening is effectively eliminated. Solvation layers initiated by hydrogen bonds around silica particles play the same role as the polymer brushes providing a repulsive force to prevent the aggregation of silica particles. That is, this thicker solvation layer suppresses the shear thickening by suppressing the formation of "hydroclusters" under high shear. For the fluorinated carbon coated silica particles, the thickness of solvation layers is approximately 5 nm, which reaches 10% of the silica particle radius. Thus, shear thickening of dispersions are strongly suppressed with only slightly viscosity increases at high shear stress. The coated silica particles KE-P10 have solvation layers (~3 nm) in the IL that are ~5% of particle radius, and concentrated dispersions show enhanced shear thickening at room temperature.



Figure 6.6: SANS profiles of (a) dilute silica particles in D_2O and $[C_4mim][BF_4]$ solvents with Schulz-distributed sphere form factor model fitting; (b) silica particles in $[C_4mim][BF_4]$ of various concentrations with "sticky" hard sphere structure factor model fitting at 25 °C.

Measurement	Particle in water	Particle in
Techniques		$[C_4 mim][BF_4]$
DLS	$63.4 \pm 4.9 \text{ nm}$	$66.9 \pm 3.7 \text{ nm}$
SANS	$62.0 \pm 6.2 \text{ nm}$	$62.0 \pm 6.2 \text{ nm}$
	<i>p</i> = 10 %	<i>p</i> = 10 %

Table 6.1:Summary of particle radii of silica particle in different solvents at 25 °Cusing various methods.

To explore this effect further, DLS measurements were performed for the particles in very dilute solution in the IL (0.1 vol %) over the temperature range of interest. The intensity averaged radii and the associated standard deviations are plotted in Figure 6.7. The mean hydrodynamic radii are intensity averaged value over 10 min of measurement directly following 10 min temperature equilibrium for each temperature. It is evident that silica particles have a narrow polydispersity at low temperatures of 10 °C and 25 °C. This confirms the stability of KE-P10 silica particles in [C₄mim][BF₄] at these temperatures, and is consistent with the steady and oscillation shear rheology. However, with increasing temperature above 40 °C the silica particles become unstable in [C₄mim][BF₄] and aggregate, which is evident in the data as an increase in both the mean particle size and the width of the size distribution as indicated by the error bar. The DLS measurements of unstable aggregation at 40 °C and above directly confirm the loss of particle stability of KE-P10 silica particles in [C₄mim][BF₄] at high temperatures. Again, this is in good agreement with the rheological results.



Figure 6.7: Hydrodynamic radius (intensity average) of KE-P10 silica particle or aggregations in $[C_4mim][BF_4]$ measured by DLS at the temperatures noted. Error bar is the standard deviation of measured particle size, indicating the polydispersity of particle or aggregations.

The solution structure measured by SANS is used to determine the interparticle interaction in [C₄mim][BF₄] at different temperatures. The scattering profiles of $\phi = 0.4$ KE-P10 silica particles in [C₄mim][BF₄] at various temperatures are shown in Figure 6.8a. Scattering intensities at high *q* range (> 10⁻² Å⁻¹) overlap for all the temperatures investigated as expected due to the essentially constant form factor of the silica particles at all temperatures. However, the intensities at low *q* (< 10⁻² Å⁻¹) increase progressively with temperature. Intensities at low *q* are correlated to large scale structure in dispersion, hence an increase of intensity at low *q* indicates an increase in structure of the particle phase in [C₄mim][BF₄].⁴¹ This structuring indicates stronger attraction at higher temperatures,⁴⁵ consistent with the previous steady shear and small amplitude oscillation shear rheological results. In addition to hydrodynamic interactions and Brownian motion, the most important interparticle forces acting between silica particles in ILs are van der Waals attractions, electrostatic repulsion,¹⁷ and the solvation repulsive force.⁶ Electrostatic forces are negligible at all but the smallest separations due to the high ion concentration in ILs,^{6, 18, 49} so the dominated inerparticle force between silica particles are van der Waals attraction and solvation layer repulsion. Model calculations of the DLVO potential between 100 nm silica particles in [C₄mim][BF₄] have been studied by Gao and Wagner.⁶ While electrostatic repulsion between silica particles in ILs can be neglected, the van der Waals attraction is always present. As the Hamaker constant of two silica particles in [C₄mim][BF₄] is relatively insensitive to temperature change, van der Waals attraction are expected to be nearly independent of temperature. Hence, the increase in strength of attraction with increasing temperature can be quantitatively linked to a reduction in the extent of stabilizing solvation layer on the particles.

The thickness of solvation layers around particle surface and strength of attraction between silica particle at different temperatures can be extracted from the SANS scattering intensities of silica particles in $[C_4 \text{mim}][BF_4]$ by fitted to a "sticky" hard sphere model in the form of a short range, square well potential. Using the same fitting method as for the sample at room temperature, which is shown in Figure 6.6b, the best fit solvation layer thickness and effective strength of attraction - stickiness parameter τ_b - are plotted as a function of temperature in Figure 6.8b. The stickiness parameter τ_b is inversely related to the strength of attraction between particles, such that the smaller the value, the stronger the interparticle attraction.⁵⁰⁻⁵² In the limit that $\tau_b \rightarrow$ infinity, hard sphere behavior is recovered, while sticky hard spheres have a

critical value for phase separation of $\tau_b = 0.11$. It is evident that τ_b decreases with increasing temperature, indicating much stronger attraction at higher temperatures, which is consistent with the increased scattering intensities at low q for high temperatures.

This enhanced attraction at high temperatures can be more quantitatively understood now from the measured thickness of solvation layers. As shown in Figure 6.8b, solvation layers of silica particles in [C₄mim][BF₄] become thinner as temperature increases, such that the van der Waals attractions between particles will become more significant. Solvation layers are composed of anion and cation layers, which are initiated by hydrogen bonds between alcohol group on particle surface and anion group of $[C_4 mim][BF_4]$. It is well known that hydrogen bonds are sensitive to temperature, especially to high temperature.⁵³⁻⁵⁴ The strength of such hydrogen bonds are expected to decrease significantly with temperature, *i.e.*, the strength and number of hydrogen bonds per water molecule are found to decrease as the temperature is increased.⁵³⁻⁵⁴ I find that the solvation layers surrounding these silica particles decrease slightly from 2.7 nm to 2.3 nm with increasing temperature from 10 °C to 25 °C. Model calculations of van der Waals forces⁶ suggest that the van der Waals attraction between two silica particles is nearly identical for these interparticle separation distances, which is twice the solvation layers thickness. This calculation is consistent with the steady shear and small amplitude oscillation shear rheology, which show that the change of rheological response simply arises from the viscosity change of the IL itself from 10 °C to 30 °C. However, van der Waals attractions increase dramatically when the minimum separation distance between two particles decreases as a result of the reduction in the solvation layers at higher temperatures. van der

Waals attraction is expected to dominate the interaction between particles at this reduced solvation layer thickness. As a result, silica particles loose stability and (reversibly) aggregate. The thinning of solvation layers and the loss of stability of silica particles results in a dramatic rise in the low shear viscosity for temperatures above 30 °C. This is the source of the significant higher TTS shift factor a_T than expected from the thermal viscosity of [C₄mim][BF₄] itself at 40 °C and above.



Figure 6.8: (a) SANS profiles of $\phi = 0.4$ silica particle – $[C_4 \text{mim}][BF_4]$ dispersion at various temperatures with "sticky" hard sphere structure factor model fittings; (b) Extracted solvation layers thickness and stickiness parameter τ_b of different temperatures from SANS fitting of $\phi = 0.4$ silica particle – $[C_4 \text{mim}][BF_4]$ dispersion.

From prior experimental and theoretical work, the liquid or gel state of dispersions also can be determined from the values of stickiness parameter τ_b , which

can be loosely interpreted as an inverse temperature or strength of attractions, and the universal adhesive hard-sphere (AHS) state diagram. The τ_b values for $\phi = 0.4$ KE-P10 silica particles in [C₄mim][BF₄] at various temperatures extracted from the SANS fitting are plotted on the state diagram of AHS particles^{52, 55} and shown in Figure 6.9. With the decrease of τ_b , the attraction between AHS particles increases and the state of the dispersion transitions from a liquid state to a gel state. The dash line in the figure is the experimentally determined gel line for 100 nm diameter AHS particle dispersions.⁵⁵ As the size of KE-P10 particles is close to 100 nm, this gel line is used as a reference to determine the state of KE-P10 silica particles in [C₄mim][BF₄]. For the $\phi = 0.4$ KE-P10 silica particles dispersed in [C₄mim][BF₄] at temperatures from 10 °C to 30 °C, the sample is predicted to be in a liquid state, in an agreement with experiments. From the SANS measurements, I anticipate that the dispersion will become a gel with increasing temperature, as observed. This liquid to gel state transition with increase of temperature is in a good agreement with rheological response of KE-P10 silica particles in [C₄mim][BF₄].



Figure 6.9: State positions of $\phi = 0.4$ silica particle – [C₄mim][BF₄] dispersion at different temperatures in state diagram of AHS particle system.⁵² The dark-yellow circle points are the experimental gel points of particles of 100 nm diameter size⁵⁵ and a gel line shown as the dash line is determined from the experimental gel points.

6.4 Conclusions

In conclusion, I demonstrate that by tuning the strength of hydrogen bonds between particle surface and anions of ionic liquid, stable shear thickening colloidal dispersions can be formulated in ILs. At room temperature, this concentrated silica nanoparticle dispersion in $[C_4mim][BF_4]$ displays stronger shear thickening than the previous studied fluorocarbon coated silica particle system. It is shown that this is a consequence of the thinner solvation layers (~3 nm) initiated by relatively weaker hydrogen bonds between the hydrocarbon surface coating and anion group in comparison to the thicker solvation layers on the fluorinated particle surface, which creates strong hydrogen bonds. This 3 nm solvation layers provides sufficient stabilization force between particles, while still allowing the particles to approach and shear-thicken at high shear rate/stress. This finding provides guidance for formulating stable, shear thickening dispersions in ILs.

Inverse melting is observed above 30 °C, indicating formation of gel. The mechanism driving increased particle attraction and loss of particle stability with increasing temperature is shown to be caused by a reduction in the solvation layer thickness, which itself is a result of the weakening of the hydrogen bonds with the surface at high temperatures. The horizontal shift factors a_{τ} determined from TTS measurements confirm the increase of particle-particle attraction and loss of stability of colloidal dispersions above 30 °C. A transition from stable dispersion to unstable gel with increasing temperature also is independently confirmed by comparison with the established AHS particle state diagram. This state transition is in the opposite direction of traditional reported gel transition for coated particles dispersions at low temperatures, such as octadecyl coated silica suspended in n-tetradecane,^{52, 55} which is due to fundamentally different stabilization mechanisms. Nevertheless, the AHS state diagram provides a good reference for this new nanoparticle dispersion in IL. This novel gel transition at high temperatures, the investigation of the surface solvation structure and the corresponding interparticle potential provide a model study for material design through surface modification and temperature control. This basic principle is expected to be applicable to dispersions of many other functional colloidal particles, such as metals, magnetic and semiconductor particles in ILs.

REFERENCES

1. Wagner, N. J.; Brady, J. F. Shear thickening in colloidal dispersions. *Physics Today* **2009**, *62* (10), 27-32.

2. Mewis, J.; Wagner, N. J. *Colloidal suspension rheology*; Cambridge University Press: New York, **2012**.

3. Maranzano, B. J.; Wagner, N. J. The effects of particle size on reversible shear thickening of concentrated colloidal dispersions. *The Journal of Chemical Physics* **2001**, *114* (23), 10514-10527.

4. Lee, Y. S.; Wetzel, E. D.; Wagner, N. J. The ballistic impact characteristics of Kevlar® woven fabrics impregnated with a colloidal shear thickening fluid. *J. Mater. Sci.* **2003**, *38* (13), 2825-2833.

5. Ueno, K.; Imaizumi, S.; Hata, K.; Watanabe, M. Colloidal interaction in ionic liquids: Effects of ionic structures and surface chemistry on rheology of silica colloidal dispersions. *Langmuir* **2009**, *25* (2), 825-831.

6. Gao, J.; Ndong, R. S.; Shiflett, M. B.; Wagner, N. J. Creating nanoparticle stability in ionic liquid [C4mim][BF4] by inducing solvation layering. *ACS Nano* **2015**, *9* (3), 3243-3253.

7. Welton, T. Room-temperature ionic liquids. Solvents for synthesis and catalysis. *Chem. Rev.* **1999**, *99* (8), 2071-2084.

8. Dupont, J.; de Souza, R. F.; Suarez, P. A. Ionic liquid (molten salt) phase organometallic catalysis. *Chem. Rev.* **2002**, *102* (10), 3667-3692.

9. Cwalina, C. D.; Dombrowski, R. D.; McCutcheon, C. J.; Christiansen, E. L.; Wagner, N. J. MMOD Puncture Resistance of EVA Suits with Shear Thickening Fluid (STF) – Armortm Absorber Layers. *Procedia Engineering* **2015**, *103*, 97-104.

10. Cwalina, C. D.; McCutcheon, C. M.; Dombrowski, R. D.; Wagner, N. J. Engineering enhanced cut and puncture resistance into the thermal micrometeoroid garment (TMG) using shear thickening fluid (STF) – Armor[™] absorber layers. *Composites Science and Technology* **2016**, *131*, 61-66.

11. Altin, E.; Gradl, J.; Peukert, W. First Studies on the Rheological Behavior of Suspensions in Ionic Liquids. *Chem. Eng. Technol.* **2006**, *29* (11), 1347-1354.

12. Wittmar, A.; Ruiz-Abad, D.; Ulbricht, M. Dispersions of silica nanoparticles in ionic liquids investigated with advanced rheology. *Journal of Nanoparticle Research C7 - 651* **2012**, *14* (2), 1-10.

13. Wittmar, A.; Gajda, M.; Gautam, D.; Dörfler, U.; Winterer, M.; Ulbricht, M. Influence of the cation alkyl chain length of imidazolium-based room temperature ionic liquids on the dispersibility of TiO2 nanopowders. *Journal of nanoparticle research* **2013**, *15* (3), 1-12.

14. Nordström, J.; Aguilera, L.; Matic, A. Effect of lithium salt on the stability of dispersions of fumed silica in the ionic liquid BMImBF4. *Langmuir* **2012**, *28* (9), 4080-4085.

15. Smith, J. A.; Werzer, O.; Webber, G. B.; Warr, G. G.; Atkin, R. Surprising particle stability and rapid sedimentation rates in an ionic liquid. *J. Phys. Chem. Lett.* **2009**, *1* (1), 64-68.

16. Ueno, K.; Watanabe, M. From colloidal stability in ionic liquids to advanced soft materials using unique media. *Langmuir* **2011**, *27* (15), 9105-9115.

17. Israelachvili, J. N. *Intermolecular and Surface Forces, 3rd Edition*; Elsevier Academic Press Inc: San Diego, **2011**. p 1-674.

18. Baldelli, S. Probing electric fields at the ionic liquid–electrode interface using sum frequency generation spectroscopy and electrochemistry. *J. Phys. Chem. B* **2005**, *109* (27), 13049-13051.

19. Feng, G.; Qiao, R.; Huang, J.; Dai, S.; Sumpter, B. G.; Meunier, V. The importance of ion size and electrode curvature on electrical double layers in ionic liquids. *Phys. Chem. Chem. Phys.* **2011**, *13* (3), 1152-1161.

20. Lynden-Bell, R. M.; Del Pópolo, M. G.; Youngs, T. G. A.; Kohanoff, J.; Hanke, C. G.; Harper, J. B.; Pinilla, C. C. Simulations of ionic liquids, solutions, and surfaces. *Acc. Chem. Res.* **2007**, *40* (11), 1138-1145.

21. Ueno, K.; Inaba, A.; Kondoh, M.; Watanabe, M. Colloidal stability of bare and polymer-grafted silica nanoparticles in ionic liquids. *Langmuir* **2008**, *24* (10), 5253-5259.

22. Hayes, R.; Warr, G. G.; Atkin, R. At the interface: solvation and designing ionic liquids. *Phys. Chem. Chem. Phys.* **2010**, *12* (8), 1709-1723.

23. Atkin, R.; Warr, G. G. Structure in confined room-temperature ionic liquids. *J. Phys. Chem. C* **2007**, *111* (13), 5162-5168.

24. Ueno, K.; Kasuya, M.; Watanabe, M.; Mizukami, M.; Kurihara, K. Resonance shear measurement of nanoconfined ionic liquids. *Phys. Chem. Chem. Phys.* **2010**, *12* (16), 4066-4071.

25. Segura, J. J.; Elbourne, A.; Wanless, E. J.; Warr, G. G.; Voïtchovsky, K.; Atkin, R. Adsorbed and near surface structure of ionic liquids at a solid interface. *Phys. Chem. Chem. Phys.* **2013**, *15* (9), 3320-3328.

26. Perkin, S.; Crowhurst, L.; Niedermeyer, H.; Welton, T.; Smith, A. M.; Gosvami, N. N. Self-assembly in the electrical double layer of ionic liquids. *Chem. Commun.* **2011**, *47* (23), 6572-6574.

27. Smith, A. M.; Lovelock, K. R. J.; Perkin, S. Monolayer and bilayer structures in ionic liquids and their mixtures confined to nano-films. *Faraday Discuss.* **2013**, *167* (0), 279-292.

28. Pinilla, C.; Del Pópolo, M. G.; Lynden-Bell, R. M.; Kohanoff, J. Structure and dynamics of a confined ionic liquid. Topics of relevance to dye-sensitized solar cells. *J. Phys. Chem. B* **2005**, *109* (38), 17922-17927.

29. Perkin, S.; Albrecht, T.; Klein, J. Layering and shear properties of an ionic liquid, 1-ethyl-3-methylimidazolium ethylsulfate, confined to nano-films between mica surfaces. *Phys. Chem. Chem. Phys.* **2010**, *12* (6), 1243-1247.

30. Guibert, C.; Dupuis, V.; Fresnais, J.; Peyre, V. Controlling nanoparticles dispersion in ionic liquids by tuning the pH. *J. Colloid Interface Sci.* **2015**, *454*, 105-111.

31. Ueno, K.; Inaba, A.; Ueki, T.; Kondoh, M.; Watanabe, M. Thermosensitive, soft glassy and structural colored colloidal array in ionic liquid: colloidal glass to gel transition. *Langmuir* **2010**, *26* (23), 18031-18038.

32. Krishnamurthy, L.-n.; Wagner, N. J. The influence of weak attractive forces on the microstructure and rheology of colloidal dispersions. *Journal of Rheology* **2005**, *49* (2), 475-499.

33. Krishnamurthy, L.-N.; Wagner, N. J.; Mewis, J. Shear thickening in polymer stabilized colloidal dispersions. *Journal of Rheology* **2005**, *49* (6), 1347-1360.

34. Gao, J.; Wagner, N. J. Non-ideal viscosity and excess molar volume of mixtures of 1-butyl-3-methylimidazolium tetrafluoroborate ([C4mim][BF4]) with water. *J. Mol. Liq.* **2016**, *223*, 678-686

35. Gurnon, A. K.; Wagner, N. J. Microstructure and rheology relationships for shear thickening colloidal dispersions. *J. Fluid Mech.* **2015**, *769*, 242-276.

36. Koga, H., Nippon Shokubai Co., Ltd, Personal communication, **2012**.

37. Bergenholtz, J.; Brady, J.; Vicic, M. The non-Newtonian rheology of dilute colloidal suspensions. *J. Fluid Mech.* **2002**, *456*, 239-275.

38. Smith, J. A.; Werzer, O.; Webber, G. B.; Warr, G. G.; Atkin, R. Surprising particle stability and rapid sedimentation rates in an ionic liquid. *The Journal of Physical Chemistry Letters* **2009**, *1* (1), 64-68.

39. Gao, J.; Wagner, N. J. Water nanocluster formation in the ionic liquid 1-Butyl-3-Methylimidazolium Tetrafluoroborate ([C4mim][BF4])–D2O mixtures. *Langmuir* **2016**, 32, (20), 5078–5084.

40. Szilagyi, I.; Szabo, T.; Desert, A.; Trefalt, G.; Oncsik, T.; Borkovec, M. Particle aggregation mechanisms in ionic liquids. *Phys. Chem. Chem. Phys.* **2014**, *16* (20), 9515-9524.

41. Kline, S. R. Reduction and analysis of SANS and USANS data using IGOR Pro. *J. Appl. Crystallogr.* **2006**, *39* (6), 895-900.

42. López-Barrón, C. R.; Li, D.; Wagner, N. J.; Caplan, J. L. Triblock Copolymer Self-Assembly in Ionic Liquids: Effect of PEO Block Length on the Self-Assembly of PEO–PPO–PEO in Ethylammonium Nitrate. *Macromolecules* **2014**, *47* (21), 7484-7495.

43. Shenoy, S. S.; Wagner, N. J. Influence of medium viscosity and adsorbed polymer on the reversible shear thickening transition in concentrated colloidal dispersions. *Rheol. Acta* **2005**, *44* (4), 360-371.

44. Maranzano, B. J.; Wagner, N. J. The effects of interparticle interactions and particle size on reversible shear thickening: Hard-sphere colloidal dispersions. *Journal of Rheology (1978-present)* **2001,** *45* (5), 1205-1222.

45. Eberle, A. P. R.; Wagner, N. J.; Castañeda-Priego, R. Dynamical Arrest Transition in Nanoparticle Dispersions with Short-Range Interactions. *Phys. Rev. Lett.* **2011**, *106* (10), 105704. 46. Wagner, N. J.; Bender, J. W. The Role of Nanoscale Forces in Colloid Dispersion Rheology. *MRS Bulletin* **2011**, *29* (2), 100-106.

47. Shikata, T.; Pearson, D. S. Viscoelastic behavior of concentrated spherical suspensions. *Journal of Rheology (1978-present)* **1994,** *38* (3), 601-616.

48. Meot-Ner, M. The ionic hydrogen bond. *Chem. Rev.* **2005**, *105* (1), 213-284.

49. Lanning, O. J.; Madden, P. A. Screening at a charged surface by a molten salt. *J. Phys. Chem. B* **2004**, *108* (30), 11069-11072.

50. Menon, S. V. G.; Manohar, C.; Rao, K. S. A new interpretation of the sticky hard sphere model. *The Journal of Chemical Physics* **1991**, *95* (12), 9186-9190.

51. Chiew, Y. C.; Glandt, E. D. Percolation behaviour of permeable and of adhesive spheres. *J. Phys. A: Math. Gen.* **1983**, *16* (11), 2599.

52. Eberle, A. P.; Castañeda-Priego, R. n.; Kim, J. M.; Wagner, N. J. Dynamical arrest, percolation, gelation, and glass formation in model nanoparticle dispersions with thermoreversible adhesive interactions. *Langmuir* **2012**, *28* (3), 1866-1878.

53. Jorgensen, W. L.; Madura, J. D. Temperature and size dependence for Monte Carlo simulations of TIP4P water. *Mol. Phys.* **1985**, *56* (6), 1381-1392.

54. Mizan, T. I.; Savage, P. E.; Ziff, R. M. Temperature dependence of hydrogen bonding in supercritical water. *J. Phys. Chem.* **1996**, *100* (1), 403-408.

55. Kim, J. M.; Fang, J.; Eberle, A. P. R.; Castañeda-Priego, R.; Wagner, N. J. Gel transition in adhesive hard-sphere colloidal dispersions: The role of gravitational effects. *Phys. Rev. Lett.* **2013**, *110* (20), 208302.

Chapter 7

PENETRATION RESISTANCE OF IONIC LIQUIDS BASED SHEAR THICKENING FLUIDS (STFS)-KEVLAR[®] NANOCOMPOSITES

The penetration resistance of ionic liquids based STFs- Kevlar[®] nanocomposites is described in this chapter. The effect of changing from the traditional STF carrier medium (PEG) to ionic liquid $[C_4mim][BF_4]$ has been studied. Optimal areal density, penetration peak force, and dissipated energy are obtained through varying the treatment condition, such as the mixing ratio of ethanol to STFs. The resistance against threats from a spike and 21 & 18 gauge needles are evaluated for the ionic liquid based STF- Kevlar[®], and the results are compared with PEG-based STF- Kevlar[®].

7.1 Introduction

Traditional personal body armors are protective covering materials used to prevent an individual from being inflicted through direct contact with weapons, such as sharp instruments or projectiles from a potentially dangerous environment or action.¹⁻³ Recent trends have led to an increasing demand for stab protection.² Stab threats include direct attacks from knives and sharpened instruments, as well as physical contact with debris, broken glass, and razor wire.⁴ Thus, it is imperative to develop body armor systems with stab-resistant capabilities. There are two types of stab threats: puncture and cut. Puncture refers to penetration caused by instruments with sharp tips but no cutting edge, for example, ice picks or awls. Cut refers to contact with knives with a continuous cutting edge.

Numerous research efforts have been carried out to develop flexible, lightweight, and wearable soft body armor.^{2-3, 5} Previous studies have used high yarn count woven aramid fabrics, such as Kevlar[®] fabric from DuPont, impregnated with a colloidal, discontinuous shear thickening fluid (STF), and their protective properties including stab^{2, 5-6} and ballistic⁷⁻⁸ protection are evaluated. These investigations have shown that STF-fabric nanocomposite offers superior protective properties as compared to the neat fabrics, without compromising the weight and flexibility of the fabrics.

STFs can exhibit orders-of-magnitude increase in viscosity at high shear rate or shear stress,⁹⁻¹⁰ thus they have been intercalated with protective fabrics, such as Kevlar[®], to create enhanced protective materials often referred as STF-ArmorTM. STF-ArmorTM has been shown to provide superior protection against ballistic, and puncture threats. STFs behave like a liquid under rest that is flowable and low viscosity. However, when high shear rate or shear stress are applied, such as applying a strong impact, STFs turn into rigid with dramatic increase in viscosity and exhibit great energy dissipation capability. Thus, STFs can be used as an additive material to improve the performance of regular body armor made by Kevlar[®] fabric, allowing the flexibility of fabrics at normal movements while providing the armor with rigidity to resist threats from bullets, stabbing knife, and so on.

The mechanisms of how the addition of STF improves penetration resistance have been studied^{3-4, 6-8, 11-12} which includes dissipation of viscous energy within the STF itself, stress-coupling between adjacent fabrics, and a reduction in yarn mobility.² Decker et al.² performed quasi-static spike tests on STF-Kevlar[®] fabrics, showing that the addition of STF significantly improves the puncture resistance of Kevlar[®] fabrics.

The enhancement is attributed to a decreased yarn mobility within the fabric, which is consistent with prior yarn pull-out¹³⁻¹⁵ and ballistic tests.¹² The STFs act to restrict the relative motion of the fibers and yarns, which prevent the sharp tip of the spike from pushing aside yarns and fibers and penetrating between them. The reduction in the mobility of yarns also has been demonstrated to improve the energy absorption, which forces a probe to engage and ultimately break more yarns.¹¹ In addition, it has been proven that the enhancement of protective properties provided by STF addition is not limited to high speed impact events, but also occurs during quasi-static puncture tests. During quasi-static puncture, localized stress generated in the STF at the point of penetration exceeds the critical shear thickening stress, which can be transmitted to the surrounding fabric, i.e., shear thickening occurs immediately in a region surrounding the point of puncture threats, resulting in enhanced protection property.

All the prior STF-fabric studies are focused on the STFs of particles dispersed in poly(ethylene glycol) (PEG). Recently, silica particles in ionic liquids (ILs) have been reported to display shear thickening behavior.¹⁶⁻¹⁸ Significant viscosity increase is observed at high shear rate for solid-spherical silica particles in the IL [C₄mim][BF₄] (more details are discussed in Chapter 6). Because of their vacuum stability property,¹⁹⁻²⁰ this ionic liquid based STF enables the use of STF-fabric soft body armor no longer limited to the application on earth, but also can be put into the extravehicular activity (EVA) suit to protect astronaut again threats from micrometeoroids and orbital debris (MMOD) particles.²¹⁻²²

In this chapter, Kevlar[®] fabrics are modified and treated with ionic liquid and its shear thickening fluids for the first time. I have investigated the performance of

stab resistance against spike puncture threats and needle cutting threats for the ionic liquid based STF-Kevlar[®]. The influence of ionic liquid itself on the penetration performance of Kevlar[®] fabric have been evaluated to ensure no lubricating or fiber damaging effect on Kevlar. The STF-Kevlar[®] treatment condition, such as the mixing ratio of ethanol to STFs, has been examined in terms of increased areal density, penetration peak force, and dissipated energy. Both of the spike and needle quasi-static penetration tests have been performed on the ionic liquids based STF-Kevlar[®] and the results are compared with traditional PEG-based STF-Kevlar[®].

7.2 Materials and Methods

7.2.1 Materials

The IL [C₄mim][BF₄] (>99% pure) was purchased from Iolitec to prepare silica particle dispersions. The poly(ethylene glycol) ($M_w = 200$, PEG-200) was obtained from Clariant. Two different sizes of silica nanoparticles KEP10 (radius a = 60 nm) and KEP50 (radius a = 260 nm) were purchased from Nippon Shokubai Co. (Tokyo, Japan) and were used to treat the Kevlar[®] fabric directly or to prepare concentrated silica particle - ionic liquid [C₄mim][BF₄] suspensions displaying shear thickening behavior. At 50 vol%, the STF sample exhibits extreme shear thickening behavior. More details about the preparation and rheological behavior of shear thickening fluids can be found in Chapter 6.

Protective Kevlar[®] 1148 fabric (untreated or treated with PEG based STF) was purchased from Barrday Inc. (Cambridge, Ontario, Canada). The areal density for untreated Kevlar[®] 1148 and treated STF- Kevlar[®] 1148 are 159 g/m² and 183 g/m², respectively. In the lab, in order to intercalate silica particle or STFs into the fabric,

the silica particles or STFs were diluted with ethanol at certain volume ratio to obtain a homogenous mixture. In this chapter, the effect of the mixing volume ratio of ethanol to silica particle/STF (e.g. 1:1, 2:1, 3:1 and 4:1) on the penetration resistance of Kevlar[®] fabric is discussed. Each individual Kevlar[®] 1148 fabric sheet (3 inch × 3 inch) was soaked into the mixture for 1 min before being squeezed through rubbercoated nip rollers to remove extra ethanol/STF mixture. Then each treated fabric sheet was hung to dry at room temperature for 5 min before dried in an oven for 1 h at 70 °C to remove the ethanol. All the fabrics including solvent treated, dry silica particle treated, and STFs treated, were dry to the touch. Fabrics that were solvent treated or STFs treated have a uniform appearance; however, the dry silica particle treated fabrics appear whitened streaky. The silica particles on the fabric can be easily peeled off when contacting with surface, leaving a white residue of dry particles. In contrast, for solvents or STFs treated fabrics, there was no obvious residue left on the contact surfaces.

Eight different fabric treatments were studied and compared in this chapter: untreated, ethanol, PEG-200, ionic liquid [C₄mim][BF₄], dry silica particle, $\phi = 0.5$ KEP50 in PEG 200, $\phi = 0.6$ KEP50 in ionic liquid [C₄mim][BF₄], and $\phi = 0.5$ KEP10 in ionic liquid [C₄mim][BF₄]. The mass or volume addition of each coating to the fabric can be controlled by tailoring the dilution ratio of treatment materials to ethanol in the soak bath. It is reported that sub-micro particles intercalate between fibers and yarns in the void space of the woven fabric.^{7, 12, 23-24} In order to have a reasonably consistent geometric packing and filling of the particles in the fabrics, the volume fraction of added particles needs to be controlled. Thus, when the fabric is treated with dry silica particles, the dilution with ethanol is twice as much so as to have

comparable particle concentrations in the coating fluid, for example, the dilution ratio of ethanol to dry particle is 2:1 in comparison to the 1:1 dilution for the STF treated fabrics.

7.2.2 Quasi-Static Puncture Tests

The puncture resistance of the Kevlar[®] fabrics with or without STFs treatments were evaluated by quasi-static puncture test. During the measurement, the ASTM F-1342 standard was followed.²⁵ An A0 6-1/4" awl spike from Malco Products. Inc. was used to simulate the threat from sharp objects without cutting edge. Two different gauges of hypodermic needles with standard bevel tips: 18G (1.270 mm outer barrel diameter), and 21 G (0.800 outer barrel diameter) were used to simulate the threat from sharp-cutting edge objects (referred to Chapter 2 Figure 2.10 for more details). Between each experiment, the needle is replaced by a fresh one to ensure the same sharpness.⁶ The spike or needle was held within the chunk that was mounted to a 500 N load cell on an Instron 5965 instrument. One single layer of Kevlar[®] fabric was used for the spike puncture tests and 4-laver assemble of Kevlar[®] fabric was used for needle penetration tests. The spike or needle was loaded at a constant quasi-static speed of 10 mm/s approaching to the fabrics. Zero displacement and zero force were defined when the probe nearly touched the fabric. The force experienced by the probe was recorded as a function of displacement. Twelve replicates were conducted on each sample at different spots to obtain a meaningful statistical value. The force resistance of the tested fabrics is evaluated by the maximum value of the force in the force-displacement curve, which is considered as the penetration peak force, and the dissipated energy until the penetration of fabric. The dissipated energy is the integral of force vs. displacement (the area underneath the force vs. displacement curve) until

reaching the penetration peak force. Combined with the areal density (weight/area) of Kevlar[®] fabric, those three factors are considered when designing the lightweight soft body armor materials.

7.3 **Results and Discussion**

7.3.1 Spike Penetration Measurements

7.3.1.1 Effect of STFs Carrier Medium

It has been known that ionic liquids have a lubricating effect when applied to the surface of various materials,²⁶⁻²⁹ especially under vacuum condition,²⁸⁻²⁹ such that, it can be used as a lubricant with broad applications. Due to the lubricating effect of ionic liquids, the penetration resistance of Kevlar[®] fabric treated by ionic liquids based STFs may be reduced due to increased mobility of yarns with the present of ionic liquid. Moreover, ionic liquids have shown an ability to dissolve a great variety of materials including various polymers.^{19, 30} They have the potential to dissolve and damage Kevlar[®] polymer fiber, which would result in a reduction in the puncture force resistance for Kevlar[®] treated by ionic liquid based STFs. Due to these properties of ionic liquids, the lubricating and dissolving effect of ionic liquids on Kevlar[®] fabric need to do examined.

Kevlar[®] 1148 fabrics (3 inch \times 3 inch, 12 pieces) were treated by ethanol, ionic liquid [C₄mim][BF₄], and PEG 200, and their puncture resistance against spike were evaluated. Their increased areal density after treatments are plotted in Figure 7.1 and compared in Table 7.1. As the ethanol evaporates from the fabrics during drying in the oven, there is nearly no density increase as shown in the Figure 7.1. The uptake of IL [C₄mim][BF₄] by Kevlar[®] is larger in comparison to PEG 200, resulting in doubled

increment of areal density than the PEG 200 treated Kevlar[®]. The force curves (12 replicates) punctured by spike for one layer Kevlar[®] fabric with and without solvents treatments are shown in Figure 7.2. The variability of force curves of twelve replicates for each treatment is due to the woven nature of the Kevlar[®] fabric. For each of the Kevlar[®] samples, including the untreated one, the results of twelve replicates can be divided to two groups: low penetration peak force and high penetration peak force. The low penetration peak force for each particular treatment condition occurs when the spike contacts the interstitial area between yarns during puncture, resulting in the penetration mainly due to "windowing" of fabric, that is, the spike penetrates though an open area by spreading apart yarns.¹⁻² In contrast, when the spike contacts at the cross-over area of yarns, fibers fracture with "windowing" occurred during the penetration, resulting in the high penetration peak force.



Figure 7.1: The increased areal density (%) of Kevlar[®] 1148 fabric (3 inch \times 3 inch) after treated by various solvents: ethanol, ionic liquid ([C₄mim][BF₄]), and PEG 200.



Figure 7.2: The force curves (force vs. displacement) of Kevlar[®] 1148 before and after treated by solvents: ethanol, ionic liquid ($[C_4mim][BF_4]$), and PEG 200 with twelve repeats.

The penetration peak force and dissipated energy for each treatment are shown as box chart in Figure 7.3. Table 7.1 summaries the statistical value of penetration peak force and dissipated energy. It can be observed that, after treatments with different solvents: ethanol, ionic liquid, or PEG 200, the penetrate peak force and the dissipated energy of Kevlar[®] 1148 during puncture tests are similar to the untreated fabric. A t-test is performed between each treatment and untreated one, and their pvalues are listed in Table 7.1. At 95% confidence, it is obvious that there is no significant difference in the penetration peak force and dissipated energy between untreated Kevlar[®] and solvent treated Kevlar[®]. Thus, it is evident that ionic liquids have no significant lubricating or dissolving effect on the Kevlar[®] fabric; thus, the puncture resistance is not adversely affected when the fabric is treated with ionic liquids.



Figure 7.3: Box charts of penetration peak forces and dissipated energies for Kevlar[®] 1148 with or without solvents treatments.

Table 7.1:Summary of increased areal density (%), penetration peak force with its
p-value of t-test (two-sample t-test compared with untreated Kevlar[®]
1148), and dissipated energy with its p-value of t-test (two-sample t-test
compared with untreated Kevlar[®] 1148) for Kevlar[®] 1148 before and
after treatment by solvents: ethanol, ionic liquid ([C₄mim][BF₄]), and
PEG 200.

Treatment	Untreated	Ethanol Treated	IL Treated	PEG 200 Treated
Increased Areal Density (%)		0.71 ± 0.62	18.2 ± 1.00	9.00 ± 0.60
Peak Force (N)	11.71 ± 5.38	13.64 ± 5.51	14.14 ± 4.47	13.77 ± 5.51
p-value (t-test)		0.39	0.24	0.36
Dissipated Energy (mJ)	8.22 ± 4.48	9.32 ± 3.79	10.27 ± 3.80	8.03 ± 3.81
p-value (t-test)		0.52	0.24	0.92

7.3.1.2 Effect of Ethanol: STFs Solution Ratio

STFs are mixed with ethanol to reduce the surface energy and viscosity of suspensions before treating the Kevlar[®] fabric. The effect of changing the mixing volume ratio of ethanol to STFs suspensions on puncture resistance is investigated. Kevlar[®] 1148 fabrics are treated by dry KEP50 silica particle, $\phi = 0.5$ KEP50-PEG STF, $\phi = 0.6$ KEP50-IL STF, and $\phi = 0.5$ KEP10-PEG STF. The increased areal densities (%) after treatments with four different mixing volume ratios of ethanol to STFs are shown in Figure 7.4. As expected, the increment of the areal density decreases with higher ethanol content due to the highly diluted STF concentration. However, the reduction in increased areal density is not significant when the mixing ratio changes from 3:1 to 4:1. For the same mixing ratio, Kevlar[®] 1148 treated by dry KEP50 silica particle has the lowest increase in weight due to the lack of non-evaporated solvents (such as ionic liquids or PEG). Dry silica particles are easily peeled off from the surface of fabrics without the adhesive effect of ionic liquids or PEG. Kevlar[®] treated by ionic liquid based STFs have slightly higher increased areal density than those treated by PEG based STF with the same particle concentration.

Due to the higher particle concentration of STFs, there is more weight gain for Kevlar[®] treated with $\phi = 0.6$ KEP50 in ionic liquid than those with lower particle concentration of $\phi = 0.5$.



Figure 7.4: The increased areal density (%) of Kevlar[®] 1148 fabric (3 inch × 3 inch) after treated by dry KEP50 particle powder, and different STFs ($\phi = 0.5$ KEP50 in PEG 200, $\phi = 0.6$ KEP50 in ionic liquid [C₄mim][BF₄], and $\phi = 0.5$ KEP10 in ionic liquid [C₄mim][BF₄]) with different volume ratios of ethanol : treating materials (particles or STFs).

Spike penetration test is performed on one layer of the untreated and treated Kevlar[®] fabrics. The penetration peak forces and dissipated energies of Kevlar[®] treated by dry KEP50 silica particles are shown as chart box in Figure 7.5, and their statistical values are listed in Table 7.2. It is evident that Kevlar[®] treated by dry silica particles possesses higher penetration resistance with higher penetration peak force and higher dissipate energy than untreated fabric. This observation is consistent with previous findings by other researchers.^{2, 7-8} When comparing the penetration resistance performance for different mixing ratios of ethanol to dry particle, the penetration peak force and dissipated energy for ratio condition of 2:1 and 3:1 are comparable and significant higher than those of 1:1 and 4:1. For more condensed mixture with the ratio of 1:1, the mixture looks like a slurry which might result in an unevenly treated Kevlar[®] surface leaving certain surface untreated. The penetration resistance performance would be lower at the untreated surface. However, for the most dilute mixture of 4:1 ratio, the low silica particle content in between the yarns causes insufficient yarn matrix connection to dissipate energy, which leads to a lower penetration resistance performance. Thus, when treated the fabrics with ethanol to dry particle ratio equal to 2:1 and 3:1, the penetration resistance performance of Kevlar[®] is optimal.



Figure 7.5: Box charts of penetration peak forces and dissipated energies during spike penetration measurements for Kevlar[®] 1148 before and after treated by dry KEP50 particle powders with various volume ratios of ethanol : KEP50.

Table 7.2:Summary of increased areal density (%), penetration peak forces, and
dissipated energies during spike penetration measurements for Kevlar[®]1148 before and after treatment by dry KEP50 particle powders with
various volume ratios of ethanol : KEP50.

	Untracted	$V_{ethanol}: V_{KEP50}$			
	Uniteated	1:1	2:1	3:1	4:1
Increased Areal Density (%)		23.59 ± 0.83	14.57 ± 1.50	6.86 ± 0.43	6.15 ± 0.46
Peak Force (N)	14.94 ± 5.25	23.82 ± 6.17	25.41 ± 10.88	25.54 ± 8.78	22.18 ± 8.38
Dissipated	12.09 ± 4.89	17.40 ± 7.93	19.49 ± 10.70	19.87 ± 7.40	16.81 ± 8.78
Energy (mJ)					

Similarly, the penetration peak force and dissipated energy of Kevlar[®] treated by $\phi = 0.5$ KEP50 in PEG 200 with varying ethanol to STFs volume ratios are shown in Figure 7.6 and their statistical value are listed in Table 7.3. The penetration peak forces with the treatment mixing condition at 1:1, 2:1 and 3:1 are significant higher than that treated at condition of 4:1, while the dissipated energy are comparable for all the conditions. The penetration resistance results of Kevlar[®] treated by $\phi = 0.6$ KEP50 in ionic liquid are shown in Figure 7.7 and Table 7.4. When the volume ratio of ethanol to STFs is equal to 3:1, both the penetration peak force and dissipated energy reach the maximum. For the Kevlar[®] treated by $\phi = 0.5$ KEP10 in ionic liquid (Figure 7.8 and Table 7.5), the peak force and dissipated energy of treated fabric at the ratio of 3:1 is slightly lower than that at condition ration of 1:1. However, considering the areal density gain is more than doubled for the condition of 1:1 than that of 3:1 shown in Table 7.5, condition of 3:1 would be the optimal condition for the purpose of lightweight fabric protection.

In all cases, STFs or particle intercalation improves the penetration force resistance and the energy dissipation capability of Kevlar[®] fabric, which are consistent with previous work suggesting improved puncture resistance with STF-treatments. Based on previous discussion, the ethanol: STFs/dry particle volume ratio = 3:1 is the optimal condition to achieve the best puncture protection without compromising the weight of the Kevlar[®] fabric. For the higher STFs/particle content (volume ratio of ethanol: STFs = 1:1 or 2:1), the increased weight of fabrics is significant higher, which is not ideal for light-weight protection purpose. What is more, uneven treated fabric is intended to happen at the high STFs content. For the lower STFs/particle content case (volume ratio of ethanol: STFs = 4:1), the small amount of STFs or particle content in between yarns would not be able to sufficiently reduce the mobility of yarns and provide enough puncture resistance.



- Figure 7.6: Box charts of penetration peak forces and dissipated energies during spike penetration measurements for Kevlar[®] 1148 before and after treated by STFs of $\phi = 0.5$ KEP50 in PEG 200 with various volume ratios of ethanol : STFs.
- Table 7.3: Summary of increased areal density (%), penetration peak forces, and dissipated energies during spike penetration measurements of Kevlar[®] 1148 before and after treatment by STFs of $\phi = 0.5$ KEP50 in PEG 200 various volume ratios of ethanol : STFs.

	Untracted	$V_{ethanol}: V_{STFs}$			
	Untreated	1:1	2:1	3:1	4:1
Increased Areal Density (%)		32.26 ± 1.60	15.08 ± 5.87	11.93 ± 1.52	8.98 ± 0.95
Peak Force (N)	14.94 ± 5.25	19.47 ± 9.10	20.09 ± 7.92	19.71 ± 6.92	16.48 ± 6.19
Dissipated	12.09 ± 4.89	15.42 ± 9.19	15.24 ± 6.29	14.45 ± 6.93	13.92 ± 7.11
Energy (mJ)					



Figure 7.7: Box charts of penetration peak forces and dissipated energies during spike penetration measurements for Kevlar[®] 1148 before and after treated by STFs of $\phi = 0.6$ KEP50 in ionic liquid [C₄mim][BF₄] with various volume ratios of ethanol : STFs.

Table 7.4: Summary of increased areal density (%), penetration peak forces, and dissipated energies during spike penetration measurements for Kevlar[®] 1148 before and after treatment by STFs of $\phi = 0.6$ KEP50 in ionic liquid [C₄mim][BF₄] with various volume ratios of ethanol : STFs.

	Untrooted	$V_{ethanol}: V_{STFs}$			
	Unifeated	1:1	2:1	3:1	4:1
Increased Areal		45.29 ± 1.53	32.78 ± 4.67	23.97 ± 4.48	20.50 ± 1.35
Density (%)					
Peak Force (N)	15.00 ± 5.37	15.11 ± 5.81	18.19 ± 6.00	22.58 ± 9.14	20.03 ± 6.29
Dissipated	11.13 ± 4.56	14.68 ± 8.26	15.11 ± 6.53	18.63 ± 10.46	15.84 ± 7.14
Energy (mJ)					


Figure 7.8: Box charts of penetration peak forces and dissipated energies during spike penetration measurements for Kevlar[®] 1148 before and after treated by STFs of $\phi = 0.5$ KEP10 in ionic liquid [C₄mim][BF₄] with various volume ratios of ethanol : STFs.

Table 7.5: Summary of increased areal density (%), penetration peak forces, and dissipated energies during spike penetration measurements for Kevlar[®] 1148 before and after treatment by STFs of $\phi = 0.5$ KEP10 in ionic liquid [C₄mim][BF₄] with various volume ratios of ethanol : STFs.

Untrooted	$V_{ethanol}: V_{STFs}$				
Uniteated	1:1	2:1	3:1	4:1	
	31.11 ± 5.21	22.15 ± 4.61	14.40 ± 0.81	14.38 ± 0.63	
15.00 ± 5.37	26.39 ± 8.00	18.44 ± 6.00	23.38 ± 7.27	21.31 ± 6.32	
11.13 ± 4.56	19.48 ± 8.17	13.29 ± 5.20	17.24 ± 5.77	15.81 ± 6.75	
	Untreated 15.00 ± 5.37 11.13 ± 4.56	Untreated $1:1$ 31.11 ± 5.21 15.00 ± 5.37 26.39 ± 8.00 11.13 ± 4.56 19.48 ± 8.17	Untreated $V_{ethanol}$ $1:1$ $2:1$ 31.11 ± 5.21 22.15 ± 4.61 15.00 ± 5.37 26.39 ± 8.00 18.44 ± 6.00 11.13 ± 4.56 19.48 ± 8.17 13.29 ± 5.20	Untreated $V_{ethanol} : V_{STFs}$ $1:1$ $2:1$ $3:1$ 31.11 ± 5.21 22.15 ± 4.61 14.40 ± 0.81 15.00 ± 5.37 26.39 ± 8.00 18.44 ± 6.00 23.38 ± 7.27 11.13 ± 4.56 19.48 ± 8.17 13.29 ± 5.20 17.24 ± 5.77	

The puncture resistance properties of untreated and STF treated Kevlar[®] 1148 at the same treatment condition of ethanol: STFs volume ratio = 3:1 are listed in Table 7.6 and the results are compared with commercial STF-Kevlar[®] 1148 product from Barrday. The STF-Kevlar[®] 1148 (Barrday) has similar areal density and comparable dissipated energy capability as the ones prepared in lab; however, the former shows a higher penetration peak force. This difference may come from different treatment protocols used as well as different STF formulations. For all the Kevlar[®] treated at lab scale, dry silica particle treated Kevlar[®] also shows a good penetration force resistance because the friction between particle and fabric reduces the mobility of the yarns. However, the yarns are more easily worn out due to the friction of dry particles, and the dry particles are easily peeled off from the fabric. For the wet-silica-Kevlar[®] treatments (STFs-Kevlar[®]), fabrics with the treatment of ionic liquid based STFs are proved to have good puncture resistance, which is even better than the one treated by PEG based STFs, making it promising to replace traditional PEG based STF-Kevlar[®] to use in space environment.

Table 7.6:Summary of increased areal density (%), penetration peak forces, and
dissipated energies during spike penetration measurements for Kevlar[®]
1148 before and after treatments.

Treatment	Increased Areal Density (%)	Peak Force (N)	Dissipated Energy (mJ)
Untreated		15.00 ± 5.37	11.13 ± 4.56
Dry KEP50	6.86 ± 0.43	25.54 ± 8.78	19.87 ± 7.40
0.5 KEP50-PEG	11.93 ± 1.52	19.71 ± 6.92	14.45 ± 6.93
0.6 KEP50-IL	23.97 ± 4.48	22.58 ± 9.14	18.63 ± 10.46
0.5 KEP10-IL	14.40 ± 0.81	23.38 ± 7.27	17.24 ± 5.77
STFs (Barrday)	14.62 ± 2.24	31.12 ± 8.04	17.41 ± 7.95

7.3.2 Needle Penetration Measurements

The penetration resistance of Kevlar[®] treated by various STFs (identical ethanol: STFs volume ratio = 3:1) is also evaluated by hypodermic needle penetration

test. The penetration force curves of 4-layers Kevlar[®] fabrics with and without various STFs treatments using 18G and 21G gauge needles are shown in Figure 7.9 and Figure 7.10, respectively. The shape of the force-displacement curves is statistical in nature.³¹⁻³² Twelve replicates are performed for each treatment to get meaningful statistical results. The averaged force values at each displacement are shown in Figure 7.11.



Figure 7.9: The force curves (force vs. displacement) of Kevlar[®] 1148 before and after treated with various STFs punctured by 18G needles with twelve repeats.



Figure 7.10: The force curves (force vs. displacement) of Kevlar[®] 1148 before and after treated with various STFs punctured by 21G needles with twelve repeats.



Figure 7.11: The averaged force curves (force vs. displacement) of Kevlar[®] 1148 before and after treated with various STFs punctured by needles of 18G (A) and 21G (B) with twelve repeats.

The average penetration peak force and dissipated energy of the Kevlar[®] fabrics during penetration process are measured to evaluate their penetration resistant capability, and the results are shown in Figure 7.12 and Figure 7.13, respectively. The statistical values of penetration resistance against 18G and 21G needles for each fabric treatment are reported in Table 7.7 and Table 7.8, respectively. For both of the 18G and 21G needle penetration threats, Kevlar[®] fabrics layers intercalated with STFs tend to have larger penetration peak force and higher dissipated energy than untreated Kevlar[®] fabric layers. This increase in penetration resistance is expected and consistent with previous research.^{2, 6, 22, 31}

Under the 18G needle penetration threats, Kevlar[®] fabrics treated by $\phi = 0.5$ KEP10 in ionic liquid have comparable penetration peak force as those of STF-Kevlar[®] (Barrday) with similar areal density, while the dissipated energy is lower than the STF-Kevlar[®] (Barrday). In contrast, after treated by $\phi = 0.6$ KEP50 in ionic liquid STF, Kevlar[®] fabrics have much higher areal density, while the penetration peak force and energy dissipated during the needle penetration are much lower than the STF-Kevlar[®] (Barrday).

For the quasi-static penetration test of smaller needle size 21G, the overall peak force and dissipated energy during penetration are significant lower than those using needle size of 18G. This reduced peak force and dissipated energy are the results of smaller contact surface area which favors occurrence of "windowing" of fabrics, i.e., needle penetrating though an open area by spreading apart of yarns.⁶ With the treatments of ionic liquid based STFs, Kevlar[®] fabrics display significant lower penetration peak force while having comparable energy dissipation capability as for PEG based STF-Kevlar[®] (Barrday).

From the previous discussions on both the 18G and 21G needle penetration tests, it is apparent that the penetration peak force and dissipated energy are decoupled, that is, high penetration peak force is not necessary coupled with high energy dissipation capability. The dissipated energy is the integral of force vs. displacement (the area underneath the force vs. displacement curve). For Kevlar[®] fabrics that are able to involve more adjacent yarns to dissipate the energy, their penetration would be delayed to even larger displacement deformation before being penetrated. Thus, a higher energy dissipation capability can be obtained.

Kevlar[®] fabrics layers intercalated with ionic liquid based STFs have significant enhanced penetration resistance than the untreated Kevlar[®] fabric layers during the needle penetration process. This enhanced needle penetration resistance allows the use of ionic liquid based STF-Kevlar[®] to provide better protection against cut hazards without compromising the weight and flexibility. Compared with the PEG based STF-Kevlar[®] nanocomposites, the ionic liquid based STF-Kevlar[®] nanocomposites show comparable but slightly lower protection capability against needle cutting threats. Similar to the spike puncture test cases, this lower needle penetration resistance phenomenon is coming from the difference in the treating process between the lab-scale and industrial-scale. However, it is promising that ionic liquid based STF-Kevlar[®] nanocomposite will have the same needle penetration resistance as the PEG based STF-Kevlar[®] when using the same treatment process.



Figure 7.12: Box charts of penetration peak forces during 18G (A) and 21G (B) needle penetration tests for Kevlar[®] 1148 before and after treated by various ionic liquid based and PEG based STFs.



Figure 7.13: Box charts of dissipated energy during 18G (A) and 21G (B) needle penetration tests for Kevlar[®] 1148 before and after treated by various ionic liquid based and PEG based STFs.

Table 7.7:Summary of increased areal density (%), penetration peak forces, and
dissipated energies during 18G needle penetration tests for Kevlar[®] 1148
before and after treatment by various ionic liquid based and PEG based
STFs.

Treatment	Untreated	0.5 KEP10 IL	0.6 KEP50 IL	STFs (Barrday)
Increased Areal		14.40 ± 0.81	23.97 ± 4.48	14.62 ± 2.24
Density (%)				
Peak Force (N)	25.40 ± 3.55	35.70 ± 5.05	31.99 ± 5.50	38.84 ± 7.02
Dissipated	40.16 ± 6.52	52.54 ± 7.47	49.64 ± 6.67	64.41 ± 14.23
Energy (mJ)				

Table 7.8:Summary of increased areal density (%), penetration peak forces, and
dissipated energies during 21G needle penetration tests for Kevlar[®] 1148
before and after treatment by various ionic liquid based and PEG based
STFs.

Treatment	Untreated	0.5 KEP10 IL	0.6 KEP50 IL	STFs (Barrday)
Increased Areal		14.40 ± 0.81	23.97 ± 4.48	14.62 ± 2.24
Density (%)				
Peak Force (N)	9.11 ± 0.95	12.05 ± 1.38	12.88 ± 1.62	17.11 ± 2.84
Dissipated	12.33 ± 3.27	18.02 ± 5.61	20.16 ± 7.68	19.44 ± 5.55
Energy (mJ)				

7.4 Conclusions

Kevlar[®] 1148 fabrics were treated by ionic liquid $[C_4mim][BF_4]$ based STFs. Because of the stability of ionic liquid under vacuum condition, the ionic liquid based STF-Kevlar[®] nanocomposites could be used in extreme space condition to provide protection against MMOD and stab threats. In this chapter, both spike and needle penetration tests were performed to evaluate the stab resistance of ionic liquid based STF-Kevlar[®] nanocomposites.

As ionic liquids are outstanding lubricants and are able to dissolve numerous polymers, there is the potential that the friction between fabric yarns could be reduced

and that the fiber could be dissolved, both of which would result in lowered penetration resistance. Therefore, the Kevlar[®] fabrics were treated with pure solvents and penetration tests were performed. The results show that the addition of ionic liquid to Kevlar[®] fabric has no obvious change in resistance capability in terms of penetration peak force and dissipated energy. Furthermore, the treatment condition for Kevlar[®] 1148 with ionic liquid based STFs has been evaluated. The dilution ratio of ethanol to STFs was found to be optimal at 3:1 volume ratio when the areal density, penetration peak force, and energy dissipation capability are considered.

Under the spike puncture threats, ionic liquid based STF-Kevlar[®] nanocomposites provide even better resistance against the threats than the PEG based STF-Kevlar[®] nanocomposites using the same treating condition in our lab. Furthermore, the lab-scale manufactured ionic liquid based STF-Kevlar[®] nanocomposites have comparable but slightly lower stab (spike puncture and needle cutting) resistant capability than the commercial PEG based STF-Kevlar[®] nanocomposites due to the potential difference in fabric treatment process. These experiments demonstrate that IL-based STF-ArmorTM has comparable performance to the PEG-based STF-ArmorTM previously used. Comparisons to commercial STF-ArmorTM formulations suggest that further improvements may be possible.

REFERENCES

1. Cheeseman, B. A.; Bogetti, T. A. Ballistic impact into fabric and compliant composite laminates. *Composite structures* **2003**, *61* (1), 161-173.

2. Decker, M.; Halbach, C.; Nam, C.; Wagner, N.; Wetzel, E. Stab resistance of shear thickening fluid (STF)-treated fabrics. *Composites Science and Technology* **2007**, *67* (3), 565-578.

3. Kang, T. J.; Kim, C. Y.; Hong, K. H. Rheological behavior of concentrated silica suspension and its application to soft armor. *J. Appl. Polym. Sci.* **2012**, *124* (2), 1534-1541.

4. Sun, L. L.; Xiong, D. S.; Xu, C. Y. Application of shear thickening fluid in ultra high molecular weight polyethylene fabric. *J. Appl. Polym. Sci.* **2013**, *129* (4), 1922-1928.

5. Mayo, J. B.; Wetzel, E. D.; Hosur, M. V.; Jeelani, S. Stab and puncture characterization of thermoplastic-impregnated aramid fabrics. *International Journal of Impact Engineering* **2009**, *36* (9), 1095-1105.

6. Houghton, J.; Schiffman, B.; Kalman, D.; Wetzel, E.; Wagner, N. Hypodermic needle puncture of shear thickening fluid (STF)-treated fabrics. *Proceedings of SAMPE* **2007**, *3*, 1-11.

7. Lee, Y. S.; Wetzel, E. D.; Wagner, N. J. The ballistic impact characteristics of Kevlar® woven fabrics impregnated with a colloidal shear thickening fluid. *J. Mater. Sci.* **2003**, *38* (13), 2825-2833.

8. Kalman, D. P.; Merrill, R. L.; Wagner, N. J.; Wetzel, E. D. Effect of particle hardness on the penetration behavior of fabrics intercalated with dry particles and concentrated particle– fluid suspensions. *ACS applied materials & interfaces* **2009**, *1* (11), 2602-2612.

9. Wagner, N. J.; Brady, J. F. Shear thickening in colloidal dispersions. *Physics Today* **2009**, *62* (10), 27-32.

10. Lee, Y. S.; Wagner, N. J. Dynamic properties of shear thickening colloidal suspensions. *Rheol. Acta* **2003**, *42* (3), 199-208.

11. Lee, B.; Walsh, T.; Won, S.; Patts, H.; Song, J.; Mayer, A. Penetration failure mechanisms of armor-grade fiber composites under impact. *Journal of Composite Materials* **2001**, *35* (18), 1605-1633.

12. Egres, R.; Lee, Y.; Kirkwood, J.; Kirkwood, K.; Wetzel, E.; Wagner, N. In *Novel flexible body armor utilizing shear thickening fluid (STF) composites*, Proceedings, 14th Int. Conf. on Composite Materials, San Diego, CA, Soc. Manufacturing Engineers, paper, 2003.

13. Kirkwood, K. M.; Kirkwood, J. E.; Lee, Y. S.; Egres, R. G.; Wagner, N. J.; Wetzel, E. D. Yarn pull-out as a mechanism for dissipating ballistic impact energy in Kevlar® KM-2 fabric part I: quasi-static characterization of yarn pull-out. *Text. Res. J.* **2004**, *74* (10), 920-928.

14. Kirkwood, J. E.; Kirkwood, K. M.; Lee, Y. S.; Egres, R. G.; Wagner, N. J.; Wetzel, E. D. Yarn pull-out as a mechanism for dissipating ballistic impact energy in Kevlar® KM-2 fabric part II: predicting ballistic performance. *Text. Res. J.* **2004**, *74* (11), 939-948.

15. Zeng, X.; Tan, V.; Shim, V. Modelling inter - yarn friction in woven fabric armour. *International Journal for Numerical Methods in Engineering* **2006**, *66* (8), 1309-1330.

16. Gao, J.; Ndong, R. S.; Shiflett, M. B.; Wagner, N. J. Creating nanoparticle stability in ionic liquid [C4mim][BF4] by inducing solvation layering. *ACS Nano* **2015**, *9* (3), 3243-3253.

17. Ueno, K.; Imaizumi, S.; Hata, K.; Watanabe, M. Colloidal interaction in ionic liquids: Effects of ionic structures and surface chemistry on rheology of silica colloidal dispersions. *Langmuir* **2009**, *25* (2), 825-831.

18. Ueno, K.; Hata, K.; Katakabe, T.; Kondoh, M.; Watanabe, M. Nanocomposite ion gels based on silica nanoparticles and an ionic liquid: ionic transport, viscoelastic properties, and microstructure. *J. Phys. Chem. B* **2008**, *112* (30), 9013-9019.

19. Welton, T. Room-temperature ionic liquids. Solvents for synthesis and catalysis. *Chem. Rev.* **1999**, *99* (8), 2071-2084.

20. Sheldon, R. Catalytic reactions in ionic liquids. *Chem. Commun.* **2001**, (23), 2399-2407.

21. Cwalina, C. D.; Dombrowski, R. D.; McCutcheon, C. J.; Christiansen, E. L.; Wagner, N. J. MMOD Puncture Resistance of EVA Suits with Shear Thickening Fluid (STF) – Armortm Absorber Layers. *Procedia Engineering* **2015**, *103*, 97-104.

22. Cwalina, C. D.; McCutcheon, C. M.; Dombrowski, R. D.; Wagner, N. J. Engineering enhanced cut and puncture resistance into the thermal micrometeoroid garment (TMG) using shear thickening fluid (STF) – ArmorTM absorber layers. *Composites Science and Technology* **2016**, *131*, 61-66.

23. Wetzel, E. D.; Lee, Y.; Egres, R.; Kirkwood, K.; Kirkwood, J.; Wagner, N. In *The effect of rheological parameters on the ballistic properties of shear thickening fluid (STF)-Kevlar composites*, AIP Conference Proceedings, **2004**; IOP INSTITUTE OF PHYSICS PUBLISHING LTD, pp 288-293.

24. Tan, V.; Tay, T.; Teo, W. Strengthening fabric armour with silica colloidal suspensions. *International journal of solids and structures* **2005**, *42* (5), 1561-1576.

25. ASTM F1342 / F1342M-05(2013)e1, Standard Test Method for Protective Clothing Material Resistance to Puncture. ASTM International: West Conshohocken, PA, **2013**.

26. Liu, W. M.; Ye, C. F.; Gong, Q. Y.; Wang, H. Z.; Wang, P. Tribological performance of room-temperature ionic liquids as lubricant. *Tribology Letters* **2002**, *13* (2), 81-85.

27. Keskin, S.; Kayrak-Talay, D.; Akman, U.; Hortacsu, O. A review of ionic liquids towards supercritical fluid applications. *J. Supercrit. Fluids* **2007**, *43* (1), 150-180.

28. Street, K. W.; Morales, W.; Koch, V. R.; Valco, D. J.; Richard, R. M.; Hanks, N. Evaluation of Vapor Pressure and Ultra-High Vacuum Tribological Properties of Ionic Liquids. *Tribol. Trans.* **2011**, *54* (6), 911-919.

29. Morales, W.; Street, K. W.; Richard, R. M.; Valco, D. J. Tribological Testing and Thermal Analysis of an Alkyl Sulfate Series of Ionic Liquids for Use as Aerospace Lubricants. *Tribol. Trans.* **2012**, *55* (6), 815-821.

30. Wasserscheid, P.; Keim, W. Ionic liquids - New "solutions" for transition metal catalysis. *Angew. Chem. Int. Ed.* **2000**, *39* (21), 3772-3789.

31. Richard, D. D. N. J. W. Measurement of needle puncture resistance using an electronic puncture detection system. In *Proceedings of SAMPE*: Baltimore, MD, **2012**.

32. Termonia, Y. Puncture resistance of fibrous structures. *International Journal of Impact Engineering* **2006**, *32* (9), 1512-1520.

Chapter 8

CONCLUSIONS AND FUTURE DIRECTIONS

8.1 Summary

Shear thickening fluid (STF)-nanocomposites (STF-ArmorTM) are a promising candidate material for use in extra-vehicular activity (EVA) suits to protect astronauts against threats of hypervelocity MMOD impacts and physical hazards from tools, and sharp edges.¹ The extreme environmental condition of space requires the development of novel STF formulations to be stable under vacuum and low temperature environment.

In this dissertation, I have developed novel STF formulations to be used in space conditions, and further, they have been successfully intercalated into aramid fabrics Kevlar[®] to form STF nanocomposites. The ionic liquid $[C_4mim][BF_4]$ is used as the carrier medium for the new developed formulation due to its low volatility and high thermal stability. The stabilization mechanism, nanostructure, and dynamics of spherical silica particles dispersed in ionic liquid $[C_4mim][BF_4]$ are investigated. With this knowledge, the primary goal of designing an ionic liquid based STF formulation is achieved. In support of this development work, the effect of water on the viscosity, density, and nanoscale structure of this ionic liquid are measured and analyzed within the framework of free volume theory. This insight provides guidance to formulate the rheological properties of ionic liquid colloidal dispersions. The major conclusions of this dissertation are summarized as follows and the directions for future research are suggested.

8.1.1 Ionic Liquids-Water Interaction and Phase Transition

Ionic liquids can easily absorb water from humid atmosphere and their thermodynamics properties and microstructure are altered significantly due to the presence of water. In order to develop ionic liquid colloidal dispersion formulations, it is thus crucial to understand the effect of water. The interaction between ionic liquid $[C_4mim][BF_4]$ and water molecules, the induced microstructure and phase behavior are examined in Chapter 3. The phase behavior of ionic liquid $[C_4mim][BF_4]$ -water mixtures covering the full composition range from high salt to low salt is investigated. With the addition of water into ionic liquid $[C_4mim][BF_4]$, there are four characteristic morphologies in the phase diagram: soluble water in ionic liquid with hydrogen bonding, water cluster formation, ionic liquid micellar aggregation, and dilute electrolyte solution.

At high salt levels, water molecularly dissolves into the network of ionic liquid $[C_4mim][BF_4]$ through hydrogen bonds, and a water-ionic liquid hydrogen bond network is formed without microphase separation. This hydrogen bonding is examined by studying the effects of added water molecules on the physical properties (density and viscosity) of ionic liquid $[C_4mim][BF_4]$ and the non-ideality of ionic liquid aqueous solutions. Densities and viscosities of the homogenous binary mixtures of $[C_4mim][BF_4]$ and water are measured for a temperature range of 283.15 to 373.15 K in the salt-rich regime. Excess molar volumes and non-idealities of the viscosities are calculated and discovered to be highly correlated. A new, semi-empirical correlation between the excess mixture viscosity and the excess molar volume in the salt-rich regime is identified, which is motivated by free volume theory. Additionally, a survey of the literature identifies ionic liquids with similar ionic structures for which this correlation is valid, which provides evidence for the role of specific molecular

interactions that govern the observed non-idealities in salt-rich mixtures of water in ionic liquids.

As the water concentration increases (over the salt-rich regime) to above ~ 70 mol%, a microphase transition to water nanoclusters resembling an inverse microemulsion is identified and characterized using small angle neutron scattering (SANS). In the salt-rich regime, water molecules are isolated and accommodated in the ionic liquid cation-anion polar network by interacting with anions and cations through hydrogen bonds. Such mixtures are a homogenous solution, with no significant changes in microstructure upon water addition as detected by SANS measurement. Water starts to form clusters when the molecular ratio of water to ionic liquid exceeds ~ 2:1, as expected from molecular structure standpoint. Distinct nanometer-sized water clusters form and grow with increase of water content, and water continues to partition between the dissolved state and microphase separated state.

On the other hand, ionic liquid $[C_4mim][BF_4]$ forms micellar aggregates at low salt regime due to the amphiphilic structural properties of cation $[C_4mim]^+$, which is revealed by small-angle x-ray scattering (SAXS). The CMC of $[C_4mim][BF_4]$ in H₂O is determined to be 0.8 mol/L, which is consistent with previous literature results.²⁻³ Spherical micelles of ionic liquid are formed at concentrations above CMC; however, micelles grow to ellipsoid rod shape as ionic liquid concentration increases. A "rod" like micellar structure is observed at higher concentration.

These observations establish a full phase diagram of ionic liquid aqueous solutions (microphase separation, phase inversion, and micelle formation) with addition of water, revealing similarities to traditional oil-water-surfactant systems. These results provide direct confirmation for recent simulations, as well insight into

the source of non-idealities in some thermophysical and transport properties (e.g. density and viscosity) of salt-rich aqueous mixtures reported in literature. These complex phase transitions of aqueous solutions of ionic liquid $[C_4mim][BF_4]$ and the non-ideal viscosity behavior provide a guidance for the following rheological measurements of ionic liquid colloidal suspensions in this dissertation. In order to reduce the effect of absorbed humidity water on the viscosity and microstructure measurements, all the suspensions are dehydrated before measurements and fresh suspensions are used for each test.

8.1.2 Stabilization Mechanism and Interfacial Structure of Silica Particles in Ionic Liquids

Dispersing spherical silica particles into ionic liquids is challenging as the electrostatic repulsive forces are screened due to the high ionic strength nature of ionic liquids; thus, silica particles intend to aggregate as a result from the dominated van der Waals attraction. However, stable particle dispersion is crucial for shear thickening properties. Herein, a method to spontaneously disperse silica particles in ionic liquid $[C_4mim][BF_4]$ is developed by inducing solvent structuring around the surface of the particle. The model system studied consists of spherical silica particles of ~ 100 nm size with 10% polydispersity, which are coated by fluorocarbon 1H,1H,9H hexadecafluoro-1-nonanol through dehydration reaction following the procedure of van Helden *et al.*⁴ The surface coating is confirmed by ¹H NMR and the grafting density is determined to be 1.4 chain/nm² with 60 % surface coverage using TGA (Chapter 4).

The stabilization mechanism and the corresponding interfacial structure of the fluorocarbon coated silica particles in ionic liquid [C₄mim][BF₄] are revealed in

Chapter 5. Solvent structured layers on the surface of particles, which are initiated by the hydrogen bonding between coated fluorocarbon and anion [BF₄]⁻, are proven to provide sufficient steric stabilization against the dispersion attractions and result in stable colloidal silica particles dispersed in ionic liquid. The thickness of the solvation layers on the surface of particles is determined to be approximately 5 nm through extensive and independent characterization techniques, including dynamic light scattering (DLS), SANS, transmission electron microscopy (TEM), and rheology. The thickness of the solvation layers, which corresponds to seven ion-pair layers, is quantitatively consistent with previous studies on ionic liquid structuring near confined flat mica surfaces. ⁵⁻⁸ Additionally, it is assumed for simplicity that the solvation layers form a well-defined, uniform shell surrounding the nanoparticles. However, the solvation layers near flat substrates are examined to become less organized and smoothly transition to the bulk liquid, as shown by experimental measurements^{5-6, 9-11} and simulations¹².

This designer surface coating on the particles alters specific molecular interactions between the particle surface and the ionic liquid. The induced solvent structuring is demonstrated as an effective method to create spontaneous dispersion and stabilization of nanoparticles that are otherwise not dispersible in ionic liquids. With the additional repulsion force from the solvation layer, silica particles are able to form stable suspensions in ionic liquid up to 61 wt% (corresponding to 45 vol%) solid content, which is a significant improvement on the current published maximum of 20 wt% particle content. These comprehensive results for fluorinated silica particle dispersions and the solvation layer mechanism not only provide a database for testing the proposed stabilization mechanism, but also act as a guidance for the development of stable ionic liquid based shear thickening fluids formulation through the design of surface coatings.

8.1.3 Ionic Liquids Based Shear Thickening Fluids and Thermal Response

Silica particles coated by fluorocarbon have been demonstrated to be stable in ionic liquid $[C_4mim][BF_4]$ with the presence of 5 nm solvation layering structure around the particle surface. However, this stable fluorinated silica particle dispersion only exhibits mild shear thickening properties, which is not sufficient for the desired protection against multiple threats. Thus, a new ionic liquid based colloidal formulation system which demonstrates significant shear thickening behavior is crucial.

In chapter 6, stable ionic liquid based shear thickening colloidal dispersions have been formulated through the design of the particle surface chemistry. An alcohol hydrocarbon coated silica particle system is formulated. At room temperature, this concentrated silica nanoparticle dispersion in $[C_4mim][BF_4]$ displays stronger shear thickening than the previous studied fluorocarbon coated silica particle system. This is shown to be a consequence of a thinner solvation layers (~3 nm) initiated by relatively weaker hydrogen bonds between the hydrocarbon surface coating and anion group. In comparison to the strong hydrogen bonds between fluorinated surface and the anion, the hydrogen bonds between the alcohol hydrocarbon surface and the anion are much weaker. This 3 nm solvation layers provides sufficient stabilization force between particles, while still allowing the particles to approach together and shearthicken at high shear rate/stress. This finding provides valuable insights for formulating stable, shear thickening dispersions in ionic liquids. Additionally, the alcohol hydrocarbon coated silica particles in ionic liquid suspension displays similar

shear thickening response as that of silica particles in polyethylene glycol (PEG), and thus, this formulation is promising to provide similar protection as PEG based STFs when intercalated with textiles.

Furthermore, the colloidal stability induced by the formation of solvation layers is found to be influenced by temperature. An increase of particle attraction and loss of stability of colloidal dispersions above 30 °C is discovered through rheological and SANS measurements. The mechanism driving increased particle attraction with increasing temperature is shown as a consequence of a reduction in the solvation layer thickness due to weakening hydrogen bonds between anion $[BF_4]^-$ and the surface. SANS and DLS measurements confirm the reduction of solvation layering thickness at elevated temperatures, revealing a transition from stable dispersion to unstable, attractive gel. This transition is compared with the established AHS particle state diagram. Because of the fundamentally different stabilization mechanisms, the gel transition at high temperature herein is in the opposite direction of traditional reported gel transition for coated particles dispersions at low temperatures.¹³⁻¹⁴ This investigation of a novel gel transition at high temperatures (*i.e.*, an inverse melting transition), the surface solvation structure, and the corresponding interparticle potential provide a model study for designing materials through surface modification and temperature control.

8.1.4 Protective Properties of Ionic Liquids Based STF-ArmorTM

With the successful development of ionic liquid based shear thickening fluid formulation as described in Chapter 6, the ionic liquid based STF-Kevlar[®] nanocomposites are fabricated and their stab resistance performance is evaluated using spike and needles (18 G, 21G) penetration tests.

Prior to the penetration tests on the ionic liquid based STF-Kevlar[®] nanocomposites, the effect of changing the solvents from traditional STF carrier medium, PEG, to ionic liquid $[C_4mim][BF_4]$ is evaluated. Ionic liquids are outstanding lubricants and can dissolve numerous polymers, thus it is possible to have dissolved fiber and a lower penetration resistance with reduced friction between fabric yarns when treated with ionic liquids. However, our study indicates no obvious change in resistance capability for the different solvents treatments.

Thus, with the proof that the addition of ionic liquid to Kevlar[®] fabric does not change the protection performance of fabric itself, the treatment condition for Kevlar[®] 1148 with ionic liquid based STFs has been evaluated. The dilution ratio of ethanol to STFs was found to be optimized at 3:1 volume ratio when the areal density, penetration peak force, and energy dissipation capability are considered.

The stab resistance of the lab-scale manufactured ionic liquid based STF-Kevlar[®] against threats from a spike and 21 & 18 gauge needles are compared with (lab-scale manufactured and commercial) PEG-based STF-Kevlar[®]. It is found that ionic liquid based STF-Kevlar[®] nanocomposites provide even better resistance than the lab-scale manufactured PEG-based STF-Kevlar[®] nanocomposites against the spike puncture hazards. Furthermore, ionic liquid based STF-Kevlar[®] nanocomposites have comparable, but slightly lower, stab (spike puncture and needle cutting) resistant capability than the commercial PEG based STF-Kevlar[®] nanocomposites due to the potential difference in fabric treatment process. These findings suggest that IL-based STF-ArmorTM has comparable performance to the PEG-based STF-ArmorTM previously used. Further improvements may be possible with implanting same fabric treatment process as commercial STF-ArmorTM. Overall, the research completed in

this dissertation provides promising new ionic liquid based STF-nanocomposites for astronaut protection.

8.2 **Recommendations for Future Work**

Ionic liquids are chosen as the carrier medium for the novel STF formulations due to their low volatility and thermal stability which ensure the ability to flow under vacuum and extreme temperatures space environmental condition. In this dissertation, shear thickening behavior have been observed when performing rheological measurements on the ionic liquid based STF formulations under atmosphere and room temperature condition. The rheological response of the ionic liquid based STF formulations at high temperatures has also been investigated. However, it is of importance to measure their accurate and precise rheological response at simulated low-pressure/low temperature environmental conditions because they are aimed to be used for application in space. Preliminary results for the rheological responses of the ionic liquid based STF are achieved at simulated extreme low temperature using environmental test chamber (ETC) oven and extra low-pressure environment using the designed vacuum cell as described in Chapter 2.

The liquid state of ionic liquids at extreme low temperature has been proven as shown in Figure 8.1. The viscosity of ionic liquid $[C_4mim][BF_4]$ is measured inside the ETC oven with temperature decreasing from 10 °C to -80 °C at a rate of 5 °C/min. The temperature inside the ETC oven is controlled by liquid nitrogen. It is obvious that when the temperature reaches -65 °C, the torque no longer increases and the shear rate drops significantly, which indicates the freezing of ionic liquid $[C_4mim][BF_4]$. Thus, $[C_4mim][BF_4]$ is shown to be in the liquid state above -65 °C; this broad liquid state range is suitable for space applications. Furthermore, shear thickening behavior

is observed for concentrated silica particles- $[C_4mim][BF_4]$ dispersions at extreme low temperature. Figure 8.2 displays an example of the rheological properties of $\phi = 0.4$ KE-P10 silica particles in $[C_4mim][BF_4]$ at temperature down to -40 °C. The dispersion exhibits shear thickening behavior at the 25 °C to -40 °C temperature range studied. The order-of magnitude higher viscosity with decreasing temperature is expected as the ionic liquid solvent is approaching its freezing point. Additionally, the relative viscosity follows a general master curve, indicating no significant change in the microstructure at the temperature range studied. Note, there are slight increases in shear thickening strength and yielding at low shear stress at -40 °C. These may come from the structure rearrangement of ionic liquid when $[C_4mim][BF_4]$ enters into glass state before freezing completely. Future researchers should study the structure of ionic liquid at the close to-freezing state, which aids to understand the particle interaction and hydrocluster formation under these extremely low temperatures.



Figure 8.1: The rheological response of ionic liquid [C₄mim][BF₄] using temperature sweep measurements with shear rate 2 s⁻¹ under atmospheric condition. Temperature starts from 10 °C and cools down to -80 °C at a cooling rate of 5 °C/min.



Figure 8.2: Steady state viscosity (A) and relative viscosity (B) as a function of shear stress at various temperature from 25 °C to -40 °C for $\phi = 0.4$ KE-P10 silica particles in [C₄mim][BF₄] under atmospheric condition.

A vacuum environment is achieved by using the vacuum cell described in Figure 2.3. Pressure as low as 10^{-6} torr is reached inside the vacuum cell within one hour using the turbo-pump station and the pressure curve is shown in Figure 8.3A.

The steady state flow curve of $\phi = 0.4$ KE-P10 silica particles in [C₄mim][BF₄] at low pressure 10⁻⁶ torr is measured at 25 °C and compared with that measured at atmosphere condition in Figure 8.3B. The dispersion at this low pressure condition displays similar shear thickening behavior as for atmospheric conditions with the same critical shear stress and shear thickening strength. However, the shear viscosity of the dispersion is systematically higher at the low pressure condition, which may be the results of evaporating residual water or air bubbles at the extreme low pressure condition. Further research will be needed to address this question and identify the source of increased viscosity. Some improvements can be made for the vacuum cell setup, such as addition of de-gas mesh screen and tubing to accelerate the de-gassing procedure and allow liquid flow back to test cell after being de-gased; installation of control unit to control the pressure inside the cell to desired value. Further improvement can be achieved by introducing low-temperature heat transfer fluids (temperature can reach as low as -110 °C) to the vacuum cell jacket, thus the rheological response of STFs can be measured at simulated low temperature/vacuum space condition.



Figure 8.3: A) The pressure change inside the vacuum cell chamber as a function of time when the turbo-pump station starts to work at 25 °C. B) Comparison of the steady state viscosity of $\phi = 0.4$ KE-P10 silica particles in [C₄mim][BF₄] at 25 °C under atmospheric and 10⁻⁶ hPa vacuum conditions.

The ionic liquid based STF formulations developed in this dissertation have shown distinguished shear thickening behavior not only at room temperature, but also under extreme low temperature and vacuum condition. However, the shear thickening property disappears at high temperatures due to the loss of particle stability. This is because the strength of the hydrogen bonds decreases with increasing temperature,¹⁵⁻¹⁶ and the solvation layers initiated by hydrogen bonds between surface coating and anion group of [C₄mim][BF₄] are also interrupted. The thermal sensitivity of colloidal stability and the corresponding rheological response can be adjusted through replacing the hydrogen bonds with other physical/chemical bonds, such as ionic bonds or covalent bonds, which are less sensitive to temperature. For example, a hydrocarbon chain with a functional group of strong charge on one chain end (either positive or negative charge) and hydroxyl group on other chain end can be covalently coated onto the silica particle surface through the same dehydration reaction⁴ discussed in this dissertation. Thus, the surface of silica particles are fully covered by positive or negative charge, which will initiate the solvation layers through ionic bonding between positive surface charge and anion $[BF_4]$ or between negative surface charge and cation $[C_4 mim]^+$. As the ionic bonds are nearly temperature independent, the solvation layers and the particle stability will be insensitive to temperature; thus, the shear thickening rheological response should also be temperature independent.

The developed novel ionic liquid based STF formulation displays similar rheological shear thickening behavior as that of PEG-based STF. Moreover, the fabricated ionic liquid based STF-Kevlar[®] nanocomposites are proved to provide even better resistance against the stab threats than the PEG based STF-Kevlar[®] nanocomposites through quasi-static puncture tests in this dissertation. As for the

space protection application against MMOD impact, their protective performance needs to be evaluated under simulated hypervelocity tests of high energy and short time scale. It has been reported that PEG based STF-ArmorTM provides enhanced protection against simulated hypervelocity MMOD impacts, with higher energy absorption capability than prototype lay-ups of the standard thermal micrometeoroid garment (TMG).¹ The damage of the PEG based STF-ArmorTM textile mainly comes from the burning and evaporating of textile layer. Compared with PEG solvent, most of ionic liquids are non-flammable and do not evaporate; thus, they have the potential to provide better protection against hypervelocity impact when using ionic liquid based STF-ArmorTM. Hypervelocity tests on ionic liquid based STF-ArmorTM will be crucial to verify this hypothesis in the future research.

A further step after the simulated MMOD hypervelocity test on STF-ArmorTM performed on Earth is having the STF-ArmorTM materials tested in the real space environment experiencing the MMOD impact threats. A proposal for NASA's Materials on International Space Station Experiment (MISSE) have been awarded to University of Delaware for the investigation of MMOD hypervelocity impact resistance of TMG prototype lay-ups with STF-ArmorTM absorber layers under the real low-earth orbit (LEO) environment. Flying over 200 miles above Earth, fixed to the exterior of the International Space Station (ISS) for a period of one year, the TMG lay-ups will endure extreme harsh environment of LEO: MMOD impact, vacuum, atomic oxygen, solar and charged-particle radiation, direct sunlight and extreme temperature. The experiment will provide a better and direct understanding of material durability and hypervelocity impact resistance of STF-ArmorTM in the real

LEO environment, which will be greatly beneficial for the designer STF-ArmorTM material for space protective application.

REFERENCES

1. Cwalina, C. D.; Dombrowski, R. D.; McCutcheon, C. J.; Christiansen, E. L.; Wagner, N. J. MMOD Puncture Resistance of EVA Suits with Shear Thickening Fluid (STF) – Armortm Absorber Layers. *Procedia Engineering* **2015**, *103*, 97-104.

2. Bowers, J.; Butts, C. P.; Martin, P. J.; Vergara-Gutierrez, M. C.; Heenan, R. K. Aggregation behavior of aqueous solutions of ionic liquids. *Langmuir* **2004**, *20* (6), 2191-2198.

3. Almasy, L.; Turmine, M.; Perera, A. Structure of aqueous solutions of ionic liquid 1-butyl-3-methylimidazolium tetrafluoroborate by small-angle neutron scattering. *J. Phys. Chem. B* **2008**, *112* (8), 2382-2387.

4. Van Helden, A.; Jansen, J. W.; Vrij, A. Preparation and characterization of spherical monodisperse silica dispersions in nonaqueous solvents. *J. Colloid Interface Sci.* **1981**, *81* (2), 354-368.

5. Ueno, K.; Watanabe, M. From colloidal stability in ionic liquids to advanced soft materials using unique media. *Langmuir* **2011**, *27* (15), 9105-9115.

6. Perkin, S.; Crowhurst, L.; Niedermeyer, H.; Welton, T.; Smith, A. M.; Gosvami, N. N. Self-assembly in the electrical double layer of ionic liquids. *Chem. Commun.* **2011**, *47* (23), 6572-6574.

7. Ueno, K.; Kasuya, M.; Watanabe, M.; Mizukami, M.; Kurihara, K. Resonance shear measurement of nanoconfined ionic liquids. *Phys. Chem. Chem. Phys.* **2010**, *12* (16), 4066-4071.

8. Perkin, S. Ionic liquids in confined geometries. *Phys. Chem. Chem. Phys.* **2012**, *14* (15), 5052-5062.

9. Smith, A. M.; Lovelock, K. R. J.; Perkin, S. Monolayer and bilayer structures in ionic liquids and their mixtures confined to nano-films. *Faraday Discuss.* **2013**, *167* (0), 279-292.

10. Hayes, R.; Warr, G. G.; Atkin, R. At the interface: solvation and designing ionic liquids. *Phys. Chem. Chem. Phys.* **2010**, *12* (8), 1709-1723.

11. Atkin, R.; Warr, G. G. Structure in confined room-temperature ionic liquids. *J. Phys. Chem. C* **2007**, *111* (13), 5162-5168.

12. Pinilla, C.; Del Pópolo, M. G.; Lynden-Bell, R. M.; Kohanoff, J. Structure and dynamics of a confined ionic liquid. Topics of relevance to dye-sensitized solar cells. *J. Phys. Chem. B* **2005**, *109* (38), 17922-17927.

13. Kim, J. M.; Fang, J.; Eberle, A. P. R.; Castañeda-Priego, R.; Wagner, N. J. Gel transition in adhesive hard-sphere colloidal dispersions: The role of gravitational effects. *Phys. Rev. Lett.* **2013**, *110* (20), 208302.

14. Eberle, A. P.; Castañeda-Priego, R. n.; Kim, J. M.; Wagner, N. J. Dynamical arrest, percolation, gelation, and glass formation in model nanoparticle dispersions with thermoreversible adhesive interactions. *Langmuir* **2012**, *28* (3), 1866-1878.

15. Jorgensen, W. L.; Madura, J. D. Temperature and size dependence for Monte Carlo simulations of TIP4P water. *Mol. Phys.* **1985**, *56* (6), 1381-1392.

16. Mizan, T. I.; Savage, P. E.; Ziff, R. M. Temperature dependence of hydrogen bonding in supercritical water. *J. Phys. Chem.* **1996**, *100* (1), 403-408.

Appendix A

SUPPORTING INFORMATION FOR CHAPTER 3

Table A.1: Summary of experimental density ρ , viscosity η , density deviation $\Delta \rho$, excess molar volume \underline{V}^{ex} and excess log viscosity $(log\eta)^{ex}$ of $[C_4 mim][BF_4]$ aqueous solutions at various temperatures.

_	Water content (wt%)							
T (K)	0.00	0.28	0.52	1.01	2.03	3.01	5.00	
$\rho (g/cm^3)$								
283.15	1.21203	1.21100	1.21008	1.20846	1.20535	1.20215	1.19586	
293.15	1.20479	1.20376	1.20281	1.20118	1.19807	1.19485	1.18852	
298.15	1.20117	1.20014	1.19920	1.19757	1.19446	1.19123	1.18486	
303.15	1.19758	1.19655	1.19561	1.19398	1.19086	1.18762	1.18122	
313.15	1.19045	1.18943	1.18848	1.18685	1.18370	1.18042	1.17395	
323.15	1.18341	1.18237	1.18141	1.17977	1.17660	1.17330	1.16673	
333.15	1.17642	1.17538	1.17441	1.17275	1.16954	1.16620	1.15953	
343.15	1.16949	1.16843	1.16746	1.16578	1.16255	1.15916	1.15240	
353.15	1.16261	1.16155	1.16056	1.15886	1.15558	1.15215	1.14532	
363.15	1.15578	1.15370	1.15370	1.15198	1.14866	1.14518	1.13827	
			Δho	(g/cm^3)				
283.15	0	-0.00031	-0.00061	-0.00098	-0.00147	-0.00217	-0.00342	
293.15	0	-0.00033	-0.00068	-0.00110	-0.00167	-0.00247	-0.00392	
298.15	0	-0.00034	-0.00069	-0.00112	-0.00173	-0.00257	-0.00413	
303.15	0	-0.00035	-0.00070	-0.00115	-0.00180	-0.00268	-0.00433	
313.15	0	-0.00036	-0.00073	-0.00120	-0.00194	-0.00290	-0.00471	
323.15	0	-0.00039	-0.00078	-0.00128	-0.00207	-0.00310	-0.00508	
333.15	0	-0.00039	-0.00081	-0.00134	-0.00220	-0.00329	-0.00543	
343.15	0	-0.00042	-0.00084	-0.00140	-0.00230	-0.00346	-0.00572	
353.15	0	-0.00042	-0.00086	-0.00145	-0.00241	-0.00362	-0.00598	
363.15	0	-0.00044	-0.00089	-0.00150	-0.00250	-0.00377	-0.00621	
\underline{V}^{ex} (cm ³ /mol)								
283.15	0	0.047	0.089	0.135	0.185	0.251	0.342	

298.15 0 0.051 0.100 0.154 0.215 0.205 0.33 298.15 0 0.052 0.102 0.158 0.222 0.303 0.4	20							
303 15 0 0.054 0.105 0.163 0.232 0.318 0.4	12							
313 15 0 0.055 0.110 0.173 0.252 0.348 0.4	88							
323 15 0 0.061 0.119 0.186 0.273 0.376 0.5	32							
333 15 0 0.063 0.124 0.197 0.294 0.404 0.5	/2 16							
343 15 0 0.067 0.131 0.208 0.310 0.430 0.6	4							
353 15 0 0.069 0.136 0.218 0.329 0.455 0.6	19							
363 15 0 0.073 0.143 0.229 0.346 0.480 0.6	32							
n (mPa s)								
$293.15 133.48 \pm 0.86 115.23 \pm 0.43 102.19 \pm 0.51 81.44 \pm 0.31 56.26 \pm 0.15 42.00 \pm 0.31 26.68 \pm 0.15$	0.30							
$298.15 101.39 \pm 0.74 88.77 \pm 0.33 79.43 \pm 0.28 64.34 \pm 0.19 45.35 \pm 0.16 34.43 \pm 0.31 22.34 \pm 0.16 24.43 \pm 0.31 22.34 \pm 0.31 23.34 23.34 23.34 \pm 0.31 23.34 $	0.33							
$303.15 78.84 \pm 0.41 69.82 \pm 0.18 63.06 \pm 0.12 51.73 \pm 0.14 37.18 \pm 0.20 28.67 \pm 0.32 18.95 \pm 0.14 18.95 18.95 18.95 18.95 18.95 18.95 18.95 $	0.34							
$313.15 50.52 \pm 0.20 45.58 \pm 0.05 41.37 \pm 0.72 35.01 \pm 0.08 26.09 \pm 0.24 20.59 \pm 0.34 14.10 \pm 0.08 26.09 \pm 0.24 20.59 \pm 0.34 20.59 \pm 0.54 20.59 \pm 0.$	0.37							
$323.15 34.52 \pm 0.10 31.56 \pm 0.13 29.13 \pm 0.23 25.01 \pm 0.20 19.17 \pm 0.28 15.49 \pm 0.36 10.89 10.89 \pm 0.36 10.89 10.89 10.89 10.89 $	0.36							
$333.15 24.86 \pm 0.13 22.98 \pm 0.22 21.40 \pm 0.27 18.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.25 14.66 \pm 0.30 12.06 \pm 0.37 8.67 \pm 0.37 $	0.33							
$343.15 18.69 \pm 0.20 17.44 \pm 0.26 16.34 \pm 0.33 14.45 \pm 0.29 11.59 \pm 0.32 9.64 \pm 0.35 7.11 \pm 0.31 \pm 0.31 14.45 \pm 0.29 11.59 \pm 0.32 14.45 \pm 0.31 14.45 \pm 0.29 11.59 \pm 0.32 14.45 \pm 0.31 14.45 1$	0.31							
$353.15 14.54 \pm 0.25 13.67 \pm 0.31 12.87 \pm 0.35 11.50 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 12.87 \pm 0.35 11.50 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 12.87 \pm 0.35 11.50 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 12.87 \pm 0.35 11.50 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 12.87 \pm 0.35 11.50 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 12.87 \pm 0.31 12.87 \pm 0.35 11.50 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 9.40 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 9.40 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 9.40 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 9.40 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 9.40 \pm 0.31 9.40 \pm 0.33 7.91 \pm 0.33 5.98 \pm 0.31 9.40 \pm 0.31 9.40$	0.30							
$363.15 11.64 \pm 0.28 11.00 \pm 0.32 10.40 \pm 0.36 9.39 \pm 0.31 7.79 \pm 0.32 6.65 \pm 0.30 5.16 \pm 0.30 5.$	0.29							
$373.15 9.55 \pm 0.30 9.06 \pm 0.32 8.60 \pm 0.34 7.83 \pm 0.31 6.60 \pm 0.31 5.71 \pm 0.30 4.54 \pm 0.51 6.60 \pm 0.31 5.71 \pm 0.30 4.54 \pm 0.31 5.71 \pm 0.30 5.71 $	0.29							
$(logn)^{ex}(log(mPas))$								
293 15 0 0.008 + 0.002 0.015 + 0.002 0.027 + 0.002 0.064 + 0.001 0.094 + 0.003 0.147 + 0.002 0.0027 + 0.002 0.064 + 0.001 0.094 + 0.003 0.147 + 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.003 0.0	0.005							
298.15 0 0.012 + 0.002 0.021 + 0.002 0.036 + 0.001 0.076 + 0.002 0.108 + 0.004 0.162 + 0.002 0.036 + 0.001 0.076 + 0.002 0.108 + 0.004 0.162 + 0.002 0.036 + 0.001 0.076 + 0.002 0.002 0.004 0.162 + 0.004 0.0162 + 0.002 0.004	0.007							
$303.15 \qquad 0 \qquad 0.015 + 0.001 \qquad 0.026 + 0.001 \qquad 0.044 + 0.001 \qquad 0.086 + 0.002 \qquad 0.121 + 0.005 \qquad 0.175 + 0.001 \qquad 0.026 + 0.001 \qquad 0.044 + 0.001 \qquad 0.086 + 0.002 \qquad 0.121 + 0.005 \qquad 0.175 + 0.001 \qquad 0.026 + 0.001 \qquad 0.044 + 0.001 \qquad 0.086 + 0.002 \qquad 0.121 + 0.005 \qquad 0.175 + 0.001 \qquad 0.044 + 0.001 \qquad 0.086 + 0.002 \qquad 0.121 + 0.005 \qquad 0.175 + 0.001 \qquad 0.044 + 0.001 \qquad 0.086 + 0.002 \qquad 0.121 + 0.005 \qquad 0.175 + 0.001 \qquad 0.044 + 0.001 \qquad 0.086 + 0.002 \qquad 0.026 + 0.001 \qquad 0.044 + 0.001 \qquad 0.086 + 0.002 \qquad 0.026 + 0.005 \qquad 0.0175 + 0.001 \qquad 0.086 + 0.002 \qquad 0.0121 + 0.005 \qquad 0.0175 + 0.001 \qquad 0.044 + 0.001 \qquad 0.086 + 0.002 \qquad 0.0121 + 0.005 \qquad 0.0175 + 0.001 \qquad 0.044 + 0.001 \qquad 0.086 + 0.002 \qquad 0.0121 + 0.005 \qquad 0.0175 + 0.001 \qquad 0.086 + 0.002 \qquad 0.0121 + 0.005 \qquad 0.0175 + 0.001 \qquad 0.086 + 0.002 \qquad 0.026 + 0.001 \qquad 0.086 + 0.002 \qquad 0.0121 + 0.005 \qquad 0.0175 + 0.001 \qquad 0.086 + 0.002 \qquad 0.026 + 0.001 \qquad 0.086 + 0.002 \qquad 0.0121 + 0.005 \qquad 0.0175 + 0.001 \qquad 0.086 + 0.002 \qquad 0.026 + 0.001 \qquad 0.086 + 0.002 \qquad 0.0121 + 0.005 \qquad 0.0175 + 0.001 \qquad 0.086 + 0.002 \qquad 0.0121 + 0.005 \qquad 0.0175 + 0.001 \qquad 0.001 + 0.001 + 0.001 \qquad 0.001 + 0.001 = 0.001 + 0.001 \qquad 0.001 + 0.001 = 0.001 + 0.001 = 0.001 + 0.001 = 0.001 + 0.001 = 0.001 + 0.001 = 0.001 + 0.001 = 0.001 + 0.001 = 0.001 + 0.001 = 0.00$	0.008							
$313.15 \qquad 0 \qquad 0.019 + 0.001 \qquad 0.030 + 0.008 \qquad 0.055 + 0.001 \qquad 0.103 + 0.004 \qquad 0.140 + 0.007 \qquad 0.198 + 0.001 \qquad 0.005 + 0.001 \qquad 0.103 + 0.004 \qquad 0.140 + 0.007 \qquad 0.198 + 0.001 \qquad 0.005 + 0.001 \qquad 0.005 + 0.001 \qquad 0.004 \qquad 0.005 = 0.004 \qquad 0.004 \qquad 0.005 = 0.004 \qquad 0.004 \qquad 0.005 =$	0.011							
323.15 0 0.022 + 0.002 0.037 + 0.003 0.064 + 0.003 0.117 + 0.006 0.157 + 0.010 0.216 + 0.003 0.064 + 0.003 0.017 + 0.006 0.007 + 0.010 0.216 + 0.007 0.007 0.007 + 0.007 0.007 0.007 + 0.007 0.007 0.007 + 0.007 0.007	0.014							
333.15 0 0.025 + 0.004 0.042 + 0.006 0.072 + 0.006 0.128 + 0.009 0.171 + 0.013 0.230 + 0.001 0.023 0.002 0.00	0.017							
343.15 0 0.027 ± 0.076 0.045 ± 0.009 0.077 ± 0.009 0.137 ± 0.012 0.180 ± 0.016 0.243 ± 0.016 0.025 ± 0.009 0.0	0.019							
$353.15 0 0.028 \pm 0.010 0.047 \pm 0.012 0.081 \pm 0.012 0.144 \pm 0.015 0.188 \pm 0.018 0.256 \pm 0.018 0.026 \pm 0.018 0.001 0.000 0.001 0.001 0.001 0.001 0.001 0.001 0.0$	0.022							
$363.15 \qquad 0 \qquad 0.029 \pm 0.013 \qquad 0.048 \pm 0.015 \qquad 0.085 \pm 0.015 \qquad 0.150 \pm 0.018 \qquad 0.197 \pm 0.020 \qquad 0.271 \pm 0.018 \qquad 0.018 \qquad 0.0197 \pm 0.020 \qquad 0.271 \pm 0.018 \qquad 0.0197 = 0$	0.025							
$373.15 \qquad 0 \qquad 0.029 \pm 0.015 \qquad 0.049 \pm 0.017 \qquad 0.088 \pm 0.017 \qquad 0.156 \pm 0.020 \qquad 0.206 \pm 0.023 \qquad 0.286 \pm 0.017 \qquad 0.010 \pm 0.020 \qquad 0.000 \pm 0.000 = 0.000 \pm 0.0000 \pm 0.00000 \pm 0.00000000$	0.028							
	\underline{V}^{ex} (cm ³ /mol)				$\Delta \rho (g$	$/cm^3$)		
--------	---	---------	-------	-------	------------------	-----------	-------	-------
T (K)	A_0	A_{l}	A_2	A_3	A_0	A_{l}	A_2	A_3
283.15	1.52	-0.05	-3.19	-3.53	0.018	0.018	0.005	0.018
293.15	1.82	0.38	-2.69	-3.35	0.021	0.024	0.001	0.016
298.15	1.94	0.54	-2.66	-3.39	0.022	0.027	0.003	0.015
303.15	2.06	0.66	-2.62	-3.40	0.024	0.029	0.005	0.015
313.15	2.28	0.80	-2.70	-3.46	0.026	0.032	0.006	0.014
323.15	2.54	1.22	-2.37	-3.48	0.028	0.039	0.012	0.012
333.15	2.80	1.72	-1.65	-3.05	0.031	0.043	0.019	0.009
343.15	2.96	1.62	-2.33	-3.64	0.032	0.044	0.017	0.012
353.15	3.15	1.91	-1.84	-3.30	0.034	0.047	0.021	0.009
363.15	3.28	1.74	-2.51	-3.82	0.035	0.047	0.017	0.012

Table A.2: Fitting parameters of the Redlich-Kister equation for excess molar volume \underline{V}^{ex} and density deviation $\Delta \rho$ of dependence on water molar fraction x_w for [C₄mim][BF₄] aqueous solutions at varying temperatures.

Table A.3: Fitting parameters of the exponential function Equation 3-4 of the viscosity dependence on water molar fraction for $[C_4mim][BF_4]$ aqueous solutions at different temperatures.

T (K)	a
293.15	0.24
298.15	0.26
303.15	0.27
313.15	0.31
323.15	0.35
333.15	0.39
343.15	0.42
353.15	0.46
363.15	0.49
373.15	0.54

Density deviations from ideal mixing with respect to water content and

temperature



Figure A.1: Density deviations $\Delta \rho$ of [C₄mim][BF₄] aqueous solutions for varying water molar fraction x_w (A) and temperature (B). The symbols represent experimental value. The solid lines represent Redlich-Kister equation fitting.

The calculated density deviation $\Delta \rho$ of the binary ionic liquid aqueous mixtures from ideal mixing are plotted in Figure A.1. The values of $\Delta \rho$ of Figure A.1A at several temperatures are fitted to the Redlich-Kister polynomial equation:

$$\Delta \rho = x_w x_{IL} \sum_{i=0}^{m} A_i (x_w - x_{IL})^i$$
 A-1

Where A_i are the fitting coefficients, *m* is the order of the Redlich-Kister polynomial equation, x_w and x_{IL} are the molar fraction of water and ionic liquid in the aqueous mixtures. The density deviations are negative for all the temperature and salt-rich compositions, and these deviations increase with water content and temperature.

Arrhenius equation fitting for viscosity dependence on temperature over various water contents for [C₄mim][BF₄] aqueous solutions

The mixture viscosities decrease with temperature and are fitted with Arrhenius equation for different water contents as shown in Figure A.2:

$$\eta = A \exp(E_a / RT)$$
 A-2

where E_a is the activation energy. Compared with Vogel-Fulcher-Tammann (VFT) equation fitting, Arrhenius fitting shown in Figure A.2 is not able to quantitatively fit the entire range and deviations are noted especially at low water concentrations. The fitted activation energy values are listed in Table A.4.



Figure A.2: The viscosities of [C₄mim][BF₄] aqueous solutions at different temperatures with Arrhenius equation fitting.

Water content	A	E_a
wt %	mPa·s	$kJ \cdot mol^{-1}$
0	1.7E-4	32.9
0.28	1.3E-4	33.4
0.52	2.7E-4	31.2
1.01	2.9E-4	30.5
2.03	6.6E-4	27.6
3.01	1.7E-4	24.6
5.00	3.5E-4	21.7

Table A.4:Fitting parameter (activation energy E_a) of the Arrhenius equation of
 $[C_4mim][BF_4]$ aqueous solutions with various water contents.

Derivation of semi-empirical linear correlation between excess molar volume and excess log viscosity from free volume theory

From free Volume theory,¹ the viscosity of fluid is related with molar free volume:

$$\log \eta = \log A + B \frac{\underline{V} - \underline{V}_f}{\underline{V}_f}$$
 A-3

As defined in the text, the excess mixture viscosity is defined as the deviation from the Arrhenius ideal solution mixing rule as:

$$\log \eta = x_{IL} \log(\eta_{IL}) + x_w \log(\eta_w) + (\log \eta)^{ex}$$
 A-4

Substituting Equation A-3 into Equation A-4

$$\log \eta = x_{IL} \log A_{IL} + x_w \log A_w + x_{IL} B_{IL} \frac{\underline{V}_{IL} - \underline{V}_{f,IL}}{\underline{V}_{f,IL}} + x_w B_w \frac{\underline{V}_w - \underline{V}_{f,w}}{\underline{V}_{f,w}} + \Delta(\log A) + \Delta(B \frac{\underline{V}_{ex}^{ex} - \underline{V}_{f}}{\underline{V}_{f}})$$
$$(\log \eta)^{ex} = \Delta(\log A) + \Delta(B \frac{\underline{V}_{ex}^{ex} - \underline{V}_{f}}{\underline{V}_{f}}) = (\Delta(\log A) - \Delta B) + \frac{B}{\underline{V}_{f}} \underline{V}_{ex}^{ex}$$

As it is global correlation, the correlation would valid for the simplest case – ideal mixing. For ideal mixing, $(\log \eta)^{ex} = 0$, $\underline{V}^{ex} = 0$, thus I have $\Delta(\log A) - \Delta B = 0$.

Then, assuming A and B are independent of composition and temperature, the linear correlation between excess viscosity and excess molar volume is obtained:

$$(\log \eta)^{ex} = \frac{B}{\underline{V}_f} \underline{V}^{ex}$$
A-5

Examples of ionic liquids aqueous solution systems invalid for semi-empirical linear correlation of Equation 3-9



Figure A.3: Plots of excess log viscosity, $(log\eta)^{ex}$, and excess molar volume, \underline{V}^{ex} , of $[C_2mim][BF_4]$ of salt-rich regime² (water molar fraction up to 40%) (A), $[BuPy][BF_4]^3$ (B), $[C_2mim][OTf]^4$ (C), $[C_2mim][EtSO_4]^4$ (D), $[C_2mim][TFA]^4$ (E), EAN⁵ (F) aqueous solutions over the entire range of water compositions.

REFERENCES

1. Doolittle, A. K.; Doolittle, D. B. Studies in Newtonian Flow. V. Further Verification of the Free - Space Viscosity Equation. *J. Appl. Phys.* **1957**, *28* (8), 901-905.

2. Zhang, S. J.; Li, M.; Chen, H. P.; Wang, J. F.; Zhang, J. M.; Zhang, M. L. Determination of physical properties for the binary system of 1-ethyl-3-methylimidazolium tetrafluoroborate + H2O. *J. Chem. Eng. Data* **2004**, *49* (4), 760-764.

3. Mokhtarani, B.; Sharifi, A.; Mortaheb, H. R.; Mirzaei, M.; Mafi, M.; Sadeghian, F. Density and viscosity of pyridinium-based ionic liquids and their binary mixtures with water at several temperatures. *J. Chem. Thermodyn.* **2009**, *41* (3), 323-329.

4. Rodriguez, H.; Brennecke, J. F. Temperature and composition dependence of the density and viscosity of binary mixtures of water plus ionic liquid. *J. Chem. Eng. Data* **2006**, *51* (6), 2145-2155.

5. Zarrougui, R.; Dhahbi, M.; Lemordant, D. Transport and Thermodynamic Properties of Ethylammonium Nitrate–Water Binary Mixtures: Effect of Temperature and Composition. *J. Solution Chem.* **2015**, *44* (3-4), 686-702.

Appendix B

SHEAR RHEOLOGY OF FUMED AMORPHOUS SILICA PARTICLES IN IONIC LIQUID

In this appendix, the shear rheological behavior of fumed amorphous silica particles in ionic liquid $[C_4mim][BF_4]$ is investigated. The effects of sample preparation (i.e., mixing) protocol, silica particle size, and surface chemical on the particle stability and hence the rheological response are studied. BeadBeater mixing method is found to be the most efficient and reproducible mixing method. Fumed amorphous hydrophilic particles show similar critical shear stress dependence on the particle size as for the hard sphere silica particles. Hydrophilic fumed silica particles are stable in ionic liquid $[C_4mim][BF_4]$ due to the solvation layers and show shear thickening behavior, whereas hydrophobic fumed silica particles are unstable and exhibit shear thinning behavior.

B.1 Materials and methods

B.1.1 Materials

The silica nanoparticles (hydrophilic particles: Aerosil 90, 200, 380 and hydrophobic particles: Aerosil R104, R805) are obtained from the Degussa Corporation. The Aerosil particles consist of primary particles with a characteristic length of approximately 12 nm which form an agglomerated secondary structure. The interparticle physical bonds of the flocculated silica network can be disrupted by shear force. All the particles are dried for 24 h in an oven at 120 °C before use. The distributions of the particle sizes have been characterized by dynamic light scattering (DLS) (Brookhaven particle size analyzer). Aerosil 90, 200, and 380 have the same hydrophilic surface chemistry - silanol groups on the surface.¹ Aerosil R104 and R805 are hydrophobic derivatives of silica particle (Aerosil 200), with the surface modified by octamethylcyclotetrasiloxane (D4) and octylsilane, respectively. As the direct derivatives of Aerosil 200, these silica particles (Aerosil 200, R104, and R805) are expected to have comparable particle sizes. Characteristics of particle, such as surface chemistry, particle size, and hydrophobicity are listed in Table B.1. Note that the hydrophobicity of the particle is determined by methanol wettability, such that higher methanol wettability indicates higher degree of hydrophobicity.

1-Butyl-3-methylimidazolium tetrafluoroborate ([C_4 mim][BF₄], 99%, Mw=226.02 g/mol) is purchased from Iolitec Ionic Liquid Technologies. This ionic liquid is dehydrated under vacuum at 70 °C for 24 h prior to use. The water content of [C_4 mim][BF₄] determined by the Karl Fisher titration is 373 ppm.

Particles	Aerosil 90	Aerosil 200	Aerosil 380	Aerosil R104	Aerosil R805
$A_{s}^{*}(m^{2}/g)$	90 ± 15	200 ± 25	380 ± 30	150 ± 25	150 ± 25
${\rm D_{h}}^{**}$ (nm)	271.7 ± 2.0	230.0 ± 0.9	201.9 ± 1.0		
Surface	-OH	-OH	-OH	octamethylcycl	octylsilane
Methanol				$\sim 33\%^{-1}$	~47% 1
wettability					

Table B.1: Characteristics of fumed amorphous particles.

* A_S is the specific surface area, obtained from ref¹

** D_h is the hydrodynamic diameter determined by DLS

B.1.2 Preparation of colloidal dispersions in [C₄mim][BF₄]

Colloidal suspensions are formulated by dispersing different silica particles in [C₄mim][BF₄]. Three standard mixing devices are employed to homogenize the silica colloidal dispersions and to investigate the impact of mixing methods on the rheological behavior: a Vortex mixer (Fisher Scientific, 150watts), a Silverson lab mixer (L4RT, 5/8" Micro Tubular frames, Silverson), and a BeadBeater (MiniBeadBeater-1, Biospec). The mixing procedures for the Vortex mixer, the Silverson lab mixer, and BeadBeater are: 3000 rpm for 5 min, 5000 rpm for 5 min, and 5000 rpm for 5 min, respectively. Homogenous dispersions are obtained by mechanical mixing followed by 10 min of degassing to remove any air bubbles in the samples. The samples are dehydrated again at 70 °C for 24 h under vacuum prior to rheological measurements.

B.1.3 Rheological measurements

For each colloidal silica dispersion, rheological measurements are performed on a stress-controlled rheometer (AR-G2, TA Instruments) with a Peltier temperature controller at 25 °C and a standard cone-plate geometry (1° cone angle, 40 mm diameter).

Previous investigations on the rheology of concentrated dispersions have shown that shear history can have a profound impact on the repeatability of results.²⁻³ To erase shear history on the samples and to establish their equilibrium structures, a steady pre-shear is applied at a shear stress of 10 Pa for 1 min followed by a 1 min equilibration period before each dynamic rheological measurement. The degree of dispersion homogeneity is assessed by the yield stress of suspensions, such that a lower yield stress indicates a better homogenous mixing. To determine the yield stress, a strain sweep oscillation test is performed at a constant frequency of 10 rad/s

272

with a strain range of 0.01% - 1000% at 25 °C. The shear thickening curve is obtained by using the following protocol: (1) a 1 min steady shear at shear stress 10 Pa followed by a 1 min equilibration; (2) a 10 min stress ramp back and forth from 0.001 Pa to 1500 Pa to remove shear history effects; (3) a 1 min steady shear at a shear stress of 10 Pa followed by a 1 min equilibration; (4) a steady state shear stress ramp back and forth sweep from 0.001 to 1500 Pa. The critical shear stress for shear thickening is defined as the shear stress value where increase in viscosity is first evident in the stress sweep curve. Finally, the slope of the viscosity versus shear stress in the shear thickening regime is used as a metric to determine the strength of the shear thickening transition.

B.2 Results and discussion

B.2.1 Effect of mixing methods

Ueno *et al.* reported that suspensions comprised of concentrated silica (Aerosil 200) in ionic liquid $[C_4mim][BF_4]$ showed shear thickening behavior.² Similar suspensions comprised of Aerosil 200 silica particles in $[C_4mim][BF_4]$ are investigated and their shear rheological behaviors are compared with reported results in Figure B.1. For 5 wt% of Aerosil 200 in $[C_4mim][BF_4]$, the suspension exhibits a weak shear thickening behavior, while a much stronger shear thickening transition is observed for 10 wt% of Aerosil 200 in $[C_4mim][BF_4]$. This increase in shear thickening behavior with particle concentration is expected and is consistent with previous studies.⁴ When subjected to low and moderate shear rates, suspensions have a relatively low viscosity and behave as a liquid. However, at higher shear rates, rheological response of

273

suspension material changes to a solid-like behavior and the viscosity increases dramatically.⁵



Figure B.1: Viscosity shear flow curves of silica particle Aerosil 200 in $[C_4mim][BF_4]$ with various particle contents and comparison with literature results by Ueno *et al.*²

As shown in Figure B.1, shear flow curves of Aerosil 200 suspensions are qualitatively comparable to the data reported by Ueno *et al.*; however, the suspension with 10 wt% silica content shows small quantitative differences with regards to the location and magnitude of the shear thickening peak. These deviations could be attributed to the different mixing protocols. Vortex mixers used for suspension preparations are quite different and less reproducible than that used by Ueno *et al.* (conditioning mixer AR-250). Therefore, investigations on the mixing methods are essential to obtain more reproducible and homogeneous suspensions. 10 wt%

suspensions Aerosil 200 in $[C_4mim][BF_4]$ are prepared and mixed by Vortex mixer, the Silverson mixer, and BeadBeater, respectively, and their steady shear flow curves are compared in Figure B.2. It is clear that suspension mixed by Beadbeater shows systematically lower value viscosity, indicating the most efficient and powerful mixing method. The flow curve of suspension mixed by Silverson mixer lays in between the ones mixed by Vorted mixer and BeadBeater. The characteristic parameters, such as critical stress, shear thickening strength from steady state shear flow curve, and yield stress from small amplitude oscillation shear curve for each samples are listed in Table B.2. As shown in Table B.2, suspension mixed by BeadBeater have the smallest yield stress than other mixing methods, which indicates a better homogeneity of suspension, and agrees with steady state flow curves discussed before. Thus, BeadBeater mixing method is used to prepare all the following Aerosil silica particle suspensions.



Figure B.2: Steady state shear flow curves of 10 wt% Aerosil 200 in [C₄mim][BF₄] suspensions mixed by various methods.

Table B.2:Comparison of characteristic parameters for 10 wt% Aerosil 200 -
[C4mim][BF4] suspensions mixed with various methods.

	Vortex mixer	Silverson mixer	BeadBeater
Critical stress (Pa)	2.7 ± 0.7	4.0 ± 0.8	3.9 ± 0.8
Shear thickening strength	0.52 ± 0.01	0.37 ± 0.01	0.30 ± 0.01
Yield stress (Pa)	2.0 ± 0.4	1.9 ± 0.4	1.6 ± 0.2

B.2.2 Effect of particle size on shear thickening behavior

The impact of particle size on the rheological response of suspensions is studied using a series of hydrophilic silica particles (Aerosil 90, 200, and 380) of the same surface chemical (silanol group on the surface), but varying the surface area (particle size). All the suspensions are prepared and mixed using the same mixing method (BeadBeater) for consistency. The steady state shear flow curves of 10 wt% silica particle suspensions prepared using Aerosil 90, 200, and 380 are shown in Figure B.3. Their characteristic parameters for each flow curve are given in Table B.3.

Suspension of larger particles size (Aerosil 90) exhibits a smaller critical stress and increased shear thickening strength, indicating an easier and more extreme shear thickening response. This is consistent with previous results of shear thickening fluids with solid sphere particles, larger particles tending to form hydroclusters at lower shear stress.⁶ However, Aerosil particle system shows a close but slightly lower critical stress when compared with solid hard sphere particle system of the equal apparent particle size under the same particle contents.³ This deviation is speculated to result from the intrinsic structure of the Aerosil particle system. The structure of the flocculated silica network can be disrupted by shear flow, resulting in new and unstable surfaces tending to form aggregations and gelation, which leads to a lower critical stress than the hard sphere system.

Table B.3:Comparison of characteristic parameters of 10 wt% hydrophilic silica
particle (Aerosil 90, 200, and 380) -[C4mim][BF4] suspensions

	Aerosil 90	Aerosil 200	Aerosil 380
Critical stress (Pa)	2.0 ± 0.3	3.9 ± 0.8	7.9 ± 1.7
Hard sphere critical stress (Pa)	~3	~5	~12
Shear thickening strength	0.31 ± 0.01	0.30 ± 0.01	0.23 ± 0.01



Figure B.3: Steady state shear flow curves of 10 wt% Aerosil hydrophilic silica particle suspensions of various particle sizes.

B.2.3 Effect of particle surface chemical on rheological behavior

Aerosil R104 and R805 are hydrophobic derivatives of Aerosil 200, with surface modified by octamethylcyclotetrasiloxane (D4) and octylsilane, respectively. Thus, fumed silica particles Aerosil 200, R104, and R805 have comparable particle sizes. The effect of particle surface chemical on suspension rheological properties can be investigated by using these particles with the same particle size. All the suspensions are mixed and prepared using the same mixer (BeadBeater) and protocol. Their rheological response of these silica particle suspensions of 10 wt% particle contents are shown in Figure B.4 and the yield stresses measured from small amplitude oscillation shear rheology are listed in Table B.4.

It is found that suspensions that are composed of more hydrophobic silica particles tend to form gelation under rest. A high yield stress value is found to be associated with more hydrophobic silica particle suspension, indicating particles aggregating in ionic liquid, rather than dispersing. These observations demonstrate a fundamentally different inter-particle interactions of hydrophilic and hydrophobic silica particle in ionic liquid [C₄mim][BF₄]. For the hydrophilic particles, the surface silanol groups form hydrogen bonds with the anionic group of [BF₄]⁻, resulting in an adsorbed anion layer on the silica particle surface. This induces a solvation layering effect which has been observed in previous studies.^{2, 7} The formation of a solvation multilayer on the silica particle surface gives rise to a repulsive solvation force and stabilizes the silica particles as shown in Figure B.5; whereas for the hydrophobic silica particles, no solvation layers are formed, such that particles are not stable and tend to aggregate in the ionic liquids.



Figure B.4: Steady state flow curves of silica particle suspensions with various surface chemical in [C₄mim][BF₄] with 10 wt% particle content.

	Aerosil 200	Aerosil R104	Aerosil R805	
Yield stress(Pa)	1.6 ± 0.2	2.3 ± 0.2	3.4 ± 0.5	

Table B.4:Yield stress of 10 wt% suspensions of hydrophilic (Aerosil 200) and
hydrophobic silica particle (Aerosil R104, R805) in [C4mim][BF4]



Figure B.5: Schematic illustration of hydrophilic Aerosil particles stabilization in [C₄mim][BF₄] by solvation layer.

B.3 Conclusions

In this work, shear thickening fluids (STFs) suspensions in an ionic liquid are successfully formulated using a variety of silica based colloidal particles. The preparation protocols for the suspensions have been systematically studied and have confirmed that the BeadBeater mixer probably produces the desired STF suspensions with respect to the efficiency and reproducibility. Several factors (*e.g.*, particle size, surface chemical, hydrophobicity) have been investigated on fumed amorphous silica particle systems. It is found that Aerosil hydrophilic particles show similar critical

shear stress dependence on particle size as for the hard sphere silica particles. Hydrophilic Aerosil silica particles are able to be well dispersed and stabilized in $[C_4mim][BF_4]$ exhibiting shear thickening behavior due to the solvation layering structure induced by hydrogen bonding. Whereas, hydrophobic Aerosil silica particles failed to form homogenous suspensions due to the absence of solvation layers and show shear thinning behavior because of the breaking of agglomerate secondary structure under shear.

REFERENCES

1. Industries, E. http://www.aerosil.com/product/aerosil/en/Pages/default.aspx.

2. Ueno, K.; Watanabe, M. From Colloidal Stability in Ionic Liquids to Advanced Soft Materials Using Unique Media. *Langmuir* **2011**, *27* (15), 9105-9115.

3. Maranzano, B. J.; Wagner, N. J. The effects of interparticle interactions and particle size on reversible shear thickening: Hard-sphere colloidal dispersions. *Journal of Rheology* **2001**, *45* (5), 1205-1222.

4. Odijk, T. STIFF CHAINS AND FILAMENTS UNDER TENSION. *Macromolecules* **1995**, *28* (20), 7016-7018.

5. Wagner, N. J.; Brady, J. F. Shear thickening in colloidal dispersions. *Physics Today* **2009**, *62* (10), 27-32.

6. Maranzano, B. J.; Wagner, N. J. The effects of particle-size on reversible shear thickening of concentrated colloidal dispersions. *Journal of Chemical Physics* **2001**, *114* (23), 10514-10527.

7. Ueno, K.; Inaba, A.; Kondoh, M.; Watanabe, M. Colloidal stability of bare and polymer-grafted silica nanoparticles in ionic liquids. *Langmuir* **2008**, *24* (10), 5253-5259.

Appendix C

TEXT AND FIGURE REPRINT PERMISSIONS

Appendix C includes the Copyright Clearance requests from RightsLink for: Figure 1.1,¹ Figure 1.3,² Figure 1.5A,³ Figure 1.5B,⁴ Figures 1.6-1.7,⁵ Figures 3.2-3.11,⁶ Figures 3.14-3.18,⁷ and Figures 4.4-4.10, 5.1-5.7.³

ROYAL SOCIETY OF CHEMISTRY LICENSE TERMS AND CONDITIONS

Dec 21, 2016

This Agreement between Jingsi Gao ("You") and Royal Society of Chemistry ("Royal Society of Chemistry") consists of your license details and the terms and conditions provided by Royal Society of Chemistry and Copyright Clearance Center.

License Number	4013730045274
License date	Dec 21, 2016
Licensed Content Publisher	Royal Society of Chemistry
Licensed Content Publication	Chemical Society Reviews
Licensed Content Title	Applications of ionic liquids in the chemical industry
Licensed Content Author	Natalia V. Plechkova,Kenneth R. Seddon
Licensed Content Date	Nov 30, 2007
Licensed Content Volume Number	37
Licensed Content Issue Number	1
Type of Use	Thesis/Dissertation
Requestor type	academic/educational
Portion	figures/tables/images
Number of figures/tables/images	1
Format	print and electronic
Distribution quantity	10
Will you be translating?	no
Order reference number	
Title of the thesis/dissertation	RHEOLOGY AND MICROSTRUCTURE OF STABLE CONCENTRATED IONIC LIQUID COLLOIDAL SUSPENSIONS
Expected completion date	Jan 2017
Estimated size	290
Requestor Location	Jingsi Gao 150 Academy Street
	NEWARK, DE 19716 United States Attn: Jingsi Gao
Billing Type	Invoice
Billing Address	Jingsi Gao 150 Academy Street
	NEWARK, DE 19716 United States Attn: Jingsi Gao

0.00 USD

Terms and Conditions

This License Agreement is between {Requestor Name} ("You") and The Royal Society of Chemistry ("RSC") provided by the Copyright Clearance Center ("CCC"). The license consists of your order details, the terms and conditions provided by the Royal Society of Chemistry, and the payment terms and conditions.

RSC / TERMS AND CONDITIONS

INTRODUCTION

The publisher for this copyrighted material is The Royal Society of Chemistry. By clicking "accept" in connection with completing this licensing transaction, you agree that the following terms and conditions apply to this transaction (along with the Billing and Payment terms and conditions established by CCC, at the time that you opened your RightsLink account and that are available at any time at .

LICENSE GRANTED

The RSC hereby grants you a non-exclusive license to use the aforementioned material anywhere in the world subject to the terms and conditions indicated herein. Reproduction of the material is confined to the purpose and/or media for which permission is hereby given. RESERVATION OF RIGHTS

The RSC reserves all rights not specifically granted in the combination of (i) the license details provided by your and accepted in the course of this licensing transaction; (ii) these terms and conditions; and (iii) CCC's Billing and Payment terms and conditions. REVOCATION

The RSC reserves the right to revoke this license for any reason, including, but not limited to, advertising and promotional uses of RSC content, third party usage, and incorrect source figure attribution.

THIRD-PARTY MATERIAL DISCLAIMER

If part of the material to be used (for example, a figure) has appeared in the RSC publication with credit to another source, permission must also be sought from that source. If the other source is another RSC publication these details should be included in your RightsLink request. If the other source is a third party, permission must be obtained from the third party. The RSC disclaims any responsibility for the reproduction you make of items owned by a third party.

PAYMENT OF FEE

If the permission fee for the requested material is waived in this instance, please be advised that any future requests for the reproduction of RSC materials may attract a fee. ACKNOWLEDGEMENT

The reproduction of the licensed material must be accompanied by the following acknowledgement:

Reproduced ("Adapted" or "in part") from {Reference Citation} (or Ref XX) with permission of The Royal Society of Chemistry.

If the licensed material is being reproduced from New Journal of Chemistry (NJC), Photochemical & Photobiological Sciences (PPS) or Physical Chemistry Chemical Physics

(PCCP) you must include one of the following acknowledgements:

For figures originally published in NJC:

Reproduced ("Adapted" or "in part") from {Reference Citation} (or Ref XX) with permission of The Royal Society of Chemistry (RSC) on behalf of the European Society for Photobiology, the European Photochemistry Association and the RSC.

For figures originally published in PPS:

Reproduced ("Adapted" or "in part") from {Reference Citation} (or Ref XX) with permission of The Royal Society of Chemistry (RSC) on behalf of the Centre National de la Recherche Scientifique (CNRS) and the RSC.

For figures originally published in PCCP:

Reproduced ("Adapted" or "in part") from {Reference Citation} (or Ref XX) with permission of the PCCP Owner Societies.

Total

HYPERTEXT LINKS

With any material which is being reproduced in electronic form, you must include a hypertext link to the original RSC article on the RSC's website. The recommended form for the hyperlink is http://dx.doi.org/10.1039/DOI suffix, for example in the link

http://dx.doi.org/10.1039/b110420a the DOI suffix is 'b110420a'. To find the relevant DOI suffix for the RSC article in question, go to the Journals section of the website and locate the article in the list of papers for the volume and issue of your specific journal. You will find the DOI suffix quoted there.

LICENSE CONTINGENT ON PAYMENT

While you may exercise the rights licensed immediately upon issuance of the license at the end of the licensing process for the transaction, provided that you have disclosed complete and accurate details of your proposed use, no license is finally effective unless and until full payment is received from you (by CCC) as provided in CCC's Billing and Payment terms and conditions. If full payment is not received on a timely basis, then any license preliminarily granted shall be deemed automatically revoked and shall be void as if never granted. Further, in the event that you breach any of these terms and conditions or any of CCC's Billing and Payment terms and conditions, the license is automatically revoked and shall be void as if never granted. Use of materials as described in a revoked license, as well as any use of the materials beyond the scope of an unrevoked license, may constitute copyright infringement and the RSC reserves the right to take any and all action to protect its copyright in the materials.

WARRANTIES

The RSC makes no representations or warranties with respect to the licensed material. INDEMNITY

You hereby indemnify and agree to hold harmless the RSC and the CCC, and their respective officers, directors, trustees, employees and agents, from and against any and all claims arising out of your use of the licensed material other than as specifically authorized pursuant to this licence.

NO TRANSFER OF LICENSE

This license is personal to you or your publisher and may not be sublicensed, assigned, or transferred by you to any other person without the RSC's written permission.

NO AMENDMENT EXCEPT IN WRITING

This license may not be amended except in a writing signed by both parties (or, in the case of "Other Conditions, v1.2", by CCC on the RSC's behalf).

OBJECTION TO CONTRARY TERMS

You hereby acknowledge and agree that these terms and conditions, together with CCC's Billing and Payment terms and conditions (which are incorporated herein), comprise the entire agreement between you and the RSC (and CCC) concerning this licensing transaction, to the exclusion of all other terms and conditions, written or verbal, express or implied (including any terms contained in any purchase order, acknowledgment, check endorsement or other writing prepared by you). In the event of any conflict between your obligations established by these terms and conditions and those established by CCC's Billing and Payment terms and conditions, these terms and conditions shall control. JURISDICTION

This license transaction shall be governed by and construed in accordance with the laws of the District of Columbia. You hereby agree to submit to the jurisdiction of the courts located in the District of Columbia for purposes of resolving any disputes that may arise in connection with this licensing transaction.

LIMITED LICENSE

The following terms and conditions apply to specific license types: Translation

This permission is granted for non-exclusive world English rights only unless your license was granted for translation rights. If you licensed translation rights you may only translate

this content into the languages you requested. A professional translator must perform all translations and reproduce the content word for word preserving the integrity of the article. Intranet

If the licensed material is being posted on an Intranet, the Intranet is to be passwordprotected and made available only to bona fide students or employees only. All content posted to the Intranet must maintain the copyright information line on the bottom of each image. You must also fully reference the material and include a hypertext link as specified above.

Copies of Whole Articles

All copies of whole articles must maintain, if available, the copyright information line on the bottom of each page.

Other Conditions

v1.2

Gratis licenses (referencing \$0 in the Total field) are free. Please retain this printable license for your reference. No payment is required.

If you would like to pay for this license now, please remit this license along with yourpayment made payable to "COPYRIGHT CLEARANCE CENTER" otherwise you will be invoiced within 48 hours of the license date. Payment should be in the form of a check or money order referencing your account number and this invoice number {Invoice Number}. Once you receive your invoice for this order, you may pay your invoice by credit card. Please follow instructions provided at that time.

Please follow instructions provided at that the

Make Payment To: Copyright Clearance Center Dept 001 P.O. Box 843006 Boston, MA 02284-3006 For suggestions or comments

For suggestions or comments regarding this order, contact Rightslink Customer Support: customercare@copyright.com or +1-855-239-3415 (toll free in the US) or +1-978-646-2777.

Questions? <u>customercare@copyright.com</u> or +1-855-239-3415 (toll free in the US) or +1-978-646-2777.







ACS Publications Title: Aost Trusted. Most Cited. Most Read Author: B. Shiflett, et al.

Creating Nanoparticle Stability in Ionic Liquid [C4mim][BF4] by Inducing Solvation Layering Jingsi Gao, Rose S. Ndong, Mark Publication: ACS Nano Publisher: American Chemical Society Mar 1, 2015 Copyright © 2015, American Chemical Society



Date:

PERMISSION/LICENSE IS GRANTED FOR YOUR ORDER AT NO CHARGE

This type of permission/license, instead of the standard Terms & Conditions, is sent to you because no fee is being charged for your order. Please note the following:

- Permission is granted for your request in both print and electronic formats, and translations.
- If figures and/or tables were requested, they may be adapted or used in part.
- Please print this page for your records and send a copy of it to your publisher/graduate school.
- Appropriate credit for the requested material should be given as follows: "Reprinted." (adapted) with permission from (COMPLETE REFERENCE CITATION). Copyright (YEAR) American Chemical Society." Insert appropriate information in place of the capitalized words.
- One-time permission is granted only for the use specified in your request. No additional uses are granted (such as derivative works or other editions). For any other uses, please submit a new request.



Copyright @ 2016 Copyright Clearance Center, Inc. All Rights Reserved. Privacy statement, Terms and Conditions, Comments? We would like to hear from you. E-mail us at customercare@copyright.com

ROYAL SOCIETY OF CHEMISTRY LICENSE TERMS AND CONDITIONS

Dec 21, 2016

This Agreement between Jingsi Gao ("You") and Royal Society of Chemistry ("Royal Society of Chemistry") consists of your license details and the terms and conditions provided by Royal Society of Chemistry and Copyright Clearance Center.

License Number	4013730729080		
License date	Dec 21, 2016		
Licensed Content Publisher	Royal Society of Chemistry		
Licensed Content Publication	Physical Chemistry Chemical Physics		
Licensed Content Title	Resonance shear measurement of nanoconfined ionic liquids		
Licensed Content Author	Kazuhide Ueno,Motohiro Kasuya,Masayoshi Watanabe,Masashi Mizukami,Kazue Kurihara		
Licensed Content Date	Mar 11, 2010		
Licensed Content Volume Number	12		
Licensed Content Issue Number	16		
Type of Use	Thesis/Dissertation		
Requestor type	academic/educational		
Portion	figures/tables/images		
Number of figures/tables/images	1		
Format	print and electronic		
Distribution quantity	10		
Will you be translating?	no		
Order reference number			
Title of the thesis/dissertation	RHEOLOGY AND MICROSTRUCTURE OF STABLE CONCENTRATED IONIC LIQUID COLLOIDAL SUSPENSIONS		
Expected completion date	Jan 2017		
Estimated size	290		
Requestor Location	Jingsi Gao 150 Academy Street		
	NEWARK, DE 19716 United States Attn: Jingsi Gao		
Billing Type	Invoice		
Billing Address	Jingsi Gao 150 Academy Street		
	NEWARK, DE 19716		

United States Attn: Jingsi Gao 0.00 USD

Terms and Conditions

Total

This License Agreement is between {Requestor Name} ("You") and The Royal Society of Chemistry ("RSC") provided by the Copyright Clearance Center ("CCC"). The license consists of your order details, the terms and conditions provided by the Royal Society of Chemistry, and the payment terms and conditions.

RSC / TERMS AND CONDITIONS INTRODUCTION

The publisher for this copyrighted material is The Royal Society of Chemistry. By clicking "accept" in connection with completing this licensing transaction, you agree that the following terms and conditions apply to this transaction (along with the Billing and Payment terms and conditions established by CCC, at the time that you opened your RightsLink account and that are available at any time at .

LICENSE GRANTED

The RSC hereby grants you a non-exclusive license to use the aforementioned material anywhere in the world subject to the terms and conditions indicated herein. Reproduction of the material is confined to the purpose and/or media for which permission is hereby given. RESERVATION OF RIGHTS

The RSC reserves all rights not specifically granted in the combination of (i) the license details provided by your and accepted in the course of this licensing transaction; (ii) these terms and conditions; and (iii) CCC's Billing and Payment terms and conditions. REVOCATION

The RSC reserves the right to revoke this license for any reason, including, but not limited to, advertising and promotional uses of RSC content, third party usage, and incorrect source figure attribution.

THIRD-PARTY MATERIAL DISCLAIMER

If part of the material to be used (for example, a figure) has appeared in the RSC publication with credit to another source, permission must also be sought from that source. If the other source is another RSC publication these details should be included in your RightsLink request. If the other source is a third party, permission must be obtained from the third party. The RSC disclaims any responsibility for the reproduction you make of items owned by a third party.

PAYMENT OF FEE

If the permission fee for the requested material is waived in this instance, please be advised that any future requests for the reproduction of RSC materials may attract a fee. ACKNOWLEDGEMENT

The reproduction of the licensed material must be accompanied by the following acknowledgement:

Reproduced ("Adapted" or "in part") from {Reference Citation} (or Ref XX) with permission of The Royal Society of Chemistry.

If the licensed material is being reproduced from New Journal of Chemistry (NJC),

Photochemical & Photobiological Sciences (PPS) or Physical Chemistry Chemical Physics (PCCP) you must include one of the following acknowledgements:

For figures originally published in NJC:

Reproduced ("Adapted" or "in part") from {Reference Citation} (or Ref XX) with permission of The Royal Society of Chemistry (RSC) on behalf of the European Society for Photobiology, the European Photochemistry Association and the RSC.

For figures originally published in PPS:

Reproduced ("Adapted" or "in part") from {Reference Citation} (or Ref XX) with

permission of The Royal Society of Chemistry (RSC) on behalf of the Centre National de la Recherche Scientifique (CNRS) and the RSC.

For figures originally published in PCCP:

Reproduced ("Adapted" or "in part") from {Reference Citation} (or Ref XX) with permission of the PCCP Owner Societies.

HYPERTEXT LINKS

With any material which is being reproduced in electronic form, you must include a hypertext link to the original RSC article on the RSC's website. The recommended form for the hyperlink is http://dx.doi.org/10.1039/DOI suffix, for example in the link

http://dx.doi.org/10.1039/b110420a the DOI suffix is 'b110420a'. To find the relevant DOI suffix for the RSC article in question, go to the Journals section of the website and locate the article in the list of papers for the volume and issue of your specific journal. You will find the DOI suffix quoted there.

LICENSE CONTINGENT ON PAYMENT

While you may exercise the rights licensed immediately upon issuance of the license at the end of the licensing process for the transaction, provided that you have disclosed complete and accurate details of your proposed use, no license is finally effective unless and until full payment is received from you (by CCC) as provided in CCC's Billing and Payment terms and conditions. If full payment is not received on a timely basis, then any license preliminarily granted shall be deemed automatically revoked and shall be void as if never granted. Further, in the event that you breach any of these terms and conditions or any of CCC's Billing and Payment terms and conditions, the license is automatically revoked and shall be void as if never granted. Use of materials as described in a revoked license, as well as any use of the materials beyond the scope of an unrevoked license, may constitute copyright infringement and the RSC reserves the right to take any and all action to protect its copyright in the materials.

WARRANTIES

The RSC makes no representations or warranties with respect to the licensed material. INDEMNITY

You hereby indemnify and agree to hold harmless the RSC and the CCC, and their respective officers, directors, trustees, employees and agents, from and against any and all claims arising out of your use of the licensed material other than as specifically authorized pursuant to this licence.

NO TRANSFER OF LICENSE

This license is personal to you or your publisher and may not be sublicensed, assigned, or transferred by you to any other person without the RSC's written permission.

NO AMENDMENT EXCEPT IN WRITING

This license may not be amended except in a writing signed by both parties (or, in the case of "Other Conditions, v1.2", by CCC on the RSC's behalf).

OBJECTION TO CONTRARY TERMS

You hereby acknowledge and agree that these terms and conditions, together with CCC's Billing and Payment terms and conditions (which are incorporated herein), comprise the entire agreement between you and the RSC (and CCC) concerning this licensing transaction, to the exclusion of all other terms and conditions, written or verbal, express or implied (including any terms contained in any purchase order, acknowledgment, check endorsement or other writing prepared by you). In the event of any conflict between your obligations established by these terms and conditions and those established by CCC's Billing and Payment terms and conditions, these terms and conditions shall control. JURISDICTION

This license transaction shall be governed by and construed in accordance with the laws of the District of Columbia. You hereby agree to submit to the jurisdiction of the courts located in the District of Columbia for purposes of resolving any disputes that may arise in connection with this licensing transaction.

LIMITED LICENSE

The following terms and conditions apply to specific license types:

Translation

This permission is granted for non-exclusive world English rights only unless your license

was granted for translation rights. If you licensed translation rights you may only translate this content into the languages you requested. A professional translator must perform all translations and reproduce the content word for word preserving the integrity of the article. Intranet

If the licensed material is being posted on an Intranet, the Intranet is to be passwordprotected and made available only to bona fide students or employees only. All content posted to the Intranet must maintain the copyright information line on the bottom of each image. You must also fully reference the material and include a hypertext link as specified above.

Copies of Whole Articles

All copies of whole articles must maintain, if available, the copyright information line on the bottom of each page.

Other Conditions

v1.2

Gratis licenses (referencing \$0 in the Total field) are free. Please retain this printable license for your reference. No payment is required.

If you would like to pay for this license now, please remit this license along with yourpayment made payable to "COPYRIGHT CLEARANCE CENTER" otherwise you will be invoiced within 48 hours of the license date. Payment should be in the form of a check or money order referencing your account number and this invoice number {Invoice Number}. Once you receive your invoice for this order, you may pay your invoice by credit card. Please follow instructions provided at that time.

Make Payment To:

Copyright Clearance Center

Dept 001

P.O. Box 843006

Boston, MA 02284-3006

For suggestions or comments regarding this order, contact Rightslink Customer Support: customercare@copyright.com or +1-855-239-3415 (toll free in the US) or +1-978-646-2777.

Questions? <u>customercare@copyright.com</u> or +1-855-239-3415 (toll free in the US) or +1-978-646-2777.

ELSEVIER LICENSE TERMS AND CONDITIONS

Dec 21, 2016

This Agreement between Jingsi Gao ("You") and Elsevier ("Elsevier") consists of your license details and the terms and conditions provided by Elsevier and Copyright Clearance Center.

License Number	4013731093493		
License date	Dec 21, 2016		
Licensed Content Publisher	Elsevier		
Licensed Content Publication	Composites Science and Technology		
Licensed Content Title	Stab resistance of shear thickening fluid (STF)-treated fabrics		
Licensed Content Author	M.J. Decker,C.J. Halbach,C.H. Nam,N.J. Wagner,E.D. Wetzel		
Licensed Content Date	March 2007		
Licensed Content Volume Number	67		
Licensed Content Issue Number	3-4		
Licensed Content Pages	14		
Start Page	565		
End Page	578		
Type of Use	reuse in a thesis/dissertation		
Portion	figures/tables/illustrations		
Number of figures/tables/illustrations	2		
Format	both print and electronic		
Are you the author of this Elsevier article?	No		
Will you be translating?	No		
Order reference number			
Original figure numbers	Figure 2, 6		
Title of your thesis/dissertation	RHEOLOGY AND MICROSTRUCTURE OF STABLE CONCENTRATED IONIC LIQUID COLLOIDAL SUSPENSIONS		
Expected completion date	Jan 2017		
Estimated size (number of pages)	290		
Elsevier VAT number	GB 494 6272 12		
Requestor Location	Jingsi Gao 150 Academy Street		
	NEWARK, DE 19716 United States Attn: Jingsi Gao		
Total	0.00 USD		

Terms and Conditions

INTRODUCTION

1. The publisher for this copyrighted material is Elsevier. By clicking "accept" in connection with completing this licensing transaction, you agree that the following terms and conditions apply to this transaction (along with the Billing and Payment terms and conditions established by Copyright Clearance Center, Inc. ("CCC"), at the time that you opened your Rightslink account and that are available at any time at http://myaccount.copyright.com).

GENERAL TERMS

2. Elsevier hereby grants you permission to reproduce the aforementioned material subject to the terms and conditions indicated.

3. Acknowledgement: If any part of the material to be used (for example, figures) has appeared in our publication with credit or acknowledgement to another source, permission must also be sought from that source. If such permission is not obtained then that material may not be included in your publication/copies. Suitable acknowledgement to the source must be made, either as a footnote or in a reference list at the end of your publication, as follows:

"Reprinted from Publication title, Vol/edition number, Author(s), Title of article / title of chapter, Pages No., Copyright (Year), with permission from Elsevier [OR APPLICABLE SOCIETY COPYRIGHT OWNER]." Also Lancet special credit - "Reprinted from The Lancet, Vol. number, Author(s), Title of article, Pages No., Copyright (Year), with permission from Elsevier."

4. Reproduction of this material is confined to the purpose and/or media for which permission is hereby given.

5. Altering/Modifying Material: Not Permitted. However figures and illustrations may be altered/adapted minimally to serve your work. Any other abbreviations, additions, deletions and/or any other alterations shall be made only with prior written authorization of Elsevier Ltd. (Please contact Elsevier at permissions@elsevier.com). No modifications can be made to any Lancet figures/tables and they must be reproduced in full.

6. If the permission fee for the requested use of our material is waived in this instance, please be advised that your future requests for Elsevier materials may attract a fee.
7. Reservation of Rights: Publisher reserves all rights not specifically granted in the combination of (i) the license details provided by you and accepted in the course of this licensing transaction, (ii) these terms and conditions and (iii) CCC's Billing and Payment terms and conditions.

8. License Contingent Upon Payment: While you may exercise the rights licensed immediately upon issuance of the license at the end of the licensing process for the transaction, provided that you have disclosed complete and accurate details of your proposed use, no license is finally effective unless and until full payment is received from you (either by publisher or by CCC) as provided in CCC's Billing and Payment terms and conditions. If full payment is not received on a timely basis, then any license preliminarily granted shall be deemed automatically revoked and shall be void as if never granted. Further, in the event that you breach any of these terms and conditions or any of CCC's Billing and Payment terms and conditions, the license is automatically revoked and shall be void as if never granted. Use of materials as described in a revoked license, as well as any use of the materials beyond the scope of an unrevoked license, may constitute copyright infringement and publisher reserves the right to take any and all action to protect its copyright in the materials.

9. Warranties: Publisher makes no representations or warranties with respect to the licensed material.

10. Indemnity: You hereby indemnify and agree to hold harmless publisher and CCC, and their respective officers, directors, employees and agents, from and against any and all claims arising out of your use of the licensed material other than as specifically authorized pursuant to this license.

11. No Transfer of License: This license is personal to you and may not be sublicensed, assigned, or transferred by you to any other person without publisher's written permission.
12. No Amendment Except in Writing: This license may not be amended except in a writing signed by both parties (or, in the case of publisher, by CCC on publisher's behalf).
13. Objection to Contrary Terms: Publisher hereby objects to any terms contained in any purchase order, acknowledgment, check endorsement or other writing prepared by you, which terms are inconsistent with these terms and conditions or CCC's Billing and Payment terms and conditions. These terms and conditions, together with CCC's Billing and Payment terms and conditions (which are incorporated herein), comprise the entire agreement between you and publisher (and CCC) concerning this licensing transaction. In the event of any conflict between your obligations established by these terms and conditions and those established by CCC's Billing and Payment terms and conditions.

14. Revocation: Elsevier or Copyright Clearance Center may deny the permissions described in this License at their sole discretion, for any reason or no reason, with a full refund payable to you. Notice of such denial will be made using the contact information provided by you. Failure to receive such notice will not alter or invalidate the denial. In no event will Elsevier or Copyright Clearance Center be responsible or liable for any costs, expenses or damage incurred by you as a result of a denial of your permission request, other than a refund of the amount(s) paid by you to Elsevier and/or Copyright Clearance Center for denied permissions.

LIMITED LICENSE

The following terms and conditions apply only to specific license types: 15. **Translation**: This permission is granted for non-exclusive world **English** rights only unless your license was granted for translation rights. If you licensed translation rights you may only translate this content into the languages you requested. A professional translator must perform all translations and reproduce the content word for word preserving the integrity of the article.

16. **Posting licensed content on any Website**: The following terms and conditions apply as follows: Licensing material from an Elsevier journal: All content posted to the web site must maintain the copyright information line on the bottom of each image; A hyper-text must be included to the Homepage of the journal from which you are licensing at

http://www.sciencedirect.com/science/journal/xxxxx or the Elsevier homepage for books at http://www.elsevier.com; Central Storage: This license does not include permission for a scanned version of the material to be stored in a central repository such as that provided by Heron/XanEdu.

Licensing material from an Elsevier book: A hyper-text link must be included to the Elsevier homepage at <u>http://www.elsevier.com</u>. All content posted to the web site must maintain the copyright information line on the bottom of each image.

Posting licensed content on Electronic reserve: In addition to the above the following clauses are applicable: The web site must be password-protected and made available only to bona fide students registered on a relevant course. This permission is granted for 1 year only. You may obtain a new license for future website posting.

17. For journal authors: the following clauses are applicable in addition to the above: **Preprints:**

A preprint is an author's own write-up of research results and analysis, it has not been peerreviewed, nor has it had any other value added to it by a publisher (such as formatting, copyright, technical enhancement etc.).

Authors can share their preprints anywhere at any time. Preprints should not be added to or enhanced in any way in order to appear more like, or to substitute for, the final versions of articles however authors can update their preprints on arXiv or RePEc with their Accepted Author Manuscript (see below).

If accepted for publication, we encourage authors to link from the preprint to their formal publication via its DOI. Millions of researchers have access to the formal publications on ScienceDirect, and so links will help users to find, access, cite and use the best available version. Please note that Cell Press, The Lancet and some society-owned have different preprint policies. Information on these policies is available on the journal homepage. **Accepted Author Manuscripts:** An accepted author manuscript is the manuscript of an article that has been accepted for publication and which typically includes author-incorporated changes suggested during submission, peer review and editor-author communications.

Authors can share their accepted author manuscript:

- immediately
 - via their non-commercial person homepage or blog
 - by updating a preprint in arXiv or RePEc with the accepted manuscript
 - via their research institute or institutional repository for internal institutional uses or as part of an invitation-only research collaboration work-group
 - directly by providing copies to their students or to research collaborators for their personal use
 - for private scholarly sharing as part of an invitation-only work group on commercial sites with which Elsevier has an agreement
- after the embargo period
 - via non-commercial hosting platforms such as their institutional repository
 - via commercial sites with which Elsevier has an agreement

In all cases accepted manuscripts should:

- link to the formal publication via its DOI
- bear a CC-BY-NC-ND license this is easy to do
- if aggregated with other manuscripts, for example in a repository or other site, be shared in alignment with our hosting policy not be added to or enhanced in any way to appear more like, or to substitute for, the published journal article.

Published journal article (JPA): A published journal article (PJA) is the definitive final record of published research that appears or will appear in the journal and embodies all value-adding publishing activities including peer review co-ordination, copy-editing, formatting, (if relevant) pagination and online enrichment.

Policies for sharing publishing journal articles differ for subscription and gold open access articles:

<u>Subscription Articles</u>: If you are an author, please share a link to your article rather than the full-text. Millions of researchers have access to the formal publications on ScienceDirect, and so links will help your users to find, access, cite, and use the best available version. Theses and dissertations which contain embedded PJAs as part of the formal submission can be posted publicly by the awarding institution with DOI links back to the formal publications on ScienceDirect.

If you are affiliated with a library that subscribes to ScienceDirect you have additional private sharing rights for others' research accessed under that agreement. This includes use for classroom teaching and internal training at the institution (including use in course packs and courseware programs), and inclusion of the article for grant funding purposes. **Gold Open Access Articles:** May be shared according to the author-selected end-user

license and should contain a <u>CrossMark logo</u>, the end user license, and a DOI link to the formal publication on ScienceDirect.

Please refer to Elsevier's posting policy for further information.

18. For book authors the following clauses are applicable in addition to the above: Authors are permitted to place a brief summary of their work online only. You are not allowed to download and post the published electronic version of your chapter, nor may you scan the printed edition to create an electronic version. **Posting to a repository:** Authors are permitted to post a summary of their chapter only in their institution's repository. 19. **Thesis/Dissertation**: If your license is for use in a thesis/dissertation your thesis may be submitted to your institution in either print or electronic form. Should your thesis be published commercially, please reapply for permission. These requirements include permission for the Library and Archives of Canada to supply single copies, on demand, of the complete thesis. Should your thesis be published commercially, please reapply for permission for Proquest/UMI to supply single copies, on demand, of the complete thesis. Should your thesis be published commercially, please reapply for permission. These sand dissertations which contain embedded PJAs as part of the formal submission can be posted publicly by the awarding institution with DOI links back to the formal publications on ScienceDirect.

Elsevier Open Access Terms and Conditions

You can publish open access with Elsevier in hundreds of open access journals or in nearly 2000 established subscription journals that support open access publishing. Permitted third party re-use of these open access articles is defined by the author's choice of Creative Commons user license. See our open access license policy for more information. **Terms & Conditions applicable to all Open Access articles published with Elsevier:** Any reuse of the article must not represent the author as endorsing the adaptation of the article nor should the article be modified in such a way as to damage the author's honour or reputation. If any changes have been made, such changes must be clearly indicated. The author(s) must be appropriately credited and we ask that you include the end user license and a DOI link to the formal publication on ScienceDirect.

If any part of the material to be used (for example, figures) has appeared in our publication with credit or acknowledgement to another source it is the responsibility of the user to ensure their reuse complies with the terms and conditions determined by the rights holder. Additional Terms & Conditions applicable to each Creative Commons user license: CC BY: The CC-BY license allows users to copy, to create extracts, abstracts and new works from the Article, to alter and revise the Article and to make commercial use of the Article (including reuse and/or resale of the Article by commercial entities), provided the user gives appropriate credit (with a link to the formal publication through the relevant DOI), provides a link to the license, indicates if changes were made and the licenser is not represented as endorsing the use made of the work. The full details of the license are available at http://creativecommons.org/licenses/by/4.0.

CC BY NC SA: The CC BY-NC-SA license allows users to copy, to create extracts, abstracts and new works from the Article, to alter and revise the Article, provided this is not done for commercial purposes, and that the user gives appropriate credit (with a link to the formal publication through the relevant DOI), provides a link to the license, indicates if changes were made and the licensor is not represented as endorsing the use made of the work. Further, any new works must be made available on the same conditions. The full details of the license are available at http://creativecommons.org/licenses/by-nc-sa/4.0. CC BY NC ND: The CC BY-NC-ND license allows users to copy and distribute the Article, provided this is not done for commercial purposes and further does not permit distribution of the Article if it is changed or edited in any way, and provided the user gives appropriate credit (with a link to the formal publication through the relevant DOI), provides a link to the license, and that the licensor is not represented as endorsing the use made of the work. The full details of the license are available at http://creativecommons.org/licenses/by-nc-nd/4.0. Any commercial reuse of Open Access articles published with a CC BY NC SA or CC BY NC ND license requires permission from Elsevier and will be subject to a fee. Commercial reuse includes:

- Associating advertising with the full text of the Article
- Charging fees for document delivery or access

- Article aggregation
- Systematic distribution via e-mail lists or share buttons

Posting or linking by commercial companies for use by customers of those companies.

20. Other Conditions:

v1.9

Questions? <u>customercare@copyright.com</u> or +1-855-239-3415 (toll free in the US) or +1-978-646-2777.




ACS Publications Title: Nost Trusted, Most Cited, Most Read

Water Nanocluster Formation in the Ionic Liquid 1-Butyl-3methylimidazolium Tetrafluoroborate ([C4mim] [BF4])-D20 Mixtures Author: Jingsi Gao, Norman J. Wagner Publication: Langmuir Publisher: American Chemical Society May 1, 2016 Copyright @ 2016, American Chemical Society



PERMISSION/LICENSE IS GRANTED FOR YOUR ORDER AT NO CHARGE

Date:

This type of permission/license, instead of the standard Terms & Conditions, is sent to you because no fee is being charged for your order. Please note the following:

- · Permission is granted for your request in both print and electronic formats, and translations.
- If figures and/or tables were requested, they may be adapted or used in part.
- Please print this page for your records and send a copy of it to your publisher/graduate school.
- Appropriate credit for the requested material should be given as follows: "Reprinted (adapted) with permission from (COMPLETE REFERENCE CITATION). Copyright (YEAR) American Chemical Society." Insert appropriate information in place of the capitalized words.
- One-time permission is granted only for the use specified in your request. No additional uses are granted (such as derivative works or other editions). For any other uses, please submit a new request.

CLOSE WINDOW BACK

Copyright © 2016 <u>Copyright Clearance Center, Inc.</u> All Rights Reserved. <u>Privacy statement</u>, <u>Terms and Conditions</u>, Comments? We would like to hear from you. E-mail us at <u>customercare@copyright.com</u>

ELSEVIER LICENSE TERMS AND CONDITIONS

Dec 21, 2016

This Agreement between Jingsi Gao ("You") and Elsevier ("Elsevier") consists of your license details and the terms and conditions provided by Elsevier and Copyright Clearance Center.

License Number	4013750312276
License date	Dec 21, 2016
Licensed Content Publisher	Elsevier
Licensed Content Publication	Journal of Molecular Liquids
Licensed Content Title	Non-ideal viscosity and excess molar volume of mixtures of 1-butyl- 3-methylimidazolium tetrafluoroborate ([C4mim][BF4]) with water
Licensed Content Author	Jingsi Gao,Norman J. Wagner
Licensed Content Date	November 2016
Licensed Content Volume Number	223
Licensed Content Issue Number	n/a
Licensed Content Pages	9
Start Page	678
End Page	686
Type of Use	reuse in a thesis/dissertation
Portion	full article
Format	both print and electronic
Are you the author of this Elsevier article?	No
Will you be translating?	No
Order reference number	
Title of your thesis/dissertation	RHEOLOGY AND MICROSTRUCTURE OF STABLE CONCENTRATED IONIC LIQUID COLLOIDAL SUSPENSIONS
Expected completion date	Jan 2017
Estimated size (number of pages)	290
Elsevier VAT number	GB 494 6272 12
Requestor Location	Jingsi Gao 150 Academy Street
	NEWARK, DE 19716 United States Attn: Jingsi Gao
Total	0.00 USD

Terms and Conditions

INTRODUCTION

1. The publisher for this copyrighted material is Elsevier. By clicking "accept" in connection with completing this licensing transaction, you agree that the following terms and conditions apply to this transaction (along with the Billing and Payment terms and conditions established by Copyright Clearance Center, Inc. ("CCC"), at the time that you opened your Rightslink account and that are available at any time at http://myaccount.copyright.com).

GENERAL TERMS

2. Elsevier hereby grants you permission to reproduce the aforementioned material subject to the terms and conditions indicated.

3. Acknowledgement: If any part of the material to be used (for example, figures) has appeared in our publication with credit or acknowledgement to another source, permission must also be sought from that source. If such permission is not obtained then that material may not be included in your publication/copies. Suitable acknowledgement to the source must be made, either as a footnote or in a reference list at the end of your publication, as follows:

"Reprinted from Publication title, Vol/edition number, Author(s), Title of article / title of chapter, Pages No., Copyright (Year), with permission from Elsevier [OR APPLICABLE SOCIETY COPYRIGHT OWNER]." Also Lancet special credit - "Reprinted from The Lancet, Vol. number, Author(s), Title of article, Pages No., Copyright (Year), with permission from Elsevier."

4. Reproduction of this material is confined to the purpose and/or media for which permission is hereby given.

5. Altering/Modifying Material: Not Permitted. However figures and illustrations may be altered/adapted minimally to serve your work. Any other abbreviations, additions, deletions and/or any other alterations shall be made only with prior written authorization of Elsevier Ltd. (Please contact Elsevier at permissions@elsevier.com). No modifications can be made to any Lancet figures/tables and they must be reproduced in full.

6. If the permission fee for the requested use of our material is waived in this instance, please be advised that your future requests for Elsevier materials may attract a fee.7. Reservation of Rights: Publisher reserves all rights not specifically granted in the combination of (i) the license details provided by you and accepted in the course of this

licensing transaction, (ii) these terms and conditions and (iii) CCC's Billing and Payment terms and conditions.

8. License Contingent Upon Payment: While you may exercise the rights licensed immediately upon issuance of the license at the end of the licensing process for the transaction, provided that you have disclosed complete and accurate details of your proposed use, no license is finally effective unless and until full payment is received from you (either by publisher or by CCC) as provided in CCC's Billing and Payment terms and conditions. If full payment is not received on a timely basis, then any license preliminarily granted shall be deemed automatically revoked and shall be void as if never granted. Further, in the event that you breach any of these terms and conditions or any of CCC's Billing and Payment terms and conditions, the license is automatically revoked and shall be void as if never granted. Use of materials as described in a revoked license, as well as any use of the materials beyond the scope of an unrevoked license, may constitute copyright infringement and publisher reserves the right to take any and all action to protect its copyright in the materials.

9. Warranties: Publisher makes no representations or warranties with respect to the licensed material.

10. Indemnity: You hereby indemnify and agree to hold harmless publisher and CCC, and their respective officers, directors, employees and agents, from and against any and all claims arising out of your use of the licensed material other than as specifically authorized pursuant to this license.

11. No Transfer of License: This license is personal to you and may not be sublicensed, assigned, or transferred by you to any other person without publisher's written permission.

12. No Amendment Except in Writing: This license may not be amended except in a writing signed by both parties (or, in the case of publisher, by CCC on publisher's behalf).
13. Objection to Contrary Terms: Publisher hereby objects to any terms contained in any purchase order, acknowledgment, check endorsement or other writing prepared by you, which terms are inconsistent with these terms and conditions or CCC's Billing and Payment terms and conditions. These terms and conditions, together with CCC's Billing and Payment terms and conditions (which are incorporated herein), comprise the entire agreement between you and publisher (and CCC) concerning this licensing transaction. In the event of any conflict between your obligations established by these terms and conditions and those established by CCC's Billing and Payment terms and conditions, these terms and conditions shall control.

14. Revocation: Elsevier or Copyright Clearance Center may deny the permissions described in this License at their sole discretion, for any reason or no reason, with a full refund payable to you. Notice of such denial will be made using the contact information provided by you. Failure to receive such notice will not alter or invalidate the denial. In no event will Elsevier or Copyright Clearance Center be responsible or liable for any costs, expenses or damage incurred by you as a result of a denial of your permission request, other than a refund of the amount(s) paid by you to Elsevier and/or Copyright Clearance Center for denied permissions.

LIMITED LICENSE

The following terms and conditions apply only to specific license types:

15. **Translation**: This permission is granted for non-exclusive world **English** rights only unless your license was granted for translation rights. If you licensed translation rights you may only translate this content into the languages you requested. A professional translator must perform all translations and reproduce the content word for word preserving the integrity of the article.

16. **Posting licensed content on any Website**: The following terms and conditions apply as follows: Licensing material from an Elsevier journal: All content posted to the web site must maintain the copyright information line on the bottom of each image; A hyper-text must be included to the Homepage of the journal from which you are licensing at

http://www.sciencedirect.com/science/journal/xxxxx or the Elsevier homepage for books at http://www.elsevier.com; Central Storage: This license does not include permission for a scanned version of the material to be stored in a central repository such as that provided by Heron/XanEdu.

Licensing material from an Elsevier book: A hyper-text link must be included to the Elsevier homepage at <u>http://www.elsevier.com</u>. All content posted to the web site must maintain the copyright information line on the bottom of each image.

Posting licensed content on Electronic reserve: In addition to the above the following clauses are applicable: The web site must be password-protected and made available only to bona fide students registered on a relevant course. This permission is granted for 1 year only. You may obtain a new license for future website posting.

17. For journal authors: the following clauses are applicable in addition to the above: **Preprints:**

A preprint is an author's own write-up of research results and analysis, it has not been peerreviewed, nor has it had any other value added to it by a publisher (such as formatting, copyright, technical enhancement etc.).

Authors can share their preprints anywhere at any time. Preprints should not be added to or enhanced in any way in order to appear more like, or to substitute for, the final versions of articles however authors can update their preprints on arXiv or RePEc with their Accepted Author Manuscript (see below).

If accepted for publication, we encourage authors to link from the preprint to their formal publication via its DOI. Millions of researchers have access to the formal publications on ScienceDirect, and so links will help users to find, access, cite and use the best available

version. Please note that Cell Press, The Lancet and some society-owned have different preprint policies. Information on these policies is available on the journal homepage. Accepted Author Manuscripts: An accepted author manuscript is the manuscript of an article that has been accepted for publication and which typically includes author-incorporated changes suggested during submission, peer review and editor-author communications.

Authors can share their accepted author manuscript:

- immediately
 - via their non-commercial person homepage or blog
 - by updating a preprint in arXiv or RePEc with the accepted manuscript
 - via their research institute or institutional repository for internal institutional uses or as part of an invitation-only research collaboration work-group
 - directly by providing copies to their students or to research collaborators for their personal use
 - for private scholarly sharing as part of an invitation-only work group on commercial sites with which Elsevier has an agreement
- after the embargo period
 - via non-commercial hosting platforms such as their institutional repository
 - via commercial sites with which Elsevier has an agreement

In all cases accepted manuscripts should:

- link to the formal publication via its DOI
- bear a CC-BY-NC-ND license this is easy to do
- if aggregated with other manuscripts, for example in a repository or other site, be shared in alignment with our hosting policy not be added to or enhanced in any way to appear more like, or to substitute for, the published journal article.

Published journal article (JPA): A published journal article (PJA) is the definitive final record of published research that appears or will appear in the journal and embodies all value-adding publishing activities including peer review co-ordination, copy-editing, formatting, (if relevant) pagination and online enrichment.

Policies for sharing publishing journal articles differ for subscription and gold open access articles:

<u>Subscription Articles</u>: If you are an author, please share a link to your article rather than the full-text. Millions of researchers have access to the formal publications on ScienceDirect, and so links will help your users to find, access, cite, and use the best available version. Theses and dissertations which contain embedded PJAs as part of the formal submission can be posted publicly by the awarding institution with DOI links back to the formal publications on ScienceDirect.

If you are affiliated with a library that subscribes to ScienceDirect you have additional private sharing rights for others' research accessed under that agreement. This includes use for classroom teaching and internal training at the institution (including use in course packs and courseware programs), and inclusion of the article for grant funding purposes. **Gold Open Access Articles:** May be shared according to the author-selected end-user

license and should contain a <u>CrossMark logo</u>, the end user license, and a DOI link to the formal publication on ScienceDirect.

Please refer to Elsevier's posting policy for further information.

18. For book authors the following clauses are applicable in addition to the above: Authors are permitted to place a brief summary of their work online only. You are not allowed to download and post the published electronic version of your chapter, nor may you scan the printed edition to create an electronic version. **Posting to a repository:** Authors are permitted to post a summary of their chapter only in their institution's repository. 19. **Thesis/Dissertation**: If your license is for use in a thesis/dissertation your thesis may be submitted to your institution in either print or electronic form. Should your thesis be published commercially, please reapply for permission. These requirements include permission for the Library and Archives of Canada to supply single copies, on demand, of the complete thesis and include permission for Proquest/UMI to supply single copies, on demand, of the complete thesis. Should your thesis be published commercially, please reapply for permission. Theses and dissertations which contain embedded PJAs as part of the formal submission can be posted publicly by the awarding institution with DOI links back to the formal publications on ScienceDirect.

Elsevier Open Access Terms and Conditions

You can publish open access with Elsevier in hundreds of open access journals or in nearly 2000 established subscription journals that support open access publishing. Permitted third party re-use of these open access articles is defined by the author's choice of Creative Commons user license. See our open access license policy for more information. **Terms & Conditions applicable to all Open Access articles published with Elsevier:** Any reuse of the article must not represent the author as endorsing the adaptation of the article nor should the article be modified in such a way as to damage the author's honour or reputation. If any changes have been made, such changes must be clearly indicated. The author(s) must be appropriately credited and we ask that you include the end user license and a DOI link to the formal publication on ScienceDirect. If any part of the material to be used (for example, figures) has appeared in our publication

with credit or acknowledgement to another source it is the responsibility of the user to ensure their reuse complies with the terms and conditions determined by the rights holder. Additional Terms & Conditions applicable to each Creative Commons user license: CC BY: The CC-BY license allows users to copy, to create extracts, abstracts and new works from the Article, to alter and revise the Article and to make commercial use of the Article (including reuse and/or resale of the Article by commercial entities), provided the user gives appropriate credit (with a link to the formal publication through the relevant DOI), provides a link to the license, indicates if changes were made and the licensor is not represented as endorsing the use made of the work. The full details of the license are available at http://creativecommons.org/licenses/by/4.0.

CC BY NC SA: The CC BY-NC-SA license allows users to copy, to create extracts, abstracts and new works from the Article, to alter and revise the Article, provided this is not done for commercial purposes, and that the user gives appropriate credit (with a link to the formal publication through the relevant DOI), provides a link to the license, indicates if changes were made and the licensor is not represented as endorsing the use made of the work. Further, any new works must be made available on the same conditions. The full details of the license are available at http://creativecommons.org/licenses/by-nc-sa/4.0. CC BY NC ND: The CC BY-NC-ND license allows users to copy and distribute the Article, provided this is not done for commercial purposes and further does not permit distribution of the Article if it is changed or edited in any way, and provided the user gives appropriate credit (with a link to the formal publication through the relevant DOI), provides a link to the license, and that the licensor is not represented as endorsing the use made of the work. The full details of the license are available at http://creativecommons.org/licenses/by-nc-nd/4.0. Any commercial reuse of Open Access articles published with a CC BY NC SA or CC BY NC ND license requires permission from Elsevier and will be subject to a fee. Commercial reuse includes:

- Associating advertising with the full text of the Article
- Charging fees for document delivery or access
- Article aggregation
- Systematic distribution via e-mail lists or share buttons

Posting or linking by commercial companies for use by customers of those companies.

20. Other Conditions:

v1.9

Questions? <u>customercare@copyright.com</u> or +1-855-239-3415 (toll free in the US) or +1-978-646-2777.

REFERENCES

1. Wagner, N. J.; Brady, J. F. Shear thickening in colloidal dispersions. *Physics Today* **2009**, *62* (10), 27-32.

2. Plechkova, N. V.; Seddon, K. R. Applications of ionic liquids in the chemical industry. *Chem. Soc. Rev.* **2008**, *37* (1), 123-150.

3. Gao, J.; Ndong, R. S.; Shiflett, M. B.; Wagner, N. J. Creating nanoparticle stability in ionic liquid [C4mim][BF4] by inducing solvation layering. *ACS Nano* **2015**, *9* (3), 3243-3253.

4. Ueno, K.; Kasuya, M.; Watanabe, M.; Mizukami, M.; Kurihara, K. Resonance shear measurement of nanoconfined ionic liquids. *Phys. Chem. Chem. Phys.* **2010**, *12* (16), 4066-4071.

5. Decker, M.; Halbach, C.; Nam, C.; Wagner, N.; Wetzel, E. Stab resistance of shear thickening fluid (STF)-treated fabrics. *Composites Science and Technology* **2007**, *67* (3), 565-578.

6. Gao, J.; Wagner, N. J. Non-ideal viscosity and excess molar volume of mixtures of 1-butyl-3-methylimidazolium tetrafluoroborate ([C4mim][BF4]) with water. *J. Mol. Liq.* **2016**, *223*, 678-686.

7. Gao, J.; Wagner, N. J. Water nanocluster formation in the ionic liquid 1-Butyl-3-Methylimidazolium Tetrafluoroborate ([C4mim][BF4])–D2O mixtures. *Langmuir* **2016**, 32, (20), 5078–5084.